UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington D.C. 20549

Form 10-Q

(Mark One) X

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2008

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE **SECURITIES EXCHANGE ACT OF 1934**

Commission file number 000-12477

Amgen Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

95-3540776 (I.R.S. Employer Identification No.)

One Amgen Center Drive, Thousand Oaks, California

(Address of principal executive offices)

(805) 447-1000

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ⊠ No □

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer 🗵	Accelerated filer \Box	Non-accelerated filer \Box	Smaller reporting company \Box
			(Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act) Yes 🗆 No 🗵

As of October 31, 2008, the registrant had 1,059,520,930 shares of common stock, \$0.0001 par value, outstanding.

91320-1799

(Zip Code)

AMGEN INC.

INDEX

PART	Г <u></u>	INANCIAL INFORMATION	Page No.
	Item 1.	Financial Statements	1
		<u>Condensed Consolidated Statements of Income –</u> <u>Three and nine months ended September 30, 2008 and 2007</u>	1
		<u>Condensed Consolidated Balance Sheets –</u> <u>September 30, 2008 and December 31, 2007</u>	2
		<u>Condensed Consolidated Statements of Cash Flows –</u> <u>Nine months ended September 30, 2008 and 2007</u>	3
		Notes to Condensed Consolidated Financial Statements	4
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	17
	Item 4.	Controls and Procedures	42
PART	II <u>(</u>	THER INFORMATION	
	Item 1.	Legal Proceedings	43
	Item 1A.	Risk Factors	43
	Item 2.	<u>Unregistered Sales of Equity Securities, Use of</u> <u>Proceeds and Issuer Purchases of Equity Securities</u>	67
	Item 6.	<u>Exhibits</u>	67
	<u>Signatures</u>		68
	Index to Exh	<u>bits</u>	69

i

PART I - FINANCIAL INFORMATION

Item 1. FINANCIAL STATEMENTS

AMGEN INC. CONDENSED CONSOLIDATED STATEMENTS OF INCOME (In millions, except per share data) (Unaudited)

		nths ended nber 30,	Nine months ended September 30,				
	2008	2007	2008	2007			
Revenues:							
Product sales	\$ 3,784	\$ 3,524	\$ 11,013	\$ 10,693			
Other revenues	91	87	239	333			
Total revenues	3,875	3,611	11,252	11,026			
Operating expenses:							
Cost of sales (excludes amortization of acquired							
intangible assets presented below)	677	792	1,738	1,942			
Research and development	729	776	2,232	2,444			
Selling, general and administrative	900	730	2,678	2,360			
Amortization of acquired intangible assets	74	76	221	224			
Write-off of acquired in-process research and development	-	590	-	590			
Other charges	12	254	306	543			
Total operating expenses	2,392	3,218	7,175	8,103			
Operating income	1,483	393	4,077	2,923			
Interest and other income and (expense), net	(12)	(21)	19	(20)			
Income before income taxes	1,471	372	4,096	2,903			
Provision for income taxes	313	171	861	572			
Net income	\$ 1,158	\$ 201	\$ 3,235	\$ 2,331			
Earnings per share:							
Basic	\$ 1.09	\$ 0.19	\$ 3.01	\$ 2.07			
Diluted	\$ 1.09	\$ 0.18	\$ 3.00	\$ 2.06			
Shares used in calculation of earnings per share:							
Basic	1,058	1,086	1,075	1,127			
Diluted	1,064	1,090	1,079	1,133			

See accompanying notes.

AMGEN INC. CONDENSED CONSOLIDATED BALANCE SHEETS (In millions, except per share data) (Unaudited)

	September 30, 2008		December 31		
			2007		
ASSETS					
Current assets:					
Cash and cash equivalents	\$	2,522	\$	2,024	
Marketable securities		7,235		5,127	
Trade receivables, net		2,114		2,101	
Inventories		2,004		2,091	
Other current assets		1,745		1,698	
Total current assets		15,620		13,041	
Property, plant and equipment, net		5,972		5,941	
Intangible assets, net		3,095		3,332	
Goodwill		11,340		11,240	
Other assets		971		1,085	
	\$	36,998	\$	34,639	

LIABILITIES AND STOCKHOLDERS' EQUITY

Current liabilities:		
Accounts payable	\$ 538 \$	378
Accrued liabilities	3,413	3,801
Current portion of other long-term debt	1,000	2,000
Total current liabilities	4,951	6,179
Deferred tax liabilities	346	480
Convertible notes	5,081	5,080
Other long-term debt	5,095	4,097
Other non-current liabilities	1,693	934

Contingencies

Stockholders' equity:			
Common stock and additional paid-in capital;			
\$0.0001 par value; 2,750 shares authorized;			
outstanding - 1,059 shares in 2008 and			
1,087 shares in 2007	25,348		24,976
Accumulated deficit	(5,519)		(7,160)
Accumulated other comprehensive income	3		53
Total stockholders' equity	 19,832		17,869
	\$ 36,998	_	\$ 34,639

See accompanying notes.

AMGEN INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In millions) (Unaudited)

	Nine mon Septem	
	2008	2007
Cash flows from operating activities:		
Net income	\$ 3,235	\$ 2,331
Write-off of acquired in-process research and development	-	590
Depreciation and amortization	799	900
Asset impairments	16	392
Other items, net	140	379
Changes in operating assets and liabilities, net of acquisitions:		
Trade receivables, net	16	(15)
Inventories	(22)	(114)
Other current assets	(29)	(68)
Accounts payable	136	(119)
Accrued income taxes	88	(934)
Deferred revenue	337	-
Other accrued liabilities	(125)	529
Net cash provided by operating activities	4,591	3,871
Cash flows from investing activities:		
Purchases of property, plant and equipment	(494)	(1,033
Cash paid for acquisitions, net of cash acquired	(50)	(698
Purchases of marketable securities	(7,794)	(4,236
Proceeds from sales of marketable securities	5,002	4,431
Proceeds from maturities of marketable securities	625	278
Other	93	(37
Net cash used in investing activities	(2,618)	(1,295)
Cash flows from financing activities:		
Net proceeds from issuance of common stock in connection with		
equity award programs	114	244
Repurchases of common stock	(1,568)	(5,000
Repayment of debt	(1,000)	(1,702
Proceeds from issuance of notes, net	992	3,982
Other	(13)	6
Net cash used in financing activities	(1,475)	(2,470
ncrease in cash and cash equivalents	498	106
Cash and cash equivalents at beginning of period	2,024	1,283
Cash and cash equivalents at end of period	\$ 2,522	\$ 1,389

See accompanying notes.

AMGEN INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS September 30, 2008 (Unaudited)

1. Summary of significant accounting policies

Business

Amgen Inc. is a global biotechnology company that discovers, develops, manufactures and markets human therapeutics based on advances in cellular and molecular biology.

Basis of presentation

The financial information for the three and nine months ended September 30, 2008 and 2007 is unaudited but includes all adjustments (consisting of only normal recurring adjustments, unless otherwise indicated), which Amgen Inc., including its subsidiaries (referred to as "Amgen," "the Company," "we," "our" or "us"), considers necessary for a fair presentation of the results of operations for those periods. Interim results are not necessarily indicative of results for the full fiscal year.

The condensed consolidated financial statements should be read in conjunction with our consolidated financial statements and the notes thereto contained in our Annual Report on Form 10-K for the year ended December 31, 2007.

Principles of consolidation

The condensed consolidated financial statements include the accounts of Amgen as well as its wholly owned subsidiaries. We do not have any significant interests in any variable interest entities. All material intercompany transactions and balances have been eliminated in consolidation.

Use of estimates

The preparation of condensed consolidated financial statements in conformity with accounting principles generally accepted in the United States ("GAAP") requires management to make estimates and assumptions that affect the amounts reported in the condensed consolidated financial statements and accompanying notes. Actual results may differ from those estimates.

Inventories

Inventories are stated at the lower of cost or market. Cost, which includes amounts related to materials, labor and overhead, is determined in a manner which approximates the first-in, first-out ("FIFO") method. During the three months ended September 30, 2008, we wrote-off \$84 million of inventory resulting from a strategic decision to change manufacturing processes. During the three months ended September 30, 2007, we wrote-off \$90 million of excess inventory principally due to changing regulatory and reimbursement environments. Inventories consisted of the following (in millions):

	September 30,	December 31,		
	2008	2007		
Raw materials	\$ 139	\$	173	
Work in process	1,434		1,246	
Finished goods	431		672	
	\$ 2,004	\$	2,091	

Property, plant and equipment, net

Property, plant and equipment are recorded at historical cost, net of accumulated depreciation of \$4.0 billion and \$3.6 billion as of September 30, 2008 and December 31, 2007, respectively.

Goodwill

Goodwill principally relates to the acquisition of Immunex Corporation ("Immunex"). The increase over the balance at December 31, 2007 is related to the goodwill associated with our acquisition of the remaining 51% ownership interest of Dompé Biotec, S.p.A ("Dompé") on January 4, 2008 (see Note 7, "*Acquisition*" for further discussion). We perform an impairment test annually and whenever events or changes in circumstances indicate that the carrying amount of goodwill may not be recoverable.

Fair value measurement

The Company adopted the provisions of the Financial Accounting Standards Board's ("FASB's") Statement of Financial Accounting Standards ("SFAS") No. 157, "*Fair Value Measurements*" ("SFAS 157"), effective January 1, 2008, for its financial assets and liabilities. The FASB delayed the effective date of SFAS 157 until January 1, 2009, with respect to the fair value measurement requirements for non-financial assets and liabilities that are not remeasured on a recurring basis. Under this standard, fair value is defined as the price that would be received to sell an asset or paid to transfer a liability (i.e., the "exit price") in an orderly transaction between market participants at the measurement date. The adoption of SFAS 157 did not have a material impact on the Company's consolidated financial statements.

In determining the fair value of its financial assets and liabilities, the Company uses various valuation approaches. SFAS 157 establishes a hierarchy for inputs used in measuring fair value that maximizes the use of observable inputs and minimizes the use of unobservable inputs by requiring that observable inputs be used when available. Observable inputs are inputs that market participants would use in pricing the asset or liability based on market data obtained from sources independent of the Company. Unobservable inputs are inputs that reflect the Company's assumptions about the inputs that market participants would use in pricing the asset or liability and are developed based on the best information available in the circumstances. The fair value hierarchy is broken down into three levels based on the source of inputs as follows:

Level 1 – Valuations based on unadjusted quoted prices in active markets for identical assets or liabilities that the Company has the ability to access

Level 2 – Valuations based on quoted prices for similar assets or liabilities in active markets, quoted prices for identical or similar assets or liabilities in markets that are not active and models for which all significant inputs are observable, either directly or indirectly

Level 3 - Valuations based on inputs that are unobservable and significant to the overall fair value measurement

The availability of observable inputs can vary among the various types of financial assets and liabilities. To the extent that the valuation is based on models or inputs that are less observable or unobservable in the market, the determination of fair value requires more judgment. In certain cases, the inputs used to measure fair value may fall into different levels of the fair value hierarchy. In such cases, for financial statement disclosure purposes, the level in the fair value hierarchy within which the fair value measurement is categorized is based on the lowest level input that is significant to the overall fair value measurement.

AMGEN INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Continued)

The Company's available-for-sale securities, substantially all of which are fixed income investments, are comprised of U.S. Treasury securities, obligations of U.S. government agencies, money market funds, corporate debt securities, other interest bearing securities and publicly traded equity investments. U.S. Treasury securities, money market funds and publicly traded equity investments are valued using quoted market prices with no valuation adjustments applied. Accordingly, these securities are categorized in Level 1. Obligations of U.S. government agencies, corporate debt securities and other interest bearing securities are valued using quoted market prices of recent transactions or are benchmarked to transactions of very similar securities. When observable price quotations are not available, cash flow models are used to incorporate benchmark yields and issuer spreads. Obligations of U.S. government agencies, corporate debt securities are categorized in Level 2.

Derivatives assets and liabilities include interest rate swaps and foreign currency forward and option contracts. The fair values of these derivatives are determined using models based on market observable inputs, including interest rate curves and both forward and spot prices for foreign currencies. All of these derivative contracts are categorized in Level 2.

The following fair value hierarchy table presents information about each major category of the Company's financial assets and liabilities measured at fair value on a recurring basis as of September 30, 2008 (in millions):

		Fair value n					
	active r identi	Quoted prices in active markets for identical assets (Level 1)		icant other servable nputs sevel 2)	Significant unobservable inputs (Level 3)		nce as of ber 30, 2008
Assets:							
Available-for-sale securities	\$	3,629	\$	6,029	\$	-	\$ 9,658
Derivatives		-		67		-	 67
Total	\$	3,629	\$	6,096	\$	-	\$ 9,725
Liabilities:							
Derivatives	\$	-	\$	44	\$	-	\$ 44
Total	\$	-	\$	44	\$	-	\$ 44

There were no material remeasurements to fair value during the three and nine months ended September 30, 2008 of financial assets and liabilities that are not measured at fair value on a recurring basis.

Product sales

Product sales primarily consist of sales of Aranesp® (darbepoetin alfa), EPOGEN® (Epoetin alfa), Neulasta® (pegfilgrastim), NEUPOGEN® (Filgrastim) and Enbrel® (etanercept).

Sales of our products are recognized when title and risk of loss have passed. Product sales are recorded net of provisions for estimated rebates, wholesaler chargebacks, discounts and other incentives (collectively "sales incentives") and returns. Taxes assessed by government authorities on the sales of the Company's products, primarily in Europe, are excluded from revenues.

AMGEN INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Continued)

We have the exclusive right to sell Epoetin alfa for dialysis, certain diagnostics and all non-human, non-research uses in the United States. We sell Epoetin alfa under the brand name EPOGEN®. We granted to Ortho Pharmaceutical Corporation (which has assigned its rights under the product license agreement to Ortho Biotech Products, L.P. ("Ortho Biotech")), a subsidiary of Johnson & Johnson ("J&J"), a license relating to Epoetin alfa for sales in the United States for all human uses except dialysis and diagnostics. This license agreement, which is perpetual, may be terminated for various reasons, including upon mutual agreement of the parties, or default. The parties are required to compensate each other for Epoetin alfa sales that either party makes into the other party's exclusive market, sometimes referred to as "spillover." Accordingly, we do not recognize product sales we make into the exclusive market of J&J and do recognize the product sales made by J&J into our exclusive market. Sales in our exclusive market are derived from our sales to our customers, as adjusted for spillover. We are employing an arbitrated audit methodology to measure each party's spillover based on estimates of and subsequent adjustments thereto of third-party data on shipments to end users and their usage.

Research and development costs

Research and development ("R&D") costs are expensed as incurred and primarily include salaries, benefits and other staff related costs; facilities and overhead costs; clinical trial and related clinical manufacturing costs; contract services and other outside costs; information systems and amortization of acquired technology used in R&D with alternative future uses. R&D expenses consist of internal R&D costs; costs incurred under R&D arrangements with our corporate partners, such as activities performed on behalf of Kirin-Amgen Inc. ("KA"), and costs associated with collaborative R&D and in-licensing arrangements, including upfront fees and milestones paid to collaboration partners in connection with technologies that have no alternative future use. Net payment or reimbursement of R&D costs for R&D collaborations are recognized as the obligation has been incurred or as we become entitled to the cost recovery.

Selling, general and administrative costs

Selling, general and administrative ("SG&A") expenses are primarily comprised of salaries and benefits associated with sales and marketing, finance, legal and other administrative personnel; outside marketing and legal expenses; overhead and facilities costs and other general and administrative costs. In connection with a co-promotion agreement, we and Wyeth market and sell ENBREL in the United States and Canada and Wyeth is paid a share of the related profits, as defined. The share of ENBREL's profits owed to Wyeth (the "Wyeth profit share expense") is included in SG&A expenses. For the three and nine months ended September 30, 2008, the Wyeth profit share expense was \$298 million and \$886 million, respectively. For the three and nine months ended September 30, 2007, the Wyeth profit share expense was \$245 million and \$719 million, respectively.

Earnings per share

Basic earnings per share ("EPS") is based upon the weighted-average number of common shares outstanding. Diluted EPS is based upon the weightedaverage number of common shares and dilutive potential common shares outstanding. Potential common shares outstanding principally include stock options, restricted stock (including restricted stock units) and other equity awards under our employee compensation plans and potential issuance of stock upon the assumed conversion of our 2011 Convertible Notes and 2013 Convertible Notes, as discussed below, and upon the assumed exercise of our warrants using the treasury stock method (collectively "Dilutive Securities"). The convertible note hedges purchased in connection with the issuance of our 2011 Convertible Notes and 2013 Convertible Notes are excluded from the calculation of diluted EPS as their impact is always anti-dilutive.

Our 2011 Convertible Notes and 2013 Convertible Notes are considered Instrument C securities as defined by Emerging Issues Task Force ("EITF") Issue No. 90-19 "*Convertible Bonds with Issuer Option to Settle for Cash upon Conversion.*" Therefore, only the shares of common stock potentially issuable with respect to the excess of the notes' conversion value over their principal amount, if any, are considered as dilutive potential common shares

for purposes of calculating diluted EPS. For the three and nine months ended September 30, 2008 and 2007, the conversion values for our convertible notes were less than the related principal amounts and, accordingly, no shares were assumed to be issued for purposes of computing diluted EPS.

The following table sets forth the computation for basic and diluted EPS (in millions, except per share information):

		nonths ended tember 30,		nonths ended tember 30,
	2008	2007	2008	2007
Income (Numerator):				
Net income for basic and diluted EPS	<u>\$1,158</u>	<u>\$ 201</u>	\$3,235	\$2,331
Shares (Denominator):				
Weighted-average shares for basic EPS	1,058	1,086	1,075	1,127
Effect of dilutive securities	6	4	4	6
Weighted-average shares for diluted EPS	1,064	1,090	1,079	1,133
Basic EPS	\$ 1.09	\$ 0.19	\$ 3.01	\$ 2.07
Diluted EPS	\$ 1.09	\$ 0.18	\$ 3.00	\$ 2.06

Recent accounting pronouncements

In June 2008, the FASB ratified EITF Issue No. 07-5, "Determining Whether an Instrument (or Embedded Feature) Is Indexed to an Entity's Own Stock" ("EITF 07-5"). Equity-linked instruments (or embedded features) that otherwise meet the definition of a derivative as outlined in SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities," are not accounted for as derivatives if certain criteria are met, one of which is that the instrument (or embedded feature) must be indexed to the entity's stock. EITF 07-5 provides guidance on how to determine if equity-linked instruments (or embedded features) such as warrants to purchase our stock, our convertible notes and convertible note hedges are considered indexed to our stock. EITF 07-5 is effective for the financial statements issued for fiscal years and interim periods within those fiscal years, beginning after December 15, 2008 and will be applied to outstanding instruments as of the beginning of the fiscal year in which it is adopted. Upon adoption, a cumulative effect adjustment will be recorded, if necessary, based on amounts that would have been recognized if this guidance had been applied from the issuance date of the affected instruments. We are currently determining the impact that EITF 07-05 will have on our financial statements, if any.

In May 2008, the FASB issued FASB Staff Position ("FSP") No. APB 14-1, "*Accounting for Convertible Debt Instruments That May Be Settled in Cash upon Conversion (Including Partial Cash Settlement*" ("FSP APB 14-1") that changes the method of accounting for convertible debt securities that require or permit settlement in cash either in whole or in part upon conversion, including our convertible debt securities (see Note 5, "*Financing arrangements*"). We will adopt FSP APB 14-1 in the first quarter of 2009 and retrospectively apply this change to prior periods, as required by this new standard. Under this new method of accounting, the debt and equity components of our convertible debt securities will be bifurcated and accounted for separately in a manner that will result in recognizing interest expense on these securities at effective rates reflective of what we would have incurred had we issued nonconvertible debt with otherwise similar terms. The equity component of our convertible debt securities will be included in the paid-in-capital section of stockholders' equity on our Consolidated Balance Sheet and, accordingly, the initial carrying values of these debt securities will be reduced. Our net income for financial reporting purposes will be reduced by recognizing the accretion of the reduced carrying values of our convertible debt securities to their face amounts as additional non-cash interest expense. We are currently determining the impact FSP APB 14-1 will have on our financial statements. We expect it will have a material adverse impact on our past and future reported financial results but will have no impact on past or future cash flows.

In December 2007, the FASB issued SFAS No. 141(R), "Business Combinations" ("SFAS 141(R)") and SFAS No. 160, "Accounting and Reporting of Noncontrolling Interests in Consolidated Financial Statements – an amendment of ARB No. 51" ("SFAS 160"). These standards will significantly change the accounting and reporting for business combination transactions and noncontrolling (minority) interests in consolidated financial statements, including capitalizing at the acquisition date the fair value of acquired in-process research and development ("IPR&D"), and testing for impairment and writing down these assets, if necessary, in subsequent periods during their development. These new standards will be applied prospectively for business combinations that occur on or after January 1, 2009, except that presentation and disclosure requirements of SFAS 160 regarding noncontrolling interests will be applied retrospectively.

2. Restructuring

On August 15, 2007, we announced a plan to restructure our worldwide operations in order to improve our cost structure while continuing to make significant R&D investments and build the framework for our future growth. This restructuring plan was primarily the result of regulatory and reimbursement developments that began in 2007 involving erythropoietic stimulating agent ("ESA") products, including our marketed ESA products Aranesp® and EPOGEN®, and the resulting impact on our operations. Our ESA products have and may continue to face regulatory and reimbursement challenges, including the potential for further revisions to product labels and loss of or restrictions on reimbursement coverage. In addition, the restructuring plan is also, to a lesser degree, the result of various challenges facing certain of our other products.

Through September 30, 2008, we have completed a majority of the actions initially included in our restructuring plan. Key components of our restructuring plan initially included: (i) worldwide staff reductions aggregating approximately 2,500 positions, (ii) rationalization of our worldwide network of manufacturing facilities in order to gain cost efficiencies while continuing to meet future commercial and clinical demand for our products and product candidates and, to a lesser degree, changes to certain R&D capital projects and (iii) abandoning leases primarily for certain R&D facilities that will not be used in our operations. Through September 30, 2008, the total cost incurred with respect to these actions was \$790 million. We have recently identified certain additional initiatives designed to further assist in improving our cost structure, including outsourcing certain non-core business functions, most notably certain of our information systems infrastructure services, as well as abandoning leases for certain additional facilities that will no longer be used in our operations. The estimated cost of these additional initiatives is \$50 million to \$100 million. As a result of the actual costs incurred to date and the addition of the recently identified initiatives, the total charges expected to be incurred in connection with our restructuring plan, including related implementation costs, has been increased to \$850 million to \$925 million, as compared to our prior estimate of \$775 million to \$825 million. We currently estimate that all remaining costs will be incurred through 2009. Such cost estimates and amounts incurred are net of amounts recovered from our ENBREL co-promotion partner, Wyeth.

The following tables summarize the charges (credits) recorded during the three and nine months ended September 30, 2008 and 2007 related to the restructuring plan by type of activity (in millions):

Three months ended September 30, 2008	Separation costs		Asset impairments		Accelerated depreciation		0	Other		Total	
Other charges	\$	-	\$	1	\$	-	\$	7	\$	8	
Interest and other income and (expense), net	-	-	-	-	Ŧ	-	-	9	+	9	
	\$	-	\$	1	\$	-	\$	16	\$	17	
Three months ended September 30, 2007											
Cost of sales (excludes amortization											
of acquired intangible assets)	\$	(1)	\$	4	\$	110	\$	-	\$	113	
R&D		(17)		35		-		-		18	
SG&A		(9)		-		-		(83)		(92)	
Other charges		104		71		-		79		254	
	\$	77	\$	110	\$	110	\$	(4)	\$	293	
Nine months ended September 30, 2008											
Cost of sales (excludes amortization											
of acquired intangible assets)	\$	-	\$	1	\$	-	\$	-	\$	1	
R&D		3		-		-		-		3	
SG&A		-		-		-		(1)		(1)	
Other charges		4		15		-		20		39	
Interest and other income and (expense), net		-	_	-		-		9		9	
	\$	7	\$	16	\$	-	\$	28	\$	51	
Nine months ended September 30, 2007											
Cost of sales (excludes amortization											
of acquired intangible assets)	\$	(1)	\$	4	\$	110	\$	-	\$	113	
R&D		(17)		35		-		-		18	
SG&A		(9)		-		-		(83)		(92)	
Other charges		107		357		-		79		543	
	\$	80	\$	396	\$	110	\$	(4)	\$	582	

As noted above, since the inception of our restructuring plan through September 30, 2008, we have incurred \$790 million of the estimated \$850 million to \$925 million of charges expected to be incurred. The charges incurred through September 30, 2008 include \$185 million of separation costs, \$424 million of asset impairments, \$148 million of accelerated depreciation, \$33 million of other charges, which primarily include \$139 million of loss accruals for leases and \$9 million with respect to the loss accrued on the disposal of certain less significant marketed products and related assets, including primarily inventory, offset by \$115 million of cost recoveries from Wyeth.

The following table summarizes the charges and spending relating to the restructuring plan (in millions):

	Sepa			
	C	osts	Other	Total
Restructuring reserves as of January 1, 2008	\$	97	\$ 102	\$ 199
Expense		7	26	33
Payments		(97)	(15)	(112)
Restructuring reserves as of September 30, 2008	\$	7	\$ 113	\$ 120

The Company records restructuring activities in accordance with SFAS 88, "Employers' Accounting for Settlements and Curtailments of Defined Benefit Pension Plans and for Termination Benefits," SFAS 144, "Accounting for the Impairment and Disposal of Long-Lived Assets" and SFAS 146, "Accounting for Costs Associated with Exit or Disposal Activities."

3. Related party transactions

We own a 50% interest in KA, a corporation formed in 1984 with Kirin Holdings Company, Limited ("Kirin") for the development and commercialization of certain products based on advanced biotechnology. We account for our interest in KA under the equity method and include our share of KA's profits or losses in "Selling, general and administrative" in the Condensed Consolidated Statements of Income. During the three and nine months ended September 30, 2008, our share of KA's profits was \$22 million and \$53 million, respectively. During the three and nine months ended September 30, 2007, our share of KA's profits was \$18 million and \$40 million, respectively. At September 30, 2008 and December 31, 2007, the carrying value of our equity method investment in KA was \$345 million and \$292 million, respectively, and is included in non-current "Other assets" in the Condensed Consolidated Balance Sheets. KA's revenues consist of royalty income related to its licensed technology rights. All of our rights to manufacture and market certain products including darbepoetin alfa, pegfilgrastim, granulocyte colony-stimulating factor ("G-CSF") and recombinant human erythropoietin are pursuant to exclusive licenses from KA, which we currently market under the brand names Aranesp®, Neulasta®, NEUPOGEN® and EPOGEN®, respectively. KA receives royalty income from us, as well as Kirin, J&J and F. Hoffmann-La Roche Ltd. ("Roche") under separate product license agreements for certain geographic areas outside of the United States. During the three and nine months ended September 30, 2008, KA earned royalties from us of \$85 million and \$243 million, respectively. During the three and nine months ended in "Cost of sales (excludes amortization of acquired intangible assets)" in the Condensed Consolidated Statements of Income. At September 30, 2008 and December 31, 2007, we owed KA \$3 million and \$91 million, respectively, which was included in "Accrued liabilities" in the Condensed Consolidated Balance Sheets.

KA's expenses primarily consist of costs related to R&D activities conducted on its behalf by Amgen and Kirin. KA pays Amgen and Kirin for such services at negotiated rates. During the three and nine months ended September 30, 2008, we earned revenues from KA of \$41 million and \$100 million, respectively, for certain R&D activities performed on KA's behalf. During the three and nine months ended September 30, 2007, we earned revenues from KA of \$39 million and \$144 million, respectively. These amounts are included in "Other revenues" in the Condensed Consolidated Statements of Income.

4. Income taxes

The effective tax rates for the three and nine months ended September 30, 2008 are lower than the statutory rate primarily as a result of indefinitely invested earnings of our foreign operations which are taxed at a lower rate than the statutory rate. The effective tax rate for the three months ended September 30, 2007 is higher than the statutory rate primarily as a result of the write-off of non-tax-deductible, acquired IPR&D expense in

AMGEN INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Continued)

connection with the acquisitions of Alantos Pharmaceuticals Holding, Inc. ("Alantos") and Ilypsa, Inc. ("Ilypsa") partially offset by indefinitely invested earnings of our foreign operations. The effective tax rate for the nine months ended September 30, 2007 is different from the statutory rate primarily as a result of these same factors as well as the favorable resolution of our federal income tax examination for prior years, which was recorded in the second quarter of 2007. We do not provide for U.S. income taxes on undistributed earnings of our controlled foreign corporations that are intended to be invested indefinitely outside the United States.

One or more of our legal entities file income tax returns in the U.S. federal jurisdiction, various U.S. state jurisdictions and certain foreign jurisdictions. Our income tax returns are routinely examined by the tax authorities in those jurisdictions. Significant disputes can arise with these tax authorities involving issues of the timing and amount of deductions and allocations of income among various tax jurisdictions because of differing interpretations of tax laws and regulations. We are no longer subject to U.S. federal income tax examinations for years ending on or before December 31, 2004 or to California state income tax examinations for years ending on or before December 31, 2003.

During the three and nine months ended September 30, 2008, the gross amount of our unrecognized tax benefits ("UTBs") increased approximately \$100 million and \$280 million, respectively, as a result of tax positions taken during the current year. During the nine months ended September 30, 2008, the gross amount of our UTBs decreased approximately \$185 million, net, related to tax positions taken in prior years, primarily as a result of an agreement with the Internal Revenue Service in the first quarter of 2008 related to certain transfer pricing positions for the years 2005 and 2006. The majority of our UTBs at September 30, 2008, if recognized, would affect our effective tax rate.

5. Financing arrangements

The following table reflects the carrying value of our long-term borrowings under our various financing arrangements as of September 30, 2008 and December 31, 2007 (in millions):

	September 30, 2008		Dec		ıber 31,)07
0.125% convertible notes due 2011 (2011 Convertible Notes)	\$	2,500	_	\$	2,500
0.375% convertible notes due 2013 (2013 Convertible Notes)		2,500			2,500
5.85% notes due 2017 (2017 Notes)		1,099			1,099
Floating rate notes due 2008 (2008 Floating Rate Notes)		1,000			2,000
4.00% notes due 2009 (2009 Notes)		1,000			999
4.85% notes due 2014 (2014 Notes)		1,000			1,000
6.375% notes due 2037 (2037 Notes)		899			899
6.15% notes due 2018 (2018 Notes)		499			-
6.90% notes due 2038 (2038 Notes)		498			-
Other		181			180
Total borrowings		11,176	-		11,177
Less current portion		1,000	_		2,000
Total non-current debt	\$	10,176	=	\$	9,177

On April 17, 2008, we filed a shelf registration statement with the Securities and Exchange Commission ("SEC"), which replaced our previous \$1.0 billion shelf registration statement and allows us to issue an unspecified amount of debt securities; common stock; preferred stock; warrants to purchase debt securities, common stock, preferred stock or depository shares; rights to purchase common stock or preferred stock; securities purchase contracts; securities purchase units and depository shares. Under this registration statement,

AMGEN INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Continued)

all of the securities available for issuance may be offered from time to time with terms to be determined at the time of issuance.

In May 2008, we increased our commercial paper program by \$1.3 billion, which provides for unsecured, short-term borrowings of up to an aggregate of \$2.5 billion. We also have a \$2.5 billion syndicated unsecured revolving credit facility which matures in November 2012 and is available for general corporate purposes, or as a liquidity backstop to our commercial paper program; however, \$178 million of such commitment was provided by a subsidiary of Lehman Brothers Holdings Inc. ("Lehman"). Lehman declared bankruptcy on September 15, 2008, and the subsidiary participant in our credit facility subsequently declared bankruptcy on October 5, 2008. As a result, we would not anticipate the ability to access this specific commitment provided by Lehman in the future. No amounts were outstanding under the commercial paper program or credit facility as of September 30, 2008.

In May 2008, we issued \$500 million aggregate principal amount of notes due in 2018 (the "2018 Notes") and \$500 million aggregate principal amount of notes due in 2038 (the "2038 Notes") in a registered offering. The 2018 Notes and 2038 Notes pay interest at fixed annual rates of 6.15% and 6.90%, respectively. Concurrent with the issuance of the 2018 Notes, we entered into interest rate swap agreements that effectively convert the payment of our fixed rate interest payments to variable rate interest payments over the life of the 2018 Notes. The 2018 Notes and 2038 Notes may be redeemed at any time at our option, in whole or in part, at 100% of the principal amount of the notes being redeemed plus accrued and unpaid interest, if any, and a "make-whole" amount, as defined in the indenture governing the notes. In the event of a change in control triggering event, as defined in the indenture governing the notes, we may be required to purchase for cash all or a portion of the 2018 Notes and the 2038 Notes at a price equal to 101% of the principal amount of the notes plus accrued interest. Debt issuance costs totaled approximately \$6 million and are being amortized over the life of the notes. Upon the receipt of the proceeds from the issuance of the 2018 Notes and 2038 Notes, we exercised our right to call \$1.0 billion of floating rate notes due November 2008, which were retired in June 2008.

6. Stockholders' equity

Stock repurchase programs

A summary of activity under our stock repurchase programs for the nine months ended September 30, 2008 and 2007 is as follows (in millions):

	20	08	2007			
	Shares	Dollars	Shares	Dollars		
First quarter	-	\$ -	8.8	\$ 537		
Second quarter	32.7	1,549 (1)	73.9 (2)	4,463		
Third quarter	-	19 (1)	2.5 (2)	-		
Total	32.7	\$ 1,568	85.2	\$ 5,000		

(1) The total cost of shares repurchased during the second quarter of 2008 excludes approximately \$19 million paid in July 2008 in connection with the final settlement of an accelerated share repurchase program ("ASR") entered into in May 2008.

(2) The total number of shares repurchased during the second quarter of 2007 excludes 2.5 million shares received in July 2007 in connection with the final settlement of an ASR entered into in May 2007.

As of September 30, 2008, \$4.9 billion remained available for stock repurchases as authorized by our Board of Directors. The manner of purchases, the amount we spend, and the number of shares repurchased will vary based on a number of factors, including the stock price, blackout periods in which we are restricted from

repurchasing shares, and our credit rating and may include private block purchases as well as market transactions.

7. Acquisition

On January 4, 2008, we completed the acquisition of Dompé, a privately held company that marketed certain of our products in Italy. This cash acquisition was accounted for as a business combination. The purchase price was approximately \$162 million, which included the carrying value of our existing 49% ownership in Dompé. The purchase price paid was preliminarily allocated to net assets acquired of approximately \$63 million based on their estimated fair values at the acquisition date and the excess of the purchase price over the fair values of net assets acquired of approximately \$99 million was assigned to goodwill. There was no material gain or loss related to the reacquisition of marketing rights previously granted to Dompé as a result of this business combination. The results of Dompé's operations have been included in the condensed consolidated financial statements commencing January 4, 2008. Pro forma results of operations for the three and nine months ended September 30, 2008 assuming the acquisition of Dompé had taken place at the beginning of 2008 would not differ significantly from the actual reported results.

8. Other charges

In the three and nine months ended September 30, 2008, we recorded loss accruals for settlements of certain commercial legal proceedings aggregating \$4 million and \$267 million, respectively. For the nine months ended September 30, 2008, the loss accruals principally related to the settlement of the Ortho Biotech antitrust suit. Such expenses are included in "Other charges" in the Condensed Consolidated Statements of Income.

For the three and nine months ended September 30, 2008, we recorded restructuring charges of \$8 million and \$39 million, respectively. For the three and nine months ended September 30, 2007, we recorded restructuring charges of \$254 million and \$543 million, respectively. Such expenses are included in "Other charges" in the Condensed Consolidated Statements of Income. (See Note 2, "*Restructuring*" for further discussion.)

9. Contingencies

In the ordinary course of business, we are involved in various legal proceedings and other matters that are complex in nature and have outcomes that are difficult to predict. In accordance with SFAS 5, "*Accounting for Contingencies*," we record accruals for such contingencies to the extent that we conclude that it is probable that a liability will be incurred and the amount of the related loss can be reasonably estimated. See Note 10, "*Contingencies*" to our Consolidated Financial Statements in our Annual Report on Form 10-K for the year ended December 31, 2007 for further discussion of certain of our legal proceedings and other matters.

Certain recent developments concerning our legal proceedings and other matters are discussed below:

Transkaryotic Therapies ("TKT") and Aventis Litigation

On October 2, 2008, the U.S. District Court for the District of Massachusetts (the "Massachusetts District Court") entered a Memorandum and Order upholding the validity of claim 1 of the '422 patent, the sole remaining issue on remand, and entered declaratory judgment enjoining TKT and Hoechst Marion Roussel, Inc. (now Aventis Pharmaceuticals Inc.) from infringing the '422 patent, the '698 patent and the '349 patent for the life of the patents, the last of which expires in 2015.

Robert J. Swanston v. TAP Pharmaceutical Products, Inc., et al.

A hearing on defendants' motion for summary judgment has been rescheduled before the Maricopa County, Arizona Superior Court for December 5, 2008.

State of Alabama v. Abbott Laboratories, Inc. ("Abbott"), et al.

Two additional trials of non-Track 4 defendants (which did not include Amgen and Immunex) were held in June 2008. Following these trials, plaintiff Alabama filed a motion to set a trial date for four additional companies, including Amgen and Immunex. The Circuit Court of Montgomery County, Alabama granted the motion and set trial for Amgen and Immunex for February 2009. The plaintiff also filed a motion to consolidate the four defendants into one trial and the motion to consolidate was granted as to two of the four defendants, which did not include Amgen or Immunex.

County of Erie v. Abbott Laboratories, Inc., et al.; County of Schenectady v. Abbott Laboratories, Inc., et al.; County of Oswego v. Abbott Laboratories, Inc., et al.

The State of New York Litigation Coordinating Panel granted defendants' motions to coordinate the Erie, Oswego and Schenectady County cases. The Erie County judge will hear all discovery disputes on November 13 and 14, 2008.

State of Kansas, ex rel Steve Six v. Amgen Inc. and Immunex Corporation

On November 3, 2008, the State of Kansas filed a complaint against Amgen and Immunex in the District Court of Wyandotte County, Kansas, Civil Court Division. Approximately forty other pharmaceutical manufacturers were also sued by the state. Plaintiff Kansas alleges that the manufacturers misrepresented product pricing information reported to the state by falsely inflating those prices.

Roche Matters

Amgen Inc. v. F. Hoffmann-La Roche Ltd., et al.

On October 2, 2008, the Massachusetts District Court entered an Order making final its ruling denying the parties' post-trial motions and upholding the jury's verdict in all respects except infringement of claim 12 of the '933 patent under the Doctrine of Equivalents. The Order also stated the Massachusetts District Court's intention to enter a permanent injunction prohibiting F. Hoffman-La Roche Ltd., Roche Diagnostics GmbH and Hoffman-La Roche, Inc. (collectively, "Roche") from infringing the '422 patent, the '933 patent, the '868 patent and the '698 patent. The Massachusetts District Court then administratively closed the case awaiting resolution of the then pending appeal of the preliminary injunction. On October 6, 2008, Roche filed a Notice of Appeal to the U.S. Court of Appeals for the Federal Circuit (the "Federal Circuit") of the Massachusetts District Court's October 2nd Order.

On October 8, 2008, the Federal Circuit held a hearing on Roche's appeal of the preliminary injunction and on October 10, 2008 the Federal Circuit entered an order affirming the grant of the preliminary injunction. On October 17, 2008, the Massachusetts District Court entered judgment that the patents in suit are valid, enforceable, and infringed and permanently enjoined Roche from infringing the `422 patent, the '933 patent, the '868 patent and the '698 patent for the remaining life of these patents. Also on October 17, 2008, Roche filed a supplemental notice of appeal to the Federal Circuit. On October 30, 2008, Amgen filed two post-judgment motions with the Massachusetts District Court seeking entry of supplemental findings of fact and conclusions of law. On October 31, 2008, the Massachusetts District Court entered a notice that it would not be entering any supplemental findings of fact and conclusions of law. On November 3, 2008, the Massachusetts District Court entered an order terminating Amgen's post-judgment motions.

Amgen Inc., et al., v. Ariad Pharmaceuticals, Inc. ("Ariad")

On September 19, 2008, the U.S. District Court for the District of Delaware (the "Delaware District Court") issued an order construing the claims of the '516 patent and granted summary judgment that ENBREL does not infringe the '516 patent. Prior to the ruling, Ariad withdrew its claims of infringement against Kineret®. Also on September 19, 2008, the Delaware District Court granted summary judgment in-part in favor of Ariad, ruling that Amgen could not prove inequitable conduct on the basis of one of its claims, but that sufficient evidence exists for a trial on inequitable conduct on Amgen's alternative bases. The Delaware District Court also dismissed Amgen's claims of invalidity on the claims of the '516 patent not asserted by Ariad to be infringed by sales of ENBREL (Ariad had asserted that only seven of the 203 patent claims were infringed), but the Delaware District Court maintained Amgen's unenforceability claims to all 203 claims of the '516 patent.

AMGEN INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Continued)

On October 3, 2008, the Delaware District Court stayed Amgen's invalidity and unenforceability claims and entered final judgment of no infringement in favor of Amgen. The Delaware District Court declared the case administratively closed, to be reopened only by the parties after a decision on appeal. The Delaware District Court canceled the previously scheduled November trial date. On October 6, 2008, Ariad filed a notice of appeal to the Federal Circuit.

Human Genome Sciences ("HGS") Litigation

HGS' motion for reconsideration before the Delaware District Court was denied on August 21, 2008, and the Federal Circuit entered an order reactivating HGS' appeal effective October 22, 2008.

Sensipar® Abbreviated New Drug Application ("ANDA") Litigation

On October 31, 2008, the Delaware District Court entered a First Scheduling Order indicating that the case will be placed in the trial pool on May 3, 2010.

Federal Antitrust Litigation

The U.S. District Court for the District of New Jersey granted Amgen's motion to dismiss in part and denied in part, permitting plaintiffs to revise their complaint. Plaintiffs elected to voluntarily dismiss their complaint and a Stipulation of Voluntary Dismissal was filed on September 22, 2008.

In the ordinary course of business, we are involved in various legal proceedings and other matters, including those discussed above. While it is not possible to accurately predict or determine the eventual outcome of these items, one or more of these items currently pending could have a material adverse effect on our consolidated results of operations, financial position or cash flows.

Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Forward looking statements

This report and other documents we file with the SEC contain forward looking statements that are based on current expectations, estimates, forecasts and projections about us, our future performance, our business or others on our behalf, our beliefs and our management's assumptions. In addition, we, or others on our behalf, may make forward looking statements in press releases or written statements, or in our communications and discussions with investors and analysts in the normal course of business through meetings, webcasts, phone calls and conference calls. Words such as "expect," "anticipate," "outlook," "could," "target," "project," "intend," "plan," "believe," "seek," "estimate," "should," "may," "assume," "continue," variations of such words and similar expressions are intended to identify such forward looking statements. These statements are not guarantees of future performance and involve certain risks, uncertainties and assumptions that are difficult to predict. We describe our respective risks, uncertainties and assumptions that could affect the outcome or results of operations in "*Item 1A. Risk Factors.*" We have based our forward looking statements on our management's beliefs and assumptions based on information available to our management at the time the statements are made. We caution you that actual outcomes and results may differ materially from what is expressed, implied or forecast by our forward looking statements. Reference is made in particular to forward looking statements regarding product sales, regulatory activities, clinical trial results, reimbursement, expenses, EPS, liquidity and capital resources and trends. Except as required under the federal securities laws and the rules and regulations of the SEC, we do not have any intention or obligation to update publicly any forward looking statements after the distribution of this report, whether as a result of new information, future events, changes in assumptions or otherwise.

Overview

The following Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A") is intended to assist the reader in understanding Amgen's business. MD&A is provided as a supplement to, and should be read in conjunction with, our condensed consolidated financial statements and accompanying notes included in this Quarterly Report on Form 10-Q and our consolidated financial statements and accompanying notes included in our Annual Report on Form 10-K for the year ended December 31, 2007.

We are a global biotechnology company that discovers, develops, manufactures and markets human therapeutics based on advances in cellular and molecular biology. Our mission is to serve patients. As a science-based, patient-focused organization, we discover and develop innovative therapies to treat grievous illness. We operate in one business segment – human therapeutics. Therefore, our results of operations are discussed on a consolidated basis.

We primarily earn revenues and income and generate cash from sales of human therapeutic products in the areas of supportive cancer care, nephrology and inflammation. Our principal products include Aranesp®, EPOGEN®, Neulasta®, NEUPOGEN® and ENBREL, all of which are sold in the United States. Aranesp® and EPOGEN® stimulate the production of red blood cells to treat anemia and belong to a class of drugs referred to as erythropoiesis-stimulating agents, or ESAs. Aranesp® is used for the treatment of anemia both in supportive cancer care and in nephrology. EPOGEN® is used to treat anemia associated with chronic renal failure ("CRF"). Neulasta® and NEUPOGEN®, which are used in supportive cancer care, selectively stimulate the production of neutrophils, one type of white blood cell that helps the body fight infections. ENBREL is marketed under a co-promotion agreement with Wyeth in the United States and Canada. ENBREL blocks the biologic activity of tumor necrosis factor ("TNF") by inhibiting TNF, a substance induced in response to inflammatory and immunological responses, such as rheumatoid arthritis and psoriasis. For both the three and nine months ended September 30, 2008, our principal products represented 94% of total worldwide product sales. Our international product sales consist principally of European sales of Aranesp®, Neulasta® and NEUPOGEN®. International product sales represented approximately 23% and 22% of total product sales for the three and nine months ended September 30, 2008, respectively. For additional information about our principal products, their

approved indications and where they are marketed, see "*Item 1. Business – Principal products*" in Part I of our Annual Report on Form 10-K for the year ended December 31, 2007.

We operate in a highly regulated industry and various U.S. and foreign regulatory bodies have substantial authority over how we conduct our business. Government authorities in the United States and in other countries regulate the manufacturing and marketing of our products and our ongoing R&D activities. The regulatory environment is evolving and there is increased scrutiny on drug safety and increased authority being granted to regulatory bodies, in particular the U.S. Food and Drug Administration ("FDA"), to assist in ensuring the safety of therapeutic products. Most patients receiving our principal products for approved indications are covered by either government or private payer health care programs. The reimbursement environment is also evolving with greater emphasis on cost containment. Further, as a result of the new U.S. presidential administration's plans for changes in the healthcare system, we believe that we and others in our industry will be under increased pressure to further demonstrate the safety, efficacy and economic value of our products. Therefore, sales of our principal products are and will continue to be affected by the availability and extent of reimbursement from third-party payers, including government and private insurance plans and administration of those programs. Further, safety signals or trends or adverse events or results from clinical trials or studies or meta-analyses (a meta-analysis is the review of studies using various statistical methods to combine results from previous separate, but related, studies) performed by us or by others (including our licensees or independent investigators) or from the marketed use of our products may expand safety labeling, restrict the use of our approved products or may result in additional regulatory requirements, such as requiring risk management activities, including a risk evaluation and mitigation strategy ("REMS"), and/or additional or more extensive clinical trials as part of postmarketing commitments ("PMCs") or a pharmacovigilance program, and may negatively impact worldwide sales or reimbursement of our products. In addition, the capital and credit markets have been experiencing extreme volatility and disruption, particularly in the past several weeks. We are working to manage our business effectively despite the unprecedented conditions in the financial markets both in the United States and around the world. However, the extent and/or the duration of any potential adverse economic impact that such financial disruption may have on our third party payers, customers, suppliers and service providers is unclear.

Total product sales for the three and nine months ended September 30, 2008 increased 7% and 3% as compared to the prior year comparative periods, respectively. Product sales in the United States for the three and nine months ended September 30, 2008 totaled \$2.9 billion and \$8.6 billion, respectively, representing an increase of 4% for the three month period and remaining essentially unchanged for the nine month period compared to the prior year comparative periods. International product sales for the three and nine months ended September 30, 2008 totaled \$855 million and \$2.5 billion, respectively, reflecting increases of 20% and 16% over the prior year comparative periods. International product sales for the three and nine months ended September 30, 2008 totaled \$855 million and \$2.5 billion, respectively, reflecting foreign currency exchange rate changes of \$78 million and \$243 million, respectively. Excluding the impact of foreign currency exchange rate changes for the three and nine months ended September 30, 2008, worldwide product sales increased 5% and 1%, respectively, and international product sales increased 9% and 4%, respectively.

Certain of our products, principally our marketed ESA products, have faced and will continue to face various challenges resulting from regulatory and reimbursement developments. Late in 2006 and throughout 2007, adverse safety results involving ESA products were observed in various studies that were performed by us and by others (including our licensees or independent investigators) that explored the use of ESAs in settings different from those outlined in the FDA approved label, including targeting higher hemoglobin ("Hb") levels and/or use in non-approved patient populations. The results of these studies culminated in significant regulatory and reimbursement developments affecting the class of ESA products, including Aranesp® and EPOGEN®. For example, in February 2007, following the reported results from our Anemia of Cancer phase 3 study (the "AoC 103 study"), the United States Pharmacopoeia Dispensing Information ("USP DI") Drug Reference Guides removed Aranesp® in the treatment of AoC. Thereafter, Aranesp® use in AoC essentially ceased. In addition, during 2007, we had discussions with the FDA and other regulatory authorities and meetings with certain of the FDA's advisory panels, namely the Oncologic Drugs Advisory Committee ("ODAC"), the Cardiovascular-Renal Drug Advisory Committee ("CRDAC") and the Drug Safety and Risk

Management Advisory Committee ("DSaRMAC"), regarding the administration of our ESA products in certain settings. These adverse safety results involving ESA products in various studies and related discussions with regulatory authorities led to several key regulatory and reimbursement developments, including safety-related revisions to ESA product labels in the United States in March and November 2007. Further, in July 2007, the Centers for Medicare and Medicaid Services ("CMS") issued its National Coverage Decision Memorandum for Use of Erythropoiesis Stimulating Agents in Cancer and Related Neoplastic Conditions (the "Decision Memorandum"). The Decision Memorandum established the ESA reimbursement policy for Medicare and other government beneficiaries who are treated for chemotherapy-induced anemia ("CIA") with ESAs. We believe that the restrictions in the Decision Memorandum changed the way ESAs are used in clinical practice, for example, by decreasing the number of treated patients, the average ESA dose and the duration of ESA therapy. These developments have had a material adverse impact on sales of our marketed ESA products, in particular Aranesp® sales in the U.S. supportive cancer care setting. Furthermore, our ESA products will continue to face future challenges, including those described below under "*ESA Developments*" and also the potential for further revisions to product labels and changes to reimbursement.

As a result of the challenges facing certain of our products and, in particular, the regulatory and reimbursement developments involving our marketed ESA products that began in 2007 and their resulting impact on our operations, on August 15, 2007, we announced a plan to restructure our worldwide operations in order to improve our cost structure while continuing to make significant R&D investments and build the framework for our future growth. Key components of our restructuring plan initially included: (i) worldwide staff reductions aggregating approximately 2,500 positions, (ii) rationalization of our worldwide network of manufacturing facilities in order to gain cost efficiencies while continuing to meet future commercial and clinical demand for our products and product candidates and, to a lesser degree, changes to certain R&D capital projects and (iii) abandoning leases primarily for certain R&D facilities that will not be used in our operations. Through September 30, 2008, we have completed a majority of these actions and incurred \$790 million of costs. We have recently identified certain additional initiatives designed to further assist in improving our cost structure, including outsourcing certain non-core business functions, most notably certain of our information systems infrastructure services, as well as abandoning leases for certain additional facilities that will no longer be used in our operations. The estimated cost of these additional initiatives is \$50 million to \$100 million. As a result of the actual costs incurred to date and the addition of the recently identified initiatives, the total charges expected to be incurred in connection with our restructuring plan, including related implementation costs, has been increased to \$850 million to \$925 million, as compared to our prior estimate of \$775 million to \$825 million. We currently estimate that all remaining costs will be incurred through 2009. Such cost estimates and amounts incurred are net of amounts recovered from our ENBREL co-promotion

The following is a discussion of select key developments affecting our business that occurred in 2008 and should be read in conjunction with "*Item 2*. *Management's Discussion and Analysis of Financial Condition and Results of Operations*" in Part I of our Form 10-Qs for the quarterly periods ended March 31, 2008 and June 30, 2008 and "*Item 1*. *Business – Key Developments*" in Part I of our Annual Report on Form 10-K for the year ended December 31, 2007.

ESA Developments

• On August 6, 2008, we finalized changes to the ESA product labeling based on a complete response letter, received on July 30, 2008, from the FDA to the revisions to the ESA labeling we proposed following the March 13, 2008 ODAC meeting. The revised labeling included, among other things, (i) the addition to the boxed warning of a statement that ESAs are not indicated for patients receiving myelosuppressive therapy when the anticipated outcome of such therapy is cure, (ii) the addition of a statement in the DOSAGE and ADMINISTRATION section of the label that ESA therapy should not be initiated at Hb levels ³ 10 grams per deciliter ("g/dL") and that dose should be adjusted to maintain the lowest Hb level sufficient to avoid red blood cell transfusions and (iii) the removal of reference to the upper safety limit of 12 g/dL. Additionally, in response to the FDA's request under authority prescribed by the Food and Drug Administration Amendments Act of 2007 (the "FDAAA"), we have recently submitted a proposed REMS program for Aranesp® in oncology. We

also continue to work with the FDA to finalize protocols for clinical trials to determine the effects of Aranesp® on survival and tumor outcomes.

- On January 1, 2008, the CMS' revisions to its Claims Monitoring Policy: Erythropoietin/darbepoetin alfa usage for beneficiaries with end stage renal disease ("EMP") became effective, which require a 50% reduction in Medicare reimbursement if a patient's Hb is above 13 g/dL for three or more consecutive months. In addition, the EMP reduces the monthly dosing limits to 400,000 international units ("IUs") of EPOGEN®, from 500,000 IUs, and to 1,200 micrograms ("mcgs") of Aranesp®, from 1,500 mcgs. We believe that the EMP implementation in January 2008 has significantly affected physician behavior resulting in declines in dosing trends as particularly noted in the quarter of implementation. However, this dose decline subsequently stabilized but may further fluctuate in the future.
- On March 5, 2008, we announced that the European Commission reached its decision to amend the product labeling for the class of ESAs, including Aranesp®. On May 15, 2008, we and other ESA marketing authorization holders participated in a closed meeting of the Scientific Advisory Group on Oncology ("SAG-O"). The marketing authorization holders were asked to provide an overview on studies that have been initiated or conducted since July 2007, as well as any other new data that can help to elucidate recent issues on the impact of ESAs on tumor progression and survival in cancer patients. These data included previously disclosed interim results from the Preoperative Epirubicin Paclitaxel Aranesp® ("PREPARE") study in neo-adjuvant breast cancer therapy; follow-up data from the Gynecologic Oncology Group study ("GOG-191 study") in cervical cancer, which were published in the February 2008 issue of Gynecologic Oncology; and the February 2008 meta-analysis by Bennett et al, which was published in the Journal of the American Medical Association. Scientific Advisory Groups ("SAGs") are established by the European Medicines Agency ("EMEA") to deliver answers, on a consultative basis, to specific questions addressed to them by the Committee for Medicinal Products for Human Use ("CHMP"). On June 26, 2008 the EMEA, based upon the CHMP's opinion which took into account the position expressed by the SAG-O, recommended updating the product information for ESAs with a new warning for their use in cancer patients. In July 2008, the EMEA requested that further clarity around the product information be provided by regulatory agencies in each European Member State country through the publication of a Dear Healthcare Professional Communication, following which we followed the necessary regulatory procedure to update the Aranesp® product information. In October 2008, we received notification that the Aranesp® product information update was approved by the European Commission. The product information for all ESAs was updated to advise that in some clinical situations blood transfusions should be the preferred treatment for the management of anemia in patients with cancer and that the decision to administer ESAs should be based on the benefit-risk assessment with the participation of the individual patient. This assessment should take into account the specific clinical context, including the type of tumor and its stage, the degree of anemia, life-expectancy, the environment in which the patient is being treated and patient preference.
 - On September 30, 2008, we announced that we had received a summary of preliminary results from the Cochrane Collaboration's independent meta-analysis of patient-level data from previously conducted, randomized, controlled, clinical studies evaluating ESAs in cancer patients, which we submitted to the FDA and EMEA. The preliminary summary includes four components: on-study deaths and overall survival in cancer patients regardless of their specific cancer treatment (chemotherapy, radiochemotherapy, radiotherapy, anemia of cancer with no treatment, other), and on-study deaths and overall survival in patients receiving chemotherapy (the only oncologic population for which ESA treatment is indicated in current FDA-approved labeling). The analysis showed that ESAs increased on-study deaths and decreased overall survival compared to controls. Although neither of these results were statistically significant, they do not exclude the potential for adverse outcomes when ESAs are used according to the current labeling. We expect to receive the complete meta-analysis results later this year and will provide this information to regulatory authorities at that time.

Other Regulatory Developments

- On June 4, 2008, the FDA issued an Early Communication regarding the ongoing safety review of TNF blockers and the possible association between the use of these medicines and the development of lymphoma and other cancers in children and young adults and stated that it had decided to conduct further analyses to evaluate the risk and benefits of TNF blockers in pediatric patients.
- Following the June 18, 2008 Dermatologic and Ophthalmic Drugs Advisory Committee ("DODAC") meeting, on July 24, 2008, the FDA requested
 additional information from us to support the supplemental biologic license application ("BLA") we submitted for the use of ENBREL in pediatric
 patients with chronic moderate to severe plaque psoriasis. We continue to work with the FDA to respond to its requests and provide it with
 information to address additional questions related to the supplemental BLA and plans for risk management activities.
- On September 4, 2008, the FDA issued a Web-Alert regarding their review of histoplasmosis and other opportunistic fungal infections in patients treated with TNF-blockers. The FDA requested that the boxed warning and warnings sections of the U.S. prescribing information ("PI") and the medication guide for ENBREL (and other TNF blockers) be strengthened to include the risk of histoplasmosis and other invasive fungal infections with the goal of increasing timely diagnosis and treatment. The FDA also requested that the approved REMS for ENBREL be modified with a communication plan to healthcare providers regarding the risk of unrecognized fungal infections. We are working with the FDA to finalize the required revisions to respond to its requests.
- On August 22, 2008, the FDA approved NplateTM (romiplostim), the first and only platelet producer for the treatment of thrombocytopenia in splenectomized (spleen removed) and non-splenectomized adults with chronic immune thrombocytopenic purpura ("ITP"). NplateTM, the first FDA approved peptibody protein, works by raising and sustaining platelet counts, representing a novel approach for the long-term treatment of this chronic disease. As part of the approval for NplateTM, a REMS was developed with the FDA to assure the safe use of NplateTM while minimizing risk. The NplateTM REMS involves, among other things, healthcare provider and patient enrollment registries, tracking of patient medical history and data and follow-up safety questionnaires to healthcare providers all of which require extensive discussion and education with healthcare providers.

Clinical Developments

Denosumab Osteoporosis

On July 25, 2008, we announced findings from the pivotal fracture trial ("Study 216") evaluating our receptor activator of nuclear factor kappa B ("RANK") ligand inhibitor, denosumab, in the treatment of postmenopausal osteoporosis. In this pivotal, three-year, international, phase 3 study of approximately 7,800 women with osteoporosis, patients were randomized to receive either denosumab, given by subcutaneous injection once every six months, or placebo injections. For the primary endpoint, treatment with denosumab resulted in a statistically significant reduction in the incidence of new vertebral fractures compared with placebo treatment. In addition, women receiving denosumab experienced a statistically significant reduction in the incidence of new non-vertebral and hip fractures (each a secondary endpoint) compared with those receiving placebo. The incidence and types of both adverse and serious adverse events observed in this study, including serious infections and neoplasms, were similar between the denosumab and placebo groups. The most common adverse events across both treatment arms were arthralgia, back pain, hypertension and nasopharyngitis.

On September 16, 2008 at the 2008 American Society of Bone and Mineral Research ("ASBMR") annual meeting in Montreal, Canada, we presented detailed results of Study 216 evaluating

denosumab in osteoporosis. Compared to placebo, patients who received denosumab had 68% fewer vertebral fractures (2.3% for denosumab versus 7.2% for placebo, p=0.0001), 40% fewer hip fractures (0.7% versus 1.2%, p=0.036) and 20% fewer non-vertebral fractures (6.5% versus 8.0%, p=0.011). At the meeting we also presented data regarding the incidence and types of adverse events and serious adverse events observed in the trial, which were similar between the placebo and denosumab groups.

In addition to the detailed results of Study 216, a pivotal fracture study, we presented the results of two non-pivotal phase 3 studies of denosumab in osteoporosis at the ASBMR. The first was a phase 3 head-to-head, double-blind trial known as the STAND (Study of Transitioning from AleNdronate to Denosumab) trial ("Study 234"). The results of this study demonstrated that subcutaneous injections of denosumab every six months achieved significantly greater increases in bone mineral density ("BMD") versus those achieved with alendronate at all sites measured. For the primary endpoint, denosumab resulted in significant increases in BMD at the total hip compared with alendronate (1.9% versus 1.05%, p<0.0001). Treatment with denosumab also resulted in significant increases in BMD compared with continued alendronate treatment at all secondary endpoints including the lumbar spine, femoral neck, hip trochanter and 1/3 radius. The incidence and types of adverse events observed in the study, including neoplasm and infection, were similar between the denosumab and alendronate treatment groups. The most common adverse events across both treatment arms were back pain, arthralgia and nasal pharyngitis.

The second non-pivotal study was a head-to-head trial comparing denosumab to weekly oral alendronate, also known as the DECIDE (Determining Efficacy: Comparison of Initiating Denosumab versus Alendronate) trial ("Study 141"). As a part of this study, patients were given a questionnaire after 12 months of treatment to gauge preference on mode of administration as well as satisfaction with frequency of dosing of twice-yearly subcutaneous injections versus weekly oral tablet. More than three-quarters of patients in both study arms preferred subcutaneous injection over oral pills (77% versus 23%, p <0.0001). In addition, significantly more patients were more satisfied with twice-yearly dosing compared to weekly dosing (80% versus 20% placebo injection versus weekly oral alendronate, and 79% versus 21% denosumab versus weekly placebo tablet, p <0.0001 for both study groups).

Denosumab Oncology

On July 14, 2008, we announced findings from a three-year pivotal phase 3 placebo-controlled trial evaluating denosumab in the treatment of bone
loss in men undergoing androgen deprivation therapy ("ADT") for non-metastatic prostate cancer. In this study of more than 1,400 men,
denosumab treatment produced statistically significantly greater increases in BMD at the lumbar spine (primary endpoint) and non-vertebral sites
compared with placebo at multiple time points. These improvements in BMD were consistent with those seen in other denosumab studies
evaluating BMD in women with breast cancer receiving aromatase inhibitor therapy, and in postmenopausal women with low bone mass. During
the thirty-six month evaluation period, men receiving denosumab experienced less than half the incidence of new vertebral fractures (a secondary
endpoint) compared with those receiving placebo, a statistically significant finding. Furthermore, in the denosumab arm there were fewer nonvertebral fractures over the thirty-six month period.

Other

• During the three months ended September 30, 2008, we received interim results from a phase 2 study of AMG 317 in patients with moderate to severe asthma. The data showed evidence of biological activity; however, the clinical efficacy from the interim analysis did not meet our expectations. The phase 2 study will be completed this year and results will be submitted to an appropriate peer-reviewed forum.

Competition

- On October 17, 2008, the Massachusetts District Court entered judgment that the patents in suit are valid and enforceable, and would be infringed by the import, use and sale of Roche's pegylated-erythropoietin product. The Massachusetts District Court permanently enjoined Roche from infringing the '422 patent, the '933 patent, the '868 patent and the '698 patent for the remaining life of these patents.
- On September 15, 2008, Ratiopharm's Ratiograstim®/Filgrastim Ratiopharm®, CT Arzneimittel's Biograstim® and Teva Generics ("Teva") Tevagrastim®, all G-CSF biosimilar products, received marketing authorization from the European Commission. Ratiopharm launched its G-CSF biosimilar product, Ratiograstim®, in the United Kingdom in October 2008, and is expected to launch it in Germany and several other European markets in the fourth quarter of 2008. CT Arzneimittel is expected to market its G-CSF biosimilar product in Germany in the fourth quarter of 2008. Teva stated that it would begin marketing its G-CSF biosimilar product throughout Europe in 2009.

Other Developments

On September 15, 2008, we announced that we have entered into an agreement with Biovitrum AB ("Biovitrum") under which Biovitrum will acquire from us the marketed biologic therapeutic products Kepivance® (palifermin) and Stemgen® (ancestim), and will also obtain from us a worldwide exclusive license to Kineret® (anakinra) for its current approved indication. In connection with entering into this agreement, we recorded a \$9 million loss accrual on the disposal in the three months ended September 30, 2008. This amount is included in "Interest and other income and (expense), net" in the Condensed Consolidated Statements of Income.

For the three and nine months ended September 30, 2008, net income was \$1.2 billion and \$3.2 billion, respectively, and diluted earnings per share was \$1.09 per share and \$3.00 per share, respectively. As of September 30, 2008, cash, cash equivalents and marketable securities were \$9.8 billion, of which approximately \$7.7 billion was generated from operations in foreign tax jurisdictions and is intended for use in our foreign operations. If these funds were repatriated for use in our U.S. operations, we would be required to pay additional U.S. federal and state income taxes at the applicable marginal tax rates. Our total debt outstanding was \$11.2 billion as of September 30, 2008, of which \$1.0 billion is scheduled to mature on November 28, 2008.

There are also many economic and industry-wide factors that affect our business generally and uniquely, including, among others, those relating to increased complexity and cost of R&D due, in part, to greater scrutiny of clinical trials with respect to safety which may lead to fewer treatments being approved by the FDA or other regulatory bodies and/or safety-related label changes for approved products; increasingly intense competition for marketed products and product candidates; reimbursement changes; healthcare provider prescribing behavior, regulatory or private healthcare organization medical guidelines and reimbursement practices; complex and expanding regulatory requirements; and intellectual property protection. See *"Item 1. Business"* in Part I of our Annual Report on Form 10-K for the year ended December 31, 2007 and *"Item 1A. Risk Factors"* in Part II herein for further information on these economic and industry-wide factors and their impact and potential impact on our business.

Reimbursement

Sales of all of our principal products are dependent, in part, on the availability and extent of reimbursement from third-party payers, including governments and private insurance plans. Generally, in Europe and other countries outside the United States, the government sponsored healthcare system is the primary payer of healthcare costs of patients. Governments may regulate access to, prices or reimbursement levels of our products to control costs or to affect levels of use of our products. Worldwide use of our products

may be affected by these cost containment pressures and cost shifting from governments and private insurers to healthcare providers or patients in response to ongoing initiatives to reduce or reallocate healthcare expenditures. Further, adverse events or results from clinical trials or studies performed by us or by others or from the marketed use of our drugs may expand the safety information in the labeling for our approved products and may negatively impact worldwide reimbursement for our products. On July 30, 2007, CMS issued its Decision Memorandum and on January 14, 2008, issued changes to its Medicare National Coverage Determinations Manual, effective for claims with dates of service on or after July 30, 2007, with an implementation date of April 7, 2008. A discussion of the Decision Memorandum follows below. (See also "*Item 1A. Risk Factors — Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market." and "— Guidelines and recommendations published by various organizations can reduce the use of our products.")*

Most patients receiving Aranesp®, Neulasta® and NEUPOGEN[®] for approved indications are covered by government and/or private payer healthcare programs. Medicare and Medicaid government healthcare programs' payment policies for drugs and biologicals are subject to various laws and regulations. Beginning in January 1, 2005 under the Medicare Prescription Drug Improvement and Modernization Act (the "MMA"), in the physician clinic setting and January 1, 2006, in the hospital outpatient and dialysis settings, Aranesp®, Neulasta® and NEUPOGEN® have been reimbursed under a Medicare Part B payment methodology that reimburses each product at 106% of its average sales price ("ASP") (sometimes referred to as "ASP+6%"). Effective January 1, 2008, Medicare payment in the hospital outpatient setting reimburses each product at 105% of its ASP and CMS has the regulatory authority to further reduce the outpatient hospital payment formula in future years. For example, effective January 1, 2009, CMS, in its Outpatient Prospective Payment System Final Rule for 2009, released on October 30, 2008, set the payment rate in the hospital outpatient setting at ASP+4% for 2009. ASP is calculated by the manufacturer based on a statutorily defined formula and submitted to CMS. A product's ASP is calculated and reported to CMS on a quarterly basis and therefore may change each quarter. The ASP in effect for a given quarter (the "Current Period") is based upon certain historical sales and sales incentive data covering a statutorily defined period of time preceding the Current Period. For example, the ASP based payment rate for Aranesp® that will be in effect for the fourth quarter of 2008 will be based in part on certain historical sales and sales incentive data for Aranesp® from July 1, 2007 through June 30, 2008. CMS publishes the ASPs for products in advance of the quarter in which they go into effect.

In the United States, dialysis providers are primarily reimbursed for EPOGEN® by the federal government through the End Stage Renal Disease ("ESRD") Program of Medicare. The ESRD Program reimburses approved providers for 80% of allowed dialysis costs; the remainder is paid by other sources, including patients, state Medicaid programs, private insurance, and to a lesser extent, state kidney patient programs. The ESRD Program reimbursement methodology is established by federal law and is monitored and implemented by CMS. Effective January 1, 2006, the payment mechanism for separately reimbursed dialysis drugs in both free-standing and hospital-based dialysis centers, including EPOGEN® and Aranesp®, is reimbursed by Medicare at ASP+6% using the same payment amounts used in the physician clinic setting. Beginning in the third quarter of 2007, based on its ongoing assessment for payment of Part B drugs, CMS instituted a single payment limit for Epoetin alfa (EPOGEN® and PROCRIT®) in all provider settings. Any changes to the ASP calculations directly affect the Medicare reimbursement for our products administered in the physician clinic setting, dialysis facility and hospital outpatient setting. These calculations are regularly reviewed for completeness and based on such review, we have revised our reported ASPs to reflect calculation changes both prospectively and retroactively. For example, partially as a result of our methodology changes, our ASP reimbursement rate for EPOGEN® was reduced for the third quarter of 2007.

Since April 1, 2006, the Medicare reimbursement for ESAs administered to dialysis patients has been subject to a revised EMP, the Medicare payment review mechanism used by CMS to monitor EPOGEN® and Aranesp® utilization and appropriate hematocrit outcomes of dialysis patients. The EMP was revised, effective January 1, 2008, requiring a 50% reduction in Medicare reimbursement if a patient's Hb is above 13 g/dL for three or more consecutive months. In addition, the revised EMP reduces the monthly dosing limits to 400,000

IUs of EPOGEN®, from 500,000 IUs, and to 1,200 mcgs of Aranesp®, from 1,500 mcgs. The implementation of the revised EMP and ESA labeling changes led to a decline in EPOGEN® sales for the first quarter of 2008 compared to the first quarter of 2007 primarily due to a decline in both overall utilization and as well as average dosing per patient. However, this dose decline subsequently stabilized but may further fluctuate in the future.

Changes resulting from the MMA, which beginning in 2005 lowered reimbursement for our products, could negatively affect product sales of some of our marketed products. However, we believe that our product sales for 2005, 2006, 2007 and for the first three quarters of 2008 were not significantly impacted by the reimbursement changes resulting from the MMA. However, additional provisions of the MMA and other regulations or legislation affecting reimbursement that have gone or may go into effect could affect our product sales in the future. For example, on July 15, 2008, the Medicare Improvements for Patients and Providers Act of 2008 became law with a number of Medicare and Medicaid reforms including a broader payment bundle for dialysis services and drugs which will require CMS, beginning in 2011, to establish a bundled Medicare payment rate that includes dialysis services and drug/labs that are currently separately billed. The new bundled rate will include dialysis services oured under the current composite rate, all ESAs and other intravenous injectable drugs and oral equivalent forms used in dialysis. The bundled reimbursement rate will be phased in over a four-year period in equal increments starting in 2011. It is possible that providers could elect to move to a full Medicare bundled payment in 2011. CMS will also be required to establish a quality incentive program that begins concurrently with bundling in 2011 which subjects facilities to up to a 2% annual reduction in Medicare reimbursement for failure to meet or exceed CMS quality performance standards, which include anemia management and dialysis adequacy. Bundling initiatives that have been implemented in other healthcare settings have occasionally resulted in lower utilization of services that had not previously been a part of the bundled payment. We are in the process of evaluating the new Medicare legislation on our business and cannot predict the full impact a bundled payments system would have on sales of EPOGEN® or Aranesp® used in the treatment

In addition, in response to CMS considering and rejecting changes to the ASP calculation methodology for accounting for discounts in multi-product contracts in the 2007 Medicare Physician Fee Schedule Final Rule, MedPAC released its second Congressionally-mandated report on December 29, 2006 on the impact of changes in Medicare payments for Part B Drugs specifically recommending that the Secretary of the Department of Health and Human Services clarify ASP reporting requirements "to ensure that ASP calculations allocate discounts to reflect the transaction price for each drug." Under the ASP system, we allocate our discounts based on the prices paid for individual drugs, according to the terms of its contracts with physicians and other purchasers, and we believe that the resulting ASPs reflect the transaction prices for individual drugs. Referencing a MedPAC December 2006 report, CMS proposed in the Medicare Physician Fee Schedule Proposed Rule for 2008 revising the methodology for calculating ASP to require the reallocation of price concessions of drugs sold under "bundled arrangements," described by CMS in part as an arrangement regardless of physical packaging under which the rebate, discount or other price concession is conditioned upon the purchase of the same drug or biological or other drugs or biologicals or some other performance requirement. In the Medicare Physician Fee Schedule Final Rule for 2008, CMS stated that it was not finalizing the proposed regulatory change at this time, based on comments recommending a delay and raising concerns about the proposal. The agency also clarified that in the absence of specific guidance, manufacturers may continue to make "reasonable assumptions" in the calculation of ASP, consistent with the general requirements and the intent of the Medicare statute and regulations and their customary business practices. The agency stated that it will continue to monitor this issue and may provide more specific guidance in the future. In the Medicare Physician Fee Schedule Fina

Other initiatives reviewing the coverage or reimbursement of our products, including those related to safety, could result in less extensive coverage or lower reimbursement and could negatively affect sales of some of our marketed products. For example, on March 14, 2007, shortly after the March 9, 2007 label changes for all ESAs, CMS announced that the agency had begun reviewing all Medicare policies related to the administration of ESAs in non-renal disease applications as part of a national coverage analysis ("NCA") which is generally CMS' first step toward developing a national coverage determination ("NCD"). Generally, a NCD is a national

policy statement granting, limiting or excluding Medicare coverage or reimbursement for a specific medical item or service. On July 30, 2007, CMS issued its Decision Memorandum which was substantially altered from the proposed NCD. On January 14, 2008, CMS issued changes to its Medicare NCD Manual, adding the ESA Decision Memorandum, effective for claims with dates of service on and after July 30, 2007 with an implementation date of April 7, 2008. In the Decision Memorandum, CMS determined that ESA treatment was not reasonable and necessary for certain clinical conditions. The Decision Memorandum established the ESA reimbursement policy for Medicare and other government beneficiaries who are treated for CIA with ESAs. We believe that the restrictions in the Decision Memorandum changed the way ESAs are used in clinical practice, for example, by decreasing the number of treated patients, the average ESA dose and the duration of ESA therapy.

We believe this restriction on reimbursement of ESAs in the Decision Memorandum has had a material adverse effect on the use, reimbursement and sales of Aranesp[®], and our business and results of operations. Additionally, based on our knowledge, although no private payers have implemented the Decision Memorandum to date, many private payers have implemented the restrictions included in the Decision Memorandum. Further, we believe many healthcare providers have reduced ESA utilization for all of their patients regardless of insurance coverage.

In addition, the FDA held a joint meeting of the CRDAC and the DSaRMAC on September 11, 2007, which evaluated the safety data on ESA use in renal disease. On July 31, 2008, CMS issued a listing of potential topics for future NCDs as a step to increase transparency in the NCD process and which included as potential topics the use of ESAs in ESRD and chronic kidney disease ("CKD"). CMS has not announced whether it will proceed to a NCD for ESAs in ESRD or CKD and we cannot predict whether ESAs in the renal setting will be the subject of a future NCD, however, any final NCD for ESAs in the renal setting, which may include non-coverage and/or new dosing and treatment restrictions similar to those proposed in Decision Memorandum for treatment of anemia in oncology with ESAs, would negatively affect use, reduce reimbursement and coverage, negatively affect product sales of our ESA products and may have a material adverse effect on our business and results of operations. In addition, on August 22, 2008 our platelet producer for the treatment of thrombocytopenia in splenectomized (spleen removed) and non-splenectomized adults with chronic ITP, NplateTM, was approved by the FDA and falls within the thrombopoiesis stimulating agents (platelet growth factors) topic that was also included on CMS' July 31, 2008 potential future NCD topic list. We cannot predict whether NplateTM will be the subject of a future NCD.

Further, the Deficit Reduction Act of 2005 ("DRA") included provisions, which are phased in over time, regarding state collection and submission of data for the purpose of collecting Medicaid drug rebates from manufacturers for physician-administered drugs. We expect that state compliance with elements of these provisions that became effective on January 1, 2006, has increased the level of Medicaid rebates paid by us.

Results of Operations

Product sales

For the three and nine months ended September 30, 2008 and 2007, worldwide product sales and total product sales by geographic region were as follows (dollar amounts in millions):

	Three months ended September 30,					Nine months ended September 30,				
	2008		2008 20		2007		Change	2008	2007	Change
Aranesp®	\$ 845		\$	818	3%	\$ 2,431	\$ 2,787	(13)%		
EPOGEN®		634		602	5%	1,810	1,851	(2)%		
Neulasta®/NEUPOGEN®		1,192		1,100	8%	3,479	3,159	10%		
ENBREL		893		821	9%	2,685	2,374	13%		
Sensipar®		161		122	32%	444	335	33%		
Vectibix®		41		41	0%	107	137	(22)%		
Other		18		20	(10)%	57	50	14%		
Total product sales	\$	3,784	\$	3,524	7%	\$ 11,013	\$ 10,693	3%		
Total U.S.	\$	2,929	\$	2,809	4%	\$ 8,560	\$ 8,572	0%		
Total International		855		715	20%	2,453	2,121	16%		
Total product sales	\$	3,784	\$	3,524	7%	\$ 11,013	\$ 10,693	3%		

Product sales are influenced by a number of factors, including demand, third-party reimbursement availability and policies, government programs, regulatory developments or guidelines, clinical trial outcomes, clinical practice, contracting and pricing strategies, wholesaler and end-user inventory management practices, patient population growth, fluctuations in foreign currency exchange rates, new product launches and indications, competitive products, product supply and acquisitions.

Total product sales for the three and nine months ended September 30, 2008 increased 7% and 3% as compared to the prior year comparative periods, respectively. Product sales in the United States for the three and nine months ended September 30, 2008 totaled \$2.9 billion and \$8.6 billion, respectively, representing an increase of 4% for the three month period and remaining essentially unchanged for the nine month period compared to the prior year comparative periods. International product sales for the three and nine months ended September 30, 2008 totaled \$855 million and \$2.5 billion, respectively, reflecting increases of 20% and 16% over the prior year comparative periods. International product sales for the three and nine months ended September 30, 2008 totaled \$855 million and \$2.5 billion, respectively, reflecting foreign currency exchange rate changes of \$78 million and \$243 million, respectively. Excluding the impact of foreign currency exchange rate changes for the three and nine months ended September 30, 2008, worldwide product sales increased 5% and 1%, respectively, and international product sales increased 9% and 4%, respectively.

Aranesp®

For the three and nine months ended September 30, 2008 and 2007, total Aranesp® sales by geographic region were as follows (dollar amounts in millions):

		e months en ptember 30		Nine months ended September 30,				
	2008 2007		Change	2008	2007	Change		
Aranesp® - U.S.	\$ 458	\$ 460	0%	\$ 1,290	\$ 1,692	(24)%		
Aranesp® - International	387	358	8%	1,141	1,095	4%		
Total Aranesp®	\$ 845	\$ 818	3%	\$ 2,431	\$ 2,787	(13)%		

U.S. sales of Aranesp® in the three months ended September 30, 2008 benefited from a \$54 million change in the accounting estimate related to its product sales returns reserve. Excluding the positive impact of the change in accounting estimate, U.S. sales of Aransep® decreased 12% compared to the three months ended September 30, 2007. This decrease in U.S. Aranesp® sales reflects the negative impact on demand, primarily in the supportive cancer care setting, of physician conformance to regulatory and reimbursement changes which principally occurred in the second half of 2007 and additional product label changes which occurred on August 6, 2008, as well as a slight decline in share. The decline in demand in the three months ended September 30, 2008 was partially offset by an increase in the average net sales price. The regulatory and reimbursement developments include in particular, (i) the CMS' Decision Memorandum issued in July 2007, which significantly restricted Medicare reimbursement for use of Aranesp® in CIA and which we believe has also negatively impacted Aranesp® use in CIA for patients covered by private insurance plans, (ii) the loss of Aranesp® for use in the treatment of AoC in 2007 and (iii) the August 6, 2008, March 7, 2008, November 8, 2007 and March 9, 2007 product safety-related label changes in the United States. During the latter part of the three months ended December 31, 2007 and during the nine months ended September 30, 2008, underlying Aranesp® demand remained relatively stable, but in September 2008 we began to see a slight reduction in weekly sales which we believe is related to the early effect of the August 6, 2008 label revision on utilization patterns. We believe that Aranesp® sales in the United States, primarily in the supportive cancer care setting, will likely further decline as a result of the August 6, 2008 label revision, our new contracts entered into with our customers and the proposed REMS that we recently submitted to the FDA.

The decrease in U.S. Aranesp® sales for the nine months ended September 30, 2008, reflects the negative impact of the above noted regulatory and reimbursement developments as well as a slight decline in share, partially offset by an increase in the average net sales price and the positive impact of changes in the estimate of its product sales returns reserve.

The increase in international Aranesp® sales for the three and nine months ended September 30, 2008 is due to changes in foreign currency exchange rates, which positively impacted sales by approximately \$35 million and \$116 million, respectively. Excluding the impact of foreign currency exchange rate changes, international Aranesp® sales for the three and nine month periods decreased 2% and 6%, respectively, reflecting the impact of pricing pressures and ESA dosing conservatism. Through September 30, 2008, biosimilars and other recently introduced marketed products in Europe have not had a significant impact on total international Aranesp® segment share.

In addition to the factors mentioned in the "Product sales" section above, future worldwide Aranesp® sales will be dependent, in part, on such factors

as:

- regulatory developments, including those resulting from:
 - o ESA product labeling changes in the United States on August 6, 2008, as the FDA directed;
 - o risk management activities, including a REMS, undertaken by us or required by the FDA or other regulatory authorities;
 - o product labeling changes occurring on March 5, 2008 in Europe for the class of ESAs, including Aranesp®, by the European Commission and the potential for further changes resulting from the EMEA's recommendation that the ESA product information be updated with a new warning for their use in cancer patients;
 - o future product label changes;
- reimbursement developments, including those resulting from:
 - o government's and/or third-party payer's reaction to recent or future product label changes;
 - o current or future cost containment pressures by third-party payers, including governments and private insurance plans;
- adverse events or results from clinical trials or studies or meta-analyses performed by us, including our pharmacovigilance clinical trials, or by
 others (including our licensees or independent investigators), which have and could further impact product safety labeling, negatively impact
 healthcare provider prescribing behavior, use of our product, regulatory or private healthcare organization medical guidelines and reimbursement
 practices;
- governmental or private organization regulations or guidelines relating to the use of our product;
- our ability to maintain worldwide segment share and differentiate Aranesp® from current and potential future competitive products, including J&J's Epoetin alfa product marketed in the United States and certain other locations outside of the United States and other competitors' products outside of the United States, including biosimilar products that have been or are expected to be launched in the future;
- our current and future contracting and related pricing strategies;
- patient population growth; and
- development of new treatments for cancer and future chemotherapy treatments. For example, targeted therapies and other treatments that are less myelosuppressive may require less Aranesp®;

certain of which could have a material adverse impact on future sales of Aranesp®.

See the "Overview" section above and "Item 1A. Risk Factors" in Part II herein for further discussion of certain of the above factors that could impact our future product sales.

EPOGEN®

For the three and nine months ended September 30, 2008 and 2007, total EPOGEN® sales were as follows (dollar amounts in millions):

	Three	Three months ended					Nine months ended					
	Se	September 30,					ıber 30,					
	2008	2007	Change	2	800		2007	Change				
EPOGEN® - U.S.	\$ 634	\$ 602	5%	\$	1,810	\$	1,851	(2)%				

EPOGEN® sales for the three months ended September 30, 2008 increased 5% primarily due to an increase in demand, reflecting an increase in the average net sales price, favorable changes in wholesaler inventory levels and revised estimates of dialysis demand (primarily spillover) for prior quarters. Increased demand due to patient population growth was offset by a decline in dose/utilization within certain settings. The decrease of 2% in EPOGEN® sales for the nine months ended September 30, 2008 was primarily due to a decrease in demand, reflecting a decrease in the average net sales price, unfavorable wholesaler inventory changes and revised estimates of dialysis demand (primarily spillover) for prior quarters. Spillover is a result of the Company's contractual relationship with J&J (see Note 1, "Summary of significant accounting policies – Product sales" to the Condensed Consolidated Financial Statements for further discussion).

In addition to the factors mentioned in the "Product sales" section above, future EPOGEN® sales will be dependent, in part, on such factors as:

- reimbursement developments, including those resulting from:
 - o changes in healthcare providers' prescribing behavior resulting in dose fluctuations due to the CMS' revisions to its EMP, which became effective January 1, 2008;
 - o the federal government's reaction to recent or future product label changes;
 - o changes in reimbursement rates or changes in the basis for reimbursement by the federal and state governments, including Medicare and Medicaid;
- regulatory developments, including those resulting from:
 - o future product label changes;
 - o risk management activities, including a REMS, undertaken by us or required by the FDA;
- governmental or private organization regulations or guidelines relating to the use of our product, including changes in medical guidelines and legislative actions;
- adverse events or results from clinical trials or studies or meta-analyses performed by us, including our pharmacovigilance clinical trials, or by
 others (including our licensees or independent investigators), which have and could further impact product safety labeling, negatively impact
 healthcare provider prescribing behavior, use of our product, regulatory or private healthcare organization medical guidelines and reimbursement
 practices;
- cost containment pressures from the federal and state governments on healthcare providers;
- our current and future contracting and related pricing strategies;

- changes in future patient population growth or dose/utilization; and
- development of new modalities to treat anemia associated with CRF;

certain of which could have a material adverse impact on future sales of EPOGEN®.

See the "Overview" section above and "Item 1A. Risk Factors" in Part II herein for further discussion of certain of the above factors that could impact our future product sales.

Neulasta®/NEUPOGEN®

For the three and nine months ended September 30, 2008 and 2007, total Neulasta®/NEUPOGEN® sales by geographic region were as follows (dollar amounts in millions):

		Thre	e mo	onths end	led	Nine months ended				
		Se	epter	nber 30,		September 30,				
	2008		2008 20		2007		Change	2008	2007	Change
Neulasta® - U.S.	\$	633	\$	598	6%	\$ 1,850	\$ 1,744	6%		
NEUPOGEN® - U.S.		223		232	(4)%	667	636	5%		
U.S. Neulasta®/NEUPOGEN® - Total		856		830	3%	2,517	2,380	6%		
Neulasta® - International		219		165	33%	620	472	31%		
NEUPOGEN® - International		117		105	11%	342	307	11%		
International Neulasta®/NEUPOGEN® - Total		336		270	24%	962	779	23%		
Total Worldwide Neulasta®/NEUPOGEN®	\$	1,192	\$	1,100	8%	\$ 3,479	\$ 3,159	10%		

The increase in U.S. sales of Neulasta®/NEUPOGEN® for the three and nine months ended September 30, 2008 primarily reflects an increase in demand for Neulasta® driven by an increase in the average net sales price. The increase in demand for the three months ended September 30, 2008 was partially offset by a decline in units sold, which we believe was primarily due to stocking by end users of our products, including healthcare providers such as physicians or their clinics and hospitals, which occurred in the three months ended June 30, 2008.

The increase in international Neulasta®/NEUPOGEN® sales for the three and nine months ended September 30, 2008 reflects changes in foreign currency exchange rates, which positively impacted combined international sales by \$33 million and \$97 million, respectively, as well as increased demand driven by continued conversion from NEUPOGEN® to Neulasta®. Excluding the favorable impact of foreign currency exchange rate changes, international Neulasta®/NEUPOGEN® sales increased 12% and 11% over the three and nine months ended September 30, 2007, respectively.

In addition to the factors mentioned in the "*Product sales*" section above, future worldwide Neulasta®/NEUPOGEN® sales growth will be dependent, in part, on such factors as:

- penetration of existing segments;
- competitive products or therapies, including biosimilar products that have been or may be approved in the European Union ("EU") and be available shortly thereafter. For example, in September 2008, Ratiopharm's Ratiograstim®/Filgrastim Ratiopharm®, CT Arzneimittel's Biograstim® and Teva's Tevagrastim®, all G-CSF biosimilar products, received marketing authorization from the European Commission. Ratiopharm has launched its G-CSF biosimilar product, Ratiograstim®, in the United Kingdom in October 2008, and is expected to launch it in Germany and several other European markets in the fourth quarter of 2008. CT Arzneimittel is

expected to market its G-CSF biosimilar product in Germany in the fourth quarter of 2008. Teva stated that it would begin marketing its G-CSF biosimilar product throughout Europe in 2009.

- reimbursement by third-party payers, including governments and private insurance plans;
- adverse events or results from clinical trials or studies or meta-analyses performed by us or by others (including our licensees or independent investigators), which could expand safety labeling and may negatively impact healthcare provider prescribing behavior, use of our products, regulatory or private healthcare organization medical guidelines and reimbursement practices;
- governmental or private organization regulations or guidelines relating to the use of our products;
- cost containment pressures from governments and private insurers on healthcare providers;
- our current and future contracting and related pricing strategies;
- patient population growth; and
- development of new treatments for cancer and future chemotherapy treatments. For example, targeted therapies and other treatments that are less myelosuppressive may require less Neulasta®/NEUPOGEN®.

See the "Overview" section above and "Item 1A. Risk Factors" in Part II herein for further discussion of certain of the above factors that could impact our future product sales.

ENBREL

For the three and nine months ended September 30, 2008 and 2007, total ENBREL sales by geographic region were as follows (dollar amounts in millions):

	Thre	ee months en	ded	Nine months ended					
	S	eptember 30,		September 30,					
	2008	2008 2007 Ch		2008 2007		Change			
ENBREL - U.S.	\$ 838	\$ 777	8%	\$ 2,531	\$ 2,247	13%			
ENBREL - International	55	44	25%	154	127	21%			
Total ENBREL	\$ 893	\$ 821	9%	\$ 2,685	\$ 2,374	13%			

ENBREL sales growth for the three months ended September 30, 2008 reflects higher demand due to increases in both average net sales price and patients. ENBREL sales growth in the three months ended September 30, 2008 was affected by share declines in the United States compared to the three months ended September 30, 2007 due to increased competitive activity. However, sales growth continued in both rheumatology and dermatology, and ENBREL continues to maintain a leading position in both segments.

ENBREL sales growth for the nine months ended September 30, 2008 reflects higher demand due to increases in both average net sales price and patients and an initial wholesaler inventory stocking of approximately \$120 million resulting from the shift to a wholesaler distribution model in the three months ended March 31, 2008. During the three months ended March 31, 2008, ENBREL's distribution model was converted from primarily being drop shipped directly to pharmacies to a wholesaler distribution model similar to our other products in the United States. We believe that this estimated initial wholesaler inventory stocking is within the expected normal inventory range. ENBREL sales growth in the nine months ended September 30, 2008 was affected by share declines in the United States compared to the nine months ended September 30, 2007 due to increased competitive activity.

In addition to the factors mentioned in the "*Product sales*" section above, future worldwide ENBREL sales growth will be dependent, in part, on such factors as:

- the effects of competing products or therapies, which may include new indications for existing products and new competitive products coming to market, such as J&J's CNTO 1275 (ustekinumab) and CNTO 148 (golimumab) and, in part, our ability to differentiate ENBREL based on its safety profile and efficacy;
- recent or future product label changes;
- risk management activities, including a REMS, undertaken by us or required by the FDA or other regulatory authorities;
- growth in the rheumatology and dermatology segments;
- the outcome of the FDA's review of the supplemental BLA for the use of ENBREL in pediatric patients with chronic moderate to severe plaque psoriasis;
- the availability, extent and access to reimbursement by government and third-party payers;
- adverse events or results from clinical trials or studies or meta-analyses performed by us or by others (including our licensees or independent investigators), which could expand safety labeling and may negatively impact healthcare provider prescribing behavior, use of our product, regulatory or private healthcare organization medical guidelines and reimbursement practices;
- governmental or private organization regulations or guidelines relating to the use of our product;
- · cost containment pressures from governments and private insurers on healthcare providers;
- current and future contracting and related pricing strategies;
- patient population growth; and
- penetration of existing and new segments, including potential expanded indications.

See the "Overview" section above and "Item 1A. Risk Factors" in Part II herein for further discussion of certain of the above factors that could impact our future product sales.

Selected operating expenses

The following table summarizes selected operating expenses for the three and nine months ended September 30, 2008 and 2007 (dollar amounts in millions):

	Three months ended September 30,						nths ended nber 30,			
		2008	08 2007 Change		Change	nange 2008		8 2007		Change
Product sales	\$	3,784	\$	3,524	7%	\$	11,013	\$	10,693	3%
Operating expenses:										
Cost of sales (excludes amortization										
of acquired intangible assets)	\$	677	\$	792	(15)%	\$	1,738	\$	1,942	(11)%
% of product sales		18%		22%			16%		18%	
Research and development	\$	729	\$	776	(6)%	\$	2,232	\$	2,444	(9)%
% of product sales		19%		22%			20%		23%	
Selling, general and administrative	\$	900	\$	730	23%	\$	2,678	\$	2,360	13%
% of product sales		24%		21%			24%		22%	
Amortization of acquired intangible assets	\$	74	\$	76	(3)%	\$	221	\$	224	(1)%
Write-off of acquired in-process										
research and development	\$	-	\$	590	(100)%	\$	-	\$	590	(100)%
Other charges	\$	12	\$	254	(95)%	\$	306	\$	543	(44)%

Cost of sales

Cost of sales, which excludes the amortization of acquired intangible assets (see "Condensed Consolidated Statements of Income"), decreased 15% for the three months ended September 30, 2008 primarily driven by prior year restructuring costs, principally accelerated depreciation, which totaled \$113 million. Absent the impact of the restructuring costs, costs of sales was relatively unchanged from the prior year as higher sales volume was offset by lower inventory reserves, lower excess capacity charges and lower ENBREL manufacturing costs. Cost of Sales for the nine months ended September 30, 2008 decreased 11% primarily as the result of charges incurred in the nine months ended September 30, 2007 including the above-mentioned restructuring charges and the write-off of a semi-completed manufacturing asset. Cost of sales for the nine months ended September 30, 2008 also reflects lower inventory reserves and lower excess capacity charges than the nine months ended September 30, 2007.

Research and development

R&D costs are expensed as incurred and primarily include salaries, benefits and other staff related costs; facilities and overhead costs; clinical trial and related clinical manufacturing costs; contract services and other outside costs; information systems and amortization of acquired technology used in R&D with alternative future uses. R&D expenses consist of internal R&D costs; costs incurred under R&D arrangements with our corporate partners, such as activities performed on behalf of KA, and costs associated with collaborative R&D and in-licensing arrangements, including upfront fees and milestones paid to collaboration partners in connection with technologies that have no alternative future use. Net payment or reimbursement of R&D costs for R&D collaborations are recognized as the obligation has been incurred or as we become entitled to the cost recovery.

R&D expenses for the three and nine months ended September 30, 2007 include \$18 million of restructuring costs, comprised of \$35 million in charges related to asset impairments offset by a \$17 million benefit associated with the reversal of previously accrued expenses for bonuses and stock-based compensation awards, which were forfeited as a result of the employees' termination. During the three months ended September 30, 2008, there were no R&D restructuring related costs. For the nine months ended September 30, 2008, restructuring related R&D costs totaled \$3 million. See Note 2, *"Restructuring"* to the Condensed Consolidated Financial Statements for further discussion.

R&D expenses decreased 6% for the three months ended September 30, 2008 due to lower clinical trial costs of \$39 million; a decrease of \$22 million in merger related expenses; the above noted decline in restructuring related costs of \$18 million and the net benefit derived from our licensing activities, including our transaction with Takeda Pharmaceutical Company Limited ("Takeda") in Japan, totaling \$11 million, partially offset by \$15 million of higher staff related costs. Our clinical trial costs were lower in the three months ended September 30, 2008 primarily due to the completion of enrollment of our large denosumab clinical trials and the related significant costs associated with site initiation and patient enrollment no longer being incurred, partially offset by the increased clinical costs for our emerging pipeline.

R&D expenses decreased 9% for the nine months ended September 30, 2008, which was primarily attributable to \$100 million of decreased staff related costs; \$88 million of lower clinical trial costs; \$82 million of cost recoveries derived from licensing transactions, primarily with Daiichi Sankyo Company and Takeda in Japan; \$24 million of decreased merger related expenses and \$15 million in reduced restructuring related costs, partially offset by a \$100 million expense in the nine months ended September 30, 2008 for the upfront payment associated with the Kyowa Hakko collaboration. Clinical trial costs decreased as some of our large clinical trials completed enrollment, as discussed above.

Selling, general and administrative

SG&A expenses are primarily comprised of salaries and benefits associated with sales and marketing, finance, legal and other administrative personnel; outside marketing and legal expenses; overhead and facilities costs and other general and administrative costs. In connection with a co-promotion agreement, we and Wyeth market and sell ENBREL in the United States and Canada and Wyeth is paid a share of the related profits, as defined. The share of ENBREL's profits owed to Wyeth (the "Wyeth profit share expense") is included in SG&A expenses.

For the three and nine months ended September 30, 2007, we recorded \$92 million in cost recoveries for certain restructuring charges, principally with respect to accelerated depreciation, in connection with our co-promotion agreement with Wyeth totaling \$83 million, and \$9 million of benefit associated with the reversal of previously accrued expenses for bonuses and stock-based compensation awards, which were forfeited as a result of the employees' termination. During the three months ended September 30, 2008, there were no SG&A restructuring related costs. For the nine months ended September 30, 2008, restructuring related SG&A costs totaled \$1 million. See Note 2, *"Restructuring"* to the Condensed Consolidated Financial Statements for further discussion.

For the three months ended September 30, 2008, the 23% increase in SG&A was due to the above noted recoveries of restructuring related costs in the prior year and higher expenses during the three months ended September 30, 2008 for Wyeth profit share expense and staff related costs of \$53 million and \$45 million, respectively, partially offset by lower litigation expense of \$28 million. For the three months ended September 30, 2008 and 2007, the Wyeth profit share expense was \$298 million and \$245 million, respectively.

For the nine months ended September 30, 2008, the 13% increase in SG&A was due to the above noted recoveries of restructuring related costs in the prior year and, higher expenses for the nine months ended September 30, 2008 for Wyeth profit share expense and staff related costs of \$167 million and \$99 million, respectively, partially offset by lower litigation expense of \$35 million. For the nine months ended September 30, 2008 and 2007, the Wyeth profit share expense was \$886 million and \$719 million, respectively.

Amortization of acquired intangible assets

Amortization of acquired intangible assets relates to the acquired product technology rights acquired in connection with the Immunex acquisition.

Other charges

As discussed in Note 2, "*Restructuring*" to the Condensed Consolidated Financial Statements, on August 15, 2007, we announced plans to restructure our worldwide operations in order to improve our cost structure while continuing to make significant R&D investments and build the framework for our future growth. As a result of this restructuring plan, we recorded the following charges during the three and nine months ended September 30, 2008: (i) staff separation costs of \$0 and \$4 million, respectively, (ii) asset impairment charges of \$1 million and \$15 million, respectively, and (iii) other charges of \$7 million and \$20 million, respectively, primarily related to loss accruals for leases for certain facilities that will not be used in our business.

In conjunction with the above noted restructuring activities, we recorded the following charges during the three and nine months ended September 30, 2007: (i) staff separation costs of \$104 and \$107 million, respectively and (ii) asset impairment charges of \$71 million and \$357 million, respectively. In addition, we recorded other charges of \$79 million during the three and nine months ended September 30, 2007 primarily related to loss accruals for leases for certain R&D facilities that will not be used in our business.

Also, in the three and nine months ended September 30, 2008, the Company recorded loss accruals for settlements of certain commercial legal proceedings aggregating \$4 million and \$267 million, respectively. Loss accruals for the nine months ended September 30, 2008 principally related to the settlement of the Ortho Biotech antitrust suit.

Interest and other income and (expense), net

Interest and other income and (expense), net for the three months ended September 30, 2008 was \$12 million of expense compared to \$21 million of expense for the three months ended September 30, 2007. This change is primarily due to higher interest income of \$17 million, primarily due to higher cash balances and lower interest expense of \$19 million, primarily due to lower interest rates, partially offset by the loss accrued in the three months ended September 30, 2008 on the sale of certain less significant marketed products and related assets of \$9 million, which was part of our restructuring plan.

Interest and other income and (expense), net for the nine months ended September 30, 2008 was \$19 million of income compared to \$20 million of expense for the nine months ended September 30, 2007. This change is primarily due to the write-off of \$51 million of deferred financing and related costs in March 2007 resulting from the repayment of certain of our convertible debt, an increase in net realized gains of approximately \$48 million primarily due to the rebalancing of investments in our marketable securities portfolio, an increase in interest income of approximately \$27 million primarily due to higher average cash balances, partially offset by incremental interest expense of approximately \$59 million primarily related to the issuance of \$4.0 billion of debt in May 2007 and \$1.0 billion of debt in May 2008, and \$9 million related to the loss accrued on the sale of certain less significant marketed products and related assets.

Income taxes

Our effective tax rates for the three and nine months ended September 30, 2008 were 21.3% and 21.0%, respectively, compared with 46.0% and 19.7%, respectively, for the same periods last year. The decrease in our effective tax rate for the three months ended September 30, 2008 compared to the same period last year was primarily due to the non-deductible, acquired IPR&D expense incurred in connection with the acquisitions of Alantos and Ilypsa in 2007, partially offset by expiration of the federal research and experimentation tax credit ("R&D credit") on December 31, 2007. The increase in our effective tax rate for the nine months ended September 30, 2008 compared to the same period last year was primarily due to the favorable resolution of our prior year's federal income tax examination in the second quarter of 2007, expiration of federal R&D credit for 2008 that was not extended prior to September 30, 2008, partially offset by the non-deductible, acquired IPR&D expense incurred in connection with Alantos and Ilypsa in 2007.

See Note 4, "Income taxes" to the Condensed Consolidated Financial Statements for further discussion.

Recent accounting pronouncements

In June 2008, the FASB ratified EITF Issue No. 07-5, "Determining Whether an Instrument (or Embedded Feature) Is Indexed to an Entity's Own Stock" ("EITF 07-5"). Equity-linked instruments (or embedded features) that otherwise meet the definition of a derivative as outlined in SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities," are not accounted for as derivatives if certain criteria are met, one of which is that the instrument (or embedded feature) must be indexed to the entity's stock. EITF 07-5 provides guidance on how to determine if equity-linked instruments (or embedded features) such as warrants to purchase our stock, our convertible notes and convertible note hedges are considered indexed to our stock. EITF 07-5 is effective for the financial statements issued for fiscal years and interim periods within those fiscal years, beginning after December 15, 2008 and will be applied to outstanding instruments as of the beginning of the fiscal year in which it is adopted. Upon adoption, a cumulative effect adjustment will be recorded, if necessary, based on amounts that would have been recognized if this guidance had been applied from the issuance date of the affected instruments. We are currently determining the impact that EITF 07-05 will have on our financial statements, if any.

In May 2008, the FASB issued FSP APB 14-1, "Accounting for Convertible Debt Instruments That May Be Settled in Cash upon Conversion (Including Partial Cash Settlement" ("FSP APB 14-1") that changes the method of accounting for convertible debt securities that require or permit settlement in cash either in whole or in part upon conversion, including our convertible debt securities (see Note 5, "Financing arrangements" to the Condensed Consolidated Financial Statements). We will adopt FSP APB 14-1 in the first quarter of 2009 and retrospectively apply this change to prior periods, as required by this new standard. Under this new method of accounting, the debt and equity components of our convertible debt securities will be bifurcated and accounted for separately in a manner that will result in recognizing interest expense on these securities at effective rates reflective of what we would have incurred had we issued nonconvertible debt with otherwise similar terms. The equity component of our convertible debt securities will be included in the paid-in-capital section of stockholders' equity on our Consolidated Balance Sheet and, accordingly, the initial carrying values of these debt securities will be reduced. Our net income for financial reporting purposes will be reduced by recognizing the accretion of the reduced carrying values of our convertible debt securities to their face amounts as additional non-cash interest expense. We are currently determining the impact FSP APB 14-1 will have on our financial statements. We expect it will have a material adverse impact on our past and future reported financial results but will have no impact on past or future cash flows.

In December 2007, the FASB issued SFAS No. 141(R), "Business Combinations" and SFAS No. 160, "Accounting and Reporting of Noncontrolling Interests in Consolidated Financial Statements – an amendment of ARB No. 51". These standards will significantly change the accounting and reporting for business combination transactions and noncontrolling (minority) interests in consolidated financial statements, including capitalizing at the acquisition date the fair value of acquired IPR&D, and testing for impairment and writing down these assets, if necessary, in subsequent periods during their development. These new standards will be applied prospectively for business combinations that occur on or after January 1, 2009, except that presentation and disclosure requirements of SFAS 160 regarding noncontrolling interests will be applied retrospectively.

Financial Condition, Liquidity and Capital Resources

The following table summarizes selected financial data (in millions):

	September 30, 2008		December 31,	
				2007
Cash, cash equivalents and marketable securities	\$	9,757	\$	7,151
Total assets		36,998		34,639
Current debt		1,000		2,000
Non-current debt		10,176		9,177
Stockholders' equity		19,832		17,869

We believe that existing funds, cash generated from operations and existing sources of and access to financing are adequate to satisfy our working capital, capital expenditure and debt service requirements for the foreseeable future. In addition, we plan to opportunistically pursue our stock repurchase programs and other business initiatives, including acquisitions and licensing activities. Our liquidity needs can be met through a variety of sources, including: cash provided by operating activities, sale of marketable securities, borrowings through commercial paper and/or our syndicated credit facility and other debt markets and equity markets. Our current financial position, liquidity and credit ratings should allow us to access the capital markets. However, due to the recent extreme volatility and disruption in the capital markets, we expect to be opportunistic in our timing for raising financing in the future. We have cash available to retire our remaining \$1.0 billion of floating rate notes due on November 28, 2008 (following the redemption of \$1.0 billion of such notes in June 2008). In addition, we anticipate that our business will generate sufficient cash for us to have flexibility to repay \$1.0 billion of our 4.00% notes due November 2009 without incurring additional indebtedness.

Cash, cash equivalents and marketable securities

Of the total cash, cash equivalents and marketable securities at September 30, 2008, approximately \$7.7 billion was generated from operations in foreign tax jurisdictions and is intended for use in our foreign operations. If these funds were repatriated for use in our U.S. operations, we would be required to pay additional U.S. federal and state income taxes at the applicable marginal tax rates.

Financing arrangements

The following table reflects the carrying value of our long-term borrowings under our various financing arrangements as of September 30, 2008 and December 31, 2007 (in millions):

	September 30, 2008		December 31, 2007	
0.125% convertible notes due 2011 (2011 Convertible Notes)	\$	2,500	\$	2,500
0.375% convertible notes due 2013 (2013 Convertible Notes)		2,500		2,500
5.85% notes due 2017 (2017 Notes)		1,099		1,099
Floating rate notes due 2008 (2008 Floating Rate Notes)		1,000		2,000
4.00% notes due 2009 (2009 Notes)		1,000		999
4.85% notes due 2014 (2014 Notes)		1,000		1,000
6.375% notes due 2037 (2037 Notes)		899		899
6.15% notes due 2018 (2018 Notes)		499		-
6.90% notes due 2038 (2038 Notes)		498		-
Other		181		180
Total borrowings	_	11,176		11,177
Less current portion		1,000		2,000
Total non-current debt	\$	10,176	\$	9,177

On April 17, 2008, we filed a shelf registration statement with the SEC, which replaced our previous \$1.0 billion shelf registration statement and allows us to issue an unspecified amount of debt securities; common stock; preferred stock; warrants to purchase debt securities, common stock, preferred stock or depository shares; rights to purchase common stock or preferred stock; securities purchase contracts; securities purchase units and depository shares. Under this registration statement, all of the securities available for issuance may be offered from time to time with terms to be determined at the time of issuance.

In May 2008, we increased our commercial paper program by \$1.3 billion, which provides for unsecured, short-term borrowings of up to an aggregate of \$2.5 billion. We also have a \$2.5 billion syndicated unsecured revolving credit facility which matures in November 2012 and is available for general corporate purposes, or as a liquidity backstop to our commercial paper program; however, \$178 million of such commitment was provided by a subsidiary of Lehman. Lehman declared bankruptcy on September 15, 2008, and the subsidiary participant in our credit facility subsequently declared bankruptcy on October 5, 2008. As a result, we would not anticipate the ability to access this specific commitment provided by Lehman in the future. No amounts were outstanding under the commercial paper program or credit facility as of September 30, 2008.

Certain of our financing arrangements contain non-financial covenants and we were in compliance with all applicable covenants as of September 30, 2008. None of our financing arrangements contain any financial covenants. Our outstanding convertible notes and our other outstanding long-term notes are rated "A+" with a stable outlook by Standard & Poor's, "A3" with a stable outlook by Moody's Investors Service, Inc. and "A" with a stable outlook by Fitch, Inc.

See Note 5, "*Financing arrangements*" to our Condensed Consolidated Financial Statements for further discussions of the transactions during the nine months ended September 30, 2008 and "*Recent accounting pronouncements*" for a discussion of future impacts to the accounting for our convertible debt.

Cash flows

The following table summarizes our cash flow activity (in millions):

	Nine months ended September 30,				
		2008		2007	
Net cash provided by operating activities	\$	4,591	\$	3,871	
Net cash used in investing activities		(2,618)		(1,295)	
Net cash used in financing activities		(1,475)		(2,470)	

Operating

Cash provided by operating activities has been and is expected to continue to be our primary recurring source of funds. Cash provided by operating activities during the nine months ended September 30, 2008 increased primarily due to the receipt of \$300 million for an upfront milestone payment related to our licensing agreement with Takeda, which is included in the "Changes in deferred revenue" in the Condensed Consolidated Statements of Cash Flows, increased cash receipts from the sale of our products, lower payments in the ordinary course of business and lower tax payments, partially offset by payments for legal settlements totaling \$283 million.

Investing

Cash used in investing activities during the nine months ended September 30, 2008 increased primarily due to the net purchases of marketable securities partially offset by a decrease in capital expenditures. Net purchases of marketable securities were \$2.2 billion in the nine months ended September 30, 2008 compared to net receipts of \$473 million for the nine months ended September 30, 2007. Capital expenditures totaled \$494 million during the nine months ended September 30, 2008 were primarily associated with manufacturing capacity expansions in Puerto Rico, Fremont and other site developments and investment in our global enterprise resource planning ("ERP") system and other information systems projects. The capital expenditures during the nine months ended September 30, 2007 were primarily associated with manufacturing capacity and site expansions in Puerto Rico and other locations and investment in our global ERP system and other information systems projects and equipment to be approximately \$750 million.

On July 16, 2007, we completed our acquisition of Alantos and pursuant to the merger agreement, we paid \$300 million in cash, net of cash acquired and transaction costs. On July 18, 2007, we completed our acquisition of Ilypsa and pursuant to the merger agreement, we paid \$398 million of cash, net of cash acquired and transaction costs of \$2 million.

Financing

In May 2008, we issued \$500 million aggregate principal amount of notes due in 2018 (the "2018 Notes") and \$500 million aggregate principal amount of notes due in 2038 (the "2038 Notes") in a registered offering. The 2018 Notes and 2038 Notes pay interest at fixed annual rates of 6.15% and 6.90%, respectively. Concurrent with the issuance of the 2018 Notes, we entered into interest rate swap agreements that effectively convert the payment of our fixed rate interest payments to variable rate interest payments over the life of the 2018 Notes. The 2018 Notes and 2038 Notes may be redeemed at any time at our option, in whole or in part, at 100% of the principal amount of the notes being redeemed plus accrued and unpaid interest, if any, and a "make-whole" amount, as defined in the indenture governing the notes. In the event of a change in control triggering event, as defined in the indenture governing the notes, we may be required to purchase for cash all or a portion of the 2018 Notes and the 2038 Notes at a price equal to 101% of the principal amount of the notes plus accrued interest. Debt issuance costs totaled approximately \$6 million and are being amortized over the life of the notes.

Upon the receipt of the proceeds from the issuance of the 2018 Notes and 2038 Notes, we exercised our right to call \$1.0 billion of floating rate notes due November 2008, which were retired in June 2008.

During the nine months ended September 30, 2008, we repurchased 32.7 million shares of our common stock at a total cost of \$1.6 billion, in connection with an ASR entered into in May 2008. During the nine months ended September 30, 2007, we repurchased 85.2 million shares of our common stock at a total cost of \$5.0 billion, in connection with an ASR entered into in May 2007. As of September 30, 2008, we had \$4.9 billion available for stock repurchases as authorized by our Board of Directors. The manner of purchases, amount we spend and the number of shares repurchased will vary based on a number of factors including the stock price, blackout periods in which we are restricted from repurchasing shares, and our credit rating and may include private block purchases as well as market transactions. Repurchases under our stock repurchase programs reflect, in part, our confidence in the long-term value of our common stock. Additionally, we believe that it is an effective way of returning cash to our stockholders.

We receive cash from the exercise of employee stock options and proceeds from the sale of stock. Employee stock option exercises provided \$114 million and \$244 million of cash during the nine months ended September 30, 2008 and 2007, respectively. Proceeds from the exercise of employee stock options will vary from period to period based upon, among other factors, fluctuations in the market value of our stock relative to the exercise price of such options.

On March 2, 2007, as a result of holders of substantially all of our outstanding 2032 Modified Convertible Notes exercising their March 1, 2007 put option, we purchased \$2.3 billion aggregate principal amount, or the majority of the then outstanding convertible notes at their then-accreted value for \$1.7 billion in cash.

In May 2007, we issued \$2.0 billion aggregate principal amount of floating rate notes due in 2008, \$1.1 billion aggregate principal amount of 5.85% notes due in 2017 and \$900 million aggregate principal amount of 6.375% notes due in 2037. The 2008 Floating Rate Notes bear interest at a rate per annum, equal to LIBOR plus 0.08%, which is reset quarterly. A total of \$3.2 billion of the net proceeds raised from the issuance of these notes were used to repurchase shares of our common stock under an ASR entered into in May 2007.

Contractual Obligations

On October 22, 2008, we entered into an agreement with International Business Machines Corporation ("IBM") for information systems infrastructure services including electronic messaging systems, networks, helpdesk support for end users, physical information technology support, servers hosting applications, information systems hardware refresh, and evolution of methods, processes and technologies being used to provide information systems infrastructure services. The term of the agreement is five years with three one-year renewal options by us, for a total of an eight year term. The cost to us for the five year term is estimated to be \$500 million with a full eight year term estimated to be \$800 million.

Item 4. CONTROLS AND PROCEDURES

We maintain "disclosure controls and procedures," as such term is defined under Exchange Act Rule 13a-15(e), that are designed to ensure that information required to be disclosed in Amgen's Exchange Act reports is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to Amgen's management, including its Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosures. In designing and evaluating the disclosure controls and procedures, Amgen's management recognized that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives and, in reaching a reasonable level of assurance, Amgen's management necessarily was required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures. We have carried out an evaluation under the supervision and with the participation of our management, including Amgen's Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of Amgen's disclosure controls and procedures. Based upon their evaluation and subject to the foregoing, the Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective as of September 30, 2008.

Management determined that, as of September 30, 2008, there were no changes in our internal control over financial reporting that occurred during the fiscal quarter then ended that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

Item 1. LEGAL PROCEEDINGS

See Note 9, "*Contingences*" to the Condensed Consolidated Financial Statements for a discussion which is limited to certain recent developments concerning our legal proceedings. This discussion should be read in conjunction with Note 10, "*Contingencies*" to our Consolidated Financial Statements in Part IV of our Annual Report on Form 10-K for the year ended December 31, 2007.

Item 1A. RISK FACTORS

This report and other documents we file with the SEC contain forward looking statements that are based on current expectations, estimates, forecasts and projections about us, our future performance, our business or others on our behalf, our beliefs and our management's assumptions. These statements are not guarantees of future performance and involve certain risks, uncertainties, and assumptions that are difficult to predict. You should carefully consider the risks and uncertainties facing our business. The risks described below are not the only ones facing us. Our business is also subject to the risks that affect many other companies, such as employment relations, general economic conditions, geopolitical events and international operations. Further, additional risks not currently known to us or that we currently believe are immaterial also may impair our business, operations, liquidity and stock price materially and adversely.

Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market.

We and certain of our licensees and partners conduct research, preclinical testing and clinical trials for our product candidates and marketed products for both their existing indications as well as for new and/or expanded indications. In addition, we manufacture and contract manufacture, and certain of our licensees and partners manufacture our products and product candidates, price, sell, distribute and market or co-market our products for their approved indications. These activities are subject to extensive regulation by numerous state and federal governmental authorities in the United States, such as the FDA and CMS, as well as in foreign countries, such as the EMEA in European countries and similar regulatory bodies in Canada and Australia. Currently, we are required in the United States and in foreign countries to obtain approval from those countries' regulatory authorities before we can manufacture (or have our third-party manufacturers produce), market and sell our products in those countries. The FDA and other U.S. and foreign regulatory agencies have substantial authority to fail to approve commencement of, suspend or terminate clinical trials, require additional testing, delay or withhold registration and marketing approval, mandate product withdrawals and require changes in labeling (including eliminating certain therapeutic indications) of our products. On September 27, 2007, President Bush signed into law the FDAAA, significantly adding to the FDA's authority including allowing the FDA to (i) require sponsors of marketed products to conduct postapproval clinical studies to assess a known serious risk, signals of serious risk or to identify an unexpected serious risk; (ii) mandate labeling changes to products, at any point in a product's lifecycle, based on new safety information and (iii) require sponsors to implement a REMS for a product which could include a medication guide, patient package insert, a communication plan to healthcare providers, or other elements as the FDA deems are necessary to assure safe use of the drug, which could include imposing certain restrictions on distribution or use of a product. Failure to comply with the new requirements, if imposed on a sponsor by the FDA under the FDAAA, could result in significant civil monetary penalties. Further, regulatory agencies could change existing, or promulgate new, regulations at any time which may affect our ability to obtain or maintain approval of our existing or future products or require significant additional costs to obtain or maintain such approvals.

In our experience, obtaining regulatory approval has been and continues to be increasingly difficult and costly and takes many years, and, after it is obtained, remains costly to maintain. With the occurrence of a number of high profile safety events with certain pharmaceutical products, regulatory authorities, and, in particular, the FDA, members of Congress, the U.S. Government Accountability Office ("GAO"), Congressional

committees, private health/science foundations and organizations, medical professionals, including physicians and investigators, and the general public are increasingly concerned about potential or perceived safety issues associated with pharmaceutical and biological products, whether under study for initial approval or already marketed. For example, we have received letters from both the House Subcommittee on Oversight and Investigation, Committee on Energy and Commerce and the United States Senate Committee on Finance with inquiries with respect to our ESA studies, promotions of our ESAs and other products, rebates and contracting strategies and our pharmacovigilance program, to which we have fully cooperated by submitting our responses and meeting with Congressional staff. To the extent that there is resulting legislation or changes in CMS or FDA policy or regulatory activity as a result of Congressional concerns, such changes could have a material or adverse effect on the use of our ESA products.

As a result of this increasing concern, potential or perceived safety signals and safety concerns, from clinical trials, use by the market or other sources, are receiving greater scrutiny, which may lead to (i) fewer treatments being approved by the FDA or other regulatory bodies, (ii) revised labeling of an approved product or a class of products for safety reasons, potentially including a boxed warning or additional limitations on the use of approved products in specific therapeutic areas (possibly until additional clinical trials can be designed and completed), (iii) mandated PMCs, pharmacovigilance programs for approved products and/or (iv) requirement of risk management activities (including a REMS) related to the promotion and sale of a product. In addition, significant concerns about the safety and effectiveness of our products could ultimately lead to the revocation of marketing approval by therapeutic area, or in total, which would have a material adverse effect on the use, sales and reimbursement of the affected products and on our business and results of operations. (See "— *Our sales depend on payment and reimbursement from third-party payers, and, to the extent that reimbursement for our products is reduced, this could negatively impact the utilization of our products.*")

Certain specific labeling or label changes of approved products or product candidates may be necessary or required for a number of reasons, including: the identification of actual or theoretical safety or efficacy concerns by regulatory agencies, the discovery of significant problems or safety signals or trends with a similar product that implicates an entire class of products, subsequent concerns about the sufficiency of the data or studies underlying the label or changes to the underlying safety/efficacy analysis related to results from clinical trials or meta-analysis of clinical trials or clinical data performed by us or others. In addition, before or after any of our products are approved for commercial use, regulatory bodies could decide that the product labels need to include certain warning language as part of an evolving label change to a particular class of products. For example, in March and November 2007, and in March and August 2008, the labels of the class of ESA products, including Aranesp® and EPOGEN®, were updated to include revised boxed warnings, restrictions on the use of ESAs in specific therapeutic areas and other safety-related product labeling changes. (See "- Recent labeling changes or risk management activities required by regulatory authorities, as well as the results or meta-analyses of clinical trials, may adversely impact the use, sales and reimbursement of our ESAs.") On March 17, 2008, we and Wyeth announced updates to the FDA approved labeling for ENBREL in which the U.S. PI now contains a boxed warning relating to the risk of infections, including tuberculosis. This information now in the boxed warning includes additional language regarding screening and monitoring patients for tuberculosis, including patients who tested negative for latent tuberculosis infection. Further, on September 4, 2008, the FDA issued a web-alert regarding their review of histoplasmosis and other opportunistic fungal infections in patients treated with TNF blockers. The FDA requested that the boxed warning and warnings sections of the U.S. PI and the medication guide for ENBREL (and other TNF blockers) be strengthened to include the risk of unrecognized histoplasmosis and other invasive fungal infections with the goal of increasing timely diagnosis and treatment. The FDA also requested that the approved REMS for ENBREL be modified with a communication plan to healthcare providers regarding the risk of unrecognized fungal infections. We are working with the FDA to finalize the required revisions to respond to its requests.

Additionally, on June 4, 2008, the FDA issued an Early Communication regarding the ongoing safety review of TNF blockers and the possible association between the use of these medicines and the development of lymphoma and other cancers in children and young adults and stated that it had decided to conduct further analyses to evaluate the risk and benefits of TNF blockers in pediatric patients. On June 18, 2008, we participated in a meeting of the DODAC to review data supporting the supplemental BLA submitted by us for

the use of ENBREL in treating pediatric patients with chronic moderate to severe plaque psoriasis, who are inadequately controlled with topical therapy or who have received systemic therapy or phototherapy and the DODAC recommended, with an 8-5 vote, to approve ENBREL in the treatment of chronic moderate to severe plaque psoriasis in children. On July 24, 2008, we received notification from the FDA through a Complete Response letter that the FDA would like additional information from us to support the use of ENBREL in pediatric patients with chronic moderate to severe plaque psoriasis. We continue to work with the FDA to respond to its requests and to provide it with information to address additional questions related to the supplemental BLA and plans for risk management activities. Although we cannot predict what the FDA's nalysis of TNF blockers and the development of lymphoma or other cancers in children and young adults may lead to or speculate on the timing of the FDA's response on the supplemental BLA, further revisions to the ENBREL label or other actions by the FDA, including additional advisory committee meetings, could have a negative impact on the use and sales of ENBREL which could have a material adverse effect on our business and results of operations.

A revision of product labeling or the regulatory actions described above could be required even if there is no clearly established connection between the product and the safety or efficacy concerns that have been raised or if the product is not indicated for a particular use. For example in October 2007, we announced that we and the FDA adopted changes to the U.S. labeling for Vectibix[®] based on the results of the Panitumumab Advanced Colorectal Cancer Evaluation ("PACCE") trial highlighting to clinicians the greater risk seen when Vectibix[®] is combined with Avastin[®] and the specific chemotherapy used in the PACCE trial to treat patients with first-line metastatic colorectal cancer ("mCRC"). Vectibix[®] is not indicated for the first-line treatment of mCRC and the new safety information applies to an unapproved use of Vectibix[®].

If we or others identify safety concerns before approval of the product or after a product is on the market, the regulatory agencies such as the FDA or EMEA may impose risk management activities upon us (including a REMS) which may require substantial costs and resources to negotiate, develop and implement, including sales force time to educate physicians on REMS requirements and compliance, and/or may require additional or more extensive clinical trials as part of a pharmacovigilance program of our product, or for approval of a new indication. Further, risk management activities, including a REMS, required by regulatory agencies such as the FDA could also modify, restrict or otherwise impact our existing promotional activities of our other products, restrict or encumber the ability of healthcare providers to prescribe, dispense or use our products or limit patient access to our products, or affect our ability to compete against products that do not have a REMS, any of which could have a negative affect on our ability to launch our product and could have a material adverse effect on sales of the affected products and on our business and results of operations. For example, as part of the approval for NplateTM, a REMS was developed with the FDA to assure the safe use of NplateTM while minimizing risk. The NplateTM REMS involves, among other things, healthcare provider and patient enrollment registries, tracking of patient medical history and data and follow-up safety questionnaires to healthcare providers all of which require extensive discussion and education with healthcare providers which has limited our ability to promote NplateTM and our other products. Further, as part of the update to the boxed warning and warnings sections of the U.S. PI and the medication guide for ENBREL, the FDA stated that it would require us and the other makers of TNF blockers to educate healthcare providers about the risk of histoplasmosis. In addition, we have ongoing PMC studies for substantially all of our marketed products other than Sensipar®. These clinical trials must be conducted by us to maintain regulatory approval and marketing authorization. For example, we have agreed with the FDA to a robust pharmacovigilance program to continue to study the safety surrounding the use of ESAs in certain cancer indications. (See "- Recent labeling changes or risk management activities required by regulatory authorities, as well as the results or meta-analyses of clinical trials, may adversely impact the use, sales and reimbursement of our ESAs.") Additionally, the approvals of Vectibix® in both the United States and EU were conditioned on us conducting additional clinical trials of the use of Vectibix[®] as a therapy in treating mCRC and our conditional approval of Vectibix[®] in the EU is currently the subject of an annual review by the CHMP. If results from clinical trials as part of a PMC or pharmacovigilance program are negative, it could result in the revocation of the marketing or conditional marketing approvals or revised labeling of our products, which could have a material adverse effect on sales of the affected products and on our business and results of operations.

Substantially all of our marketed products are currently approved in the United States and most are approved in Europe and in other foreign countries for specific uses. However, later discovery of unknown problems with our products could result in the regulatory activities described above or even the potential withdrawal of the product in certain therapeutic areas or certain product presentations, or completely, from the market. If new medical data suggests an unacceptable safety risk or previously unidentified side-effects, we may voluntarily withdraw, or regulatory authorities may mandate we withdraw such product in certain therapeutic areas, or completely recall a product presentation from the market for some period or permanently. For example in 2006, we initiated a voluntary recall of the Neulasta® SureClickTM pre-filled pen in Europe because of the potential risk to patients of receiving an incomplete dose and we conducted a voluntary wholesaler recall of a limited number of lots of ENBREL as a result of a small number of reports of missing, detached or loose rubber caps on the needleless syringe filled with diluent liquid by a third-party contract manufacturer and packaged with the vials of ENBREL. In addition in August 2008, we voluntary recalled two manufacturing lots of EPOGEN® and our licensee, Ortho Biotech, voluntarily recalled one manufacturing lot of PROCRIT® (Epoetin alfa) that was manufactured in our manufacturing facilities after having identified cracks in the necks of a small number of vials upon post-manufacturing inspection. Although there have been no observable adverse event trends associated with the Neulasta® SureClick[™] pre-filled pen, with the reports of missing, detached or loose rubber caps on the needleless syringe packaged with the ENBREL vials or with the cracks in the neck of vials of Epoetin alfa, we may experience the same or other problems in the future resulting in broader product recalls or adverse event trends. Additionally, if other parties (including our licensees, such as J&J and Wyeth, or independent investigators) report or fail to effectively report to regulatory agencies side effects or other safety concerns that occur from their use of our products in clinical trials or studies or from marketed use, regulatory approval may be withdrawn for a product for the therapeutic area in question, or completely, or other risk management activities may be required by regulators.

If regulatory authorities determine that we or our licensees or partners conducting R&D activities on our behalf have not complied with regulations in the R&D of a product candidate, new indication for an existing product or information to support a current indication, then they may not approve the product candidate or new indication or maintain approval of the current indication in its current form or at all, and we will not be able to market and sell it. If we were unable to market and sell our products or product candidates, our business and results of operations would be materially and adversely affected. Additionally, safety signals or adverse events or results from clinical trials or studies performed by us or by others (including our licensees or independent investigators) from the marketed use of our drugs that resulted in revised safety-related labeling or restrictions on the use of our approved products could negatively impact healthcare provider prescribing behavior, use of our products, regulatory or private health organization medical guidelines and reimbursement for our products all of which would have a material adverse effect on our business and results of operations. (See "— *Our sales depend on payment and reimbursement from third-party payers, and, to the extent that reimbursement for our products is reduced, this could negatively impact the utilization of our products.*" and "— *Guidelines and recommendations published by various organizations can reduce the use of our products.*")

Recent labeling changes or risk management activities required by regulatory authorities, as well as the results or meta-analyses of clinical trials, may adversely impact the use, sales and reimbursement of our ESAs.

On March 9, 2007, based upon data from our AoC 103 Study, J&J's Correction of Hemoglobin and Outcomes in Renal Insufficiency ("CHOIR") study, and preliminary data from the third-party investigator Danish Head and Neck Cancer ("DAHANCA") 10 Study, among others, the FDA approved updated safety information, including a boxed warning, in the labeling for the class of ESAs, including Aranesp® and EPOGEN®. On May 10, 2007, the ODAC held a panel meeting to discuss the safety/efficacy profile of ESA use in oncology. Responding to questions posed by the FDA, the ODAC recommended that more restrictions be added to ESA labeling and that additional clinical trials be conducted by companies with currently approved ESAs, including us, although no specific restrictions or studies were recommended at the ODAC meeting. The committee is advisory and FDA officials are not bound to or limited by its recommendations although, the FDA has commonly followed the recommendations of its advisory panels. The FDA also held a joint meeting of the CRDAC and the DSaRMAC on September 11, 2007, which evaluated the safety data on ESA use in renal disease. On November 8, 2007, in recognition of the input from the May 2007 ODAC and September 2007 joint

CRDAC/DSaRMAC meetings, we announced additional updates to the Aranesp[®] and EPOGEN[®]/PROCRIT[®] labeling which reflected ongoing interactions with the FDA regarding the safety and benefit/risk profile of ESAs and included modifications to the boxed warnings of the ESA labeling. Additionally, based on safety data from the PREPARE interim study results in neo-adjuvant breast cancer and the data from the GOG-191 study in cervical cancer, on March 7, 2008, we announced that the FDA approved updated safety information, including the boxed warning in the labeling information for the class of ESAs, including Aranesp[®] and EPOGEN[®]. On March 13, 2008, the FDA held a follow-up ODAC panel meeting to discuss cumulative data, including recent study results, on the risks of ESAs when used in the oncology setting.

On July 30, 2008, we received a complete response letter from the FDA to the revisions to the ESA labeling we proposed following the March 13, 2008 ODAC meeting. The letter included, among other things, (i) the addition to the boxed warning of a statement that ESAs are not indicated for patients receiving myelosuppressive therapy when the anticipated outcome of such therapy is cure, (ii) the addition of a statement in the DOSAGE and ADMINISTRATION section of the label that ESA therapy should not be initiated at Hb levels ³ 10 g/dL and that dose should be adjusted to maintain the lowest Hb level sufficient to avoid red blood cell transfusions and (iii) the removal of reference to the upper safety limit of 12 g/dL. We finalized the ESA labeling on August 6, 2008, as the FDA directed. Although we cannot predict what impact the final ESA labels would have on our business, the final ESA labeling could have a material adverse impact on the reimbursement, use and sales of our ESA products, which would have a material adverse effect on our business and results of operations.

Additionally, we continue to work closely with the FDA to develop a REMS program for Aranesp[®] in oncology under authority prescribed by the FDAAA. We have submitted a proposed REMS responsive to the FDA's requests, although we cannot predict what risk management activities the FDA may require of us. A REMS program for Aranesp[®] could have a material adverse impact on the reimbursement, use and sales of our ESA products, which would have a material adverse effect on our business and results of operations. (See "— Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market" and "— Our sales depend on payment and reimbursement from third-party payers, and, to the extent that reimbursement for our products is reduced, this could negatively impact the utilization of our products.") We also continue to work with the FDA to finalize protocols for clinical trials to determine the effects of Aranesp[®] on survival and tumor outcomes. The addition of these clinical trials to our pharmacovigilance program and any additional clinical trials required by the FDA could result in substantial additional expense, and their outcomes could result in additional label restrictions or the loss of regulatory approval for an approved indication, each of which may have a material adverse effect on our business and results of operations. (See "— Before we commercialize and sell any of our product candidates or existing products for new indications, we must conduct clinical trials in humans; if we fail to adequately manage these trials we may not be able to sell future products and our sales could be adversely affected.")

Further on March 5, 2008, we announced that the European Commission reached its decision to amend the product labeling for the class of ESAs, including Aranesp[®], based on the positive opinion from the CHMP in January 2008, which was consistent with the EMEA's October 23, 2007 press release stipulating a uniform target Hb range for all ESAs of 10 g/dL to 12 g/dL with guidance to avoid sustained Hb levels above 12 g/dL. Following the March 13, 2008 ODAC, we have continued to share additional ESA safety data with the EMEA as it has become available. On May 15, 2008, we and other ESA marketing authorization holders participated in a closed meeting of the SAG-O. The marketing authorization holders were asked to provide an overview on studies that have been initiated or conducted since July 2007, as well as any other new data that can help to elucidate recent issues on the impact of ESAs on tumor progression and survival in cancer patients. These data included previously disclosed interim results from the PREPARE study in neo-adjuvant breast cancer therapy; follow-up data from the GOG-191 study in cervical cancer, which were published in the February 2008 issue of Gynecologic Oncology; and the February 2008 meta-analysis by Bennett et al, which was published in the Journal of the American Medical Association. SAGs are established by the EMEA to deliver answers, on a

consultative basis, to specific questions addressed to them by the CHMP. On June 26, 2008 the EMEA, based upon the CHMP's opinion which took into account the position expressed by the SAG-O, recommended updating the product information for ESAs with a new warning for their use in cancer patients. In July 2008, the EMEA requested that further clarity around the product information be provided by regulatory agencies in each European Member State country through the publication of a Dear Healthcare Professional Communication, following which we followed the necessary regulatory procedure to update the Aranesp® product information. In October 2008, we received notification that the Aranesp® product information update was approved by the European Commission. The product information for all ESAs was updated to advise that in some clinical situations blood transfusions should be the preferred treatment for the management of anemia in patients with cancer and that the decision to administer ESAs should be based on the benefit-risk assessment with the participation of the individual patient, which should take into account the specific clinical context and that factors that should be considered in the assessment should include the type of tumor and its stage, the degree of anemia, life-expectancy, the environment in which the patient is being treated and patient preference. Although we cannot predict what impact the final EU ESA product information will have on our business, the reimbursement, use and sales of Aranesp® in Europe could be materially adversely affected, which would have a material adverse effect on our business and results of operations.

Further, we continue to receive results from meta-analyses or previously initiated clinical trials using ESAs. For example, on September 30, 2008, we announced that we had received a summary of preliminary results from the Cochrane Collaboration's independent meta-analysis of patient-level data from previously conducted, randomized, controlled, clinical studies evaluating ESAs in cancer patients which we submitted to the FDA and EMEA. The preliminary summary includes four components: on-study deaths and overall survival in cancer patients regardless of their specific cancer treatment (chemotherapy, radiochemotherapy, radiotherapy, anemia of cancer with no treatment, other), and on-study deaths and overall survival in patients receiving chemotherapy (the only oncologic population for which ESA treatment is indicated in current FDA-approved labeling). The analysis showed that ESAs increased on-study deaths and decreased overall survival compared to controls. Although neither of these results were statistically significant, they do not exclude the potential for adverse outcomes when ESAs are used according to the current labeling. We expect to receive the complete meta-analysis results later this year and will provide this information to regulatory authorities at that time. Additionally, our Trial to Reduce cardiovascular Events with Aranesp Therapy ("TREAT"), a large 4,000 patient multi-center, randomized, double-blind, controlled phase 3 trial designed to determine the impact of anemia therapy with Aranesp® on mortality and non-fatal cardiovascular events in patients with CKD, anemia and type 2 diabetes, continues to progress. The independent data safety monitoring committee completed on October 30, 2008, a pre-specified, unblinded review of the TREAT data at a point where 80% of the targeted number of fully adjudicated events had been recorded and recommended that the study continue without modification. We also expect the interim data from the ARA-PLUS breast cancer adjuvant chemotherapy study to be presented at the San Antonio Breast Cancer Symposium in December 2008. The ARA-PLUS study is an investigator sponsored study and is part of our Aranesp® pharmacovigilance program. Although we cannot predict the results of meta-analyses or the outcomes of these clinical trials, we cannot exclude the possibility that adverse results could have a material adverse impact on the reimbursement, use and sales of our ESAs which would have a material adverse effect on our business and results of operations.

Before we commercialize and sell any of our product candidates or existing products for new indications, we must conduct clinical trials in humans; if we fail to adequately manage these trials we may not be able to sell future products and our sales could be adversely affected.

Before we can sell any products, we must conduct clinical trials which demonstrate that our product candidates are safe and effective for use in humans for the indications sought or our existing products are safe and effective for use in humans in new indications sought. Additionally, we may be required to conduct additional trials as a condition of the approval of our label or as a result of perceived or existing safety concerns. The results of these clinical trials are used as the basis to obtain regulatory approval from regulatory authorities such as the FDA. Clinical trials are experiments conducted using our product candidates in human patients having the diseases or medical conditions we are trying to address. Conducting clinical trials is a complex, time-consuming and expensive process. We are required to conduct clinical trials using an appropriate number of trial sites and patients to support the product label claims we are seeking or to support our existing label. The length of time, number of trial sites and patients required for clinical trials vary substantially according to the type, complexity, novelty and intended use of the product candidate or the extent of the safety concerns, post-

marketing issues and/or exposure to patients and therefore, we may spend several years and incur substantial expense in completing certain trials. Our ability to complete our clinical trials in a timely fashion depends in large part on a number of key factors including protocol design, regulatory and institutional review board approval, availability of clinical study material and the rate of patient enrollment in clinical trials. Patient enrollment is a function of several factors, including the size and location of the patient population, enrollment criteria and competition with other clinical trials for eligible patients. As such, there may be limited availability of patients who meet the criteria for certain clinical trials. Delays in planned clinical trials can result in increased development costs, delays in regulatory approvals, associated delays in product candidates reaching the market and revisions to existing product labels. In addition, in order to increase the number of patients available for enrollment for our clinical trials, we have and will continue to open clinical sites and enroll patients in a number of new geographic locations where our experience conducting clinical trials is more limited, including Russia, China, India and some Central and South American countries either through utilization of third-party contract clinical trial providers entirely or in combination with local staff. Conducting clinical trials in locations where we have limited experience requires substantial time and resources to identify and understand the unique regulatory environments of individual countries. If we fail to adequately manage the design, execution and regulatory aspects of our large, complex and regulatory diverse clinical trials, our clinical trials and corresponding regulatory approvals may be delayed or we may fail to gain approval for our product candidates altogether or could lose our ability to market existing products in certain therapeutic areas or altogether. If we are unable to market and sell our product candidates or are unable to obtain approvals in the timeframe needed to execute our product strategies, our business and results of operations would be materially adversely affected. Additional information on our clinical trials can be found on our website at (http://www.amgen.com). (This website address is not intended to function as a hyperlink, and the information contained on our website is not intended to be a part of this filing.)

Patients may also suffer adverse medical events or side effects in the course of our, our licensees, partners or independent investigator's clinical trials of our products or product candidates that may delay the clinical program, require additional or longer trials to gain approval, prohibit regulatory approval of our product candidates or additional indications for our currently approved products, or may render the product candidate commercially unfeasible or limit our ability to market existing products in certain therapeutic areas or at all. For example, as a result of observing an increased frequency of cholecystitis (inflammation of the gall bladder) in patients treated with our late-stage product candidate motesanib diphosphate, we delayed our phase 3 trial in first-line non-small cell lung cancer, which was previously expected to begin in the fourth quarter of 2006, until the second half of 2007. Clinical trials must be designed based on the current standard of medical care. However in certain diseases, such as cancer, the standard of care is evolving rapidly. In these diseases, the duration of time needed to complete certain clinical trials may result in the design of such clinical trials being based on an out of date standard of medical care, limiting the utility and application of such trials. Of course, even if we successfully manage our clinical trials, we may not obtain favorable clinical trial results and may not be able to obtain regulatory approval for new product candidates, product label extensions or maintenance of our current labels on this basis. Further, clinical trials conducted by others, including our licensees, partners or independent investigators, may result in unfavorable clinical trials results that may call into question the safety of our products in off-label or on label uses that may result in label restrictions and/or additional trials.

In connection with our efforts to improve our cost structure, we refocused our spending on critical R&D and operational priorities and sought greater efficiencies in how we conduct our business, including optimizing ongoing clinical trials and trial initiation. To the extent future sales are negatively affected as a result of additional regulatory and reimbursement developments or other challenges, we may be required to further adjust our R&D investment plans. Such actions could result in delays in obtaining approval or reductions in the number of indications and market potential of our product candidates.

Our sales depend on payment and reimbursement from third-party payers, and, to the extent that reimbursement for our products is reduced, this could negatively impact the utilization of our products.

Sales of all of our principal products are dependent, in part, on the availability and extent of reimbursement from third-party payers, including governments and private insurance plans. Generally, in Europe and other countries outside the United States, the government sponsored healthcare system is the primary payer of healthcare costs of patients. Governments may regulate access to, prices or reimbursement levels of our products to control costs or to affect levels of use of our products. Worldwide use of our products may be affected by these cost containment pressures and cost shifting from governments and private insurers to healthcare providers or patients in response to ongoing initiatives to reduce or reallocate healthcare expenditures. Further, adverse events or results from clinical trials or studies performed by us or by others or from the marketed use of our drugs may expand the safety information in the labeling for our approved products and may negatively impact worldwide reimbursement for our products. On July 30, 2007, CMS issued its Decision Memorandum and on January 14, 2008, issued changes to its Medicare National Coverage Determinations Manual, effective for claims with dates of service on or after July 30, 2007, with an implementation date of April 7, 2008. A discussion of the Decision Memorandum follows below. (See also "— *Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market." and "— <i>Guidelines and recommendations published by various organizations can reduce the use of our products.*")

Most patients receiving Aranesp[®], Neulasta[®] and NEUPOGEN[®] for approved indications are covered by government and/or private payer healthcare programs. Medicare and Medicaid government healthcare programs' payment policies for drugs and biologicals are subject to various laws and regulations. Beginning in January 1, 2005 under the MMA, in the physician clinic setting and January 1, 2006, in the hospital outpatient and dialysis settings, Aranesp[®], Neulasta[®] and NEUPOGEN[®] have been reimbursed under a Medicare Part B payment methodology that reimburses each product at 106% of its ASP (sometimes referred to as "ASP+6%"). Effective January 1, 2008, Medicare payment in the hospital outpatient setting reimburses each product at 105% of its ASP and CMS has the regulatory authority to further reduce the outpatient hospital payment formula in future years. For example, effective January 1, 2009, CMS, in its Outpatient Prospective Payment System Final Rule for 2009, released on October 30, 2008, set the payment rate in the hospital outpatient setting at ASP+4% for 2009. ASP is calculated by the manufacturer based on a statutorily defined formula and submitted to CMS. A product's ASP is calculated and reported to CMS on a quarterly basis and therefore may change each quarter. The ASP in effect for a given quarter (the "Current Period") is based upon certain historical sales and sales incentive data covering a statutorily defined period of time preceding the Current Period. For example, the ASP based payment rate for Aranesp[®] that will be in effect for the fourth quarter of 2008 will be based in part on certain historical sales and sales incentive data for Aranesp[®] from July 1, 2007 through June 30, 2008. CMS publishes the ASPs for products in advance of the quarter in which they go into effect.

In the United States, dialysis providers are primarily reimbursed for EPOGEN[®] by the federal government through the ESRD Program of Medicare. The ESRD Program reimburses approved providers for 80% of allowed dialysis costs; the remainder is paid by other sources, including patients, state Medicaid programs, private insurance, and to a lesser extent, state kidney patient programs. The ESRD Program reimbursement methodology is established by federal law and is monitored and implemented by CMS. Effective January 1, 2006, the payment mechanism for separately reimbursed dialysis drugs in both free-standing and hospital-based dialysis centers, including EPOGEN[®] and Aranesp[®], is reimbursed by Medicare at ASP+6% using the same payment amounts used in the physician clinic setting. Beginning in the third quarter of 2007, based on its ongoing assessment for payment of Part B drugs, CMS instituted a single payment limit for Epoetin alfa (EPOGEN[®] and PROCRIT[®]) in all provider settings. Although we cannot predict the payment levels of EPOGEN[®] in future quarters or whether Medicare payments for dialysis drugs may be modified by future federal legislation, a decrease in the reimbursement rate for EPOGEN[®] may have a material adverse effect on our business and results of operations. Any changes to the ASP calculations directly affect the Medicare reimbursement for our products administered in the physician clinic setting, dialysis facility and hospital

outpatient setting. These calculations are regularly reviewed for completeness and based on such review, we have revised our reported ASPs to reflect calculation changes both prospectively and retroactively. For example, partially as a result of our methodology changes, our ASP reimbursement rate for EPOGEN[®] was reduced for the third quarter of 2007.

Since April 1, 2006, the Medicare reimbursement for ESAs administered to dialysis patients has been subject to a revised EMP, the Medicare payment review mechanism used by CMS to monitor EPOGEN® and Aranesp® utilization and appropriate hematocrit outcomes of dialysis patients. The EMP was revised, effective January 1, 2008, requiring a 50% reduction in Medicare reimbursement if a patient's Hb is above 13 g/dL for three or more consecutive months. In addition, the revised EMP reduces the monthly dosing limits to 400,000 IUs of EPOGEN®, from 500,000 IUs, and to 1,200 mcgs of Aranesp®, from 1,500 mcgs. The implementation of the revised EMP and ESA labeling changes led to a decline in EPOGEN® sales for the first quarter of 2008 compared to the first quarter of 2007 primarily due to a decline in both overall utilization and as well as average dosing per patient. However, this dose decline subsequently stabilized but may further fluctuate in the future. Further fluctuations in dosing of EPOGEN® as a result of the revised EMP are possible and could have a material adverse effect on the sales of EPOGEN® and our business and results of operations.

Changes resulting from the MMA, which beginning in 2005 lowered reimbursement for our products, could negatively affect product sales of some of our marketed products. However, we believe that our product sales for 2005, 2006, 2007 and for the first three quarters of 2008 were not significantly impacted by the reimbursement changes resulting from the MMA. However, additional provisions of the MMA and other regulations or legislation affecting reimbursement that have gone or may go into effect could affect our product sales in the future. For example, on July 15, 2008, the Medicare Improvements for Patients and Providers Act of 2008 became law with a number of Medicare and Medicaid reforms including a broader payment bundle for dialysis services and drugs which will require CMS, beginning in 2011, to establish a bundled Medicare payment rate that includes dialysis services and drug/labs that are currently separately billed. The new bundled rate will include dialysis services ourder the current composite rate, all ESAs and other intravenous injectable drugs and oral equivalent forms used in dialysis. The bundled reimbursement rate will be phased in over a four-year period in equal increments starting in 2011. It is possible that providers could elect to move to a full Medicare bundled payment in 2011. CMS will also be required to establish a quality incentive program that begins concurrently with bundling in 2011 which subjects facilities to up to a 2% annual reduction in Medicare reimbursement for failure to meet or exceed CMS quality performance standards, which include anemia management and dialysis adequacy. Bundling initiatives that have been implemented in other healthcare settings have occasionally resulted in lower utilization of services that had not previously been a part of the bundled payment. We are in the process of evaluating the new Medicare legislation on our business and cannot predict the full impact a bundled payments system would have on sales of EPOGEN[®] or Aranesp[®] used in the treatment

In addition, in response to CMS considering and rejecting changes to the ASP calculation methodology for accounting for discounts in multi-product contracts in the 2007 Medicare Physician Fee Schedule Final Rule, MedPAC released its second Congressionally-mandated report on December 29, 2006 on the impact of changes in Medicare payments for Part B Drugs specifically recommending that the Secretary of the Department of Health and Human Services clarify ASP reporting requirements "to ensure that ASP calculations allocate discounts to reflect the transaction price for each drug." Under the ASP system, we allocate our discounts based on the prices paid for individual drugs, according to the terms of its contracts with physicians and other purchasers, and we believe that the resulting ASPs reflect the transaction prices for individual drugs. Referencing a MedPAC December 2006 report, CMS proposed in the Medicare Physician Fee Schedule Proposed Rule for 2008 revising the methodology for calculating ASP to require the reallocation of price concessions of drugs sold under "bundled arrangements," described by CMS in part as an arrangement regardless of physical packaging under which the rebate, discount or other price concession is conditioned upon the purchase of the same drug or biological or other drugs or biologicals or some other performance requirement. In the Medicare Physician Fee Schedule Final Rule for 2008, CMS stated that it was not finalizing the proposed regulatory change at this time, based on comments recommending a delay and raising concerns about the proposal. The agency also clarified that in the absence of specific guidance, manufacturers may continue to make "reasonable

assumptions" in the calculation of ASP, consistent with the general requirements and the intent of the Medicare statute and regulations and their customary business practices. The agency stated that it will continue to monitor this issue and may provide more specific guidance in the future. In the Medicare Physician Fee Schedule Final Rule for 2009 released on October 30, 2008, the agency did not address the topic of bundled price concessions.

Other initiatives reviewing the coverage or reimbursement of our products, including those related to safety, could result in less extensive coverage or lower reimbursement and could negatively affect sales of some of our marketed products. For example, on March 14, 2007, shortly after the March 9, 2007 label changes for all ESAs, CMS announced that the agency had begun reviewing all Medicare policies related to the administration of ESAs in non-renal disease applications as part of a NCA which is generally CMS' first step toward developing a NCD. Generally, a NCD is a national policy statement granting, limiting or excluding Medicare coverage or reimbursement for a specific medical item or service. On July 30, 2007, CMS issued its Decision Memorandum which was substantially altered from the proposed NCD. On January 14, 2008, CMS issued changes to its Medicare NCD Manual, adding the ESA Decision Memorandum, effective for claims with dates of service on and after July 30, 2007 with an implementation date of April 7, 2008. In the Decision Memorandum, CMS determined that ESA treatment was not reasonable and necessary for certain clinical conditions. The Decision Memorandum established the ESA reimbursement policy for Medicare and other government beneficiaries who are treated for CIA with ESAs. We believe that the restrictions in the Decision Memorandum changed the way ESAs are used in clinical practice, for example, by decreasing the number of treated patients, the average ESA dose and the duration of ESA therapy.

We believe this restriction on reimbursement of ESAs in the Decision Memorandum has had and may continue to have a material adverse effect on the use, reimbursement and sales of Aranesp[®], and our business and results of operations. Additionally, based on our knowledge, although no private payers have implemented the Decision Memorandum to date, many private payers have implemented the restrictions included in the Decision Memorandum. Further, we believe many healthcare providers have reduced ESA utilization for all of their patients regardless of insurance coverage. While we cannot fully predict the further impact of the Decision Memorandum on how, or under what circumstances, healthcare providers will prescribe or administer our ESAs, it had a significant impact to our business in 2007 and 2008 and believe that it may continue to impact us in the future.

In addition, the FDA held a joint meeting of the CRDAC and the DSaRMAC on September 11, 2007, which evaluated the safety data on ESA use in renal disease. On July 31, 2008, CMS issued a listing of potential topics for future NCDs as a step to increase transparency in the NCD process and which included as potential topics the use of ESAs in ESRD and CKD. CMS has not announced whether it will proceed to a NCD for ESAs in ESRD or CKD and we cannot predict whether ESAs in the renal setting will be the subject of a future NCD, however, any final NCD for ESAs in the renal setting, which may include non-coverage and/or new dosing and treatment restrictions similar to those proposed in Decision Memorandum for treatment of anemia in oncology with ESAs, would negatively affect use, reduce reimbursement and coverage, negatively affect product sales of our ESA products and may have a material adverse effect on our business and results of operations. In addition, on August 22, 2008 our platelet producer for the treatment of thrombocytopenia in splenectomized (spleen removed) and non-splenectomized adults with chronic ITP, NplateTM, was approved by the FDA and falls within the thrombopoiesis stimulating agents (platelet growth factors) topic that was also included on CMS' July 31, 2008 potential future NCD topic list. We cannot predict whether NplateTM will be the subject of a future NCD.

Further, the DRA included provisions, which are phased in over time, regarding state collection and submission of data for the purpose of collecting Medicaid drug rebates from manufacturers for physician-administered drugs. We expect that state compliance with elements of these provisions that became effective on January 1, 2006, has increased the level of Medicaid rebates paid by us. We continue to evaluate the impact of the DRA and cannot predict what impact the DRA will have on our business.

If, and when, reimbursement rates or availability for our marketed products changes adversely or if we fail to obtain adequate reimbursement for our current or future products, healthcare providers may limit how much or under what circumstances they will prescribe or administer them, which could reduce the use of our products or cause us to reduce the price of our products. This could result in lower product sales, which could have a material adverse effect on us and our results of operations. For example, the use of EPOGEN[®] in the United States in connection with treatment for ESRD is funded primarily by the U.S. federal government. In early 1997, CMS, formerly known as Healthcare Financing Administration ("HCFA"), instituted a reimbursement change for EPOGEN[®], which materially and adversely affected our EPOGEN[®] sales until the policies were revised. In addition, following the update to the ESA labeling and associated revisions in compendia, nearly all Medicare contractors dropped reimbursement for Aranesp[®] for anemia of cancer. (See "— *Guidelines and recommendations published by various organizations can reduce the use of our products.*") Also, we believe the increasing emphasis on cost-containment initiatives in the United States, Europe and other countries has and will continue to put pressure on the price and usage of our products, which may adversely impact product sales. Further, when a new therapeutic product is approved, the governmental and/or private coverage and reimbursement for that product is uncertain and a failure to demonstrate clear clinical and/or comparative value associated with the use of a new therapeutic product as compared to existing therapeutic products or products or products are may result in inadequate or no reimbursement. We cannot predict the availability or amount of reimbursement for our approved products or products or products are and will be affected by government and private payer reimbursement policies. Reduction in reimbursement for our approved could have a mat

If our intellectual property positions are challenged, invalidated, circumvented or expire, or if we fail to prevail in present and future intellectual property litigation, our business could be adversely affected.

The patent positions of pharmaceutical and biotechnology companies can be highly uncertain and often involve complex legal, scientific and factual questions. To date, there has emerged no consistent policy regarding breadth of claims allowed in such companies' patents. Third parties may challenge, invalidate or circumvent our patents and patent applications relating to our products, product candidates and technologies. In addition, our patent positions might not protect us against competitors with similar products or technologies because competing products or technologies may not infringe our patents. For certain of our product candidates, there are third parties who have patents or pending patent applications that they may claim prevent us from commercializing these product candidates in certain territories. Patent disputes are frequent, costly and can preclude or delay commercialization of products. We are currently, and in the future may be, involved in patent litigation. However, a patent dispute or litigation may not discourage a potential violator from bringing the product that is alleged to infringe to market and we may be subject to competition during certain periods of litigation. Further, under the Hatch-Waxman Act, products approved by the FDA under a new drug application ("NDA") may be the subject of patent litigation with generic competitors before the five year period of data exclusivity provided for under the Hatch-Waxman Act has expired and prior to the expiration of the patent term of product. For example, on July 25, 2008, we, NPS Pharmaceuticals and Brigham and Women's Hospital, filed a lawsuit against Teva and Barr for infringement of four Sensipar® patents. The lawsuit is based on ANDA filed by Teva and Barr which seek approval to market generic versions of Sensipar® before expiration of the patents. This lawsuit is described in Note 9, "*Contingencies*" to the Condensed Consolidated Financial Statements. If we lose or settle current or future litigations at certain stages or rentrely, we c

Our success depends in part on our ability to obtain and defend patent rights and other intellectual property rights that are important to the commercialization of our products and product candidates. We have filed applications for a number of patents and have been granted patents or obtained rights relating to erythropoietin, natural and recombinant G-CSF, darbepoetin alfa, pegfilgrastim, etanercept, cinacalcet, panitumumab and our other products and potential products. We market our erythropoietin, recombinant G-

CSF, darbepoetin alfa, pegfilgrastim, etanercept, cinacalcet and panitumumab products as EPOGEN[®] (Epoetin alfa), NEUPOGEN[®] (Filgrastim), Aranesp[®] (darbepoetin alfa), Neulasta[®] (pegfilgrastim), Enbrel[®] (etanercept), Sensipar[®]/Mimpara[®] (cinacalcet) and Vectibix[®] (panitumumab), respectively. With respect to our material patents, we have had a number of G-CSF patent expiries in the United States.

We also have been granted or obtained rights to patents in Europe relating to erythropoietin; G-CSF; pegfilgrastim (pegylated G-CSF); etanercept; two relating to darbepoetin alfa; hyperglycosylated erythropoietic proteins; and cinacalcet. Our principal European patent relating to erythropoietin expired on December 12, 2004 and our principal European patent relating to G-CSF expired on August 22, 2006. As these patents have expired, some companies have and we believe others may receive approval for and market biosimilar (as they are generally known in the EU) and other products to compete with these products in the EU presenting additional competition to our products. (See "— Our marketed products face substantial competition and other companies may discover, develop, acquire or commercialize products before or more successfully than we do.")

We may experience difficulties, delays or unexpected costs and not achieve or maintain anticipated cost savings from our restructuring plan.

As a result of various regulatory and reimbursement developments that began in 2007 and, in particular those affecting our marketed ESA products, on August 15, 2007, we announced a plan to restructure our worldwide operations in order to improve our cost structure while continuing to make significant R&D investments and build the framework for our future growth. As part of the restructuring plan, we reduced staff, made changes to certain capital projects and closed certain production operations. As a result of our restructuring plan, we have reduced costs in 2008. Our ability to maintain these savings is dependent upon various future developments, some of which are beyond our control. If we are unable to maintain all of the resulting savings or benefits to our business or other unforeseen events occur, our business and results of operations may be adversely affected. Further, if we were to experience additional changes to our business or redesign certain processes to achieve increased efficiencies, we may face further restructuring and/or reorganization activities in the future.

In addition, our reduction of staff was completed through a combination of a voluntary transition program and an involuntary reduction in force. In order to be successful and build our framework for future growth, we must continue to execute and deliver on our core business initiatives with fewer human resources and losses of intellectual capital. We must also attract, retain and motivate key employees including highly qualified management, scientific, manufacturing and sales and marketing personnel who are critical to our business. We may not be able to attract, retain or motivate qualified employees in the future and our inability to do so may adversely affect our business.

Guidelines and recommendations published by various organizations can reduce the use of our products.

Government agencies promulgate regulations and guidelines directly applicable to us and to our products. However, professional societies, practice management groups, insurance carriers, physicians, private health/science foundations and organizations involved in various diseases from time to time may also publish guidelines or recommendations to healthcare providers, administrators and payers, and patient communities. Recommendations of government agencies or these other groups/organizations may relate to such matters as usage, dosage, route of administration and use of related therapies and reimbursement of our products by government and private payers. (See "— Our sales depend on payment and reimbursement from third-party payers, and, to the extent that reimbursement for our products is reduced, this could negatively impact the utilization of our products.") Organizations like these have in the past made recommendations about our products. Recommendations or guidelines that are followed by patients and healthcare providers could result in decreased use and/or dosage of our products. Some examples of agency and organizational guidelines include:

 On August 30, 2007, the National Kidney Foundation (the "NKF") distributed to the nephrology community final updated Kidney Disease Outcomes Quality Initiative ("KDOQI") clinical practice guidelines and clinical practice recommendations for anemia in CKD. The NKF's Anemia Work Group conducted an extensive review of results from 26 new and existing randomized controlled trials, comparing the risks and benefits of a range of Hb therapeutic targets in CKD patients. Based on this review, the NKF-KDOQITM Anemia Work Group recommended in their 2007 Update to the NKF-KDOQITM Anemia Management Guidelines that physicians target Hb in the range of 11 g/dL to 12 g/dL, and also stipulated that the target not be above 13 g/dL.

On February 2, 2007, following the reported results from our AoC 103 Study, the USP DI Drug Reference Guides removed Aranesp[®] in the treatment of AoC. Thereafter, Aranesp[®] use in AoC essentially ceased.

Any recommendations or guidelines that result in decreased use, dosage or reimbursement of our products could adversely affect our product sales and operating results materially. In addition, the perception by the investment community or stockholders that such recommendations or guidelines will result in decreased use and dosage of our products could adversely affect the market price for our common stock.

We may not be able to develop commercial products.

We intend to continue to make significant R&D investments. Successful product development in the biotechnology industry is highly uncertain, and very few R&D projects produce a commercial product. Product candidates or new indications for existing products (collectively, "product candidates") that appear promising in the early phases of development, such as in early human clinical trials, may fail to reach the market for a number of reasons, such as:

- the product candidate did not demonstrate acceptable clinical trial results even though it demonstrated positive preclinical trial results
- the product candidate was not effective or more effective than currently available therapies in treating a specified condition or illness
- the product candidate had harmful side effects in humans or animals
- the necessary regulatory bodies, such as the FDA, did not approve our product candidate for an intended use
- the product candidate was not economical for us to manufacture and commercialize
- other parties have or may have proprietary rights to our product candidate, such as patent rights, and will not let us sell it on reasonable terms, or at all
- the product candidate is not cost effective in light of existing therapeutics
- we and certain of our licensees, partners or independent investigators may fail to effectively conduct clinical development or clinical manufacturing activities
- the regulatory pathway to approval for product candidates is uncertain or not well-defined

For example, we announced that after discussions with the FDA we have decided not to file for approval of motesanib diphosphate in refractory thyroid cancer until there is more clarity on what would constitute an appropriate regulatory filing package for that indication. We believe that the safety concerns around our ESAs expressed by the FDA must be addressed to the agency's satisfaction before new indications or expanded labeling of our ESA products will likely be approved.

Further, several of our product candidates have failed or been discontinued at various stages in the product development process, including, but not limited to, Brain Derived Neurotrophic Factor ("BDNF"), Megakaryocyte Growth and Development Factor ("MGDF") and Glial Cell Lined-Derived Neurotrophic Factor ("GDNF"). For example, in 1997, we announced the failure of BDNF for the treatment of amyotrophic lateral sclerosis, or Lou Gehrig's Disease, because the product candidate, when administered by injection, did not produce acceptable clinical results for a specific use after a phase 3 trial, even though BDNF had progressed successfully through preclinical and earlier clinical trials. In addition, in 1998, we discontinued development of MGDF, a novel platelet growth factor, at the phase 3 trial stage after several people in platelet donation trials developed low platelet counts and neutralizing antibodies. Also, in June 2004, we announced that the phase 2 study of GDNF for the treatment of advanced Parkinson's disease did not meet the primary study endpoint upon completion of nine months of the double-blind treatment phase of the study even though a small phase 1 pilot investigator-initiated open-label study over a three year period appeared to result in improvements for advanced Parkinson's disease patients. Subsequently, in the fall of 2004 we discontinued clinical development of GDNF in patients with advanced Parkinson's disease after several patients in the phase 2 study developed neutralizing antibodies and new preclinical data in rhesus monkeys showed that GDNF caused irreversible damage to the area of the brain critical to movement control and coordination. On February 11, 2005, we confirmed our previous decision to halt clinical trials and, as a part of that decision and based on thorough scientific review, we also concluded that we will not provide GDNF to the 48 patients who participated in clinical trials that were terminated in the fall of 2004. Of course, there may be other factors that prevent us from marketing a product. We cannot guarantee we will be able to produce or manufacture commercially successful products. (See "- Difficulties, disruptions or delays in manufacturing or failure to comply with manufacturing regulations may limit supply of our products and limit our product sales."; "--- Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market." and "- Before we commercialize and sell any of our product candidates or existing products for new indications, we must conduct clinical trials in humans; if we fail to adequately manage these trials we may not be able to sell future products and our sales could be adversely affected.")

Our business may be affected by government investigations or litigation.

We and certain of our subsidiaries are involved in legal proceedings relating to various patent matters, government investigations, our business operations, government requests for information and other legal proceedings that arise from time to time in the ordinary course of our business. Matters required to be disclosed by us are set forth in Note 10, "*Contingencies*" to the Consolidated Financial Statements in our 2007 Form 10-K and are updated as required in subsequently filed Form 10-Qs. Litigation is inherently unpredictable, and the outcome can result in excessive verdicts and/or injunctive relief that affects how we operate our business. Consequently, it is possible that we could, in the future, incur judgments or enter into settlements of claims for monetary damages or change the way we operate our business, which could have a material adverse effect on our results of operations, financial position or cash flows.

We have received subpoenas from a number of government entities, including the U.S. Attorney's Offices for the Eastern District of New York and the Western District of Washington, as well as the Attorneys General of New York and New Jersey. The federal subpoenas have been issued pursuant to the Health Insurance Portability and Accountability Act of 1996 (18 U.S.C. 3486), while the Attorneys General subpoenas have been issued pursuant to state specific statutes relating to consumer fraud laws and state false claims acts. In general, the subpoenas request documents relating to the sales and marketing of our products, and our collection and dissemination of information reflecting clinical research as to the safety and efficacy of our ESAs. To the extent it is alleged in a proceeding that we are in violation of the various federal and state laws that govern the sales and marketing of its products, then a decision adverse to our interests could result in federal criminal liability or federal or state civil or administrative liability, and thus could result in substantial financial damages or criminal penalties that could have a material adverse effect on our results of operations, financial position or cash flows in the period in which such liabilities are incurred.

We may be required to defend lawsuits or pay damages for product liability claims.

Product liability is a major risk in testing and marketing biotechnology and pharmaceutical products. We may face substantial product liability exposure in human clinical trials and for products that we sell after regulatory approval. Product liability claims, regardless of their merits, could be costly and divert management's attention, and adversely affect our reputation and the demand for our products. Amgen and Immunex have previously been named as defendants in product liability actions for certain of our products.

Our revenues may fluctuate and our operating results are subject to fluctuations and these fluctuations could cause financial results to be below expectations and our stock price is volatile, which could adversely affect your investment.

Our revenues and operating results may fluctuate from period to period for a number of reasons, some of which we cannot control. For example, primarily as a result of various regulatory and reimbursement developments involving ESA products that began in 2007, our anemia product sales, in particular sales of Aranesp[®], for 2007 were materially adversely impacted. Even a relatively small revenue shortfall may cause financial results for a period to be below our expectations or projections as some of our operating expenses are fixed in the short term and cannot be reduced within a short period of time to offset reductions in revenue. Further, primarily as a result of the various regulatory and reimbursement developments impacting ESA products, on August 15, 2007, we announced a plan to restructure our worldwide operations in order to improve our cost structure. As of September 30, 2008, we have completed a majority of the actions initially included in our restructuring plan and have incurred approximately \$790 million in charges. We have recently identified certain additional initiatives designed to further assist in improving our cost structure. The estimated cost of these additional initiatives is \$50 million to \$100 million. As a result of the actual costs incurred to date and the addition of the recently identified initiatives, the total charges expected to be incurred in connection with our restructuring plan, including implementation costs, is \$850 million to \$925 million. Our operating results have and may continue to fluctuate and be adversely impacted as a result of these restructuring charges. (See "- We may experience difficulties, delays or unexpected costs and not achieve or maintain anticipated cost savings from our restructuring plan.") In addition, in the event that the actual restructuring charges exceed our latest estimate, this may cause our operating results for a period to be below our expectations or projections. As a result of the above or other challenges, including further label revisions to our ESAs, our revenues and operating results and, in turn, our stock price may be subject to significant fluctuations. Changes in credit ratings issued by nationally recognized statistical ratings organizations could adversely affect our cost of financing and have an adverse effect on the market price of our securities. Additionally, our stock price, like that of other biotechnology companies, is volatile. For example, in the fifty-two weeks prior to September 30, 2008, the trading price of our common stock has ranged from a high of \$65.89 per share to a low of \$39.97 per share.

Our revenues, operating results and stock price may be affected by a number of factors, such as:

- adverse developments regarding the safety or efficacy of our products
- changes in the government's or private payers' reimbursement policies, particularly for supportive cancer care products, or prescribing guidelines for our products
- current volatility and disruption of the financial markets
- · evolving medical care in treating cancer requiring less use of supportive cancer care products
- inability to maintain regulatory approval of marketed products or manufacturing facilities
- · actual or anticipated clinical trial results of ours or our licensees, partners or independent investigators
- business development or licensing activities

- product development or other business announcements by us or our competitors
- regulatory matters or actions, such as label changes or risk management activities, including a REMS
- lower than expected demand for our products or a change in product mix either or both of which may result in less than optimal utilization of our manufacturing facilities and the potential to incur excess capacity or impairment charges
- changes in our product contracting and related pricing strategies
- changes in wholesaler buying patterns
- increased competition from new or existing products
- fluctuations in foreign currency exchange rates
- · announcements in the scientific and research community
- intellectual property and legal matters
- actual or anticipated product supply constraints
- broader economic, industry and market trends unrelated to our performance

Of course, there may be other factors that affect our revenues, operating results and stock price in any given period. In addition, if our revenues, earnings or other financial results in any period fail to meet the investment community's expectations, there could be an immediate adverse impact on our stock price.

Current levels of market volatility are unprecedented and adverse capital and credit market conditions may affect our ability to access cost-effective sources of funding and our investment in marketable securities may be subject to market, interest and credit risk that could reduce their value.

The capital and credit markets have been experiencing extreme volatility and disruption which, particularly in the past several weeks, has led to uncertainty and liquidity issues for both borrowers and investors. We currently have sufficient cash to repay our floating rate notes due November 28, 2008 and we anticipate that our business will generate sufficient cash for us to repay the \$1.0 billion of our 4.00% notes due in November 2009. Historically, we have occasionally and opportunistically accessed the capital markets to support certain business activities including acquisitions, in-licensing activities, share repurchases and to refinance existing debt. In the future, we may not be able to obtain capital market financing on similar favorable terms, or at all, which could have a material adverse effect on our business and results of operations.

We have some exposure to financial institutions which have come under pressure as a result of the current credit crisis. For example, we have historically had 16 financial institutions participate in our \$2.5 billion revolving credit facility including a subsidiary of Lehman, which had a \$178 million commitment. Lehman declared bankruptcy on September 15, 2008, and the subsidiary participant in our credit facility subsequently declared bankruptcy on October 5, 2008. Although we have never drawn on our credit facilities and do not currently anticipate any need to do so, we would not anticipate the ability to access this specific commitment provided by Lehman in the future. Additionally, the conversion feature of our 0.125% Convertible Senior Notes due 2011 and our 0.375% Convertible Senior Notes due 2013 are hedged pursuant to transactions entered into with two financial institutions. We have also entered into interest rate swap agreements for certain of our outstanding debt and routinely enter into foreign currency exchange contracts with financial institutions as counterparties. Additional bankruptcies in the financial sector could limit our ability to replace these

transactions on favorable terms, or at all, or to manage the risks inherent in our business which could have a material adverse effect on our business and results of operations.

Additionally, we maintain a significant portfolio of fixed-income based investments disclosed as cash equivalents and marketable securities on our Condensed Consolidated Balance Sheet. The value of our investments may be adversely affected by interest rate fluctuations, downgrades in credit ratings, illiquidity in the capital markets and other factors which may result in other than temporary declines in the value of our investments. Any of these events could cause us to record impairment charges with respect to our investment portfolio or to realize losses on the sale of investments. We seek to mitigate these risks with the help of our investment advisors by generally investing in high quality securities and continuously monitoring the overall risk of our portfolio. To date, we have not realized any material impairments within our investment portfolio.

The volatility of the current financial markets may magnify certain risks that affect our business.

Sales of our principal products are dependent, in part, on the availability and extent of reimbursement from third-party payers, including governments and private insurance plans. (See "— Our sales depend on payment and reimbursement from third-party payers, and, to the extent that reimbursement for our products is reduced, this could negatively impact the utilization of our products.") As a result of the volatility of the current financial markets, our third-party payers may delay or be unable to satisfy their reimbursement obligations. A reduction in the availability or extent of reimbursement from government programs, including Medicare and Medicaid, and/or private payer healthcare programs could have a material adverse affect on the sales of our products, our business and results of operations.

Additionally, we rely upon third-parties for certain parts of our business, including licensees and partners, wholesale distributors of our products, contract clinical trial providers, contract manufacturers and single third-party suppliers. Because of the recent volatility in the financial markets, there may be a disruption or delay in the performance or satisfaction of commitments to us by these third-parties which could have a material adverse affect on our business and results of operations.

We rely on single third-party suppliers for some of our raw materials, medical devices and components; if these third-parties fail to supply these items, we may be unable to supply our products.

Certain raw materials necessary for commercial manufacturing and formulation of our products are provided by single-source unaffiliated third-party suppliers. Also, certain medical devices and components necessary for formulation, fill and finish of our products are provided by single-source unaffiliated third-party suppliers. Certain of these raw materials, medical devices and components are the proprietary products of these unaffiliated third-party suppliers and, in some cases, such proprietary products are specifically cited in our drug application with the FDA so that they must be obtained from that specific sole source and could not be obtained from another supplier unless and until the FDA approved that other supplier. We would be unable to obtain these raw materials, medical devices or components for an indeterminate period of time if these third-party single-source suppliers were to cease or interrupt production or otherwise fail to supply these materials or products to us for any reason, including:

- regulatory requirements or action by the FDA or others
- adverse financial developments at or affecting the supplier
- unexpected demand for or shortage of raw materials, medical devices or components
- labor disputes or shortages, including the effects of an avian or pandemic flu outbreak, or otherwise
- failure to comply with our quality standards which results in quality failures, product contamination and/or recall

These events could adversely affect our ability to satisfy demand for our products, which could adversely affect our product sales and operating results materially. For example, we have experienced shortages in certain components necessary for the formulation, fill and finish of certain of our products in our Puerto Rico facility without impact on our ability to supply these products. However, we may experience these or other shortages in the future resulting in delayed shipments, supply constraints and/or stock-outs of our products.

Also, certain of the raw materials required in the commercial manufacturing and the formulation of our products are sourced from other countries and/or derived from biological sources, including mammalian tissues, bovine serum and human serum albumin ("HSA"). We are also investigating alternatives to certain biological sources and alternative manufacturing processes that do not require the use of certain biologically-sourced raw materials as such raw materials may be subject to contamination and/or recall. Also, some countries in which we market our products may restrict the use of certain biologically derived substances in the manufacture of drugs. A material shortage, contamination, recall and/or restriction of the use of certain biologically derived substances or other raw materials, which may be sourced from other countries, used in the manufacture of our products could adversely impact or disrupt our commercial manufacturing of our products or could result in a mandated withdrawal of our products from the market. This could adversely affect our ability to satisfy demand for our products, which could adversely affect our product sales and operating results materially. Further, any disruptions or delays by us or by third-party suppliers or partners in converting to alternatives to certain biological sources and alternative manufacturing processes or our ability to gain regulatory approval for the alternative materials and manufacturing processes could increase our associated costs or result in the recognition of an impairment in the carrying value of certain related assets, which could have a material and adverse affect on our results of operations.

Difficulties, disruptions or delays in manufacturing or failure to comply with manufacturing regulations may limit supply of our products and limit our product sales.

We currently manufacture and market all our principal products, and we plan to manufacture and market many of our product candidates. Manufacturing biologic human therapeutic products is difficult, complex and highly regulated. (See "— *Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market.") We currently manufacture our products and product candidates at our manufacturing facilities located in Thousand Oaks and Fremont, California; Boulder and Longmont, Colorado; West Greenwich, Rhode Island; Bothell, Washington and Juncos, Puerto Rico. (See "— <i>We manufacture and formulate, fill and finish substantially all our products at our Puerto Rico manufacturing facility; if significant natural disasters or produce or assist in the production of ENBREL, Sensipar®/Mimpara® and NplateTM and plan to use contract manufacturers to produce a number of our late-stage product candidates. (See "— <i>We are dependent on third parties for a significant portion of our bulk supply and the formulation, fill and finish of ENBREL.*") Our ability to adequately and timely manufacture and supply our products is dependent on the uninterrupted and efficient operation of our facilities which is impacted by many manufacturing variables including:

- availability or contamination of raw materials and components used in the manufacturing process, particularly those for which we have no other source or supplier
- facility capacity of our contract manufacturers
- facility contamination by microorganisms or viruses
- · labor disputes or shortages, including the effects of an avian or pandemic flu outbreak
- compliance with regulatory requirements

- changes in forecasts of future demand
- timing and actual number of production runs
- production success rates and bulk drug yields
- timing and outcome of product quality testing

If we have problems in one or more of these or other manufacturing variables, we may experience delayed shipments, supply constraints, stock-outs and/or recalls of our products. For example, in the second quarter of 2002, the prior co-marketers with respect to ENBREL experienced a brief period where no ENBREL was available to fill new patient prescriptions, primarily due to variation in the expected production yield from Boehringer Ingelheim Pharma KG ("BI Pharma"). If we are at any time unable to provide an uninterrupted supply of our products to patients, we may lose patients, physicians may elect to prescribe competing therapeutics instead of our products, and sales of our products will be adversely affected, which could materially and adversely affect our product sales and results of operations.

We manufacture and contract manufacture, price, sell, distribute and market or co-market our products for their approved indications. These activities are subject to extensive regulation by numerous state and federal governmental authorities in the United States, such as the FDA and CMS, as well as in foreign countries, including European countries, Canada, Australia and Japan. Although we have obtained regulatory approval for our marketed products, these products and our manufacturing processes and those of our third-party contract manufacturers must undergo a potentially lengthy FDA or other regulatory approval process and are subject to continued review by the FDA and other regulatory authorities. It can take longer than five years to build and license a new manufacturing plant and it can take longer than three years to qualify and license a new contract manufacturer. In order to maintain supply, mitigate risks associated with the vast majority of our formulation, fill and finish operations being performed in a single facility and to adequately prepare to launch a number of our late-stage product candidates, we must successfully implement a number of manufacturers and maintain a state of regulatory compliance. Key manufacturing projects include: (i) expansion of our existing bulk protein facilities at our Puerto Rico site for the production of our late-stage product candidate denosumab; (ii) construction, qualification and licensure of a new formulation and filling facility at our Puerto Rico site and (iii) expansion of our Fremont, California facility to support future product launches.

If regulatory authorities determine that we or our third-party contract manufacturers or third-party service providers have violated regulations or if they restrict, suspend or revoke our prior approvals, they could prohibit us from manufacturing our products or conducting clinical trials or selling our marketed products until we or our third-party contract manufacturers or third-party service providers comply, or indefinitely. Because our third-party contract manufacturers and third-party service providers are subject to FDA and foreign regulatory authorities, alternative qualified third-party contract manufacturers and service providers may not be available on a timely basis or at all. If we or our third-party contract manufacturers and third-party service providers fail to supply materials, products or services to us for any reason, we may experience delayed shipments, supply constraints, stock-outs and/or recalls of our products. If we are unable to manufacture, market and sell our products, our business and results of operations would be materially and adversely affected.

We manufacture and formulate, fill and finish substantially all our products at our Puerto Rico manufacturing facility; if significant natural disasters or production failures occur at this facility, we may not be able to supply these products.

We currently perform all of the formulation, fill and finish for EPOGEN[®], Aranesp[®], Neulasta[®] and NEUPOGEN[®], some formulation, fill and finish operations for ENBREL, and all of the bulk manufacturing for Aranesp[®], Neulasta[®] and NEUPOGEN[®] at our manufacturing facility in Juncos, Puerto Rico. Our global supply

of these products is significantly dependent on the uninterrupted and efficient operation of this facility. A number of factors could adversely affect our operations, including:

- power failures
- breakdown, failure or substandard performance of equipment
- improper installation or operation of equipment
- · labor disputes or shortages, including the effects of an avian or pandemic flu outbreak
- · inability of third-party suppliers to provide raw materials and components
- natural or other disasters, including hurricanes
- failures to comply with regulatory requirements, including those of the FDA

For example, this facility in Puerto Rico has experienced manufacturing component shortages and there was evidence of adverse trends in the microbial bioburden of the production environment that reduced the production output in the past. Although these experiences in Puerto Rico have not impacted our ability to supply product in the past, the same or other problems may result in our being unable to supply these products, which could adversely affect our product sales and operating results materially. Although we have obtained limited insurance to protect against certain business interruption losses, there can be no assurance that such coverage will be adequate or that such coverage will continue to remain available on acceptable terms, if at all. The extent of the coverage of our insurance could limit our ability to mitigate for lost sales and could result in such losses adversely affecting our product sales and operating results materially. (See "— *Difficulties, disruptions or delays in manufacturing or failure to comply with manufacturing regulations may limit supply of our products and limit our product sales.*")

We are dependent on third parties for a significant portion of our bulk supply and the formulation, fill and finish of ENBREL.

Under a collaboration and global supply agreement, we and Wyeth share the total worldwide bulk supply of ENBREL produced by our Rhode Island manufacturing facility, BI Pharma's manufacturing facility in Germany and Wyeth's manufacturing facility in Ireland. Our ENBREL supply forecasts rely on certain assumptions of how much ENBREL each of these manufacturing facilities is expected to produce. If any of these manufacturing facilities are unable to produce in accordance with our or Wyeth's expectations, the worldwide supply of ENBREL could be adversely affected materially. In such cases, we may be required to allocate supply for Wyeth's benefit. To the extent that there is a shortfall in worldwide production, our supply of ENBREL could be adversely affected. Additionally, the costs associated with a shortfall or failure in production of ENBREL would be borne by both parties.

We currently produce a substantial portion of the annual ENBREL supply at our Rhode Island manufacturing facility. However, we also depend on third parties for a significant portion of our ENBREL bulk supply as well as for some of the formulation, fill and finish of ENBREL that we manufacture. BI Pharma is our third-party contract manufacture of ENBREL bulk drug; accordingly, our U.S. and Canadian supply of ENBREL is currently significantly dependent on BI Pharma's production schedule for ENBREL. We would be unable to produce ENBREL in sufficient quantities to substantially offset shortages in BI Pharma's scheduled production if BI Pharma or other third-party contract manufacturers used for the formulation, fill and finish of ENBREL bulk drug were to cease or interrupt production or services or otherwise fail to supply materials, products or services to us for any reason, including labor shortages or disputes, regulatory requirements or action or contamination of product lots or product recalls. For example, in the second quarter of 2002, the prior co-marketers with respect to ENBREL experienced a brief period where no ENBREL was available to fill new patient prescriptions, primarily due to variation in the expected production yield from BI Pharma. We cannot

guarantee that an alternative third-party contract manufacturer would be available on a timely basis or at all. This in turn could materially reduce our ability to satisfy demand for ENBREL, which could materially and adversely affect our operating results.

Among the factors that could affect our actual supply of ENBREL at any time include, without limitation, BI Pharma's and our Rhode Island facility's bulk drug production scheduling. For example, BI Pharma does not produce ENBREL continuously; rather, it produces the bulk drug substance through a series of periodic campaigns throughout the year. Our Rhode Island manufacturing facility is currently dedicated to ENBREL production. The amount of commercial inventory available to us at any time depends on a variety of factors, including the timing and actual number of BI Pharma's production runs, the actual number of runs at our Rhode Island manufacturing facility, and, for either the Rhode Island or BI Pharma facilities, the level of production yields and success rates, the timing and outcome of product quality testing and the amount of formulation, fill and finish capacity. We are also dependent on third-parties for some formulation, fill and finish of ENBREL bulk drug substance manufactured at our Rhode Island facility. If third-party formulation, fill and finish manufacturers are unable to provide sufficient capacity or are otherwise unable to provide services to us, the supply of ENBREL could be adversely affected materially.

Our marketed products face substantial competition and other companies may discover, develop, acquire or commercialize products before or more successfully than we do.

We operate in a highly competitive environment. Our products compete with other products or treatments for diseases for which our products may be indicated. For example, ENBREL competes in certain circumstances with products marketed by J&J, Abbott, Biogen IDEC Inc., Genentech, Inc., Bristol-Myers Squibb Corporation, Novartis AG and Sanofi-Aventis, as well as the generic drug methotrexate, and may face competition from other potential therapies being developed, including J&J's CNTO 1275 (ustekinumab) and CNTO 148 (golimumab). Additionally, in the first quarter of 2008 Abbott received approval from the FDA to market HUMIRA[®] as a treatment for adult patients with moderate to severe chronic plaque psoriasis and HUMIRA[®] now competes with ENBREL in both the rheumatology and dermatology segments and ENBREL has experienced and continues to experience share loss to competitors. Further, Aranesp® competes with J&J's PROCRIT® in the U.S. in the oncology setting and effective October 1, 2008, Amgen restructured its oncology clinic contracts which may adversely impact sales of Aranesp[®] which could have a material adverse effect on our business and results of operations.

Additionally, Aranesp[®] competes or will potentially compete in the EU with:

Product	Company	Key Countries Launched
EPREX®	J&J	EU
Neorecormon®	Roche	EU
Biosimilar Erythropoietin	Sandoz with co-marketers Hexal and Medice	Austria, Germany, UK, Netherlands, Finland, France, Ireland, Italy
Biosimilar Erythropoietin	Hospira/Stada	Germany, Austria, Greece, Ireland, Netherlands, Sweden, UK
peg-EPO/MIRCERA®	Roche	Across international countries except for Italy, Portugal, Australia

In addition, several companies are developing potentially competing therapies. For example, Affymax Inc./Takeda are co-developing, HematideTM, an erythropoietin mimetic for the treatment of anemia. Further, if our currently marketed products are approved for new uses, or if we sell new products, or our competitors get new or expanded indications, we may face new, additional competition that we do not face today. Further, adverse clinical developments for our current products could limit our ability to compete. (See "— Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market.") Our products may compete against products that have lower prices, equivalent or superior performance, are easier to administer or that are otherwise competitive with our products.

Our principal European patent relating to erythropoietin expired on December 12, 2004 and our principal European patent relating to G-CSF expired on August 22, 2006. As these patents have expired, some companies have and other companies may receive approval for and market biosimilar or other products to compete with our products in the EU, presenting additional competition to our products. For example, on September 15, 2008, the European Commission issued marketing authorizations for the first G-CSF biosimilar products to Ratiopharm's Ratiograstim®/Filgrastim Ratiopharm®, CT Arzneimittel's Biograstim® and Teva's Tevagrastim®. Ratiopharm launched its G-CSF biosimilar product, Ratiograstim®, in the United Kingdom in October 2008, and is expected to launch it in Germany and several other European markets in the fourth quarter of 2008. CT Arzneimittel is expected to market its G-CSF biosimilar product in Germany in the fourth quarter of 2008. Teva stated that it would begin marketing its G-CSF biosimilar product throughout Europe in 2009. These G-CSF biosimilar products would compete with Neulasta® and NEUPOGEN®. We cannot predict to what extent the entry of biosimilar products or other competing products will impact future Aranesp®, Neulasta® or NEUPOGEN® sales in the EU. Our inability to compete effectively could reduce sales which could have a material adverse effect on our results of operations.

In 2006, the EMEA developed and issued final regulatory guidelines related to the development and approval of biosimilar products. The final guidelines included clinical trial guidance for certain biosimilar products including erythropoietins and G-CSFs, which guidance recommends that applicants seeking approval of such biosimilar products conduct fairly extensive pharmacodynamic, toxicological, clinical safety studies and a pharmacovigilance program. In the United States, there currently is no legal approval pathway for the approval of BLAs for biosimilars. A number of events would need to occur before these products could enter the market, including passage of legislation by Congress to create a new approval pathway and, depending on the specific provisions of any such legislation, promulgation of associated regulations or guidance by the FDA. In 2007, several members of Congress expressed interest in the issue, a number of bills were introduced, the House of Representatives and the Senate held hearings on biosimilars, and the Senate Committee on HELP voted on legislation in June 2007. In 2008, additional legislation was introduced in the House of Representatives. To date, however, no final legislation has been considered or passed in either chamber of Congress. Given the continuing interest of Congress in the issue, it is possible but not likely that legislation on biosimilars will be finalized in 2008. It is unknown what type of regulatory framework, what legal provisions, and what timeframes for issuance of regulations or guidance any final legislation would contain. Until such legislation is created, we cannot predict when biosimilars could appear in the United States.

Certain of our competitors, including biotechnology and pharmaceutical companies, market products or are actively engaged in R&D in areas where we have products or where we are developing product candidates or new indications for existing products. In the future, we expect that our products will compete with new drugs currently in development, drugs approved for other indications that may be approved for the same indications as those of our products and drugs approved for other indications that are used off-label. Large pharmaceutical corporations may have greater clinical, research, regulatory, manufacturing, marketing, financial and human resources than we do. In addition, some of our competitors may have technical or competitive advantages over us for the development of technologies and processes. These resources may make it difficult for us to compete with them to successfully discover, develop and market new products and for our current products to compete with new products or new product indications that these competitors may bring to market. Business

combinations among our competitors may also increase competition and the resources available to our competitors.

We must build the framework for our future growth, and if we fail to execute on our initiatives our business could be adversely affected.

As a result of developments in 2007 and, in particular the regulatory and reimbursement changes to our ESA products, on August 15, 2007, we announced a plan to restructure our worldwide operations in order to improve our cost structure while continuing to make significant R&D investments and build the framework for our future growth. We face a number of risks, some of which we cannot completely control. For example:

- we will need to manage complexities associated with a large and geographically diverse organization
- we will need to manage and execute large, complex and global clinical trials
- we will need to significantly expand our sales and marketing resources to launch our late-stage product candidate, denosumab
- we will need to accurately anticipate demand for the products we manufacture and maintain adequate manufacturing capacity for both commercial and clinical supply
- we have implemented a new ERP system to support our increasing complex business and business processes and need to ensure that the new
 system continues to operate without disruptions to our operations

Of course, there may be other risks and we cannot guarantee that we will be able to successfully manage these or other risks. If we fail to execute on our initiatives in these ways or others, such failure could result in a material adverse effect on our business and results of operations.

Concentration of sales at certain of our wholesaler distributors and consolidation of free-standing dialysis clinic businesses may negatively impact our bargaining power and profit margins.

The substantial majority of our U.S. product sales are made to three pharmaceutical product wholesaler distributors, AmerisourceBergen Corporation, Cardinal Health, Inc. and McKesson Corporation. These distributors, in turn, sell our products to their customers, which include physicians or their clinics, dialysis centers, hospitals and pharmacies. One of these products, EPOGEN®, is primarily sold to free-standing dialysis clinics, which have experienced significant consolidation. Two organizations, DaVita Inc. and Fresenius Medical Care North America, Inc. ("Fresenius") own or manage a large number of the outpatient dialysis facilities located in the United States and account for a significant majority of all EPOGEN® sales in the free-standing dialysis clinic setting. In October 2006, we entered into a five-year sole sourcing and supply agreement with an affiliate of Fresenius, on its behalf and on behalf of certain of its affiliates, to purchase, and we have agreed to supply, all of Fresenius' commercial requirements for ESAs for use in managing the anemia of its hemodialysis patients in the United States and Puerto Rico, based on forecasts provided by Fresenius and subject to the terms and conditions of the agreement.

These entities' purchasing leverage has increased due to this concentration and consolidation which may put pressure on our pricing by their potential ability to extract price discounts on our products or fees for other services, correspondingly negatively impacting our bargaining position and profit margins. The results of these developments may have a material adverse effect on our product sales and results of operations.

Our marketing of ENBREL is dependent in part upon Wyeth.

Under a co-promotion agreement, we and Wyeth market and sell ENBREL in the United States and Canada. A management committee comprised of an equal number of representatives from us and Wyeth is responsible for overseeing the marketing and sales of ENBREL including strategic planning, the approval of an annual marketing plan, product pricing and the establishment of a brand team. The brand team, with equal representation from us and Wyeth, prepares and implements the annual marketing plan, which includes a minimum level of financial and sales personnel commitment from each party, and is responsible for all sales activities. If Wyeth fails to effectively deliver on its marketing commitments to us or if we and Wyeth fail to coordinate our efforts effectively, our sales of ENBREL may be adversely affected materially.

Our corporate compliance program cannot guarantee that we are in compliance with all potentially applicable U.S. federal and state regulations and all potentially applicable foreign regulations.

The development, manufacturing, distribution, pricing, sales, marketing and reimbursement of our products, together with our general operations, are subject to extensive federal and state regulation in the United States and to extensive regulation in foreign countries. (See " — Our current products and products in development cannot be sold if we do not gain or maintain regulatory approval of our products and we may be required to perform additional clinical trials or change the labeling of our products or conduct other potentially limiting or costly risk management activities if we or others identify side effects or safety concerns after our products are on the market." and " — Difficulties, disruptions or delays in manufacturing or failure to comply with manufacturing regulations may limit supply of our products and limit our product sales.") While we have developed and instituted a corporate compliance program, we cannot guarantee you that we, our employees, our consultants or our contractors are or will be in compliance with all potentially applicable U.S. federal and state regulations and/or laws or all potentially applicable foreign regulations and/or laws. If we fail to comply with any of these regulations and/or laws, a range of actions could result, including, but not limited to, the termination of clinical trials, the failure to approve a product candidate, restrictions on our products or manufacturing processes, withdrawal of our products from the market, significant fines, exclusion from government healthcare programs or other sanctions or litigation.

Continual process improvement efforts may result in the carrying value of certain existing manufacturing facilities or other assets becoming impaired or other related charges being incurred.

In connection with our continuous process improvement activities, we evaluate our processes and procedures in order to identify opportunities to achieve greater efficiencies in how we conduct our business in order to reduce costs. In particular, we evaluate our manufacturing practices and related processes to increase production yields and/or success rates as well as capacity utilization to gain increased cost efficiencies. Depending on the timing and outcomes of these process improvement initiatives, the carrying value of certain manufacturing or other assets may not be fully recoverable and could result in the recognition of impairment charges and/or the recognition of other related charges. The recognition of such charges, if any, could have a material and adverse affect on our results of operations.

Item 2. UNREGISTERED SALES OF EQUITY SECURITIES, USE OF PROCEEDS AND ISSUER PURCHASES OF EQUITY SECURITIES

As of September 30, 2008, we had one outstanding stock repurchase program. The manner of purchases, the amount we spend and the number of shares repurchased will vary based on a number of factors including the stock price, blackout periods in which we are restricted from repurchasing shares, and our credit rating and may include private block purchases as well as market transactions. Repurchases under our stock repurchase programs reflect, in part, our confidence in the long-term value of our common stock. Additionally, we believe that it is an effective way of returning cash to our stockholders. A summary of our repurchase activity for the three months ended September 30, 2008 is as follows:

			Total number of	Maximum \$ value
	Total number	Average	shares purchased	that may yet be
	of shares	price paid	as part of publicly	purchased under the
	purchased	per share	announced programs	programs ⁽¹⁾
July 1 - July 31	89	\$ 62.07 (2)	-	\$ 4,871,328,709 (2)
August 1 - August 31	8,321	63.12	-	4,871,328,709
September 1 - September 30	759	61.83	-	4,871,328,709
	9,169 (3)	63.00	- (3)	

⁽¹⁾ In July 2007, the Board of Directors authorized us to repurchase up to an additional \$5.0 billion of our common stock.

(2) The total cost of shares repurchased during the three months ended September 30, 2008 includes \$19,060,523 paid in July 2008 in connection with the final settlement of an ASR entered into in May 2008.

⁽³⁾ The difference between total number of shares purchased and the total number of shares purchased as part of publicly announced programs is due to shares of common stock withheld by us for the payment of taxes upon vesting of certain employees' restricted stock.

Item 6. EXHIBITS

(a) Reference is made to the Index to Exhibits included herein.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Quarterly Report to be signed on its behalf by the undersigned, thereunto duly authorized.

Amgen Inc. (Registrant)

Date: November 7, 2008

/s/ ROBERT A. BRADWAY Robert A. Bradway Executive Vice President and Chief Financial Officer

68

By:

AMGEN INC.

INDEX TO EXHIBITS

Exhibit No.	Description
3.1	Restated Certificate of Incorporation (As Restated December 6, 2005). (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
3.2	Certificate of Amendment of the Restated Certificate of Incorporation (As Amended May 24, 2007). (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2007 on August 9, 2007 and incorporated herein by reference.)
3.3	Certificate of Correction of the Restated Certificate of Incorporation (As Corrected May 24, 2007). (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2007 on August 9, 2007 and incorporated herein by reference.)
3.4	Amended and Restated Bylaws of Amgen Inc. (As Amended and Restated February 14, 2007). (Filed as an exhibit to Form 8-K filed on February 20, 2007 and incorporated herein by reference.)
3.5	Amendment to Amended and Restated Bylaws of Amgen Inc. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2007 on August 9, 2007 and incorporated herein by reference.)
4.1	Form of stock certificate for the common stock, par value \$.0001 of the Company. (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 1997 on May 13, 1997 and incorporated herein by reference.)
4.2	Form of Indenture, dated January 1, 1992. (Filed as an exhibit to Form S-3 Registration Statement filed on December 19, 1991 and incorporated herein by reference.)
4.3	Agreement of Resignation, Appointment and Acceptance dated February 15, 2008. (Filed as an exhibit to Form 10-K for the year ended December 31, 2007 on February 28, 2008 and incorporated herein by reference.)
4.4	First Supplemental Indenture, dated February 26, 1997. (Filed as an exhibit to Form 8-K on March 14, 1997 and incorporated herein by reference.)
4.5	8-1/8% Debentures due April 1, 2097. (Filed as an exhibit to Form 8-K filed on April 8, 1997 and incorporated herein by reference.)
4.6	Officer's Certificate, dated as of January 1, 1992, as supplemented by the First Supplemental Indenture, dated as of February 26, 1997, establishing a series of securities entitled "8 1/8% Debentures due April 1, 2097." (Filed as an exhibit to Form 8-K filed on April 8, 1997 and incorporated herein by reference.)
4.7	Form of Liquid Yield Option™ Note due 2032. (Filed as an exhibit to Form 8-K on March 1, 2002 and incorporated herein by reference.)
4.8	Indenture, dated as of March 1, 2002. (Filed as an exhibit to Form 8-K on March 1, 2002 and incorporated herein by reference.)
4.9	First Supplemental Indenture, dated March 2, 2005. (Filed as an exhibit to Form 8-K filed on March 4, 2005 and incorporated herein by reference.)
4.10	Indenture, dated as of August 4, 2003. (Filed as an exhibit to Form S-3 Registration Statement on August 4, 2003 and incorporated herein by reference.)
4.11	Form of 4.00% Senior Note due 2009. (Filed as an exhibit to Form 8-K on November 19, 2004 and incorporated herein by reference.)
4.12	Form of 4.85% Senior Notes due 2014. (Filed as an exhibit to Form 8-K on November 19, 2004 and incorporated herein by reference.)
4.13	Officers' Certificate, dated November 18, 2004, including forms of the 4.00% Senior Notes due 2009 and 4.85% Senior Notes due 2014. (Filed as an exhibit to Form 8-K on November 19, 2004 and incorporated herein by reference.)
4.14	Registration Rights Agreement, dated as of November 18, 2004, among Amgen Inc. and Morgan Stanley & Co. Incorporated and Merrill Lynch.

4.14 Registration Rights Agreement, dated as of November 18, 2004, among Amgen Inc. and Morgan Stanley & Co. Incorporated and Merrill Lynch, Pierce, Fenner & Smith Incorporated. (Filed as an exhibit to Form 8-K on November 19, 2004 and incorporated herein by reference.)

4.15 Form of Zero Coupon Convertible Note due 2032. (Filed as an exhibit to Form 8-K on May 6, 2005 and incorporated herein by reference.)

Exhibit No.	Description
4.16	Indenture, dated as of May 6, 2005. (Filed as an exhibit to Form 8-K on May 6, 2005 and incorporated herein by reference.)
4.17	Indenture, dated as of February 17, 2006 and First Supplemental Indenture, dated as of June 8, 2006 (including form of 0.125% Convertible Senior Note due 2011). (Filed as exhibit to Form 10-Q for the quarter ended June 30, 2006 on August 9, 2006 and incorporated herein by reference.)
4.18	Indenture, dated as of February 17, 2006 and First Supplemental Indenture, dated as of June 8, 2006 (including form of 0.375% Convertible Senior Note due 2013). (Filed as exhibit to Form 10-Q for the quarter ended June 30, 2006 on August 9, 2006 and incorporated herein by reference.)
4.19	Registration Rights Agreement, dated as of February 17, 2006, among Amgen Inc. and Merrill Lynch, Pierce, Fenner & Smith Incorporated, Morgan Stanley & Co. Incorporated, Citigroup Global Markets Inc., JPMorgan Securities Inc., Lehman Brothers Inc., Bear, Stearns & Co. Inc., Credit Suisse Securities (USA) LLC. (Filed as an exhibit to Form 8-K on February 21, 2006 and incorporated herein by reference.)
4.20	Corporate Commercial Paper - Master Note between and among Amgen Inc., as Issuer, Cede & Co., as Nominee of The Depository Trust Company, and Citibank, N.A., as Paying Agent. (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 1998 on May 13, 1998 and incorporated herein by reference.)
4.21	The instruments defining the rights of holders of the long-term debt securities of Abgenix, Inc. and its subsidiaries are omitted pursuant to section (b)(4)(iii)(A) of Item 601 of Regulation S-K. Amgen Inc. hereby agrees to furnish copies of these instruments to the Securities and Exchange Commission upon request.
4.22	Officers' Certificate of Amgen Inc. dated as of May 30, 2007, including forms of the Company's Senior Floating Rate Notes due 2008, 5.85% Senior Notes due 2017 and 6.375% Senior Notes due 2037. (Filed as an exhibit to Form 8-K on May 30, 2007 and incorporated herein by reference.)
4.23	Registration Rights Agreement, dated as of May 30, 2007, among Amgen Inc. and Morgan Stanley & Co. Incorporated, Merrill Lynch, Pierce, Fenner & Smith Incorporated, Barclays Capital Inc., Credit Suisse Securities (USA) LLC, Goldman, Sachs & Co., Citigroup Global Markets Inc., J.P. Morgan Securities Inc. and Lehman Brothers Inc. (Filed as an exhibit to Form 8-K on May 30, 2007 and incorporated herein by reference.)
10.1+*	Amgen Inc. Amended and Restated 1991 Equity Incentive Plan (As Amended and Restated October 1, 2008).
10.2+	Amgen Inc. Amended and Restated Director Equity Incentive Program (As Amended and Restated December 10, 2007) and forms of Stock Option Grant Agreement and Restricted Stock Unit Agreement for the Amgen Inc. Amended and Restated Director Equity Incentive Program. (Filed as an exhibit to Form 10-K for the year ended December 31, 2007 on February 28, 2008 and incorporated herein by reference.)
10.3+*	Amgen Inc. Amended and Restated 1999 Equity Incentive Plan (As Amended and Restated of October 1, 2008).
10.4+*	Amgen Inc. Amended and Restated 1999 Incentive Stock Plan (As Amended and Restated October 1, 2008).
10.5+*	Forms of Stock Option Grant Agreement and Restricted Stock Unit Agreement for the Amgen Inc. Amended and Restated 1991 Equity Incentive Plan, the Amgen Inc. Amended and Restated 1999 Equity Incentive Plan and the Amgen Inc. Amended and Restated 1999 Incentive Stock Plan.
10.6+	Amgen Inc. Amended and Restated Employee Stock Purchase Plan. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2000 on August 1, 2000 and incorporated herein by reference.)
10.7+	First Amendment to the Amgen Inc. Amended and Restated Employee Stock Purchase Plan (As Amended and Restated July 12, 2005). (Filed as an exhibit to Form 8-K on July 14, 2005 and incorporated herein by reference.)
10.8+	Second Amendment to the Amgen Inc. Amended and Restated Employee Stock Purchase Plan (As Amended and Restated July 12, 2005). (Filed as an exhibit to Form 10-K for the year ended December 31, 2007 on February 28, 2008 and incorporated herein by reference.)

10.9+* Amgen Supplemental Retirement Plan (As Amended and Restated January 1, 2009.)

Exhibit No.	Description
10.10+	Amgen Inc. Change of Control Severance Plan. (Filed as an exhibit to Form 10-K for the year ended December 31, 1998 on March 16, 1999 and incorporated herein by reference.)
10.11+	First Amendment to Amgen Inc. Change of Control Severance Plan (As Amended May 10, 2000). (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2000 on August 1, 2000 and incorporated herein by reference.)
10.12+	Second Amendment to the Amgen Inc. Change in Control Severance Plan (As Amended October 16, 2001). (Filed as an exhibit to Form 10-Q for the quarter ended September 30, 2001 on October 26, 2001 and incorporated herein by reference.)
10.13+	Third Amendment to the Amgen Inc. Change of Control Severance Plan (As Amended January 1, 2004). (Filed as an exhibit to Form 10-K for the year ended December 31, 2004 on March 9, 2005 and incorporated herein by reference.)
10.14+	Fourth Amendment to the Amgen Inc. Change of Control Severance Plan (As Amended June 1, 2004). (Filed as an exhibit to Form 10-K for the year ended December 31, 2004 on March 9, 2005 and incorporated herein by reference.)
10.15+	Fifth Amendment to the Amgen Inc. Change of Control Severance Plan (As Amended December 6, 2004). (Filed as an exhibit to Form 8-K on December 9, 2004 and incorporated herein by reference.)
10.16+	Sixth Amendment to the Amgen Inc. Change of Control Severance Plan (As Amended May 10, 2006). (Filed as an exhibit to Form 8-K on May 16, 2006 and incorporated herein by reference.)
10.17+	Seventh Amendment to the Amgen Inc. Change of Control Severance Plan (As Amended October 4, 2006). (Filed as exhibit to Form 8-K on October 6, 2006 and incorporated herein by reference.)
10.18+	Eighth Amendment to the Amgen Inc. Change of Control Severance Plan (As Amended December 15, 2006). (Filed as an exhibit to Form 10-K for the year ended December 31, 2006 on February 28, 2007 and incorporated herein by reference.)
10.19+*	Amgen Inc. Executive Incentive Plan. (As Amended and Restated January 1, 2009.)
10.20+*	Amgen Inc. Executive Nonqualified Retirement Plan. (As Amended and Restated January 1, 2009.)
10.21+*	Amgen Nonqualified Deferred Compensation Plan (As Amended and Restated effective January 1, 2009.)
10.22+*	Amended and Restated Amgen Inc. Performance Award Program (As Amended and Restated October 1, 2008.)
10.23+	Form of Performance Unit Agreement. (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2008 on May 12, 2008 and incorporated herein by reference.)
10.24+	2002 Special Severance Pay Plan for Amgen Employees. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2002 on August 13, 2002 and incorporated herein by reference.)
10.25+	Agreement, dated March 2, 2001, between Amgen Inc. and Mr. George J. Morrow. (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2001 on May 14, 2001 and incorporated herein by reference.)
10.26+	Agreement, dated March 2, 2001 between Amgen Inc. and Dr. Roger M. Perlmutter, M.D., Ph.D. (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2001 on May 14, 2001 and incorporated herein by reference.)
10.27+	Agreement, dated May 2, 2001, between Amgen Inc. and Mr. Brian McNamee. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2001 on July 27, 2001 and incorporated herein by reference.)
10.28+	Restricted Stock Purchase Agreement, dated March 3, 2003, between Amgen Inc. and Brian M. McNamee. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2003 on July 30, 2003 and incorporated herein by reference.)
10.29+	Agreement, dated May 14, 2001, between Amgen Inc. and Mr. Richard Nanula. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2001 on July 27, 2001 and incorporated herein by reference.)
10.30+	Promissory Note, dated June 27, 2001, of Mr. Richard Nanula. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2001 on July 27, 2001 and incorporated herein by reference.)
10 31+	Amendment to Promissory Note dated August 31, 2007 to Promissory Note dated June 27

10.31-Amendment to Promissory Note, dated August 31, 2007 to Promissory Note, dated June 27,

Exhibit No.			
	2001, of Mr. Richard Nanula. (Filed as an exhibit to Form 10-Q for the quarter ended September 30, 2007 on November 9, 2007 and incorporated		
	herein by reference.)		
10.32+	Agreement, dated February 11, 2004, between Amgen Inc. and David J. Scott. (Filed as an exhibit to Form 10-K for the year ended December 31,		
	2003 on March 11, 2004 and incorporated herein by reference.)		
10.33+	Product License Agreement, dated September 30, 1985, and Technology License Agreement, dated, September 30, 1985 between Amgen and		
	Ortho Pharmaceutical Corporation. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2000 on August 1, 2000 and incorporated		
	herein by reference.)		
10.34+	Shareholders' Agreement, dated May 11, 1984, among Amgen, Kirin Brewery Company, Limited and Kirin-Amgen, Inc. (Filed as an exhibit to		
	Form 10-K for the year ended December 31, 2000 on March 7, 2001 and incorporated herein by reference.)		
10.35+	Amendment No. 1 dated March 19, 1985, Amendment No. 2 dated July 29, 1985 (effective July 1, 1985), and Amendment No. 3, dated		
	December 19, 1985, to the Shareholders' Agreement dated May 11, 1984. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2000		
10.20	on August 1, 2000 and incorporated herein by reference.)		
10.36+	Amendment No. 4 dated October 16, 1986 (effective July 1, 1986), Amendment No. 5 dated December 6, 1986 (effective July 1, 1986),		
	Amendment No. 6 dated June 1, 1987, Amendment No. 7 dated July 17, 1987 (effective April 1, 1987), Amendment No. 8 dated May 28, 1993		
	(effective November 13, 1990), Amendment No. 9 dated December 9, 1994 (effective June 14, 1994), Amendment No. 10 effective March 1, 1996, and Amendment No. 11 effective March 20, 2000 to the Shareholders' Agreement, dated May 11, 1984. (Filed as exhibits to Form 10-K for		
	the year ended December 31, 2000 on March 7, 2001 and incorporated herein by reference.)		
10.37+	Amendment No. 12 to the Shareholders' Agreement, dated January 31, 2001. (Filed as an exhibit to Form 10-Q for the quarter ended June 30,		
10.57	2005 on August 8, 2005 and incorporated herein by reference.)		
10.38+	Amendment No. 13 to the Shareholders' Agreement, dated June 28, 2007 (with certain confidential information deleted therefrom). (Filed as an		
10.00	exhibit to Form 10-Q for the quarter ended June 30, 2007 on August 9, 2007 and incorporated herein by reference.)		
10.39+	Product License Agreement, dated September 30, 1985, and Technology License Agreement, dated September 30, 1985, between Kirin-Amgen,		
	Inc. and Ortho Pharmaceutical Corporation. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2000 on August 1, 2000 and		
	incorporated herein by reference.)		
10.40 +	Research, Development Technology Disclosure and License Agreement: PPO, dated January 20, 1986, by and between Kirin Brewery Co., Ltd.		
	and Amgen Inc. (Filed as an exhibit to Amendment No. 1 to Form S-1 Registration Statement on March 11, 1986 and incorporated herein by		
	reference.)		
10.41+	Amendment Agreement, dated June 30, 1988, to Research, Development, Technology Disclosure and License Agreement: GM-CSF dated		
	March 31, 1987, between Kirin Brewery Company, Limited and Amgen Inc. (Filed as an exhibit to Form 8 amending the Quarterly Report on		
	Form 10-Q for the quarter ended June 30, 1988 on August 25, 1988 and incorporated herein by reference.)		
10.42+	Assignment and License Agreement, dated October 16, 1986 (effective July 1, 1986, between Amgen and Kirin-Amgen, Inc. (Filed as an exhibit		
	to Form 10-K for the year ended December 31, 2000 on March 7, 2001 and incorporated herein by reference.)		
10.43+	G-CSF United States License Agreement, dated June 1, 1987 (effective July 1, 1986), Amendment No. 1, dated October 20, 1988, and		
	Amendment No. 2, dated October 17, 1991 (effective November 13, 1990), between Kirin-Amgen, Inc. and Amgen Inc. (Filed as exhibits to		

Form 10-K for the year ended December 31, 2000 on March 7, 2001 and incorporated herein by reference.)
 G-CSF European License Agreement, dated December 30, 1986, between Kirin-Amgen and Amgen, Amendment No. 1 to Kirin-Amgen, Inc. / Amgen G-CSF European License Agreement, dated June 1, 1987, Amendment No. 2 to Kirin-Amgen, Inc. / Amgen G-CSF European License Agreement, dated March 15, 1998, Amendment No. 3 to Kirin-Amgen, Inc. / Amgen G-CSF European License Agreement, dated October 20,

72

1988, and Amendment No. 4 to Kirin-Amgen,

Exhibit No.	t No. Description		
	Inc. / Amgen G-CSF European License Agreement, dated December 29, 1989, between Kirin-Amgen, Inc. and Amgen Inc. (Filed as exhibits to		
	Form 10-K for the year ended December 31, 2000 on March 7, 2001 and incorporated herein by reference.)		
10.45	Enbrel® Supply Agreement among Immunex Corporation, American Home Products Corporation and Boehringer Ingelheim Pharma KG, dated		
	as of November 5, 1998 (with certain confidential information deleted therefrom). (Filed as an exhibit to the Immunex Corporation Annual		
	Report on Form 10-K for the year ended December 31, 1998 on March 23, 1998 and incorporated herein by reference.)		
10.46	Amendment No. 1 to the Enbrel [®] Supply Agreement, dated June 27, 2000, among Immunex Corporation, American Home Products Corporation		
	and Boehringer Ingelheim Pharma KG, (with certain confidential information deleted therefrom). (Filed as an exhibit to the Immunex		
	Corporation Form 10-Q for the quarter ended June 30, 2000 on August 11, 2000 and incorporated herein by reference.)		
10.47	Amendment No. 2 to the Enbrel [®] Supply Agreement, dated June 3, 2002, among Immunex Corporation, Wyeth (formerly known as American		
	Home Products Corporation) and Boehringer Ingelheim Pharma KG (with certain confidential information deleted therefrom). (Filed as an		
	exhibit to Form 10-Q for the quarter ended June 30, 2002 on August 13, 2002 and incorporated herein by reference.)		
10.48	Amendment No. 1 to Amendment No. 2 to the Enbrel® Supply Agreement, dated June 23, 2008, among Immunex Corporation, Wyeth (formerly		
	"American Home Products Corporation") and Boehringer Ingelheim Pharma KG (Filed as an exhibit to Form 10-Q for the quarter ended June 30,		
10.10	2008 on August 8, 2008 and incorporated herein by reference.)		
10.49	Amendment No. 3 to the Enbrel® Supply Agreement, dated December 18, 2002, among Immunex Corporation, Wyeth (formerly, "American		
	Home Products Corporation") and Boehringer Ingelheim Pharma KG (with certain confidential information deleted therefrom). (Filed as an archibitete Form 10 K for the sum and a Descendar 21, 2002 or March 10, 2002 and in sum and the sum formation being formed as a sum of the s		
10.50	exhibit to Form 10-K for the year ended December 31, 2002 on March 10, 2003 and incorporated herein by reference.)		
10.50	Amendment No. 4 to the Enbrel [®] Supply Agreement, dated May 21, 2004, among Immunex Corporation, Wyeth (formerly, "American Home Products Corporation") and Boehringer Ingelheim Pharma KG. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2005 on August		
	8, 2005 and incorporated herein by reference.)		
10.51	Amendment No. 5 to the Enbrel [®] Supply Agreement, dated August 30, 2005, among Immunex Corporation, Wyeth (formerly, "American Home		
10.51	Products Corporation") and Boehringer Ingelheim Pharma KG. (Filed as an exhibit to Form 10-Q for the quarter ended September 30, 2005 on		
	November 9, 2005 and incorporated herein by reference.)		
10.52	Amendment No. 6 to the Enbrel [®] Supply Agreement, dated November 27, 2007, among Immunex Corporation, Wyeth (formerly, "American		
	Home Products Corporation") and Boehringer Ingelheim Pharma KG (with certain confidential information deleted therefrom) (Filed as an		
	exhibit to Form 10-K for the year ended December 31, 2007 on February 28, 2008 and incorporated herein by reference.)		
10.53*	Amendment No. 2 to Amendment No. 6, dated August 26, 2008, to the Enbrel® Supply Agreement, dated November 27, 2007, among Immunex		
	Corporation, Wyeth (formerly, "American Home Products Corporation") and Boehringer Ingelheim Pharma KG (with certain confidential		
	information deleted therefrom).		
10.54	Agreement Regarding Governance and Commercial Matters, dated December 16, 2001, by and among American Home Products Corporation,		
	American Cyanamid Company and Amgen Inc. (with certain confidential information deleted therefrom). (Filed as an exhibit to Amendment No.		
	1 to Form S-4 Registration Statement on March 22, 2002 and incorporated herein by reference.)		
10.55	Amended and Restated Promotion Agreement, dated as of December 16, 2001, by and among Immunex Corporation, American Home Products		
	Corporation and Amgen Inc. (with certain confidential information deleted therefrom). (Filed as an exhibit to Amendment No. 1 to Form S-4		
10 - 0	Registration Statement on March 22, 2002 and incorporated herein by reference.)		

10.56 Description of Amendment No. 1 to Amended and Restated Promotion Agreement, effective as

Exhibit No.	Description
	of July 8, 2003, among Wyeth, Amgen Inc. and Immunex Corporation, (with certain confidential information deleted therefrom). (Filed as an exhibit to Form 10-K for the year ended December 31, 2003 on March 11, 2004 and incorporated herein by reference.)
10.57	Description of Amendment No. 2 to Amended and Restated Promotion Agreement, effective as of April 20, 2004, by and among Wyeth, Amgen Inc. and Immunex Corporation. (Filed as an exhibit to Form S-4/A on June 29, 2004 and incorporated herein by reference.)
10.58	Amendment No. 3 to Amended and Restated Promotion Agreement, effective as of January 1, 2005, by and among Wyeth, Amgen Inc. and Immunex Corporation (with certain confidential information deleted therefrom). (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2005 on May 4, 2005 and incorporated herein by reference.)
10.59	Purchase Agreement, dated as of November 15, 2004, among Amgen Inc. and Morgan Stanley & Co. Incorporated and Merrill Lynch, Pierce, Fenner & Smith Incorporated, as representatives of the several initial purchasers. (Filed as an exhibit to Form 8-K on November 19, 2004 and incorporated herein by reference.)
10.60	Purchase Agreement, dated as of February 14, 2006, among Amgen Inc., Merrill Lynch, Pierce, Fenner & Smith Incorporated, Morgan Stanley & Co. Incorporated, Citigroup Global Markets Inc., JP Morgan Securities, Inc., Lehman Brothers Inc, Bear, Stearns & Co. Inc., Credit Suisse Securities (USA) LLC. (Filed as an exhibit to Form 8-K on February 21, 2006 and incorporated herein by reference.)
10.61	Confirmation of OTC Convertible Note Hedge related to 2011 Notes, dated February 14, 2006, to Amgen Inc. from Merrill Lynch International related to the 0.125% Convertible Senior Notes Due 2011. (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
10.62	Confirmation of OTC Convertible Note Hedge related to 2013 Notes, dated February 14, 2006, to Amgen Inc. from Merrill Lynch International related to 0.375% Convertible Senior Notes Due 2013. (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
10.63	Confirmation of OTC Convertible Note Hedge related to 2011 Notes, dated February 14, 2006, to Amgen Inc. from Morgan Stanley & Co. International Limited related to the 0.125% Convertible Senior Notes Due 2011 Notes. (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
10.64	Confirmation of OTC Warrant Transaction, dated February 14, 2006, to Amgen Inc. from Merrill Lynch International for warrants expiring in 2011. (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
10.65	Confirmation of OTC Warrant Transaction, dated February 14, 2006, to Amgen Inc. from Merrill Lynch International for warrants expiring in 2013. (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
10.66	Confirmation of OTC Warrant Transaction, dated February 14, 2006, to Amgen Inc. from Morgan Stanley & Co. International Limited for warrants maturing in 2011. (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
10.67	Purchase Agreement, dated February 16, 2006, between Amgen Inc. and Citigroup Global Markets Inc. (Filed as an exhibit to Form 10-K for the year ended December 31, 2005 on March 10, 2006 and incorporated herein by reference.)
10.68	Purchase Agreement, dated May 24, 2007, among Amgen Inc., Morgan Stanley & Co. Incorporated, Merrill Lynch, Pierce, Fenner & Smith Incorporated and the Initial Purchasers Names in Schedule A thereof. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2007 on August 9, 2007 and incorporated herein by reference.)
10.69	Purchase Agreement, dated May 29, 2007, between Amgen Inc. and Merrill Lynch International. (Filed as an exhibit to Form 10-Q for the quarter

- ended June 30, 2007 on August 9, 2007, between Anger Inte, and Merrin Eynen International. (Thed as an exhibit to Form Fo-Q for the quarter ended June 30, 2007 on August 9, 2007 and incorporated herein by reference.)
- 10.70 Collaboration Agreement, dated July 11, 2007, between Amgen Inc. and Daiichi Sankyo Company (with certain confidential information deleted therefrom). (Filed as an exhibit to Form 10-Q for the quarter ended September 30, 2007 on November 9, 2007 and incorporated herein by

Exhibit No.	Description	
	reference.)	
10.71	Credit Agreement, dated November 2, 2007, among Amgen Inc., with Citicorp USA, Inc., as administrative agent, Barclays Bank PLC, as syndication agent, Citigroup Global Markets, Inc. and Barclays Capital, as joint lead arrangers and joint book runners, and the other banks party thereto. (Filed as an exhibit to Form 8-K filed on November 2, 2007 and incorporated herein by reference.)	
10.72	Multi-product License Agreement with Respect to Japan between Amgen Inc. and Takeda Pharmaceutical Company Limited dated February 1, 2008 (with certain confidential information deleted therefrom). (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2008 on May 12, 2008 and incorporated herein by reference.)	
10.73	License Agreement for motesanib diphosphate between Amgen Inc. and Takeda Pharmaceutical Company Limited dated February 1, 2008 (with certain confidential information deleted therefrom). (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2008 on May 12, 2008 and incorporated herein by reference.)	
10.74	Supply Agreement between Amgen Inc. and Takeda Pharmaceutical Company Limited dated February 1, 2008 (with certain confidential information deleted therefrom). (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2008 on May 12, 2008 and incorporated herein by reference.)	
10.75	Sale and Purchase Agreement between Amgen Inc. and Takeda Pharmaceutical Company Limited dated February 1, 2008 (with certain confidential information deleted therefrom). (Filed as an exhibit to Form 10-Q for the quarter ended March 31, 2008 on May 12, 2008 and incorporated herein by reference.)	
10.76	Variable Term Accelerated Share Repurchase Transaction dated May 28, 2008, between Amgen Inc. and Lehman Brothers, Inc. acting as Agent Lehman Brothers OTC Derivatives Inc., acting as Principal. (Filed as an exhibit to Form 10-Q for the quarter ended June 30, 2009 on August 8, 2008 and incorporated herein by reference.)	
31*	Rule 13a-14(a) Certifications.	
32**	Section 1350 Certifications.	

^{(* =} filed herewith)

^{(**=} furnished herewith and not "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended)

^{(+ =} management contract or compensatory plan or arrangement.)

AMGEN INC.

AMENDED AND RESTATED 1991 EQUITY INCENTIVE PLAN

1. <u>PURPOSE</u>.

(a) The purpose of the Amended and Restated 1991 Equity Incentive Plan as amended and restated in December 2005 (the "Plan") is to provide a means by which employees or directors of and consultants to Amgen Inc., a Delaware corporation (the "Company"), and its Affiliates, as defined in paragraph 1(b), directly, or indirectly through Trusts, may be given an opportunity to benefit from increases in value of the stock of the Company through the granting of (i) incentive stock options, (ii) nonqualified stock options, (iii) stock bonuses, and (iv) rights to purchase restricted stock, all as defined below. For purposes of the incentive stock option rules of Section 422 of the Internal Revenue Code of 1986, as amended (the "Code"), the Plan is a new plan.

(b) The word "Affiliate" as used in the Plan means (a) any parent corporation or subsidiary corporation of the Company, as those terms are defined in Sections 424(e) and (f), respectively, of the Internal Revenue Code of 1986, as amended (together with the regulations and official guidance promulgated thereunder, the "Code") and (b) any domestic eligible entity that is disregarded, under Treasury Regulation Section 301.7701-3, as an entity separate from either (i) the Company or (ii) any parent corporation or subsidiary corporation, as those terms are defined in Sections 424(e) and (f), respectively, of the Code.

(c) The Company, by means of the Plan, seeks to retain the services of persons now employed by or serving as directors or consultants to the Company, to secure and retain the services of persons capable of filling such positions, and to provide incentives for such persons to exert maximum efforts for the success of the Company.

(d) The Company intends that the rights issued under the Plan shall, in the discretion of the Board of Directors of the Company (the "Board") or any committee to which responsibility for administration of the Plan has been delegated pursuant to paragraph 2(c), be either (i) stock options granted pursuant to Sections 5 or 6 hereof, including incentive stock options as that term is used in Section 422 of the Code ("Incentive Stock Options"), or options which do not qualify as Incentive Stock Options ("Nonqualified Stock Options") (together hereinafter referred to as "Options"), or (ii) stock bonuses or rights to purchase restricted stock granted pursuant to Section 7 hereof (all such rights included in (i) and (ii), collectively "Stock Awards").

(e) The word "Trust" as used in the Plan shall mean a trust created for the benefit of the employee, director or consultant, his or her spouse, or members of their immediate family. The word optionee shall mean the person to whom the option is granted or the employee, director or consultant for whose benefit the option is granted to a Trust, as the context shall require.

2. ADMINISTRATION.

(a) The Plan shall be administered by the Board unless and until the Board delegates administration to a committee, as provided in paragraph 2(c).

(b) The Board shall have the power, subject to, and within the limitations of, the express provisions of the Plan:

(1) To determine from time to time which of the persons eligible under the Plan shall be granted Stock Awards; when and how Stock Awards shall be granted; whether a Stock Award will be an Incentive Stock Option, a Nonqualified Stock Option, a stock bonus, a right to purchase restricted stock, or a combination of the foregoing; the provisions of each Stock Award granted (which need not be identical), including the time or times when a person shall be permitted to purchase or receive stock pursuant to a Stock Award; and the number of shares with respect to which Stock Awards shall be granted to each such person.

(2) To construe and interpret the Plan and Stock Awards granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any Stock Award, in a manner and to the extent it shall deem necessary or expedient to make the Plan fully effective.

(3) To amend the Plan as provided in Section 14.

(4) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company.

(c) The Board may delegate administration of the Plan to a committee composed of not fewer than two (2) members of the Board (the "Committee"). One or more of these members may be non-employee directors and outside directors, if required and as defined by the provisions of paragraphs 2(e) and 2(f). If administration is delegated to a Committee, the Committee shall have, in connection with the administration of the Plan, the powers theretofore possessed by the Board (except amendment of Section 6 or the options granted thereunder shall only be by action taken by the Board or a committee of one or more members of the Board to which such authority has been specifically delegated by the Board), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. Notwithstanding anything else in this paragraph 2(c) to the contrary, at any time the Board or the Committee may delegate to a committee of one or more members of the Board the authority to grant or amend Stock Awards to all employees, directors or consultants or any portion or class thereof.

(d) Notwithstanding anything else in the Plan to the contrary, at any time the Board or the Committee may authorize by duly adopted resolution one or more Officers (as defined below) (each a "Delegated Officer") to take the actions described in paragraph 2(b)(1) of the Plan with respect to

Options only, subject to, and within the limitations of, the express provisions of the Plan; <u>provided</u>, <u>however</u>, that a Delegated Officer shall not have the power to (1) grant any Options to himself, any non-employee director, consultant, Trust, other Delegated Officer or Officer, (2) determine the time or times when a person shall be permitted to purchase stock pursuant to the exercise of an Option (i.e., vesting), (3) determine the exercise price of an Option, or (4) grant any Option to a parent corporation of the Company, as defined in Section 424(e) of the Code. The resolution authorizing a Delegated Officer to act as such shall specify the total number of shares of Common Stock that a Delegated Officer may grant with respect to Options. The exercise price, which shall be not less than 100% of the closing price of the Common Stock of the Company as quoted on the NASDAQ system on the grant date, or in the Board or the Committee's sole discretion, otherwise determined in accordance with applicable provisions of Code Section 409A (the "Option Fair Market Value") and the time or times when a person shall be permitted to purchase stock pursuant to the exercise of an Option shall, however, be set by the Board or the Committee and not by a Delegated Officer to the extent required by Delaware General Corporation Law Section 157 or any other applicable law. The term "Officer" shall include any natural person who is elected as a corporate officer of the Company by the Board.

(e) The term "non-employee director" shall mean a member of the Board who (i) is not currently an officer of the Company or a parent or subsidiary of the Company (as defined in Rule 16a-1(f) promulgated by the Securities and Exchange Commission under Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) or an employee of the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company of the Company of the Company for services rendered in any capacity other than as a member of the Board (including a consultant) in an amount required to be disclosed to the Company's stockholders under Rule 404 of Regulation S-K promulgated by the Securities and Exchange Commission ("Rule 404"); (iii) does not possess an interest in any other transaction required to be disclosed under Rule 404, as all of these provisions are interpreted by the Securities and Exchange Commission under Rule 16b-3 promulgated under the Exchange Act.

(f) The term "outside director," as used in this Plan, shall mean an administrator of the Plan, whether a member of the Board or of any Committee to which responsibility for administration of the Plan has been delegated pursuant to paragraph 2(c), who is considered to be an "outside director" in accordance with the rules, regulations or interpretations of Section 162(m) of the Code.

(g) Any requirement that an administrator of the Plan be a "non-employee director" or "outside director" shall not apply if the Board or the Committee expressly declares that such requirement shall not apply.

3. SHARES SUBJECT TO THE PLAN.

(a) Subject to the provisions of Section 11 relating to adjustments upon changes in stock, the stock that may be issued pursuant to Stock Awards granted under the Plan shall not exceed in the aggregate One Hundred Ninety-Two Million (192,000,000) shares of the Company's \$.0001 par value common stock (the "Common Stock"). If any Stock Award granted under the Plan shall for any reason expire or otherwise terminate without having been exercised in full, the Common Stock not purchased under such Stock Award shall again become available for the Plan. Shares repurchased by the Company pursuant to any repurchase rights reserved by the Company pursuant to the Plan shall not be available for subsequent issuance under the Plan.

(b) The Common Stock subject to the Plan may be unissued shares or reacquired shares, bought on the market or otherwise.

(c) An Incentive Stock Option may be granted to an eligible person under the Plan only if the aggregate fair market value (determined at the time the Incentive Stock Option is granted) of the Common Stock with respect to which incentive stock options (as defined by the Code) are exercisable for the first time by such optionee during any calendar year under all such plans of the Company and its Affiliates does not exceed one hundred thousand dollars (\$100,000). If it is determined that an entire Option or any portion thereof does not qualify for treatment as an Incentive Stock Option by reason of exceeding such maximum, such Option or the applicable portion shall be considered a Nonqualified Stock Option.

4. <u>ELIGIBILITY</u>.

(a) Incentive Stock Options may be granted only to employees (including officers) of the Company or its Affiliates. A director of the Company shall not be eligible to receive Incentive Stock Options unless such director is also an employee of the Company or any Affiliate. Stock Awards other than Incentive Stock Options may be granted to employees (including officers) or directors of or consultants to the Company or any Affiliate or to Trusts of any such employee, director or consultant.

(b) A director shall in no event be eligible for the benefits of the Plan (other than from a Director NQSO under Section 6 of the Plan) unless and until such director is expressly declared eligible to participate in the Plan by action of the Board or the Committee, and only if, at any time discretion is exercised by the Board or the Committee in the selection of a director as a person to whom Stock Awards may be granted, or in the determination of the number of shares which may be covered by Stock Awards granted to a director, the Plan complies with the requirements of Rule 16b-3 promulgated under the Exchange Act, as from time to time in effect. The Board shall otherwise comply with the requirements of Rule 16b-3 promulgated under the Exchange Act, as from time to time in effect. Notwithstanding the foregoing, the restrictions set forth in this paragraph 4(b) shall not apply if the Board or Committee expressly declares that such restrictions shall not apply.

(c) No person shall be eligible for the grant of an Incentive Stock Option under the Plan if, at the time of grant, such person owns (or is deemed to own pursuant to Section 424(d) of the Code) stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or of any of its Affiliates unless the exercise price of such Incentive Stock Option is at least one hundred and ten percent (110%) of the fair market value of the Common Stock at the date of grant and the Incentive Stock Option is not exercisable after the expiration of five (5) years from the date of grant.

(d) Stock Awards shall be limited to a maximum of 2,000,000 shares of Common Stock per person per calendar year.

5. TERMS OF DISCRETIONARY STOCK OPTIONS.

An option granted pursuant to this Section 5 (a "Discretionary Stock Option") shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate. The provisions of separate Options need not be identical, but each Option shall include (through incorporation of provisions hereof by reference in the Option or otherwise) the substance of each of the following provisions:

(a) No Option shall be exercisable after the expiration of ten (10) years from the date it was granted.

(b) The exercise price of each Incentive Stock Option and each Nonqualified Stock Option shall be not less than one hundred percent (100%) of the Option Fair Market Value of the Common Stock subject to the Option on the date the Option is granted.

(c) The purchase price of Common Stock acquired pursuant to an Option shall be paid, to the extent permitted by applicable statutes and regulations, either: (i) in cash at the time the Option is exercised; or (ii) at the discretion of the Board or the Committee, either at the time of grant or exercise of the Option (A) by delivery to the Company of shares of Common Stock that have been held for the period required to avoid a charge to the Company's reported earnings and valued at the fair market value on the date of exercise, (B) according to a deferred payment or other arrangement with the person to whom the Option is granted or to whom the Option is transferred pursuant to paragraph 5(d), or (C) in any other form of legal consideration that may be acceptable to the Board or the Committee in their discretion; including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price to the Company from the sales proceeds before Common Stock is issued.

In the case of any deferred payment arrangement, interest shall be payable at least annually and shall be charged at not less than the minimum rate of interest necessary to avoid the treatment as

interest, under any applicable provisions of the Code, of any amounts other than amounts stated to be interest under the deferred payment arrangement.

(d) An Option granted to a natural person shall be exercisable during the lifetime of such person only by such person, provided that such person during such person's lifetime may designate a Trust to be such person's beneficiary with respect to any Incentive Stock Options granted after February 25, 1992 and with respect to any Nonqualified Stock Options, and such beneficiary shall, after the death of the person to whom the Option was granted, have all the rights that such person has while living, including the right to exercise the Option. In the absence of such designation, after the death of the person to whom the Option is granted, the Option shall be exercisable by the person or persons to whom the optionee's rights under such Option pass by will or by the laws of descent and distribution.

(e) The total number of shares of Common Stock subject to an Option may, but need not, be allotted in periodic installments (which may, but need not, be equal). From time to time during each of such installment periods, the Option may become exercisable ("vest") with respect to some or all of the shares allotted to that period, and may be exercised with respect to some or all of the shares allotted to such period and/or any prior period as to which the Option was not fully exercised. During the remainder of the term of the Option (if its term extends beyond the end of the installment periods), the Option may be exercised from time to time with respect to any shares then remaining subject to the Option. The provisions of this paragraph 5(e) are subject to any Option provisions governing the minimum number of shares as to which an Option may be exercised.

(f) The Company may require any optionee, or any person to whom an Option is transferred under paragraph 5(d), as a condition of exercising any such Option: (i) to give written assurances satisfactory to the Company as to such person's knowledge and experience in financial and business matters and/or to employ a purchaser representative who has such knowledge and experience in financial and business matters, and that such person is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Option; and (ii) to give written assurances satisfactory to the Company stating that such person is acquiring the Common Stock subject to the Option for such person's own account and not with any present intention of selling or otherwise distributing the Common Stock. These requirements, and any assurances given pursuant to such requirements, shall be inoperative if: (x) the issuance of the shares upon the exercise of the Option has been registered under a then currently effective registration statement under the Securities Act of 1933, as amended (the "Securities Act"); or (y) as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities law.

(g) An Option shall terminate three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate, unless the

Option by its term specifies either (i) that it shall terminate sooner than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate; or (ii) that it may be exercised more than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate. This paragraph 5(g) shall not be construed to extend the term of any Option or to permit anyone to exercise the Option after expiration of its term, nor shall it be construed to increase the number of shares as to which any Option is exercisable from the amount exercisable on the date of termination of the optionee's employment or relationship as a consultant or director.

(h) The Option may, but need not, include a provision whereby the optionee may elect at any time during the term of the optionee's employment or relationship as a consultant or director with the Company or any Affiliate to exercise the Option as to any part or all of the shares subject to the Option prior to the stated vesting dates of the Option. Any shares so purchased from any unvested installment or Option may be subject to a repurchase right in favor of the Company or to any other restriction the Board or the Committee determines to be appropriate.

(i) To the extent provided by the terms of an Option, each optionee may satisfy any federal, state or local tax withholding obligation relating to the exercise of such Option by any of the following means or by a combination of such means: (i) tendering a cash payment; (ii) authorizing the Company to withhold from the shares of the Common Stock otherwise issuable to the optionee as a result of the exercise of the Option a number of shares having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding; or (iii) delivering to the Company owned and unencumbered shares of the Common Stock having a fair market value less than or equal to the amount of the Company's required minimum statutory of the Company's required minimum statutory withholding.

(j) Without in any way limiting the authority of the Board or Committee to make or not to make grants of Discretionary Stock Options under this Section 5, the Board or Committee shall have the authority (but not an obligation) to include as part of any Option agreement a provision entitling the optionee to a further Option (a "Re-Load Option") in the event the optionee exercises the Option evidenced by the Option agreement, in whole or in part, by surrendering other shares of Common Stock in accordance with this Plan and the terms and conditions of the Option agreement. Any such Re-Load Option (i) shall be for a number of shares equal to the number of shares surrendered as part or all of the exercise price of such Option; (ii) shall have an expiration date which is the same as the expiration date of the Option fair Market Value of the Common Stock subject to the Re-Load Option on the date of exercise of the original Option or, in the case of a Re-Load Option which is an Incentive Stock Option and which is granted to a 10% stockholder (as defined in paragraph 4(c)), shall have an exercise price which is equal to one hundred and ten percent (110%) of the Option Fair Market Value of the Option Fair Market Value of the Option Fair Market Value of the Option and which is granted to a 10% stockholder (as defined in paragraph 4(c)), shall have an exercise price which is equal to one hundred percent (110%) of the Option Fair Market Value of the Common Stock subject to the Re-Load Option on the date of exercise of the original Option or, in the case of a Re-Load Option which is an Incentive Stock Option and which is granted to a 10% stockholder (as defined in paragraph 4(c)), shall have an exercise price which is equal to one hundred and

the date of exercise of the original Option.

Any such Re-Load Option may be an Incentive Stock Option or a Nonqualified Stock Option, as the Board or Committee may designate at the time of the grant of the original Option, provided, however, that the designation of any Re-Load Option as an Incentive Stock Option shall be subject to the one hundred thousand dollars (\$100,000) annual limitation on exercisability of Incentive Stock Options described in paragraph 3(c) of the Plan and in Section 422(d) of the Code. There shall be no Re-Load Option on a Re-Load Option. Any such Re-Load Option shall be subject to the availability of sufficient shares under paragraph 3(a) and shall be subject to such other terms and conditions as the Board or Committee may determine.

6. TERMS OF NON-DISCRETIONARY OPTIONS

(a) Prior to December 9, 2003, on January 27 of each year, each person who is at that time an Eligible Director of the Company, (as defined in paragraph 6(k)), shall automatically be granted under the Plan, without further action by the Company, the Board, or the Company's stockholders, a Nonqualified Stock Option (a "Director NQSO") to purchase sixteen thousand (16,000) shares of Common Stock on the terms and conditions set forth herein. An Eligible Director may designate that such Director NQSO be granted in the name of a Trust instead of in the name of such Eligible Director. The Director NQSO shall be on the terms and conditions set forth herein and should the date of grant set forth above be a Saturday, Sunday or legal holiday, such grant shall be made on the next business day. Notwithstanding anything else in the Plan to the contrary, this paragraph 6(a) shall be of no force and effect from and after December 9, 2003.

(b) Prior to December 9, 2003, each person who becomes an Eligible Director, shall, upon the date such person first becomes an Eligible Director, automatically be granted under the Plan, without further action by the Company, the Board, or the Company's stockholders, a Director NQSO to purchase sixty thousand (60,000) shares of Common Stock on the terms and conditions set forth herein. An Eligible Director may designate that such Director NQSO be granted in the name of a Trust instead of in the name of such Eligible Director. The Director NQSO shall be on the terms and conditions set forth herein and should the date of grant set forth above be a Saturday, Sunday or legal holiday, such grant shall be made on the next business day. Notwithstanding anything else in the Plan to the contrary, this paragraph 6(b) shall be of no force and effect from and after December 9, 2003.

(c) Each Director NQSO granted pursuant to this Section 6 (or any Director Re-Load Option granted pursuant to paragraph 6(j)) shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate. The provisions of separate Director NQSO's need not be identical, but each Director NQSO shall include (through incorporation of provisions hereof by reference in the Director NQSO or otherwise) the substance of each of the following provisions as set forth in paragraphs 6(d) through 6(j), inclusive.

(d) The term of each Director NQSO shall be ten (10) years from the date it was granted.

(e) The exercise price of each Director NQSO shall be one hundred percent (100%) of the Option Fair Market Value of the Common Stock subject to such Director NQSO on the date such Director NQSO is granted.

(f) The purchase price of Common Stock acquired pursuant to a Director NQSO shall be paid, to the extent permitted by applicable statutes and regulations, either (i) in cash at the time the Director NQSO is exercised; (ii) by delivery to the Company of shares of Common Stock that have been held for the period required to avoid a charge to the Company's reported earnings and valued at their fair market value on the date of exercise; or (iii) pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instructions to pay the aggregate exercise price to the Company from the sales proceeds before Common Stock is issued.

(g) A Director NQSO shall be exercisable during the lifetime of the Eligible Director with respect to whom it was granted only by the person to whom it was granted (whether the Eligible Director or a Trust), provided that such person during the Eligible Director's lifetime may designate a Trust to be a beneficiary with respect to the Director NQSO, and such beneficiary shall, after the death of the Eligible Director to whom the Director NQSO was granted, have all of the rights designated for such beneficiary. In the absence of such designation, after the death of the Eligible Director with respect to whom the Director NQSO was granted to the Eligible Director, the Director NQSO shall be exercisable by the person or persons to whom the optionee's rights under such option pass by will or by the laws of descent and distribution.

(h) A Director NQSO shall not vest with respect to an Eligible Director, or the affiliate of such Eligible Director, as the case may be, (i) unless the Eligible Director, has, at the date of grant, provided three (3) years of prior continuous service as an Eligible Director, or (ii) until the date upon which such Eligible Director has provided one year of continuous service as an Eligible Director following the date of grant of such Director NQSO, whereupon such Director NQSO shall become fully vested and exercisable in accordance with its terms.

(i) The Company may require any optionee under this Section 6, or any person to whom a Director NQSO is transferred under paragraph 6(g), as a condition of exercising any such option: (i) to give written assurances satisfactory to the Company as to such person's knowledge and experience in financial and business matters and/or to employ a purchaser representative who has such knowledge and experience in financial and business matters, and that such person is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Director NQSO; and (ii) to give written assurances satisfactory to the Company stating that such person is acquiring the Common Stock subject to the Director NQSO for such person's own account and not

with any present intention of selling or otherwise distributing the stock. These requirements, and any assurances given pursuant to such requirements, shall be inoperative if (i) the issuance of the shares upon the exercise of the Director NQSO has been registered under a then currently effective registration statement under the Securities Act of 1933, as amended (the "Securities Act"), or (ii), as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities laws.

(j) Subject to the last sentence of this paragraph 6(j), each Director NQSO shall include a provision entitling the optionee to a further Nonqualified Stock Option (a "Director Re-Load Option") in the event the optionee exercises the Director NQSO evidenced by the Director NQSO grant, in whole or in part, by surrendering other shares of Common Stock in accordance with the Plan and the terms of the Director NQSO grant. Any such Director Re-Load Option (i) shall be for a number of shares equal to the number of shares surrendered as part or all of the exercise price of the original Director NQSO; (ii) shall have an expiration date which is the same as the expiration date of the original Director NQSO; and (iii) shall have an exercise price which is equal to one hundred percent (100%) of the Option Fair Market Value of the Common Stock subject to the Director Re-Load Option on the date of exercise of the original Director Re-Load Option on a Director Re-Load Option. Notwithstanding anything else in the Plan to the contrary, this paragraph 6(j) shall be of no force and effect from and after June 23, 1998.

(k) For purposes of this Section 6, the term "Eligible Director" shall mean a member of the Board who is not an employee of the Company or any Affiliate, and the term "affiliate" shall mean a person that directly or indirectly controls, is controlled by, or is under common control with, the Eligible Director.

7. TERMS OF STOCK BONUSES AND PURCHASES OF RESTRICTED STOCK.

Each stock bonus or restricted stock purchase agreement shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate. The terms and conditions of stock bonus or restricted stock purchase agreements may change from time to time, and the terms and conditions of separate agreements need not be identical, but each stock bonus or restricted stock purchase agreement shall include (through incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions as appropriate:

(a) The purchase price under each stock purchase agreement shall be such amount as the Board or Committee shall determine and designate in such agreement. Notwithstanding the foregoing, the Board or the Committee may determine that eligible participants in the Plan may be awarded stock pursuant to a stock bonus agreement or stock purchase agreement in consideration for

future services to be rendered or past services actually rendered to the Company or for its benefit.

(b) No rights under a stock bonus or restricted stock purchase agreement shall be assignable by any participant under the Plan, either voluntarily or by operation of law, except where such assignment is required by law or expressly authorized by the terms of the applicable stock bonus or restricted stock purchase agreement.

(c) The purchase price of stock acquired pursuant to a stock purchase agreement shall be paid either: (i) in cash at the time of purchase; (ii) at the discretion of the Board or the Committee, according to a deferred payment or other arrangement with the person to whom the Common Stock is sold; or (iii) in any other form of legal consideration that may be acceptable to the Board or the Committee in their discretion; including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price of the Company from the sales proceeds before Common Stock is issued. Notwithstanding the foregoing, the Board or the Committee to which administration of the Plan has been delegated may award Common Stock pursuant to a stock bonus agreement or stock purchase agreement in consideration for future services to be rendered or past services actually rendered to the Company or for its benefit.

(d) Shares of Common Stock sold or awarded under the Plan may, but need not, be subject to a repurchase option in favor of the Company in accordance with a vesting schedule to be determined by the Board or the Committee.

(e) In the event a person ceases to be an employee of or ceases to serve as a director or consultant to the Company or an Affiliate, the Company may repurchase or otherwise reacquire any or all of the shares of Common Stock held by that person which have not vested as of the date of termination under the terms of the stock bonus or restricted stock purchase agreement between the Company and such person.

(f) To the extent provided by the terms of stock bonus or restricted stock purchase agreement, a participant may satisfy any federal, state or local tax withholding obligation relating to the lapsing of a repurchase option in favor of the Company or vesting of a stock bonus or a restricted stock award by any of the following means or by a combination of such means: (i) tendering a cash payment; (ii) authorizing the Company to withhold from the shares of the Common Stock otherwise deliverable to a participant as a result of the lapsing of a repurchase option in favor of the Company or the vesting of a stock bonus or a restricted stock award a number of shares having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding; or (iii) delivering to the Company owned and unencumbered shares of the Common Stock having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding.

8. <u>COVENANTS OF THE COMPANY</u>.

(a) During the terms of the Stock Awards granted under the Plan, the Company shall keep available at all times the number of shares of Common Stock required to satisfy such Stock Awards up to the number of shares of Common Stock authorized under the Plan.

(b) The Company shall seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to issue and sell shares of Common Stock under the Stock Awards granted under the Plan; provided, however, that this undertaking shall not require the Company to register under the Securities Act either the Plan, any Stock Award granted under the Plan or any Common Stock issued or issuable pursuant to any such Stock Award. If, after reasonable efforts, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under the Plan, the Company shall be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Stock Awards unless and until such authority is obtained.

9. USE OF PROCEEDS FROM COMMON STOCK.

Proceeds from the sale of Common Stock pursuant to Stock Awards granted under the Plan shall constitute general funds of the Company.

10. MISCELLANEOUS.

(a) The Board or Committee shall have the power to accelerate the time during which a Stock Award may be exercised or the time during which a Stock Award or any part thereof will vest, notwithstanding the provisions in the Stock Award stating the time during which it may be exercised or the time during which it will vest. Each Discretionary Stock Option providing for vesting pursuant to paragraph 5(e) may also provide that if the employee's employment or a director's or consultant's affiliation with the Company or an Affiliate of the Company is terminated by reason of death or disability, then the vesting schedule of Discretionary Stock Options granted to such employee, director or consultant or to the Trusts of such employee, director or consultant may be accelerated.

(b) Neither an optionee nor any person to whom an Option is transferred under the provisions of the Plan shall be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares subject to such Option unless and until such person has satisfied all requirements for exercise of the Option pursuant to its terms.

(c) Nothing in the Plan or any instrument executed or Stock Award granted pursuant thereto shall confer upon any eligible employee, consultant, director, optionee or holder of Stock Awards under the Plan any right to continue in the employ of the Company or any Affiliate or to continue acting as a consultant or director or shall affect the right of the Company or any Affiliate to terminate the employment or consulting relationship or directorship of any eligible employee,

consultant, director, optionee or holder of Stock Awards under the Plan with or without cause. In the event that a holder of Stock Awards under the Plan is permitted or otherwise entitled to take a leave of absence, the Company shall have the unilateral right to (i) determine whether such leave of absence will be treated as a termination of employment or relationship as consultant or director for purposes hereof, and (ii) suspend or otherwise delay the time or times at which exercisability or vesting would otherwise occur with respect to any outstanding Stock Awards under the Plan.

(d) Notwithstanding any provision of the Plan to the contrary, the Board or the Committee shall have the power to condition the grant or vesting of stock bonuses and rights to purchase restricted stock under the Plan upon the attainment of performance goals, determined by the Board or the Committee in their respective sole discretion, with respect to any one or more of the following business criteria with respect to the Company, any Affiliate, any division, any operating unit or any product line: (i) return on capital, assets or equity, (ii) sales or revenue, (iii) net income, (iv) cash flow, (v) earnings per share, (vi) adjusted earnings or adjusted net income as defined below, (vii) working capital, (vii) total shareholder return, (ix) economic value or (x) product development, research, in-licensing, out-licensing, litigation, human resources, information services, manufacturing, manufacturing capacity, production, inventory, site development, plant, building or facility development, government relations, product market share, mergers, acquisitions or sales of assets or subsidiaries. "Adjusted net income" and "adjusted earnings" shall mean net income or earnings, as the case may be, for the relevant performance period computed in accordance with accounting principles generally accepted in the U.S. which may be adjusted by the Committee, as specified in writing, for such performance period, at the time a performance goal is established for the performance period, for the following: (a) any item of significant gain or loss for the performance period determined to be related to a change in accounting principle as reflected in the Company's audited consolidated financial statements, (b) amortization expenses associated with acquired intangible assets, (c) expenses associated with acquired in-process research and development and (d) any other items of significant income or expense which are determined to be appropriate adjustments and are specified in writing by the Committee at the time the goal is established for the performance period. With respect to any stock bonuses or rights to purchase restricted stock granted to persons who are or who may be "covered employees" within the meaning of Section 162(m) of the Code, the Board or the Committee shall have the power to grant such awards upon terms and conditions that qualify such awards as "qualified performance-based compensation" within the meaning of Section 162(m) of the Code. Stock bonuses and rights to purchase restricted stock made in accordance with this paragraph 10(d) shall contain the terms and conditions of Section 7 above.

11. ADJUSTMENTS UPON CHANGES IN COMMON STOCK.

If any change is made in the Common Stock subject to the Plan, or subject to any Stock Award granted under the Plan (through merger, consolidation, reorganization, recapitalization, stock dividend, dividend in property other than cash, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other transaction not involving the receipt of consideration by the Company), the Plan and outstanding Stock Awards will be appropriately adjusted in the class(es) and maximum number of shares subject to the Plan, the maximum number of shares which may be granted to a participant in a calendar year, the class(es) and number of shares and price per share of stock subject to outstanding Stock Awards, and the number of shares of Common Stock to be granted as provided for in paragraphs 6(a) and 6(b). Such adjustment shall be made by the Board or the Committee, the determination of which shall be final, binding and conclusive. (The conversion of any convertible securities of the Company shall not be treated as a "transaction not involving the receipt of consideration".) The Board or the Committee, in its sole discretion, may accomplish any such adjustment in a manner calculated not to constitute a "modification" of any such Stock Awards (within the meaning of Code Section 409A) that would cause any such Stock Award to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A).

12. CHANGE OF CONTROL

(a) Notwithstanding anything to the contrary in this Plan, in the event of a Change in Control (as hereinafter defined), then, to the extent permitted by applicable law: (i) the time during which Stock Awards become vested shall automatically be accelerated so that the unvested portions of all Stock Awards shall be vested prior to the Change in Control and (ii) the time during which the Options may be exercised shall automatically be accelerated to prior to the Change in Control and (ii) the time during which the Options may be exercised shall automatically be accelerated to prior to the Change in Control. Upon and following the acceleration of the vesting and exercise periods, at the election of the holder of the Stock Award, the Stock Award may be: (x) exercised (with respect to Options) or, if the surviving or acquiring corporation agrees to assume the Stock Awards or substitute similar stock awards, (y) assumed; or (z) replaced with substitute stock awards. Options not exercised, substituted or assumed prior to or upon the Change in Control shall be terminated. The Board or the Committee, in its sole discretion, may cause any such assumption or substitution to be conducted in a manner so as not to constitute an "extension," "renewal" or "modification" (each within the meaning of Code Section 409A) of any such Stock Award that would cause any such Stock Award to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A).

(b) For purposes of the Plan, a "Change of Control" shall be deemed to have occurred at any of the following times:

(i) upon the acquisition (other than from the Company) by any person, entity or "group," within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act (excluding, for

this purpose, the Company or its affiliates, or any employee benefit plan of the Company or its affiliates which acquires beneficial ownership of voting securities of the Company), of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of fifty percent (50%) or more of either the then outstanding shares of Common Stock or the combined voting power of the Company's then outstanding voting securities entitled to vote generally in the election of directors; or

(ii) at the time individuals who, as of April 2, 1991, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board, provided that any person becoming a director subsequent to April 2, 1991, whose election, or nomination for election by the Company's stockholders, was approved by a vote of at least a majority of the directors then comprising the Incumbent Board (other than an election or nomination of an individual whose initial assumption of office is in connection with an actual or threatened election contest relating to the election of the Directors of the Company, as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) shall be, for purposes of the Plan, considered as though such person were a member of the Incumbent Board; or

(iii) immediately prior to the consummation by the Company of a reorganization, merger, consolidation, (in each case, with respect to which persons who were the stockholders of the Company immediately prior to such reorganization, merger or consolidation do not, immediately thereafter, own more than fifty percent (50%) of the combined voting power entitled to vote generally in the election of directors of the reorganized, merged or consolidated company's then outstanding voting securities) or a liquidation or dissolution of the Company or of the sale of all or substantially all of the assets of the Company; or

(iv) the occurrence of any other event which the Incumbent Board in its sole discretion determines constitutes a Change of Control.

13. QUALIFIED DOMESTIC RELATIONS ORDERS

(a) Anything in the Plan to the contrary notwithstanding, the Board or the Committee, in its sole discretion, may determine that rights under Stock Awards may be assigned to an Alternate Payee to the extent that a QDRO so provides, and in the event of such determination, the provisions of this Section 13 shall apply. (The terms "Alternate Payee" and "QDRO" are defined in paragraph 13(c) below.) The assignment of a Stock Award to an Alternate Payee pursuant to a QDRO shall not be treated as having caused a new grant. The transfer of an Incentive Stock Option to an Alternate Payee may, however, cause it to fail to qualify as an Incentive Stock Option. If a Stock Award is assigned to an Alternate Payee, the Alternate Payee generally has the same rights as the grantee under the terms of the Plan; provided however, that (i) the Stock Award shall be subject to the same vesting terms and exercise period as if the Stock Award were still held by the grantee, (ii) an Alternate Payee may not transfer a Stock Award and (iii) an Alternate Payee is ineligible for Re-Load Options described at paragraph 5(j) or

Director Re-Load Options described at paragraph 6(j).

(b) In the event of the Plan administrator's receipt of a domestic relations order or other notice of adverse claim by an Alternate Payee of a grantee of a Stock Award, transfer of the proceeds of the exercise of such Stock Award, whether in the form of cash, stock or other property, may be suspended. Such proceeds shall thereafter be transferred pursuant to the terms of a QDRO or other agreement between the grantee and Alternate Payee. A grantee's ability to exercise a Stock Award may be barred if the Plan administrator receives a court order directing the Plan administrator not to permit exercise.

(c) The word "QDRO" as used in the Plan shall mean a court order (i) that creates or recognizes the right of the spouse, former spouse or child (an "Alternate Payee") of an individual who is granted a Stock Award to an interest in such Stock Award relating to marital property rights or support obligations and (ii) that the administrator of the Plan determines would be a "qualified domestic relations order," as that term is defined in section 414(p) of the Code and section 206(d) of the Employee Retirement Income Security Act ("ERISA"), but for the fact that the Plan is not a plan described in section 3(3) of ERISA.

14. AMENDMENT OF THE PLAN.

(a) The Board at any time, and from time to time, may amend the Plan. However, except as provided in Section 10 relating to adjustments upon changes in the Common Stock, no amendment shall be effective unless approved by the stockholders of the Company within twelve (12) months before or after the adoption of the amendment, where the amendment will:

(i) increase the number of shares reserved for Stock Awards under the Plan;

(ii) modify the requirements as to eligibility for participation in the Plan (to the extent such modification requires stockholder approval in order for the Plan to satisfy the requirements of Section 422(b) of the Code); or

(iii) modify the Plan in any other way if such modification requires stockholder approval in order for the Plan to satisfy the requirements of Section 422(b) of the Code.

(b) The Board may in its sole discretion submit any other amendment to the Plan for stockholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 162(m) of the Code and the regulations promulgated thereunder regarding the exclusion of performance-based compensation from the limit on corporate deductibility of compensation to certain executive officers.

(c) It is expressly contemplated that the Board may amend the Plan in any respect the Board deems necessary or advisable to provide optionees with the maximum benefits provided or to be provided under the provisions of the Code and the regulations promulgated thereunder relating to

employee Incentive Stock Options and/or to bring the Plan and/or Options granted under it into compliance therewith.

(d) Rights and obligations under any Stock Award granted before amendment of the Plan shall not be impaired by any amendment of the Plan, unless: (i) the Company requests the consent of the person to whom the Stock Award was granted; and (ii) such person consents in writing.

(e) Any amendment of the Plan may be accomplished in a manner calculated to cause such amendment not to constitute an "extension," "renewal" or "modification" (each within the meaning of Code Section 409A) of any Stock Awards that would cause such Stock Awards to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A). Notwithstanding the foregoing, if at any time the Board or the Committee determines that any Stock Award may be subject to Code Section 409A, the Board or the Committee shall have the right, in its sole discretion, and without a Participant's prior consent to amend the Plan or any Stock Award as it may determine is necessary or desirable either for the Plan and Stock Awards to be exempt from the application of Section 409A or to satisfy the requirements of Section 409A, including by adding conditions with respect to the vesting and/or the payment of the Stock Awards.

15. TERMINATION OR SUSPENSION OF THE PLAN.

(a) The Board may suspend or terminate the Plan at any time. No Stock Awards may be granted under the Plan while the Plan is suspended or after it is terminated. No Incentive Stock Options may be granted under the Plan after February 22, 2009.

(b) Rights and obligations under any Stock Awards granted while the Plan is in effect shall not be impaired by suspension or termination of the Plan, except with the consent of the person to whom the Stock Award was granted.

16. EFFECTIVE DATE OF PLAN.

The Plan shall become effective as determined by the Board.

17. <u>CODE SECTION 409A</u>.

Except as may be expressly provided with respect to any Stock Award granted under the Plan, the Plan and the Stock Awards are not intended to constitute a "nonqualified deferred compensation plan" within the meaning of Code Section 409A, but rather are intended to be exempt from the application of Code Section 409A. To the extent that the Plan and/or Stock Awards are nevertheless deemed to be subject to Code Section 409A, the Plan and Stock Awards shall be interpreted in accordance with Code Section 409A and Department of Treasury regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the grant of any Stock Award. Notwithstanding any provision of the Plan or any

Stock Award to the contrary, in the event that the Committee determines that any Stock Award may be or become subject to Code Section 409A, the Committee may adopt such amendments to the Plan and the affected Stock Award (as described above) or adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), or take any other actions, that the Committee determines are necessary or appropriate to (a) exempt the Plan and any Stock Award from the application of Code Section 409A and/or preserve the intended tax treatment of the benefits provided with respect to the Stock Award, or (b) comply with the requirements of Code Section 409A.

AMGEN INC.

AMENDED AND RESTATED 1999 EQUITY INCENTIVE PLAN

Amgen Inc. has adopted this Amended and Restated 1999 Equity Incentive Plan (the "Plan"), effective as of December 5, 2005. This Plan amends and restates in its entirety the Amended and Restated 1999 Equity Incentive Plan, as previously amended and restated on July 15, 2002 (the "Restatement Date"), which amended and restated in its entirety the Immunex Corporation 1999 Stock Option Plan, as amended (the "Original Plan").

ARTICLE I.

PROVISIONS APPLICABLE TO OPTIONS GRANTED PRIOR TO RESTATEMENT DATE

The following provisions of this Article I shall govern awards granted under the Plan prior to the Restatement Date:

SECTION 1. PURPOSE.

The purpose of Article I of the Plan is to enhance the long-term stockholder value of Amgen Inc., a Delaware corporation (the "Company"), by offering opportunities to selected employees, officers and directors to participate in the Company's growth and success, and to encourage them to remain in the service of the Company and its Related Corporations (as defined in Article I, Section 2) and to acquire and maintain stock ownership in the Company.

SECTION 2. DEFINITIONS.

For purposes of the Plan, the following terms shall be defined as set forth below:

"Board" means the Board of Directors of the Company.

"Cause" means dishonesty, fraud, misconduct, unauthorized use or disclosure of confidential information or trade secrets, or conviction or confession of a crime punishable by law (except minor violations), in each case as determined by the Plan Administrator, and its determination shall be conclusive and binding.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"Common Stock" means the common stock, par value \$.0001 per share, of the Company.

"Disability," unless otherwise defined by the Plan Administrator, means a mental or physical impairment of the Optionee that is expected to result in death or that has lasted or is expected to last for a continuous period of 12 months or more and that causes the Optionee to be unable, in the opinion of the Company and one independent physician selected by the Company, to perform his or her duties for the Company or a Related Corporation and to be engaged in any substantial gainful activity.

"Effective Date" means the date on which the Plan was adopted by the Board of Directors of Immunex Corporation ("Immunex"), provided that it was approved by Immunex's stockholders at any time within 12 months of such adoption.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Exchange Stock" has the meaning set forth in Article I, Section 11.3.

"Fair Market Value" shall be as established in good faith by the Plan Administrator or (a) if the Common Stock is listed on the Nasdaq National Market, the closing per share sales prices for the Common Stock as reported by the Nasdaq National Market for a single trading day or (b) if the Common Stock is listed on the New York Stock Exchange or the American Stock Exchange, the closing per share sales prices for the Common Stock as such price is officially quoted in the composite tape of transactions on such exchange for a single trading day. If there is no such reported price for the Common Stock for the date in question, then such price on the last preceding date for which such price exists shall be determinative of Fair Market Value.

"Grant Date" means the date on which the Plan Administrator completes the corporate action relating to the grant of an Option and all conditions precedent to the grant have been satisfied, provided that conditions to the exercisability or vesting of Options shall not defer the Grant Date.

"Incentive Stock Option" means an Option to purchase Common Stock granted under Article I, Section 7 with the intention that it qualify as an "incentive stock option" as that term is defined in Section 422 of the Code.

"Nonqualified Stock Option" means an Option to purchase Common Stock granted under Article I, Section 7 other than an Incentive Stock Option.

"Option" means the right to purchase Common Stock granted under Article I, Section 7.

"Optionee" means (a) the person to whom an Option is granted; (b) for an Optionee who has died, the personal representative of the Optionee's estate, the person(s) to whom the Optionee's rights under the Option have passed by will or by the applicable laws of descent and distribution, or the beneficiary designated in accordance with Article I, Section 10; or (c) the person(s) to whom an Option has been transferred in accordance with Article I, Section 10.

"Option Term" has the meaning set forth in Article I, Section 7.3.

"Parent," except as provided in Article I, Section 8.3 in connection with Incentive Stock Options, means any entity, whether now or hereafter existing, that directly or indirectly controls the Company.

"Plan Administrator" means the Board or any committee or committees designated by the Board or any person to whom the Board has delegated authority to administer the Plan under Article I, Section 3.1.

"Related Corporation" means any Parent or Subsidiary of the Company.

"Retirement" means retirement as of the individual's normal retirement date under the Amgen Inc. Profit Sharing 401(k) Plan and Trust or other similar successor plan applicable to salaried employees, unless otherwise defined by the Plan Administrator from time to time for purposes of Article I of the Plan.

"Securities Act" means the Securities Act of 1933, as amended.

"Subsidiary," except as provided in Article I, Section 8.3 in connection with Incentive Stock Options, means any entity that is directly or indirectly controlled by the Company.

"Termination Date" has the meaning set forth in Article I, Section 7.6.

SECTION 3. ADMINISTRATION.

3.1 Plan Administrator.

The Plan shall be administered by the Board and/or a committee or committees (which term includes subcommittees) appointed by, and consisting of two or more members of, the Board (a "Plan Administrator"). If and so long as the Common Stock is registered under Section 12(b) or 12(g) of the Exchange Act, the Board shall consider in selecting the members of any committee acting as Plan

Administrator, with respect to any persons subject or likely to become subject to Section 16 of the Exchange Act, the provisions regarding (a) "outside directors" as contemplated by Section 162(m) of the Code and (b) "nonemployee directors" as contemplated by Rule 16b-3 under the Exchange Act. The Board may delegate the responsibility for administering the Plan with respect to designated classes of eligible persons to different committees consisting of two or more members of the Board, subject to such limitations as the Board deems appropriate. Committee members shall serve for such term as the Board may determine, subject to removal by the Board at any time. To the extent consistent with applicable law, the Board may authorize one or more senior executive officers of the Company to grant Options to specified eligible persons, within the limits specifically prescribed by the Board.

3.2 Administration and Interpretation by Plan Administrator.

Except for the terms and conditions explicitly set forth in the Plan, the Plan Administrator shall have exclusive authority, in its discretion, to determine all matters relating to Options under the Plan, including the selection of individuals to be granted Options, the type of Options, the number of shares of Common Stock subject to an Option, all terms, conditions, restrictions and limitations, if any, of an Option and the terms of any instrument that evidences the Option. The Plan Administrator shall also have exclusive authority to interpret the Plan and may from time to time adopt, and change, rules and regulations of general application for the Plan's administrator. The Plan Administrator's interpretation of the Plan and its rules and regulations, and all actions taken and determinations made by the Plan Administrator pursuant to the Plan, shall be conclusive and binding on all parties involved or affected. The Plan Administrator may delegate administrative duties to such of the Company's officers as it so determines.

SECTION 4. STOCK SUBJECT TO THE PLAN.

4.1 Shares Available for Issuance.

Subject to adjustment from time to time as provided in Article I, Section 11.1, shares of Common Stock shall be available for issuance under the Plan. Shares issued under the Plan shall be drawn from authorized and unissued shares or shares now held or subsequently acquired by the Company.

4.2 Reuse of Shares.

Any shares of Common Stock that have been made subject to an Option that cease to be subject to the Option (other than by reason of exercise of the Option to the extent it is exercised for shares) shall again be available for issuance in connection with future grants of Options under the Plan; provided,

however, that for purposes of any individual award limit under the Plan, any such shares shall be counted in accordance with the requirements of Section 162(m) of the Code.

SECTION 5. ELIGIBILITY.

Options may be granted under the Plan to those officers, directors and employees of the Company and its Related Corporations as the Plan Administrator from time to time selects.

SECTION 6. ACQUIRED COMPANY OPTIONS.

Notwithstanding anything in the Plan to the contrary, the Plan Administrator may grant Options under the Plan in substitution for awards issued under other plans, or assume under the Plan awards issued under other plans, if the other plans are or were plans of other acquired entities ("Acquired Entities") (or the parent of the Acquired Entity) and the new Option is substituted, or the old option is assumed, by reason of a merger, consolidation, acquisition of property or of stock, reorganization or liquidation (the "Acquisition Transaction"). In the event that a written agreement pursuant to which the Acquisition Transaction is completed is approved by the Board and said agreement sets forth the terms and conditions of the substitution for or assumption of outstanding options of the Acquired Entity, said terms and conditions shall be deemed to be the action of the Plan Administrator without any further action by the Plan Administrator, except as may be required for compliance with Rule 16b-3 under the Exchange Act, and the persons holding such awards shall be deemed to be Optionees.

SECTION 7. TERMS AND CONDITIONS OF OPTIONS.

7.1 Grant of Options.

The Plan Administrator is authorized under the Plan, in its sole discretion, to issue Options as Incentive Stock Options or as Nonqualified Stock Options, which shall be appropriately designated.

7.2 Option Exercise Price.

The exercise price for shares purchased under an Option shall be as determined by the Plan Administrator, but shall not be less than 100% of the Fair Market Value of the Common Stock on the Grant Date with respect to Incentive Stock Options and not less than 85% of the Fair Market Value of the Common Stock on the Grant Date with respect to Nonqualified Stock Options. For Incentive Stock Options granted to a more than 10% stockholder, the Option exercise price shall be as specified in Article I, Section 8.2.

7.3 Term of Options.

The term of each Option (the "Option Term") shall be as established by the Plan Administrator or, if not so established, shall be 10 years from the Grant Date. For Incentive Stock Options, the maximum Option Term shall be as specified in Article I, Sections 8.2 and 8.4.

7.4 Exercise of Options.

The Plan Administrator shall establish and set forth in each instrument that evidences an Option the time at which, or the installments in which, the Option shall vest and become exercisable, which provisions may be waived or modified by the Plan Administrator at any time. If not so established in the instrument evidencing the Option, the Option shall vest and become exercisable according to the following schedule, which may be waived or modified by the Plan Administrator at any time:

Period of Optionee's Continuous Employment or Service With the Company or Its Related Corporations From the Option Grant Date	Portion of Total Option That Is Vested and Exercisable
After one year	20%
After two years	40%
After three years	60%
After four years	80%
After five years	100%

Notwithstanding the foregoing, an Option granted under Article I of the Plan shall become 100% vested and exercisable on the date of termination of an Optionee's employment or service relationship with the Company or a Related Corporation on account of the Optionee's death, provided that the Optionee has been in the continuous employment of or service to the Company or a Related Corporation for at least two years at the date of such Optionee's death.

The Plan Administrator may adjust the vesting schedule of an Option held by an Optionee who works less than "full-time" as that term is defined by the Plan Administrator.

To the extent that the right to purchase shares has accrued thereunder, an Option may be exercised from time to time by delivery to the Company of a stock option exercise agreement or notice, in

a form and in accordance with procedures established by the Plan Administrator, setting forth the number of shares with respect to which the Option is being exercised, the restrictions imposed on the shares purchased under such exercise agreement, if any, and such representations and agreements as may be required by the Company, accompanied by payment in full as described in Article I, Section 7.5. An Option may not be exercised as to less than a reasonable number of shares at any one time, as determined by the Plan Administrator.

7.5 Payment of Exercise Price.

The purchase price of Common Stock acquired pursuant to an Option shall be paid, to the extent permitted by applicable statutes and regulations, either: (i) in cash at the time the Option is exercised; or (ii) at the discretion of the Plan Administrator, either at the time of grant or exercise of the Option (A) by delivery to the Company of shares of Common Stock that have been held for the period required to avoid a charge to the Company's reported earnings and valued at the fair market value on the date of exercise, (B) according to a deferred payment or other arrangement with the person to whom the Option is granted or to whom the Option is transferred pursuant to Article I, Section 10, or (C) in any other form of legal consideration that may be acceptable to the Plan Administrator in its discretion; including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price to the Company from the sales proceeds before Common Stock is issued.

In the case of any deferred payment arrangement, interest shall be payable at least annually and shall be charged at not less than the minimum rate of interest necessary to avoid the treatment as interest, under any applicable provisions of the Code, of any amounts other than amounts stated to be interest under the deferred payment arrangement.

7.6 Post-Termination Exercises.

The Plan Administrator shall establish and set forth in each instrument that evidences an Option whether the Option shall continue to be exercisable, and the terms and conditions of such exercise, if an Optionee ceases to be employed by, or to provide services to, the Company or its Related Corporations, which provisions may be waived or modified by the Plan Administrator at any time. If not so established in the instrument evidencing the Option, the Option shall be exercisable according to the following terms and conditions, which may be waived or modified by the Plan Administrator at any time.

(a) Any portion of an Option that is not vested and exercisable on the date of termination of the Optionee's employment or service relationship (the "Termination Date") shall expire on such date, unless the Plan Administrator determines otherwise.

(b) Any portion of an Option that is vested and exercisable on the Termination Date shall expire upon the earliest to occur of:

(i) the last day of the Option Term;

(ii) if the Optionee's Termination Date occurs for reasons other than Cause, Disability, death or Retirement, the three-month anniversary of such Termination Date; and

(iii) if the Optionee's Termination Date occurs by reason of Disability, death or Retirement, the one-year anniversary of such Termination Date.

Notwithstanding the foregoing, if the Optionee dies after the Termination Date while the Option is otherwise exercisable, the Option shall expire upon the earlier to occur of (y) the last day of the Option Term and (z) the first anniversary of the date of death.

Also notwithstanding the foregoing, in case of termination of the Optionee's employment or service relationship for Cause, the Option shall automatically expire upon first notification to the Optionee of such termination, unless the Plan Administrator determines otherwise. If an Optionee's employment or service relationship with the Company is suspended pending an investigation of whether the Optionee shall be terminated for Cause, all the Optionee's rights under any Option likewise shall be suspended during the period of investigation.

An Optionee's transfer of employment or service relationship between or among the Company and its Related Corporations, or a change in status from an employee to a consultant that is evidenced by a written agreement between an Optionee and the Company or a Related Corporation, shall not be considered a termination of employment or service relationship for purposes of this Article I, Section 7. Employment or service relationship shall be deemed to continue while the Optionee is on a bona fide leave of absence, if such leave was approved by the Company or a Related Corporation in writing and if continued crediting of service for purposes of this Article I, Section 7 is expressly required by the terms of such leave or by applicable law (as determined by the Company). The effect of a Company-approved leave of absence on the terms and conditions of an Option shall be determined by the Plan Administrator, in its sole discretion.

SECTION 8. INCENTIVE STOCK OPTION LIMITATIONS.

To the extent required by Section 422 of the Code, Incentive Stock Options shall be subject to the following additional terms and conditions:

8.1 Dollar Limitation.

To the extent the aggregate Fair Market Value (determined as of the Grant Date) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time during any calendar year (under the Plan and all other stock option plans of the Company) exceeds \$100,000, such portion in excess of \$100,000 shall be treated as a Nonqualified Stock Option. In the event the Optionee holds two or more such Options that become exercisable for the first time in the same calendar year, such limitation shall be applied on the basis of the order in which such Options are granted.

8.2 More Than 10% Stockholders.

If an individual owns more than 10% of the total voting power of all classes of the Company's stock, then the exercise price per share of an Incentive Stock Option shall not be less than 110% of the Fair Market Value of the Common Stock on the Grant Date and the Option Term shall not exceed five years. The determination of more than 10% ownership shall be made in accordance with Section 422 of the Code.

8.3 Eligible Employees.

Individuals who are not employees of the Company or one of its parent corporations or subsidiary corporations may not be granted Incentive Stock Options. For purposes of this Article I, Section 8.3, "parent corporation" and "subsidiary corporation" shall have the meanings attributed to those terms for purposes of Section 422 of the Code.

8.4 <u>Term</u>.

Except as provided in Article I, Section 8.2, the Option Term shall not exceed 10 years.

8.5 Exercisability.

An Option designated as an Incentive Stock Option shall cease to qualify for favorable tax treatment as an Incentive Stock Option to the extent it is exercised (if permitted by the terms of the



Option) (a) more than three months after the Termination Date for reasons other than death or Disability, (b) more than one year after the Termination Date by reason of Disability, or (c) after the Optionee has been on leave of absence for more than 90 days, unless the Optionee's reemployment rights are guaranteed by statute or contract.

For purposes of this Article I, Section 8.5, Disability shall mean "disability" as that term is defined for purposes of Section 422 of the Code.

8.6 Taxation of Incentive Stock Options.

In order to obtain certain tax benefits afforded to Incentive Stock Options under Section 422 of the Code, the Optionee must hold the shares issued upon the exercise of an Incentive Stock Option for two years after the Grant Date and one year from the date of exercise. An Optionee may be subject to the alternative minimum tax at the time of exercise of an Incentive Stock Option. The Optionee shall give the Company prompt notice of any disposition of shares acquired by the exercise of an Incentive Stock Option prior to the expiration of such holding periods.

SECTION 9. WITHHOLDING.

The Company may require the Optionee to pay to the Company the amount of any withholding taxes that the Company is required to withhold with respect to the grant, vesting or exercise of any Option. Subject to the Plan and applicable law, the Plan Administrator may, in its sole discretion, permit the Optionee to satisfy withholding obligations, in whole or in part, by paying cash, by electing to have the Company withhold shares of Common Stock or by transferring shares of Common Stock to the Company, in such amounts as are equivalent to the Fair Market Value of the withholding obligation. The Company shall have the right to withhold from any Option or any shares of Common Stock issuable pursuant to an Option or from any cash amounts otherwise due or to become due from the Company to the Optionee an amount equal to such taxes. The Company may also deduct from any Option any other amounts due from the Optionee to the Company or a Related Corporation.

SECTION 10. ASSIGNABILITY.

Options granted under Article I of the Plan and any interest therein may not be assigned, pledged or transferred by the Optionee and may not be made subject to attachment or similar proceedings otherwise than by will or by the applicable laws of descent and distribution, and, during the Optionee's lifetime, such Options may be exercised only by the Optionee. Notwithstanding the foregoing, and to the extent permitted by Section 422 of the Code, the Plan Administrator, in its sole discretion, may permit

such assignment, transfer and exercisability and may permit an Optionee to designate a beneficiary who may exercise the Option or receive compensation under the Option after the Optionee's death; <u>provided</u>, <u>however</u>, that any Option so assigned or transferred shall be subject to all the same terms and conditions contained in the instrument evidencing the Option.

SECTION 11. ADJUSTMENTS UPON CHANGES IN CAPITALIZATION.

11.1 Adjustment of Shares.

The aggregate number and class of shares for which Options may be granted under the Plan, the number and class of shares covered by each outstanding Option and the exercise price per share thereof (but not the total price), shall all be proportionately adjusted for any increase or decrease in the number of issued shares of Common Stock resulting from a split-up or consolidation of shares or any like capital adjustment, or the payment of any stock dividend (not including the stock dividend approved by the Board of Directors of Immunex on February 23, 1999).

11.2 Cash, Stock or Other Property for Stock.

Except as provided in Article I, Section 11.3, upon a merger (other than a merger of the Company in which the holders of Common Stock immediately prior to the merger have the same proportionate ownership of Common Stock in the surviving corporation immediately after the merger), consolidation, acquisition of property or stock, separation, reorganization (other than a mere reincorporation or the creation of a holding company) or liquidation of the Company, as a result of which the stockholders of the Company receive cash, stock or other property in exchange for or in connection with their shares of Common Stock, any Option granted hereunder shall terminate, but the Optionee shall have the right immediately prior to any such merger, consolidation, acquisition of property or stock, liquidation or reorganization to exercise such Option in whole or in part whether or not the vesting requirements set forth in the Option agreement have been satisfied.

11.3 Conversion of Options on Stock for Stock Exchange.

If the stockholders of the Company receive capital stock of another corporation ("Exchange Stock") in exchange for their shares of Common Stock in any transaction involving a merger (other than a merger of the Company in which the holders of Common Stock immediately prior to the merger have the same proportionate ownership of Common Stock in the surviving corporation immediately after the merger), consolidation, acquisition of property or stock, liquidation or reorganization (other than a mere reincorporation or the creation of a holding company), the Company and the corporation issuing the

Exchange Stock, in their sole discretion, may determine that all Options granted hereunder shall be converted into options to purchase shares of Exchange Stock instead of terminating in accordance with the provisions of Article I, Section 11.2. The amount and price of converted options shall be determined by adjusting the amount and price of the Options granted hereunder in the same proportion as used for determining the number of shares of Exchange Stock the holders of the Common Stock receive in such merger, consolidation, acquisition of property or stock, liquidation or reorganization. Unless accelerated by the Board, the vesting schedule set forth in the Option agreement shall continue to apply to the options granted for the Exchange Stock. The aggregate number and kind of shares for which options may be granted under this Plan shall be proportionately adjusted in the event of such merger, consolidation, acquisition of property or stock, liquidation or reorganization.

11.4 Fractional Shares.

In the event of any adjustment in the number of shares covered by any Option, any fractional shares resulting from such adjustment shall be disregarded and each such Option shall cover only the number of full shares resulting from such adjustment.

11.5 Determination of Board to Be Final.

All Article I, Section 11 adjustments shall be made by the Plan Administrator, and its determination as to what adjustments shall be made, and the extent thereof, shall be final, binding and conclusive. Unless an Optionee agrees otherwise, any change or adjustment to an Incentive Stock Option shall be made in such a manner so as not to constitute a "modification" as defined in Section 424(h) of the Code and so as not to cause his or her Incentive Stock Option issued hereunder to fail to continue to qualify as an "incentive stock option" as defined in Section 422(b) of the Code.

11.6 Limitations.

The grant of Options shall in no way affect the Company's right to adjust, reclassify, reorganize or otherwise change its capital or business structure or to merge, consolidate, dissolve, liquidate or sell or transfer all or any part of its business or assets.

SECTION 12. AMENDMENT AND TERMINATION OF PLAN.

12.1 Amendment of Plan.

The Plan may be amended only by the Board in such respects as it shall deem advisable; <u>provided</u>, <u>however</u>, that to the extent required for compliance with Section 422 of the Code or any applicable law or regulation, stockholder approval shall be required for any amendment that would (a) increase the total number of shares available for issuance under the Plan, (b) modify the class of persons eligible to receive Options, or (c) otherwise require stockholder approval under any applicable law or regulation. Any amendment made to the Plan that would constitute a "modification" to Incentive Stock Options outstanding on the date of such amendment shall not, without the consent of the Optionee, be applicable to such outstanding Incentive Stock Options but shall have prospective effect only.

12.2 Termination of Plan.

The Board may suspend or terminate the Plan at any time. The Plan shall have no fixed expiration date; <u>provided</u>, <u>however</u>, that no Incentive Stock Options may be granted more than 10 years after the later of (a) the Plan's adoption by the Board of Directors of Immunex and (b) the adoption by the Board of Directors of Immunex of any amendment to the Plan that constitutes the adoption of a new plan for purposes of Section 422 of the Code.

12.3 Consent of Optionee.

The amendment or termination of the Plan or the amendment of an outstanding Option shall not, without the Optionee's consent, impair or diminish any rights or obligations under any Option theretofore granted to the Optionee under the Plan. Except as otherwise provided in the Plan, no outstanding Option shall be terminated without the consent of the Optionee. Any change or adjustment to an outstanding Incentive Stock Option shall not, without the consent of the Optionee, be made in a manner so as to constitute a "modification" that would cause such Incentive Stock Option to fail to continue to qualify as an Incentive Stock Option.

SECTION 13. GENERAL

13.1 Evidence of Options.

Options granted under the Plan shall be evidenced by a written instrument that shall contain such terms, conditions, limitations and restrictions as the Plan Administrator shall deem advisable and that are not inconsistent with the Plan.

13.2 No Individual Rights.

Nothing in the Plan or any Option granted under the Plan shall be deemed to constitute an employment contract or confer or be deemed to confer on any Optionee any right to continue in the employ of, or to continue any other relationship with, the Company or any Related Corporation or limit in any way the right of the Company or any Related Corporation of the Company to terminate an Optionee's employment or other relationship at any time, with or without Cause.

13.3 Registration.

Notwithstanding any other provision of the Plan, the Company shall have no obligation to issue or deliver any shares of Common Stock under the Plan or make any other distribution of benefits under the Plan unless such issuance, delivery or distribution would comply with all applicable laws (including, without limitation, the requirements of the Securities Act), and the applicable requirements of any securities exchange or similar entity.

The Company shall be under no obligation to any Optionee to register for offering or resale or to qualify for exemption under the Securities Act, or to register or qualify under state securities laws, any shares of Common Stock, security or interest in a security paid or issued under, or created by, the Plan, or to continue in effect any such registrations or qualifications if made. The Company may issue certificates for shares with such legends and subject to such restrictions on transfer and stop-transfer instructions as counsel for the Company deems necessary or desirable for compliance by the Company with federal and state securities laws.

To the extent that the Plan or any instrument evidencing an Option provides for issuance of stock certificates to reflect the issuance of shares of Common Stock, the issuance may be effected on a noncertificated basis, to the extent not prohibited by applicable law or the applicable rules of any stock exchange.

13.4 No Rights as a Stockholder.

No Option shall entitle the Optionee to any cash dividend, voting or other right of a stockholder unless and until the date of issuance under the Plan of the shares that are the subject of such Option.

13.5 Compliance With Laws and Regulations.

Notwithstanding anything in the Plan to the contrary, the Plan Administrator, in its sole discretion, may bifurcate the Plan so as to restrict, limit or condition the use of any provision of the Plan to Optionees who are officers or directors subject to Section 16 of the Exchange Act without so restricting, limiting or conditioning the Plan with respect to other Optionees. Additionally, in interpreting and applying the provisions of the Plan, any Option granted as an Incentive Stock Option pursuant to the Plan shall, to the extent permitted by law, be construed as an "incentive stock option" within the meaning of Section 422 of the Code.

13.6 Optionees in Foreign Countries.

The Plan Administrator shall have the authority to adopt such modifications, procedures and subplans as may be necessary or desirable to comply with provisions of the laws of foreign countries in which the Company or its Related Corporations may operate to assure the viability of the benefits from Options granted to Optionees employed in such countries and to meet the objectives of the Plan.

13.7 No Trust or Fund.

The Plan is intended to constitute an "unfunded" plan. Nothing contained herein shall require the Company to segregate any monies or other property, or shares of Common Stock, or to create any trusts, or to make any special deposits for any immediate or deferred amounts payable to any Optionee, and no Optionee shall have any rights that are greater than those of a general unsecured creditor of the Company.

13.8 Severability.

If any provision of the Plan or any Option is determined to be invalid, illegal or unenforceable in any jurisdiction, or as to any person, or would disqualify the Plan or any Option under any law deemed applicable by the Plan Administrator, such provision shall be construed or deemed amended to conform to applicable laws, or, if it cannot be so construed or deemed amended without, in the Plan Administrator's determination, materially altering the intent of the Plan or the Option, such provision shall be stricken as to such jurisdiction, person or Option, and the remainder of the Plan and any such Option shall remain in full force and effect.

13.9 Choice of Law.

The Plan and all determinations made and actions taken pursuant hereto, to the extent not otherwise governed by the laws of the United States, shall be governed by the laws of the State of Washington without giving effect to principles of conflicts of laws.

SECTION 14. EFFECTIVE DATE.

The Effective Date of the Original Plan was the date on which it was adopted by the Board of Directors of Immunex, provided that it was approved by Immunex's stockholders at any time within 12 months of such adoption.

SECTION 15. ADDENDUM TO ARTICLE I OF THE PLAN.

Notwithstanding anything in Article I of the Plan or any program adopted under the Original Plan to the contrary, effective as of the Effective Time (as defined in the Amended and Restated Agreement and Plan of Merger by and between the Company, AMS Acquisition Inc. and Immunex dated as of December 16, 2001, as amended by that certain First Amendment to Amended and Restated Agreement and Plan of Merger dated as of July 15, 2002 (as amended, the "Merger Agreement")), the following provisions shall constitute an addendum (the "Addendum") to Article I of the Plan:

15.1. At the Effective Time, each option granted pursuant Article I of the Plan shall be treated in accordance with the applicable terms of the Merger Agreement.

15.2. In the event that an optionee's employment with Immunex or the Company is terminated by the optionee for Good Reason or by Immunex or the Company without Cause during the fifteen (15) months following the Effective Time, each option held by such optionee for Common Stock that was granted pursuant to the Merger Agreement with respect to (a) a Cancelled Company Option (as defined in the Merger Agreement) or (b) an option for common stock of Immunex that was granted after December 16, 2001, shall immediately vest in full and shall remain exercisable until the earlier of (x) the first anniversary of the optionee's termination of employment or (y) the end of the term of such option.

15.3. In the event that an optionee who is a nonemployee director of Immunex immediately prior to the Effective Time ceases to be a director of Immunex or the Company for any reason



immediately prior to, at, or during the fifteen (15) months following the Effective Time, each option held by such optionee for Common Stock shall immediately vest in full and shall remain exercisable until the earlier of (x) the first anniversary of the date such optionee ceases to be a director of Immunex or the Company or (y) the end of the term of such option.

15.4. For purposes of this Addendum only, "Good Reason" shall mean the occurrence on or after the Effective Time and without the optionee's consent of, (a) a reduction in the optionee's annual base salary or wages, other than as part of a general reduction applicable to substantially all employees of Immunex or the Company employed in the United States or (ii) the relocation of the optionee's principal place of employment to a location more than fifty (50) miles from the optionee's principal place of employment prior to the Effective Time.

15.5. For purposes of this Addendum only, "Cause" shall mean (a) the willful and continued failure by the optionee to substantially perform the optionee's duties with Immunex or the Company (other than such failure resulting form the optionee's incapacity due to physical or mental illness) or (b) the willful engaging by the optionee in conduct which is demonstrably and materially injurious to Immunex or the Company, monetarily or otherwise. For purposes of this definition, no act, or failure to act, on the optionee's part shall be deemed willful unless done, or omitted to be done, by the optionee not in good faith or without reasonable belief that the optionee's act, or failure to act, was in the best interest of Immunex or the Company.

ARTICLE II.

PROVISIONS APPLICABLE TO OPTIONS GRANTED ON OR AFTER RESTATEMENT DATE

The following provisions of this Article II shall govern awards granted under the Plan on or after the Restatement Date:

SECTION 1. PURPOSE.

(a) The purpose of Article II of the Plan is to provide a means by which employees or directors of and consultants to Amgen Inc., a Delaware corporation (the "Company"), and its Affiliates, as defined in Article II, paragraph 1(b), directly, or indirectly through Trusts, may be given an opportunity to benefit from increases in value of the stock of the Company through the granting of (i) incentive stock options, (ii) nonqualified stock options, (iii) stock bonuses, and (iv) rights to purchase

restricted stock, all as defined below.

(b) The word "Affiliate" as used in Article II of the Plan means (a) any parent corporation or subsidiary corporation of the Company, as those terms are defined in Sections 424(e) and (f), respectively, of the Internal Revenue Code of 1986, as amended (together with the regulations and other official guidance promulgated thereunder, the "Code") and (b) any domestic eligible entity that is disregarded, under Treasury Regulation Section 301.7701-3, as an entity separate from either (i) the Company or (ii) any parent corporation or subsidiary corporation, as those terms are defined in Sections 424(e) and (f), respectively, of the Code.

(c) The Company, by means of Article II of the Plan, seeks to retain the services of persons now employed by or serving as directors or consultants to the Company, to secure and retain the services of persons capable of filling such positions, and to provide incentives for such persons to exert maximum efforts for the success of the Company.

(d) The Company intends that the rights issued under Article II of the Plan shall, in the discretion of the Board of Directors of the Company (the "Board") or any committee to which responsibility for administration of the Plan has been delegated pursuant to Article II, paragraph 2(c), be either (i) stock options granted pursuant to Article II, Sections 5 or 6 hereof, including incentive stock options as that term is used in Section 422 of the Code ("Incentive Stock Options"), or options which do not qualify as Incentive Stock Options ("Nonqualified Stock Options") (together hereinafter referred to as "Options"), or (ii) stock bonuses or rights to purchase restricted stock granted pursuant to Article II, Section 7 hereof (all such rights included in (i) and (ii), collectively "Stock Awards").

(e) The word "Trust" as used in Article II of the Plan shall mean a trust created for the benefit of the employee, director or consultant, his or her spouse, or members of their immediate family. The word optionee shall mean the person to whom the option is granted or the employee, director or consultant for whose benefit the option is granted to a Trust, as the context shall require.

SECTION 2. ADMINISTRATION.

(a) The Plan shall be administered by the Board unless and until the Board delegates administration to a committee, as provided in Article II, paragraph 2(c).

(b) The Board shall have the power, subject to, and within the limitations of, the express provisions of the Plan:

(1) To determine from time to time which of the persons eligible under the Plan shall be granted Stock Awards; when and how Stock Awards shall be granted; whether a Stock Award will be an Incentive Stock Option, a Nonqualified Stock Option, a stock bonus, a right to purchase restricted stock, or a combination of the foregoing; the provisions of each Stock Award granted (which need not be identical), including the time or times when a person shall be permitted to purchase or receive stock pursuant to a Stock Award; and the number of shares with respect to which Stock Awards shall be granted to each such person.

(2) To construe and interpret the Plan and Stock Awards granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any Stock Award, in a manner and to the extent it shall deem necessary or expedient to make the Plan fully effective.

(3) To amend the Plan as provided in Article II, Section 14.

(4) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company.

(c) The Board may delegate administration of the Plan to a committee composed of not fewer than two (2) members of the Board (the "Committee"). One or more of these members may be non-employee directors and outside directors, if required and as defined by the provisions of Article II, paragraphs 2(e) and 2(f). If administration is delegated to a Committee, the Committee shall have, in connection with the administration of the Plan, the powers theretofore possessed by the Board (except amendment of any program adopted pursuant to Article II, Section 6 or any Non-Discretionary Director Awards granted thereunder shall only be by action taken by the Board or a committee of one or more members of the Board to which such authority has been specifically delegated by the Board), subject, however, to such resolutions, not inconsistent with the provisions of Article II of the Plan, as may be adopted from time to time by the Board. Notwithstanding anything else in this Article II, paragraph 2(c) to the contrary, at any time the Board or the Committee may delegate to a committee of one or more members of the Board or the Committee may delegate to a committee of one or more members of the Board or the Committee may delegate to a committee of one or more members of the Board or the Committee may delegate to a committee of one or more members of the Board or the Committee may delegate to a committee of one or more members of the Board or the Committee may delegate to a committee of one or more members of the Board the authority to grant or amend Stock Awards to all employees, directors or consultants or any portion or class thereof.

(d) Notwithstanding anything else in the Plan to the contrary, at any time the Board or the Committee may authorize by duly adopted resolution one or more Officers (as defined below) (each a "Delegated Officer") to take the actions described in Article II, paragraph 2(b)(1) of the Plan with respect to Options only, subject to, and within the limitations of, the express provisions of Article II of the Plan; <u>provided</u>, <u>however</u>, that a Delegated Officer shall not have the power to (1) grant any Options to himself,

any non-employee director, consultant, Trust, other Delegated Officer or Officer, (2) determine the time or times when a person shall be permitted to purchase stock pursuant to the exercise of an Option (i.e., vesting), (3) determine the exercise price of an Option, or (4) grant any Option to a parent corporation of the Company, as defined in Section 424(e) of the Code. The resolution authorizing a Delegated Officer to act as such shall specify the total number of shares of Common Stock that a Delegated Officer may grant with respect to Options. The exercise price, which shall be not less than 100% of the closing price of the Common Stock of the Company as quoted on the NASDAQ system on the grant date, or in the Board or the Committee's sole discretion, otherwise determined in accordance with applicable provisions of Code Section 409A (the "Option Fair Market Value") and the time or times when a person shall be permitted to purchase stock pursuant to the exercise of an Option shall, however, be set by the Board or the Committee and not by a Delegated Officer to the extent required by Delaware General Corporation Law Section 157 or any other applicable law. The term "Officer" shall include any natural person who is elected as a corporate officer of the Company by the Board.

(e) The term "non-employee director" shall mean a member of the Board who (i) is not currently an officer of the Company or a parent or subsidiary of the Company (as defined in Rule 16a-1(f) promulgated by the Securities and Exchange Commission under Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) or an employee of the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company; (ii) does not receive compensation ("Rule 404"); (iii) does not possess an interest in any other transaction required to be disclosed under Rule 404; or (iv) is not engaged in a business relationship required to be disclosed under Rule 404, as all of these provisions are interpreted by the Securities and Exchange Commission under Rule 16b-3 promulgated under the Exchange Act.

(f) The term "outside director," as used in Article II of this Plan, shall mean an administrator of the Plan, whether a member of the Board or of any Committee to which responsibility for administration of the Plan has been delegated pursuant to Article II, paragraph 2(c), who is considered to be an "outside director" in accordance with the rules, regulations or interpretations of Section 162(m) of the Code.

(g) Any requirement that an administrator of the Plan be a "non-employee director" or "outside director" shall not apply if the Board or the Committee expressly declares that such requirement shall not apply.

SECTION 3. SHARES SUBJECT TO THE PLAN.

(a) Subject to the provisions of Article II, Section 11 relating to adjustments upon changes in stock, the stock that may be issued pursuant to Stock Awards granted under the Plan shall not exceed in the aggregate 19,273,852 shares of the Company's \$.0001 par value common stock (the "Common Stock"). If any Stock Award granted under the Plan shall for any reason expire or otherwise terminate without having been exercised in full, the Common Stock not purchased under such Stock Award shall again become available for the Plan. Shares repurchased by the Company pursuant to any repurchase rights reserved by the Company pursuant to the Plan shall not be available for subsequent issuance under the Plan.

(b) The Common Stock subject to the Plan may be unissued shares or reacquired shares, bought on the market or otherwise.

(c) An Incentive Stock Option may be granted to an eligible person under the Plan only if the aggregate fair market value (determined at the time the Incentive Stock Option is granted) of the Common Stock with respect to which incentive stock options (as defined by the Code) are exercisable for the first time by such optionee during any calendar year under all such plans of the Company and its Affiliates does not exceed one hundred thousand dollars (\$100,000). If it is determined that an entire Option or any portion thereof does not qualify for treatment as an Incentive Stock Option by reason of exceeding such maximum, such Option or the applicable portion shall be considered a Nonqualified Stock Option.

SECTION 4. ELIGIBILITY.

(a) Incentive Stock Options may be granted only to employees (including officers) of the Company or its Affiliates. A director of the Company shall not be eligible to receive Incentive Stock Options unless such director is also an employee of the Company or any Affiliate. Stock Awards other than Incentive Stock Options may be granted to employees (including officers) or directors of or consultants to the Company or any Affiliate or to Trusts of any such employee, director or consultant.

(b) A director shall in no event be eligible for the benefits of the Plan (other than Non-Discretionary Director Awards, as defined in Article II, Section 6) unless and until such director is expressly declared eligible to participate in the Plan by action of the Board or the Committee, and only if, at any time discretion is exercised by the Board or the Committee in the selection of a director as a person to whom Stock Awards may be granted, or in the determination of the number of shares which may be covered by Stock Awards granted to a director, the Plan complies with the requirements of Rule 16b-3

promulgated under the Exchange Act, as from time to time in effect. The Board shall otherwise comply with the requirements of Rule 16b-3 promulgated under the Exchange Act, as from time to time in effect. Notwithstanding the foregoing, the restrictions set forth in this Article II, paragraph 4(b) shall not apply if the Board or Committee expressly declares that such restrictions shall not apply.

(c) No person shall be eligible for the grant of an Incentive Stock Option under the Plan if, at the time of grant, such person owns (or is deemed to own pursuant to Section 424(d) of the Code) stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or of any of its Affiliates unless the exercise price of such Incentive Stock Option is at least one hundred and ten percent (110%) of the fair market value of the Common Stock at the date of grant and the Incentive Stock Option is not exercisable after the expiration of five (5) years from the date of grant.

(d) Stock Awards shall be limited to a maximum of 649,455 shares of Common Stock per person per calendar year.

SECTION 5. TERMS OF DISCRETIONARY STOCK OPTIONS.

An option granted pursuant to this Article II, Section 5 (a "Discretionary Stock Option") shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate. The provisions of separate Options need not be identical, but each Option shall include (through incorporation of provisions hereof by reference in the Option or otherwise) the substance of each of the following provisions:

(a) No Option shall be exercisable after the expiration of ten (10) years from the date it was granted.

(b) The exercise price of each Incentive Stock Option and each Nonqualified Stock Option shall be not less than one hundred percent (100%) of the fair market value of the Common Stock subject to the Option on the date the Option is granted.

(c) The purchase price of Common Stock acquired pursuant to an Option shall be paid, to the extent permitted by applicable statutes and regulations, either: (i) in cash at the time the Option is exercised; or (ii) at the discretion of the Board or the Committee, either at the time of grant or exercise of the Option (A) by delivery to the Company of shares of Common Stock that have been held for the period required to avoid a charge to the Company's reported earnings and valued at the fair market value on the date of exercise, (B) according to a deferred payment or other arrangement with the person to whom the Option is granted or to whom the Option is transferred pursuant to Article II, paragraph 5(d), or (C) in

any other form of legal consideration that may be acceptable to the Board or the Committee in their discretion; including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price to the Company from the sales proceeds before Common Stock is issued.

In the case of any deferred payment arrangement, interest shall be payable at least annually and shall be charged at not less than the minimum rate of interest necessary to avoid the treatment as interest, under any applicable provisions of the Code, of any amounts other than amounts stated to be interest under the deferred payment arrangement.

(d) An Option granted to a natural person shall be exercisable during the lifetime of such person only by such person, provided that such person during such person's lifetime may designate a Trust to be such person's beneficiary with respect to any Incentive Stock Options and with respect to any Nonqualified Stock Options, and such beneficiary shall, after the death of the person to whom the Option was granted, have all the rights that such person has while living, including the right to exercise the Option. In the absence of such designation, after the death of the person to whom the Option is granted, the Option shall be exercisable by the person or persons to whom the optionee's rights under such Option pass by will or by the laws of descent and distribution.

(e) The total number of shares of Common Stock subject to an Option may, but need not, be allotted in periodic installments (which may, but need not, be equal). From time to time during each of such installment periods, the Option may become exercisable ("vest") with respect to some or all of the shares allotted to that period, and may be exercised with respect to some or all of the shares allotted to such period and/or any prior period as to which the Option was not fully exercised. During the remainder of the term of the Option (if its term extends beyond the end of the installment periods), the Option may be exercised from time to time with respect to any shares then remaining subject to the Option. The provisions of this Article II, paragraph 5(e) are subject to any Option provisions governing the minimum number of shares as to which an Option may be exercised.

(f) The Company may require any optionee, or any person to whom an Option is transferred under Article II, paragraph 5(d), as a condition of exercising any such Option: (i) to give written assurances satisfactory to the Company as to such person's knowledge and experience in financial and business matters and/or to employ a purchaser representative who has such knowledge and experience in financial and business matters, and that such person is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Option; and (ii) to give written assurances satisfactory to the Company stating that such person is acquiring the Common Stock subject

to the Option for such person's own account and not with any present intention of selling or otherwise distributing the Common Stock. These requirements, and any assurances given pursuant to such requirements, shall be inoperative if: (x) the issuance of the shares upon the exercise of the Option has been registered under a then currently effective registration statement under the Securities Act of 1933, as amended (the "Securities Act"); or (y) as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities law.

(g) An Option shall terminate three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate, unless the Option by its term specifies either (i) that it shall terminate sooner than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate; or (ii) that it may be exercised more than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate; or (ii) that it may be exercised more than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate. This Article II, paragraph 5(g) shall not be construed to extend the term of any Option or to permit anyone to exercise the Option after expiration of its term, nor shall it be construed to increase the number of shares as to which any Option is exercisable from the amount exercisable on the date of termination of the optionee's employment or relationship as a consultant or director.

(h) The Option may, but need not, include a provision whereby the optionee may elect at any time during the term of the optionee's employment or relationship as a consultant or director with the Company or any Affiliate to exercise the Option as to any part or all of the shares subject to the Option prior to the stated vesting dates of the Option. Any shares so purchased from any unvested installment or Option may be subject to a repurchase right in favor of the Company or to any other restriction the Board or the Committee determines to be appropriate.

(i) To the extent provided by the terms of an Option, each optionee may satisfy any federal, state or local tax withholding obligation relating to the exercise of such Option by any of the following means or by a combination of such means: (i) tendering a cash payment; (ii) authorizing the Company to withhold from the shares of the Common Stock otherwise issuable to the optionee as a result of the exercise of the Option a number of shares having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding; or (iii) delivering to the Company owned and unencumbered shares of the Common Stock having a fair market value less than or equal to the amount of the Company's required minimum statutory is required minimum statutory withholding.

SECTION 6. NON-DISCRETIONARY DIRECTOR AWARDS.

The Board may from time to time adopt award programs under the Plan providing for the grant of formula or non-discretionary Stock Awards to directors of the Company who are not employees of the Company or any Affiliate ("Non-Discretionary Director Awards"). The terms and conditions of any such program shall be established by the Board in its sole discretion, subject to the terms and conditions of the Plan.

SECTION 7. TERMS OF STOCK BONUSES AND PURCHASES OF RESTRICTED STOCK.

Each stock bonus or restricted stock purchase agreement shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate. The terms and conditions of stock bonus or restricted stock purchase agreements may change from time to time, and the terms and conditions of separate agreements need not be identical, but each stock bonus or restricted stock purchase agreement shall include (through incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions as appropriate:

(a) The purchase price under each stock purchase agreement shall be such amount as the Board or Committee shall determine and designate in such agreement. Notwithstanding the foregoing, the Board or the Committee may determine that eligible participants in the Plan may be awarded stock pursuant to a stock bonus agreement in consideration for past services actually rendered to the Company or for its benefit.

(b) No rights under a stock bonus or restricted stock purchase agreement shall be assignable by any participant under the Plan, either voluntarily or by operation of law, except where such assignment is required by law or expressly authorized by the terms of the applicable stock bonus or restricted stock purchase agreement.

(c) The purchase price of stock acquired pursuant to a stock purchase agreement shall be paid either: (i) in cash at the time of purchase; (ii) at the discretion of the Board or the Committee, according to a deferred payment or other arrangement with the person to whom the Common Stock is sold; or (iii) in any other form of legal consideration that may be acceptable to the Board or the Committee in their discretion; including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price of the Company from the sales proceeds before Common Stock is issued. Notwithstanding the foregoing, the Board or the Committee to which

administration of the Plan has been delegated may award Common Stock pursuant to a stock bonus agreement in consideration for past services actually rendered to the Company or for its benefit.

(d) Shares of Common Stock sold or awarded under the Plan may, but need not, be subject to a repurchase option in favor of the Company in accordance with a vesting schedule to be determined by the Board or the Committee.

(e) In the event a person ceases to be an employee of or ceases to serve as a director or consultant to the Company or an Affiliate, the Company may repurchase or otherwise reacquire any or all of the shares of Common Stock held by that person which have not vested as of the date of termination under the terms of the stock bonus or restricted stock purchase agreement between the Company and such person.

(f) To the extent provided by the terms of a stock bonus or restricted stock purchase agreement, a participant may satisfy any federal, state or local tax withholding obligation relating to the lapsing of a repurchase option in favor of the Company or vesting of a stock bonus or a restricted stock award by any of the following means or by a combination of such means: (i) tendering a cash payment; (ii) authorizing the Company to withhold from the shares of the Common Stock otherwise deliverable to a participant as a result of the lapsing of a repurchase option in favor of the Company or the vesting of a stock bonus or a restricted stock award a number of shares having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding; or (iii) delivering to the Company owned and unencumbered shares of the Common Stock having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding.

SECTION 8. COVENANTS OF THE COMPANY.

(a) During the terms of the Stock Awards granted under the Plan, the Company shall keep available at all times the number of shares of Common Stock required to satisfy such Stock Awards up to the number of shares of Common Stock authorized under the Plan.

(b) The Company shall seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to issue and sell shares of Common Stock under the Stock Awards granted under the Plan; <u>provided</u>, <u>however</u>, that this undertaking shall not require the Company to register under the Securities Act either the Plan, any Stock Award granted under the Plan or any Common Stock issued or issuable pursuant to any such Stock Award. If, after reasonable efforts, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under

the Plan, the Company shall be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Stock Awards unless and until such authority is obtained.

SECTION 9. USE OF PROCEEDS FROM COMMON STOCK.

Proceeds from the sale of Common Stock pursuant to Stock Awards granted under the Plan shall constitute general funds of the Company.

SECTION 10. MISCELLANEOUS.

(a) The Board or Committee shall have the power to accelerate the time during which a Stock Award may be exercised or the time during which a Stock Award or any part thereof will vest, notwithstanding the provisions in the Stock Award stating the time during which it may be exercised or the time during which it will vest. Each Discretionary Stock Option providing for vesting pursuant to Article II, paragraph 5(e) may also provide that if the employee's employment or a director's or consultant's affiliation with the Company or an Affiliate of the Company is terminated by reason of death or disability, then the vesting schedule of Discretionary Stock Options granted to such employee, director or consultant or to the Trusts of such employee, director or consultant may be accelerated.

(b) Neither an optionee nor any person to whom an Option is transferred under the provisions of the Plan shall be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares subject to such Option unless and until such person has satisfied all requirements for exercise of the Option pursuant to its terms.

(c) Nothing in the Plan or any instrument executed or Stock Award granted pursuant thereto shall confer upon any eligible employee, consultant, director, optionee or holder of Stock Awards under the Plan any right to continue in the employ of the Company or any Affiliate or to continue acting as a consultant or director or shall affect the right of the Company or any Affiliate to terminate the employment or consulting relationship or directorship of any eligible employee, consultant, director, optionee or holder of Stock Awards under the Plan with or without cause. In the event that a holder of Stock Awards under the Plan is permitted or otherwise entitled to take a leave of absence, the Company shall have the unilateral right to (i) determine whether such leave of absence will be treated as a termination of employment or relationship as consultant or director for purposes hereof, and (ii) suspend or otherwise delay the time or times at which exercisability or vesting would otherwise occur with respect to any outstanding Stock Awards under the Plan.

SECTION 11. ADJUSTMENTS UPON CHANGES IN COMMON STOCK.

If any change is made in the Common Stock subject to the Plan, or subject to any Stock Award granted under the Plan (through merger, consolidation, reorganization, recapitalization, stock dividend, dividend in property other than cash, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other transaction not involving the receipt of consideration by the Company), the Plan and outstanding Stock Awards will be appropriately adjusted in the class(es) and maximum number of shares subject to the Plan, the maximum number of shares which may be granted to a participant in a calendar year, the class(es) and number of shares and price per share of stock subject to outstanding Stock Awards, and the number of shares of Common Stock to be granted as Non-Discretionary Director Awards, if any. Such adjustment shall be made by the Board or the Committee, the determination of which shall be final, binding and conclusive. (The conversion of any convertible securities of the Company shall not be treated as a "transaction not involving the receipt of consideration".) The Board or the Committee, in its sole discretion, may accomplish any such adjustment in a manner calculated not to constitute a "modification" of any such Stock Awards (within the meaning of Code Section 409A) that would cause any such Stock Award to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A).

SECTION 12. CHANGE OF CONTROL.

(a) Notwithstanding anything to the contrary in this Plan, in the event of a Change in Control (as hereinafter defined), then, to the extent permitted by applicable law: (i) the time during which Stock Awards become vested shall automatically be accelerated so that the unvested portions of all Stock Awards shall be vested prior to the Change in Control and (ii) the time during which the Options may be exercised shall automatically be accelerated to prior to the Change in Control and (ii) the time during which the Options may be exercised shall automatically be accelerated to prior to the Change in Control and (ii) the time during which the Options may be exercised shall automatically be accelerated to prior to the Change in Control. Upon and following the acceleration of the vesting and exercise periods, at the election of the holder of the Stock Award, the Stock Award may be: (x) exercised (with respect to Options) or, if the surviving or acquiring corporation agrees to assume the Stock Awards or substitute similar stock awards, (y) assumed; or (z) replaced with substitute stock awards. Options not exercised, substituted or assumed prior to or upon the Change in Control shall be terminated. The Board or the Committee, in its sole discretion, may cause any such assumption or substitution to be conducted in a manner so as not to constitute an "extension," "renewal" or "modification" (each within the meaning of Code Section 409A) of any such Stock Award that would cause any such Stock Award to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A).

(b) For purposes of Article II of the Plan, a "Change of Control" shall be deemed to have occurred at any of the following times:

(i) upon the acquisition (other than from the Company) by any person, entity or "group," within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act (excluding, for this purpose, the Company or its affiliates, or any employee benefit plan of the Company or its affiliates which acquires beneficial ownership of voting securities of the Company), of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of fifty percent (50%) or more of either the then outstanding shares of Common Stock or the combined voting power of the Company's then outstanding voting securities entitled to vote generally in the election of directors; or

(ii) at the time individuals who, as of July 15, 2002, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board, provided that any person becoming a director subsequent to July 15, 2002, whose election, or nomination for election by the Company's stockholders, was approved by a vote of at least a majority of the directors then comprising the Incumbent Board (other than an election or nomination of an individual whose initial assumption of office is in connection with an actual or threatened election contest relating to the election of the Directors of the Company, as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) shall be, for purposes of Article II of the Plan, considered as though such person were a member of the Incumbent Board; or

(iii) immediately prior to the consummation by the Company of a reorganization, merger, consolidation, (in each case, with respect to which persons who were the stockholders of the Company immediately prior to such reorganization, merger or consolidation do not, immediately thereafter, own more than fifty percent (50%) of the combined voting power entitled to vote generally in the election of directors of the reorganized, merged or consolidated company's then outstanding voting securities) or a liquidation or dissolution of the Company or of the sale of all or substantially all of the assets of the Company; or

(iv) the occurrence of any other event which the Incumbent Board in its sole discretion determines constitutes a Change of Control.

SECTION 13. QUALIFIED DOMESTIC RELATIONS ORDERS

(a) Anything in the Plan to the contrary notwithstanding, the Board or the Committee, in its sole discretion, may determine that rights under Stock Awards may be assigned to an Alternate Payee to the extent that a QDRO so provides, and in the event of such determination, the provisions of this Section 13 shall apply. (The terms "<u>Alternate Payee</u>" and "<u>QDRO</u>" are defined in Article II, paragraph 13(c) below.) The assignment of a Stock Award to an Alternate Payee pursuant to a QDRO shall not be treated as

having caused a new grant. The transfer of an Incentive Stock Option to an Alternate Payee may, however, cause it to fail to qualify as an Incentive Stock Option. If a Stock Award is assigned to an Alternate Payee, the Alternate Payee generally has the same rights as the grantee under the terms of the Plan; provided however, that (i) the Stock Award shall be subject to the same vesting terms and exercise period as if the Stock Award were still held by the grantee and (ii) an Alternate Payee may not transfer a Stock Award.

(b) In the event of the Plan administrator's receipt of a domestic relations order or other notice of adverse claim by an Alternate Payee of a grantee of a Stock Award, transfer of the proceeds of the exercise of such Stock Award, whether in the form of cash, stock or other property, may be suspended. Such proceeds shall thereafter be transferred pursuant to the terms of a QDRO or other agreement between the grantee and Alternate Payee. A grantee's ability to exercise a Stock Award may be barred if the Plan administrator receives a court order directing the Plan administrator not to permit exercise.

(c) The word "QDRO" as used in Article II of the Plan shall mean a court order (i) that creates or recognizes the right of the spouse, former spouse or child (an "Alternate Payee") of an individual who is granted a Stock Award to an interest in such Stock Award relating to marital property rights or support obligations and (ii) that the administrator of the Plan determines would be a "qualified domestic relations order," as that term is defined in section 414(p) of the Code and section 206(d) of the Employee Retirement Income Security Act ("ERISA"), but for the fact that the Plan is not a plan described in section 3(3) of ERISA.

SECTION 14. AMENDMENT OF THE PLAN.

(a) The Board at any time, and from time to time, may amend the Plan. However, except as provided in Article II, Section 11 relating to adjustments upon changes in the Common Stock, no amendment shall be effective unless approved by the stockholders of the Company within twelve (12) months before or after the adoption of the amendment, where the amendment will:

(i) increase the number of shares reserved for Stock Awards under the Plan;

(ii) modify the requirements as to eligibility for participation in the Plan (to the extent such modification requires stockholder approval in order for the Plan to satisfy the requirements of Section 422(b) of the Code); or

(iii) modify the Plan in any other way if such modification requires stockholder approval in order for the Plan to satisfy the requirements of Section 422(b) of the Code.

(b) The Board may in its sole discretion submit any other amendment to the Plan for stockholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 162(m) of the Code and the regulations promulgated thereunder regarding the exclusion of performance-based compensation from the limit on corporate deductibility of compensation to certain executive officers.

(c) It is expressly contemplated that the Board may amend the Plan in any respect the Board deems necessary or advisable to provide optionees with the maximum benefits provided or to be provided under the provisions of the Code and the regulations promulgated thereunder relating to employee Incentive Stock Options and/or to bring the Plan and/or Options granted under it into compliance therewith.

(d) Rights and obligations under any Stock Award granted before amendment of the Plan shall not be impaired by any amendment of the Plan, unless: (i) the Company requests the consent of the person to whom the Stock Award was granted; and (ii) such person consents in writing.

(e) Any amendment of the Plan may be accomplished in a manner calculated to cause such amendment not to constitute an "extension," "renewal" or "modification" (each within the meaning of Code Section 409A) of any Stock Awards that would cause such Stock Awards to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A). Notwithstanding the foregoing, if at any time the Board or the Committee determines that any Stock Award may be subject to Code Section 409A, the Board or the Committee may, in its sole discretion, and without a Participant's prior consent to amend the Plan or any Stock Award as it may determine is necessary or desirable either for the Plan and Stock Awards to be exempt from the application of Section 409A or to satisfy the requirements of Section 409A, including by adding conditions with respect to the vesting and/or the payment of the Stock Awards.

SECTION 15. TERMINATION OR SUSPENSION OF THE PLAN.

(a) The Board may suspend or terminate the Plan at any time. No Stock Awards may be granted under the Plan while the Plan is suspended or after it is terminated. No Incentive Stock Options may be granted under the Plan after February 22, 2009.

(b) Rights and obligations under any Stock Awards granted while the Plan is in effect shall not be impaired by suspension or termination of the Plan, except with the consent of the person to whom the Stock Award was granted.

SECTION 16. CODE SECTION 409A.

Except as may be expressly provided with respect to any Stock Award granted under the Plan, the Plan and the Stock Awards are not intended to constitute a "nonqualified deferred compensation plan" within the meaning of Code Section 409A, but rather are intended to be exempt from the application of Code Section 409A. To the extent that the Plan and/or Stock Awards are nevertheless deemed to be subject to Code Section 409A, the Plan and Stock Awards shall be interpreted in accordance with Code Section 409A and Department of Treasury regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the grant of any Stock Award. Notwithstanding any provision of the Plan or any Stock Award to the contrary, in the event that the Committee determines that any Stock Award may be or become subject to Code Section 409A, the Committee may adopt such amendments to the Plan and the affected Stock Award (as described above) or adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), or take any other actions, that the Committee determines are necessary or appropriate to (a) exempt the Plan and any Stock Award from the application of Code Section 409A and/or preserve the intended tax treatment of the benefits provided with respect to the Stock Award, or (b) comply with the requirements of Code Section 409A.

AMGEN INC.

AMENDED AND RESTATED 1999 INCENTIVE STOCK PLAN

Amgen Inc. has adopted this Amended and Restated 1999 Incentive Stock Plan (the "Plan"), effective as of April 1, 2006 (the "Restatement Date"). The Plan amends and restates in its entirety the Abgenix, Inc. Amended and Restated 1999 Nonstatutory Stock Option Plan (the "Original Plan").

ARTICLE I.

PROVISIONS APPLICABLE TO AWARDS GRANTED PRIOR TO RESTATEMENT DATE

The following provisions of this Article I shall govern awards granted under the Plan prior to the effective time (the "Effective Time") of the merger of Athletics Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of Amgen Inc., a Delaware corporation, with and into Abgenix, Inc., a Delaware corporation, pursuant to the Agreement and Plan of Merger, dated as of December 14, 2005:

1. <u>Purposes of the Plan</u>. The purposes of this Plan are: to attract and retain the best available personnel for positions of substantial responsibility, to provide additional incentive to Employees and Consultants, and to promote the success of the Company's business.

Options granted under the Plan will be Nonstatutory Stock Options.

2. <u>Definitions</u>. As used herein, the following definitions shall apply:

(a) "Administrator" means the Board or any of its Committees as shall be administering the Plan, in accordance with Article I, Section 4 of the Plan.

(b) "Applicable Laws" means the requirements relating to the administration of stock option plans under U.S. state corporate laws, U.S. federal and state securities laws, the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any foreign country or jurisdiction where Options are, or will be, granted under the Plan.

(c) "Board" means the Board of Directors of the Company.

(d) "Code" means the Internal Revenue Code of 1986, as amended.

(e) "Committee" means a committee of Directors appointed by the Board in accordance with Article I, Section 4 of the Plan.

(f) "Common Stock" means the common stock of the Company.

(g) "Company" means Amgen Inc., a Delaware corporation.

(h) "Consultant" means any person, including an advisor, engaged by the Company or a Parent or Subsidiary to render services to such entity.

(i) "Director" means a member of the Board.

(j) "Disability" means total and permanent disability as defined in Section 22(e)(3) of the Code.

(k) "Employee" means any person, including Officers, employed by the Company or any Parent or Subsidiary of the Company. A Service Provider shall not cease to be an Employee in the case of (i) any leave of absence approved by the Company or (ii) transfers between locations of the Company or between the Company, its Parent, any Subsidiary, or any successor. Neither service as a Director nor payment of a director's fee by the Company shall be sufficient to constitute "employment" by the Company.

(l) "Exchange Act" means the Securities Exchange Act of 1934, as amended.

(m) "Fair Market Value" means, as of any date, the value of Common Stock determined as follows:

(i) If the Common Stock is listed on any established stock exchange or a national market system, including without limitation the Nasdaq National Market or The Nasdaq SmallCap Market of The Nasdaq Stock Market, its Fair Market Value shall be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or system on the day of determination (the most recent day prior to the day of determination, if the

day of determination is not a day on which reported sales and bids occurred), as reported in The Wall Street Journal or such other source as the Administrator deems reliable;

(ii) If the Common Stock is regularly quoted by a recognized securities dealer but selling prices are not reported, the Fair Market Value of a Share of Common Stock shall be the mean between the high bid and low asked prices for the Common Stock on the last market trading day prior to the day of determination, as reported in The Wall Street Journal or such other source as the Administrator deems reliable; or

(iii) In the absence of an established market for the Common Stock, the Fair Market Value shall be determined in good faith by the Administrator.

(n) "Notice of Grant" means a written or electronic notice evidencing certain terms and conditions of an individual Option grant. The Notice of Grant is part of the Option Agreement.

(o) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder.

(p) "Option" means a nonstatutory stock option granted pursuant to the Plan, that is not intended to qualify as an incentive stock option within the meaning of Section 422 of the Code and the regulations promulgated thereunder.

(q) "Option Agreement" means an agreement between the Company, as successor in interest to Abgenix, Inc., and an Optionee evidencing the terms and conditions of an individual Option grant. The Option Agreement is subject to the terms and conditions of the Plan.

(r) "Option Exchange Program" means a program whereby outstanding Options are surrendered in exchange for Options with a lower exercise price.

(s) "Optioned Stock" means the Common Stock subject to an Option.

(t) "Optionee" means the holder of an outstanding Option granted under the Plan.

(u) "Parent" means (i) any parent corporation of the Company, as such term is defined in Section 424(e) of the Code, or (ii) any domestic eligible entity that is disregarded under

Treasury Regulation Section 301.7701-3, as an entity separate from either (I) the Company or (II) any parent corporation of the Company, as such term is defined in Section 424(e) of the Code.

(v) "Plan" means this Amended and Restated 1999 Nonstatutory Stock Option Plan.

(w) "Service Provider" means an Employee including an Officer or Consultant who is not also a Director.

(x) "Share" means a share of Common Stock, as adjusted in accordance with Article I, Section 11 of the Plan.

(y) "Subsidiary" means (i) any subsidiary corporation of the Company, as such term is defined in Sections 424(f) of the Code, or (ii) any domestic eligible entity that is disregarded under Treasury Regulation Section 301.7701-3, as an entity separate from either (I) the Company or (II) any subsidiary corporation of the Company, as such term is defined in Sections 424(f) of the Code.

3. <u>Stock Subject to the Plan</u>. Subject to the provisions of Article I, Section 11 of the Plan, the maximum aggregate number of Shares which may be issued pursuant to Options granted under the Plan before the Effective Time shall not exceed 1,421,576 Shares of the Company. The Shares may be authorized, but unissued, or reacquired Common Stock.

If an Option expires or becomes unexercisable without having been exercised in full, or is surrendered pursuant to an Option Exchange Program, the unpurchased Shares which were subject thereto shall become available for future grant or sale under Article II of the Plan (unless the Plan has terminated).

4. Administration of the Plan.

(a) *Administration*. The Plan shall be administered by (i) the Board or (ii) a Committee, which committee shall be constituted to satisfy Applicable Laws.

(b) *Powers of the Administrator*. Subject to the provisions of the Plan, and in the case of a Committee, subject to the specific duties delegated by the Board to such Committee, the Administrator shall have the authority, in its discretion:

(i) to determine the Fair Market Value of the Common Stock;

(ii) to determine the terms and conditions, not inconsistent with the terms of the Plan, of any award granted hereunder. Such terms and conditions include, but are not limited to, the exercise price, the time or times when Options may be exercised (which may be based on performance criteria), any vesting acceleration or waiver of forfeiture restrictions, and any restriction or limitation regarding any Option or the shares of Common Stock relating thereto, based in each case on such factors as the Administrator, in its sole discretion, shall determine;

(iii) to institute an Option Exchange Program;

(iv) to construe and interpret the terms of the Plan and awards granted pursuant to the Plan;

(v) to prescribe, amend and rescind rules and regulations relating to the Plan, including rules and regulations relating to sub-plans established for the purpose of qualifying for preferred tax treatment under foreign tax laws;

(vi) to modify or amend each Option (subject to Article I, subsection 17(b) of the Plan), including the discretionary authority to extend the post-termination exercisability period of Options longer than is otherwise provided for in the Plan; <u>provided</u>, <u>however</u>, that the Board shall not have the power to reprice Options or Stock Purchase Rights once granted, except for adjustments resulting from a stock split, reverse stock split or similar change to the outstanding capital stock;

(vii) to authorize any person to execute on behalf of the Company any instrument required to effect the grant of an Option previously granted by the Administrator;

(viii) to determine the terms and restrictions applicable to Options;

(ix) to allow Optionees to satisfy withholding tax obligations by electing to have the Company withhold from the Shares to be issued upon exercise of an Option that

number of Shares having a Fair Market Value equal to the amount required to be withheld. The Fair Market Value of the Shares to be withheld shall be determined on the date that the amount of tax to be withheld is to be determined. All elections by an Optionee to have Shares withheld for this purpose shall be made in such form and under such conditions as the Administrator may deem necessary or advisable; and

(x) to make all other determinations deemed necessary or advisable for administering the Plan.

(c) *Effect of Administrator's Decision*. The Administrator's decisions, determinations and interpretations shall be final and binding on all Optionees and any other holders of Options.

5. <u>Eligibility</u>. Options may be granted to Service Providers other than Officers (except as set forth in this Article I, Section 5 below). Officers and Directors shall not be eligible to receive Options under this Plan; <u>provided</u>, <u>however</u>, that, notwithstanding anything to the contrary contained in the Plan, Options may be granted to an Officer not previously employed by the Company, as an inducement essential to the individual's entering into an employment contract with the Company.

6. <u>Limitation</u>. Neither the Plan nor any Option shall confer upon an Optionee any right with respect to continuing the Optionee's relationship as a Service Provider with the Company, nor shall they interfere in any way with the Optionee's right or the Company's right to terminate such relationship at any time, with or without cause.

7. Term of Option. The term of each Option shall be stated in the Option Agreement.

8. Option Exercise Price and Consideration.

(a) *Exercise Price*. The per share exercise price for the Shares to be issued pursuant to exercise of an Option shall be determined by the Administrator.

(b) *Waiting Period and Exercise Dates*. At the time an Option is granted, the Administrator shall fix the period within which the Option may be exercised and shall determine any conditions which must be satisfied before the Option may be exercised.

(c) *Form of Consideration*. The Administrator shall determine the acceptable form of consideration for exercising an Option, including the method of payment. Such consideration may consist entirely of:

(i) cash;

(ii) check;

(iii) promissory note;

(iv) other Shares which (A) in the case of Shares acquired upon exercise of an option, have been owned by the Optionee for more than six months on the date of surrender, and (B) have a Fair Market Value on the date of surrender equal to the aggregate exercise price of the Shares as to which said Option shall be exercised;

(v) a reduction in the amount of any Company liability to the Optionee, including any liability attributable to the Optionee's participation in any Company-sponsored deferred compensation program or arrangement;

(vi) such other consideration and method of payment for the issuance of Shares to the extent permitted by Applicable Laws, including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price to the Company from the sales proceeds before stock is issued; or

(vii) any combination of the foregoing methods of payment.

9. Exercise of Option.

(a) *Procedure for Exercise; Rights as a Shareholder.* Any Option granted hereunder shall be exercisable according to the terms of the Plan and at such times and under such conditions as determined by the Administrator and set forth in the Option Agreement. An Option may not be exercised for a fraction of a Share.

An Option shall be deemed exercised when the Company receives: (i) written or electronic notice of exercise (in accordance with the Option Agreement) from the person entitled to exercise the Option, and (ii) full payment for the Shares with respect to which the Option is exercised. Full payment may consist of any consideration and method of payment authorized by the Administrator and permitted by the Option Agreement and the Plan. Shares issued upon exercise of an Option shall be issued in the name of the Optionee or, if requested by the Optionee, in the name of the Optionee and his or her spouse. Until the Shares are issued (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to the Optioned Stock, notwithstanding the exercise of the Option. The Company shall issue (or cause to be issued) such Shares promptly after the Option is exercised. No adjustment will be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Article I, Section 11 of the Plan.

Exercising an Option in any manner shall decrease the number of Shares thereafter available, both for purposes of the Plan and for sale under the Option, by the number of Shares as to which the Option is exercised.

(b) *Termination of Relationship as a Service Provider*. If an Optionee ceases to be a Service Provider, other than upon the Optionee's death or Disability, the Optionee may exercise his or her Option, but only within such period of time as is specified in the Option Agreement, and only to the extent that the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for three (3) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified by the Administrator, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

(c) *Disability of Optionee*. If an Optionee ceases to be a Service Provider as a result of the Optionee's Disability, the Optionee may exercise his or her Option within such period of time as is specified in the Option Agreement, to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for twelve (12) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified herein, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

(d) *Death of Optionee*. If an Optionee dies while a Service Provider, the Option may be exercised within such period of time as is specified in the Option Agreement (but in no event later than the expiration of the term of such Option as set forth in the Notice of Grant), by the Optionee's estate or by a person who acquires the right to exercise the Option by bequest or inheritance, but only to the extent that the Option is vested on the date of death. In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for twelve (12) months following the Optionee's termination. If, at the time of death, the Optione is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall immediately revert to the Plan. The Option may be exercised by the executor or administrator of the Optionee's estate or, if none, by the person(s) entitled to exercise the Option under the Optionee's will or the laws of descent or distribution. If the Option is not so exercised within the time specified herein, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

(e) *Buyout Provisions*. The Administrator may at any time offer to buy out for a payment in cash or Shares, an Option previously granted based on such terms and conditions as the Administrator shall establish and communicate to the Optionee at the time that such offer is made.

10. <u>Non-Transferability of Options</u>. Unless determined otherwise by the Administrator, an Option may not be sold, pledged, assigned, hypothecated, transferred, or disposed of in any manner other than by will or by the laws of descent or distribution and may be exercised, during the lifetime of the Optionee, only by the Optionee. If the Administrator makes an Option transferable, such Option shall contain such additional terms and conditions as the Administrator deems appropriate.

11. Adjustments Upon Changes in Capitalization, Dissolution, Merger or Asset Sale.

(a) *Changes in Capitalization*. Subject to any required action by the shareholders of the Company, the number of shares of Common Stock covered by each outstanding Option, as well as the price per share of Common Stock covered by each such outstanding Option, shall be proportionately adjusted for any increase or decrease in the number of issued shares of Common Stock resulting from a stock split, reverse stock split, stock dividend, combination or reclassification of the Company; provided, however, that conversion of any convertible securities of the Company shall not be deemed to have been "effected without receipt of consideration." Such adjustment shall be made by the Board, whose determination in that respect shall be final, binding and conclusive. Except as expressly provided herein, no issuance by the Company of shares of stock of any class, or securities convertible into shares of stock of any class, shall affect, and no adjustment by reason thereof shall be made with respect to, the number or price of shares of Common Stock subject to an Option.

(b) *Dissolution or Liquidation*. In the event of the proposed dissolution or liquidation of the Company, the Administrator shall notify each Optionee as soon as practicable prior to the effective date of such proposed transaction. The Administrator in its discretion may provide for an Optionee to have the right to exercise his or her Option until ten (10) days prior to such transaction as to all of the Optioned Stock covered thereby, including Shares as to which the Option would not otherwise be exercisable. In addition, the Administrator may provide that any Company repurchase option applicable to any Shares purchased upon exercise of an Option shall lapse as to all such Shares, <u>provided</u> that the proposed dissolution or liquidation takes place at the time and in the manner contemplated. To the

¹⁰

extent it has not been previously exercised, an Option will terminate immediately prior to the consummation of such proposed action.

(c) *Merger or Asset Sale*. In the event of a merger of the Company with or into another corporation, or the sale of substantially all of the assets of the Company, each outstanding Option shall be assumed or an equivalent option or right substituted by the successor corporation or a Parent or Subsidiary of the successor corporation. In the event that the successor corporation refuses to assume or substitute for the Option, the Optionee shall fully vest in and have the right to exercise the Option as to all of the Optioned Stock, including Shares as to which it would not otherwise be vested or exercisable. If an Option becomes fully vested and exercisable in lieu of assumption or substitution in the event of a merger or sale of assets, the Administrator shall notify the Optionee in writing or electronically that the Option shall be fully vested and exercisable for a period of fifteen (15) days from the date of such notice, and the Option shall terminate upon the expiration of such period. For the purposes of this paragraph, the Optioned Stock, immediately prior to the merger or sale of assets, the option or right confers the right to purchase or receive, for each Share of Optioned Stock, immediately prior to the merger or sale of assets, the consideration (whether stock, cash, or other securities or property) received in the merger or sale of assets by holders of Common Stock for each Share held on the effective date of the transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding Shares); provided, however, that if such consideration received in the merger or sale of assets is not solely common stock of the successor corporation or its Parent, the Administrator may, with the consent of the successor corporation, provide for the consideration to be received upon the exercise of the Option, for each Share of Optioned Stock to be solely common stock of the successor corporation or its Parent equal in fair market value to th

12. *Date of Grant*. The date of grant of an Option shall be, for all purposes, the date on which the Administrator makes the determination granting such Option, or such other later date as is

determined by the Administrator. Notice of the determination shall be provided to each Optionee within a reasonable time after the date of such grant.

13. Conditions Upon Issuance of Shares.

(a) *Legal Compliance*. Shares shall not be issued pursuant to the exercise of an Option unless the exercise of such Option and the issuance and delivery of such Shares shall comply with Applicable Laws and shall be further subject to the approval of counsel for the Company with respect to such compliance.

(b) *Investment Representations*. As a condition to the exercise of an Option the Company may require the person exercising such Option to represent and warrant at the time of any such exercise that the Shares are being purchased only for investment and without any present intention to sell or distribute such Shares if, in the opinion of counsel for the Company, such a representation is required.

14. <u>Inability to Obtain Authority</u>. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained.

15. <u>Reservation of Shares</u>. The Company, during the term of this Plan, will at all times reserve and keep available such number of Shares as shall be sufficient to satisfy the requirements of the Plan.

16. <u>Qualified Domestic Relations Orders</u>.

(a) Anything in the Plan to the contrary notwithstanding, rights under an Option may be assigned to an Alternate Payee to the extent that a QDRO so provides. (The terms "Alternate Payee" and "QDRO" are defined in Article I, subsection 16(c) below.) The assignment of an Option to an Alternate Payee pursuant to a QDRO shall not be treated as having caused a new grant. If an Option is assigned to an Alternate Payee, the Alternate Payee generally has the same rights as the grantee under

the terms of the Plan; <u>provided</u>, <u>however</u>, that (i) the Option shall be subject to the same vesting terms and exercise period as if the Option were still held by the grantee, and (ii) an Alternate Payee may not transfer an Option.

(b) In the event of the Administrator's receipt of a domestic relations order or other notice of adverse claim by an Alternate Payee of a grantee of an Option, transfer of the proceeds of the exercise of such Option, whether in the form of cash, stock or other property, may be suspended. Such proceeds shall thereafter be transferred pursuant to the terms of a QDRO or other agreement between the grantee and Alternate Payee. A grantee's ability to exercise an Option may be barred if the Administrator receives a court order directing the Administrator not to permit exercise.

(c) The word "QDRO" as used in Article I of the Plan shall mean a court order (i) that creates or recognizes the right of the spouse, former spouse or child (an "Alternate Payee") of an individual who is granted an Option to an interest in such Option relating to marital property rights or support obligations and (ii) that the Administrator determines would be a "qualified domestic relations order," as that term is defined in Section 414(p) of the Code and Section 206(d) of the Employee Retirement Income Security Act ("ERISA"), but for the fact that the Plan is not a plan described in Section 3(3) of ERISA.

17. Amendment of Options.

(a) Rights and obligations under any Option granted before amendment of the Plan shall not be impaired by any amendment of the Plan unless (i) the Company requests the consent of the person to whom the Option was granted and (ii) such person consents in writing.

(b) The Board at any time, and from time to time, may amend the terms of any one or more Option; <u>provided</u>, <u>however</u>, that the rights and obligations under any Option shall not be impaired by any such amendment unless (i) the Company requests the consent of the person to whom the Option was granted and (ii) such person consents in writing.

¹³

ARTICLE II.

PROVISIONS APPLICABLE TO OPTIONS GRANTED ON OR AFTER RESTATEMENT DATE

The following provisions of this Article II shall govern awards granted under the Plan after the Effective Time:

1. Purpose.

(a) The purpose of Article II of the Plan is to provide a means by which employees or directors of and consultants to Amgen Inc., a Delaware corporation (the "Company"), and its Affiliates, as defined in Article II, subsection 1(b), directly, or indirectly through Trusts (as defined in Article II, subsection 1(e)), may be given an opportunity to benefit from increases in value of the stock of the Company through the granting of (i) incentive stock options, (ii) nonqualified stock options, (iii) stock bonuses, and (iv) rights to purchase restricted stock, all as defined below.

(b) The word "Affiliate" as used in Article II of the Plan means (i) any parent corporation or subsidiary corporation of the Company, as those terms are defined in Sections 424(e) and (f), respectively, of the Internal Revenue Code of 1986, as amended (together with the regulations and other official guidance promulgated thereunder, the "Code"), or (ii) any domestic eligible entity that is disregarded under Treasury Regulation Section 301.7701-3, as an entity separate from either (I) the Company or (II) any parent corporation or subsidiary corporation of the Company, as those terms are defined in Sections 424(e) and (f), respectively, of the Code.

(c) The Company, by means of Article II of the Plan, seeks to retain the services of persons now employed by or serving as directors or consultants to the Company, to secure and retain the services of persons capable of filling such positions, and to provide incentives for such persons to exert maximum efforts for the success of the Company.

(d) The Company intends that the rights issued under Article II of the Plan shall, in the discretion of the Board of Directors of the Company (the "Board") or any committee to which

responsibility for administration of the Plan has been delegated pursuant to Article II, subsection 2(c), be either (i) stock options granted pursuant to Article II, Sections 5 or 6 hereof, including incentive stock options as that term is used in Section 422 of the Code ("Incentive Stock Options"), or options which do not qualify as Incentive Stock Options ("Nonqualified Stock Options") (together hereinafter referred to as "Options"), or (ii) stock bonuses or rights to purchase restricted stock granted pursuant to Article II, Section 7 hereof (all such rights included in (i) and (ii), collectively "Stock Awards").

(e) The word "Trust" as used in Article II of the Plan shall mean a trust created for the benefit of the employee, director or consultant, his or her spouse, or members of their immediate family. The word optionee shall mean the person to whom the option is granted or the employee, director or consultant for whose benefit the option is granted to a Trust, as the context shall require.

2. Administration.

(a) The Plan shall be administered by the Board unless and until the Board delegates administration to a committee, as provided in Article II, subsection 2(c).

(b) The Board shall have the power, subject to, and within the limitations of, the express provisions of the Plan:

(i) To determine from time to time which of the persons eligible under the Plan shall be granted Stock Awards; when and how Stock Awards shall be granted; whether a Stock Award will be an Incentive Stock Option, a Nonqualified Stock Option, a stock bonus, a right to purchase restricted stock, or a combination of the foregoing; the provisions of each Stock Award granted (which need not be identical), including the time or times when a person shall be permitted to purchase or receive stock pursuant to a Stock Award; and the number of shares with respect to which Stock Awards shall be granted to each such person.

(ii) To construe and interpret the Plan and Stock Awards granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any

Stock Award, in a manner and to the extent it shall deem necessary or expedient to make the Plan fully effective.

(iii) To amend the Plan as provided in Article II, Section 14.

(iv) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company.

(c) The Board may delegate administration of the Plan to a committee composed of not fewer than two (2) members of the Board (the "Committee"). One or more of these members may be non-employee directors and outside directors, if required and as defined by the provisions of Article II, subsections 2(e) and 2(f). If administration is delegated to a Committee, the Committee shall have, in connection with the administration of the Plan, the powers theretofore possessed by the Board (except amendment of any program adopted pursuant to Article II, Section 6 or any Non-Discretionary Director Awards granted thereunder shall only be by action taken by the Board or a committee of one or more members of the Board to which such authority has been specifically delegated by the Board), subject, however, to such resolutions, not inconsistent with the provisions of Article II of the Plan, as may be adopted from time to time by the Board. Notwithstanding anything else in this Article II, subsection 2(c) to the contrary, at any time the Board or the Committee may delegate to a committee of one or more members of consultants or any portion or class thereof, or amend such Stock Awards.

(d) Notwithstanding anything else in the Plan to the contrary, at any time the Board or the Committee may authorize by duly adopted resolution one or more Officers (as defined below) (each a "Delegated Officer") to take the actions described in Article II, subsection 2(b)(i) of the Plan with respect to Options only, subject to, and within the limitations of, the express provisions of Article II of the Plan; <u>provided</u>, <u>however</u>, that a Delegated Officer shall not have the power to (1) grant any Options to himself, any non-employee director, consultant, Trust, other Delegated Officer or Officer, (2) determine the time or times when a person shall be permitted to purchase stock pursuant to the exercise of an Option (i.e., vesting), (3) determine the exercise price of an Option, or (4) grant any Option to a parent

corporation of the Company, as defined in Section 424(e) of the Code. The resolution authorizing a Delegated Officer to act as such shall specify the total number of shares of Common Stock that a Delegated Officer may grant with respect to Options. The exercise price, which shall be not less than 100% of the closing price of the Common Stock of the Company as quoted on the NASDAQ system on the grant date, or in the Board or the Committee's sole discretion, otherwise determined in accordance with applicable provisions of Code Section 409A (the "Option Fair Market Value") and the time or times when a person shall be permitted to purchase stock pursuant to the exercise of an Option shall, however, be set by the Board or the Committee and not by a Delegated Officer to the extent required by Delaware General Corporation Law Section 157 or any other applicable law. The term "Officer" shall include any natural person who is elected as a corporate officer of the Company by the Board.

(e) The term "non-employee director" shall mean a member of the Board who (i) is not currently an officer of the Company or a parent or subsidiary of the Company (as defined in Rule 16a-1(f) promulgated by the Securities and Exchange Commission under Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) or an employee of the Company or a parent or subsidiary of the Company; (ii) does not receive compensation from the Company or a parent or subsidiary of the Company of the Company for services rendered in any capacity other than as a member of the Board (including a consultant) in an amount required to be disclosed to the Company's stockholders under Rule 404 of Regulation S-K promulgated by the Securities and Exchange Commission ("Rule 404"); (iii) does not possess an interest in any other transaction required to be disclosed under Rule 404, as all of these provisions are interpreted by the Securities and Exchange Commission under Rule 16b-3 promulgated under the Exchange Act.

(f) The term "outside director," as used in Article II of the Plan, shall mean an administrator of the Plan, whether a member of the Board or of any Committee to which responsibility for administration of the Plan has been delegated pursuant to Article II, subsection 2(c), who is

considered to be an "outside director" in accordance with the rules, regulations or interpretations of Section 162(m) of the Code.

(g) Any requirement that an administrator of the Plan be a "non-employee director" or "outside director" shall not apply if the Board or the Committee expressly declares that such requirement shall not apply.

3. Shares Subject to the Plan.

(a) Subject to the provisions of Article II, Section 11 relating to adjustments upon changes in stock, the stock that may be issued pursuant to Stock Awards granted under the Plan after the Effective Time shall not exceed in the aggregate 1,950,597 shares of the Company's common stock (the "Common Stock"), plus any forfeited shares and any shares which revert to and become available for issuance under Article II of the Plan pursuant to Article I, Section 3. For purposes of this Article II, subsection 3(a), "forfeited shares" means any shares issued pursuant to Stock Awards made under the Plan which are forfeited to the Company pursuant to the Stock Award's terms and conditions; <u>provided</u>, <u>however</u>, that the term "forfeited shares" shall not include shares as to which the original recipient received any benefits of ownership (other than voting rights).

(b) If any Stock Award shall for any reason expire or otherwise terminate, in whole or in part, without having been exercised in full, the Common Stock not acquired under such Stock Award shall revert to and again become available for issuance under Article II of the Plan.

(c) For purposes of Article II, subsection 3(a), except as to forfeited shares, the payment of cash dividends and dividend equivalents in conjunction with outstanding awards shall not be counted against the shares available for issuance.

(d) An Incentive Stock Option may be granted to an eligible person under the Plan only if the aggregate fair market value (determined at the time the Incentive Stock Option is granted) of the Common Stock with respect to which incentive stock options (as defined by the Code) are exercisable for the first time by such optionee during any calendar year under all such plans of the Company and its Affiliates does not exceed one hundred thousand dollars (\$100,000). If it is determined that an entire

Option or any portion thereof does not qualify for treatment as an Incentive Stock Option by reason of exceeding such maximum, such Option or the applicable portion shall be considered a Nonqualified Stock Option. Notwithstanding anything to the contrary, no Incentive Stock Options shall be granted under Article II of the Plan unless the Company's stockholders approve the Plan within twelve (12) months after the Restatement Date.

4. Eligibility.

(a) Incentive Stock Options may be granted only to employees (including officers) of the Company or its Affiliates. A director of the Company shall not be eligible to receive Incentive Stock Options unless such director is also an employee of the Company or any Affiliate. Stock Awards other than Incentive Stock Options may be granted to employees (including officers) or directors of or consultants to the Company or any Affiliate or to Trusts of any such employee, director or consultant. Notwithstanding any provision of the Plan to the contrary, no Stock Award may be granted to any person who is an employee or director of or consultant to the Company or its Affiliates (other than Abgenix, Inc. or any of its subsidiaries) at the Effective Time.

(b) A director shall in no event be eligible for the benefits of the Plan (other than Non-Discretionary Director Awards, as defined in Article II, Section 6) unless and until such director is expressly declared eligible to participate in the Plan by action of the Board or the Committee, and only if, at any time discretion is exercised by the Board or the Committee in the selection of a director as a person to whom Stock Awards may be granted, or in the determination of the number of shares which may be covered by Stock Awards granted to a director, the Plan complies with the requirements of Rule 16b-3 promulgated under the Exchange Act, as from time to time in effect. The Board shall otherwise comply with the requirements of Rule 16b-3 promulgated under the Exchange Act, as from time to time in effect. Notwithstanding the foregoing, the restrictions set forth in this Article II, subsection 4(b) shall not apply if the Board or Committee expressly declares that such restrictions shall not apply.

(c) No person shall be eligible for the grant of an Incentive Stock Option under the Plan if, at the time of grant, such person owns (or is deemed to own pursuant to Section 424(d) of the

Code) stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or of any of its Affiliates unless the exercise price of such Incentive Stock Option is at least one hundred and ten percent (110%) of the fair market value of the Common Stock at the date of grant and the Incentive Stock Option is not exercisable after the expiration of five (5) years from the date of grant.

(d) Subject to the provisions of Article II, Section 11 relating to adjustments upon changes in Common Stock, no person shall be eligible to be granted Stock Awards covering more than 2,000,000 shares of Common Stock per person per calendar year.

5. Terms of Discretionary Stock Options.

An option granted pursuant to this Article II, Section 5 (a "Discretionary Stock Option") shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate. The provisions of separate Options need not be identical, but each Option shall include (through incorporation of provisions hereof by reference in the Option or otherwise) the substance of each of the following provisions:

(a) No Option shall be exercisable after the expiration of ten (10) years from the date it was granted.

(b) The exercise price of each Incentive Stock Option and each Nonqualified Stock Option shall be not less than one hundred percent (100%) of the Option Fair Market Value of the Common Stock subject to the Option on the date the Option is granted.

(c) The purchase price of Common Stock acquired pursuant to an Option shall be paid, to the extent permitted by applicable statutes and regulations, either: (i) in cash at the time the Option is exercised; or (ii) at the discretion of the Board or the Committee, either at the time of grant or exercise of the Option (A) by delivery to the Company of shares of Common Stock that have been held for the period required to avoid a charge to the Company's reported earnings and valued at the fair market value on the date of exercise, (B) according to a deferred payment or other arrangement with the person to whom the Option is granted or to whom the Option is transferred pursuant to Article II,

²⁰

subsection 5(d), or (C) in any other form of legal consideration that may be acceptable to the Board or the Committee in their discretion, including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price to the Company from the sales proceeds before Common Stock is issued.

In the case of any deferred payment arrangement, interest shall be payable at least annually and shall be charged at not less than the minimum rate of interest necessary to avoid the treatment as interest, under any applicable provisions of the Code, of any amounts other than amounts stated to be interest under the deferred payment arrangement.

(d) An Option granted to a natural person shall be exercisable during the lifetime of such person only by such person, provided that such person during such person's lifetime may designate a Trust to be such person's beneficiary with respect to any Incentive Stock Options and with respect to any Nonqualified Stock Options, and such beneficiary shall, after the death of the person to whom the Option was granted, have all the rights that such person has while living, including the right to exercise the Option. In the absence of such designation, after the death of the person to whom the Option is granted, the Option shall be exercisable by the person or persons to whom the optionee's rights under such Option pass by will or by the laws of descent and distribution.

(e) The total number of shares of Common Stock subject to an Option may, but need not, be allotted in periodic installments (which may, but need not, be equal). From time to time during each of such installment periods, the Option may become exercisable ("vest") with respect to some or all of the shares allotted to that period, and may be exercised with respect to some or all of the shares allotted to such period and/or any prior period as to which the Option was not fully exercised. During the remainder of the term of the Option (if its term extends beyond the end of the installment periods), the Option may be exercised from time to time with respect to any shares then remaining subject to the

Option. The provisions of this Article II, subsection 5(e) are subject to any Option provisions governing the minimum number of shares as to which an Option may be exercised.

(f) The Company may require any optionee, or any person to whom an Option is transferred under Article II, subsection 5(d), as a condition of exercising any such Option: (i) to give written assurances satisfactory to the Company as to such person's knowledge and experience in financial and business matters and/or to employ a purchaser representative who has such knowledge and experience in financial and business matters, and that such person is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Option; and (ii) to give written assurances satisfactory to the Company stating that such person is acquiring the Common Stock subject to the Option for such person's own account and not with any present intention of selling or otherwise distributing the Common Stock. These requirements, and any assurances given pursuant to such requirements, shall be inoperative if: (x) the issuance of the shares upon the exercise of the Option has been registered under a then currently effective registration statement under the Securities Act of 1933, as amended (the "Securities Act"); or (y) as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities law.

(g) An Option shall terminate three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate, unless the Option by its term specifies either (i) that it shall terminate sooner than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate; or (ii) that it may be exercised more than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate; or (ii) that it may be exercised more than three (3) months after termination of the optionee's employment or relationship as a consultant or director with the Company or an Affiliate. This Article II, subsection 5(g) shall not be construed to extend the term of any Option or to permit anyone to exercise the Option after expiration of its term, nor shall it be construed to increase the number of shares as to which any Option is exercisable from the amount exercisable on the date of termination of the optionee's employment or relationship as a consultant or director.

(h) The Option may, but need not, include a provision whereby the optionee may elect at any time during the term of the optionee's employment or relationship as a consultant or director with the Company or any Affiliate to exercise the Option as to any part or all of the shares subject to the Option prior to the stated vesting dates of the Option. Any shares so purchased from any unvested installment or Option may be subject to a repurchase right in favor of the Company or to any other restriction the Board or the Committee determines to be appropriate.

(i) To the extent provided by the terms of an Option, each optionee may satisfy any federal, state or local tax withholding obligation relating to the exercise of such Option by any of the following means or by a combination of such means: (i) tendering a cash payment; (ii) authorizing the Company to withhold from the shares of the Common Stock otherwise issuable to the optionee as a result of the exercise of the Option a number of shares having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding; or (iii) delivering to the Company owned and unencumbered shares of the Common Stock having a fair market value less than or equal to the amount of the Company's required minimum statutory of the Company's required minimum statutory withholding.

6. Non-Discretionary Director Awards.

The Board may from time to time adopt award programs under the Plan providing for the grant of formula or non-discretionary Stock Awards to directors of the Company who are not employees of the Company or any Affiliate ("Non-Discretionary Director Awards"). The terms and conditions of any such program shall be established by the Board in its sole discretion, subject to the terms and conditions of the Plan.

7. Terms of Stock Bonuses and Purchases of Restricted Stock.

Each stock bonus or restricted stock purchase agreement shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate. The terms and conditions of stock bonus or restricted stock purchase agreements may change from time to time, and the terms and conditions of separate agreements need not be identical, but each stock bonus or restricted

²³

stock purchase agreement shall include (through incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions as appropriate:

(a) The purchase price under each stock purchase agreement shall be such amount as the Board or Committee shall determine and designate in such agreement, but the purchase price shall not be less than fifty percent (50%) of the fair market value of the Common Stock on the date such award is made. Notwithstanding the foregoing, the Board or the Committee may determine that eligible participants in the Plan may be awarded stock pursuant to a stock bonus agreement in consideration for past services actually rendered to the Company or for its benefit.

(b) No rights under a stock bonus or restricted stock purchase agreement shall be assignable by any participant under the Plan, either voluntarily or by operation of law, except where such assignment is required by law or expressly authorized by the terms of the applicable stock bonus or restricted stock purchase agreement.

(c) The purchase price of stock acquired pursuant to a stock purchase agreement shall be paid either: (i) in cash at the time of purchase; (ii) at the discretion of the Board or the Committee, according to a deferred payment or other arrangement with the person to whom the Common Stock is sold; or (iii) in any other form of legal consideration that may be acceptable to the Board or the Committee in their discretion; including but not limited to payment of the purchase price pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board which results in the receipt of cash (or a check) by the Company before Common Stock is issued or the receipt of irrevocable instruction to pay the aggregate exercise price of the Company from the sales proceeds before Common Stock is issued. Notwithstanding the foregoing, the Board or the Committee to which administration of the Plan has been delegated may award Common Stock pursuant to a stock bonus agreement in consideration for past services actually rendered to the Company or for its benefit.

(d) Shares of Common Stock sold or awarded under the Plan may, but need not, be subject to a repurchase option in favor of the Company in accordance with a vesting schedule to be determined by the Board or the Committee.

(e) In the event a person ceases to be an employee of or ceases to serve as a director or consultant to the Company or an Affiliate, the Company may repurchase or otherwise reacquire any or all of the shares of Common Stock held by that person which have not vested as of the date of termination under the terms of the stock bonus or restricted stock purchase agreement between the Company and such person.

(f) To the extent provided by the terms of a stock bonus or restricted stock purchase agreement, a participant may satisfy any federal, state or local tax withholding obligation relating to the lapsing of a repurchase option in favor of the Company or vesting of a stock bonus or a restricted stock award by any of the following means or by a combination of such means: (i) tendering a cash payment; (ii) authorizing the Company to withhold from the shares of the Common Stock otherwise deliverable to a participant as a result of the lapsing of a repurchase option in favor of the Company or the vesting of a stock bonus or a restricted stock award a number of shares having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding; or (iii) delivering to the Company owned and unencumbered shares of the Common Stock having a fair market value less than or equal to the amount of the Company's required minimum statutory withholding.

8. Covenants of the Company.

(a) During the terms of the Stock Awards granted under the Plan, the Company shall keep available at all times the number of shares of Common Stock required to satisfy such Stock Awards up to the number of shares of Common Stock authorized under the Plan.

(b) The Company shall seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to issue and sell shares of Common Stock under the Stock Awards granted under the Plan; <u>provided</u>, <u>however</u>, that this undertaking shall not require the Company to register under the Securities Act either the Plan, any Stock Award granted under the Plan or any Common Stock issued or issuable pursuant to any such Stock Award. If, after reasonable efforts, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under

²⁵

the Plan, the Company shall be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Stock Awards unless and until such authority is obtained.

9. Use of Proceeds from Common Stock.

Proceeds from the sale of Common Stock pursuant to Stock Awards granted under the Plan shall constitute general funds of the Company.

10. Miscellaneous.

(a) The Board or Committee shall have the power to accelerate the time during which a Stock Award may be exercised or the time during which a Stock Award or any part thereof will vest, notwithstanding the provisions in the Stock Award stating the time during which it may be exercised or the time during which it will vest. Each Discretionary Stock Option providing for vesting pursuant to Article II, subsection 5(e) may also provide that if the employee's employment or a director's or consultant's affiliation with the Company or an Affiliate of the Company is terminated by reason of death or disability, then the vesting schedule of Discretionary Stock Options granted to such employee, director or consultant or to the Trusts of such employee, director or consultant may be accelerated.

(b) Neither an optionee nor any person to whom an Option is transferred under the provisions of the Plan shall be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares subject to such Option unless and until such person has satisfied all requirements for exercise of the Option pursuant to its terms.

(c) Nothing in the Plan or any instrument executed or Stock Award granted pursuant thereto shall confer upon any eligible employee, consultant, director, optionee or holder of Stock Awards under the Plan any right to continue in the employ of the Company or any Affiliate or to continue acting as a consultant or director or shall affect the right of the Company or any Affiliate to terminate the employment or consulting relationship or directorship of any eligible employee, consultant, director, optionee or holder of Stock Awards under the Plan with or without cause. In the event that a holder of Stock Awards under the Plan is permitted or otherwise entitled to take a leave of absence, the Company shall have the unilateral right to (i) determine whether such leave of absence will be treated as

a termination of employment or relationship as consultant or director for purposes hereof, and (ii) suspend or otherwise delay the time or times at which exercisability or vesting would otherwise occur with respect to any outstanding Stock Awards under the Plan.

(d) Notwithstanding any provision of the Plan to the contrary, the Board or the Committee shall have the power to condition the grant or vesting of stock bonuses and rights to purchase restricted stock under the Plan upon the attainment of performance goals, determined by the Board or the Committee in their respective sole discretion, with respect to any one or more of the following business criteria with respect to the Company, any Affiliate, any division, any operating unit or any product line: (i) return on capital, assets or equity, (ii) sales or revenue, (iii) net income, (iv) cash flow, (v) earnings per share, (vi) adjusted earnings or adjusted net income as defined below, (vii) working capital, (vii) total shareholder return, (ix) economic value or (x) product development, research, in-licensing, out-licensing, litigation, human resources, information services, manufacturing, manufacturing capacity, production, inventory, site development, plant, building or facility development, government relations, product market share, mergers, acquisitions or sales of assets or subsidiaries. "Adjusted net income" and "adjusted earnings" shall mean net income or earnings, as the case may be, for the relevant performance period computed in accordance with accounting principles generally accepted in the U.S. which may be adjusted by the Committee, as specified in writing, for such performance period, at the time a performance goal is established for the performance period, for the following: (a) any item of significant gain or loss for the performance period determined to be related to a change in accounting principle as reflected in the Company's audited consolidated financial statements, (b) amortization expenses associated with acquired intragresses research and development and (d) any other items of significant income or expense which are determined to be appropriate adjustments and are specified in writing by the Committee at the time the goal is established for the performance period. With respect to any stock bo

2	-
2	1
_	1

conditions that qualify such awards as "qualified performance-based compensation" within the meaning of Section 162(m) of the Code. Stock bonuses and rights to purchase restricted stock made in accordance with this Article II, subsection 10(d) shall contain the terms and conditions of Article II, Section 7 above.

11. Adjustments upon Changes in Common Stock.

If any change is made in the Common Stock subject to the Plan, or subject to any Stock Award granted under the Plan (through merger, consolidation, reorganization, recapitalization, stock dividend, dividend in property other than cash, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other transaction not involving the receipt of consideration by the Company), the Plan and outstanding Stock Awards will be appropriately adjusted in the class(es) and maximum number of shares subject to the Plan, the maximum number of shares which may be granted to a participant in a calendar year, the class(es) and number of shares and price per share of stock subject to outstanding Stock Awards, and the number of shares of Common Stock to be granted as Non-Discretionary Director Awards, if any. Such adjustment shall be made by the Board or the Committee, the determination of which shall be final, binding and conclusive. (The conversion of any convertible securities of the Company shall not be treated as a "transaction not involving the receipt of consideration".) The Board or the Committee, in its sole discretion, may accomplish any such adjustment in a manner calculated not to constitute a "modification" of any such Stock Awards (within the meaning of Code Section 409A) that would cause any such Stock Award to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A).

12. Change of Control.

(a) Notwithstanding anything to the contrary in the Plan, in the event of a Change in Control (as hereinafter defined), then, to the extent permitted by applicable law: (i) the time during which Stock Awards become vested shall automatically be accelerated so that the unvested portions of all Stock Awards shall be vested prior to the Change in Control and (ii) the time during which the Options may be exercised shall automatically be accelerated to prior to the Change in Control. Upon and following the acceleration of the vesting and exercise periods, at the election of the holder of the Stock Award, the Stock

Award may be: (x) exercised (with respect to Options) or, if the surviving or acquiring corporation agrees to assume the Stock Awards or substitute similar stock awards, (y) assumed; or (z) replaced with substitute stock awards. Options not exercised, substituted or assumed prior to or upon the Change in Control shall be terminated. The Board or the Committee, in its sole discretion, may cause any such assumption or substitution to be conducted in a manner so as not to constitute an "extension," "renewal" or "modification" (each within the meaning of Code Section 409A) of any such Stock Award that would cause any such Stock Award to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A).

(b) For purposes of Article II of the Plan, a "Change of Control" shall be deemed to have occurred at any of the following times:

(i) upon the acquisition (other than from the Company) by any person, entity or "group," within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act (excluding, for this purpose, the Company or its Affiliates, or any employee benefit plan of the Company or its Affiliates which acquires beneficial ownership of voting securities of the Company), of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of fifty percent (50%) or more of either the then outstanding shares of Common Stock or the combined voting power of the Company's then outstanding voting securities entitled to vote generally in the election of directors; or

(ii) at the time individuals who, as of the Restatement Date, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board, provided that any person becoming a director subsequent to the Restatement Date, whose election, or nomination for election by the Company's stockholders, was approved by a vote of at least a majority of the directors then comprising the Incumbent Board (other than an election or nomination of an individual whose initial assumption of office is in connection with an actual or threatened election contest relating to the election of the Directors of the Company, as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) shall be, for

purposes of Article II of the Plan, considered as though such person were a member of the Incumbent Board; or

(iii) immediately prior to the consummation by the Company of a reorganization, merger, consolidation (in each case, with respect to which persons who were the stockholders of the Company immediately prior to such reorganization, merger or consolidation do not, immediately thereafter, own more than fifty percent (50%) of the combined voting power entitled to vote generally in the election of directors of the reorganized, merged or consolidated company's then outstanding voting securities) or a liquidation or dissolution of the Company or of the sale of all or substantially all of the assets of the Company; or

(iv) the occurrence of any other event which the Incumbent Board in its sole discretion determines constitutes a Change of Control.

13. Qualified Domestic Relations Orders.

(a) (a) Anything in the Plan to the contrary notwithstanding, the Board or the Committee, in its sole discretion, may determine that rights under Stock Awards may be assigned to an Alternate Payee to the extent that a QDRO so provides, and in the event of such determination, the provisions of this Section 13 shall apply. (The terms "<u>Alternate Payee</u>" and "<u>QDRO</u>" are defined in Article II, paragraph 13(c) below.) The assignment of a Stock Award to an Alternate Payee pursuant to a QDRO shall not be treated as having caused a new grant. The transfer of an Incentive Stock Option to an Alternate Payee may, however, cause it to fail to qualify as an Incentive Stock Option. If a Stock Award is assigned to an Alternate Payee, the Alternate Payee generally has the same rights as the grantee under the terms of the Plan; provided however, that (i) the Stock Award shall be subject to the same vesting terms and exercise period as if the Stock Award were still held by the grantee and (ii) an Alternate Payee may not transfer a Stock Award.

(b) In the event of the Plan administrator's receipt of a domestic relations order or other notice of adverse claim by an Alternate Payee of a grantee of a Stock Award, transfer of the proceeds of the exercise of such Stock Award, whether in the form of cash, stock or other property, may

be suspended. Such proceeds shall thereafter be transferred pursuant to the terms of a QDRO or other agreement between the grantee and Alternate Payee. A grantee's ability to exercise a Stock Award may be barred if the Plan administrator receives a court order directing the Plan administrator not to permit exercise.

(c) The word "QDRO" as used in Article II of the Plan shall mean a court order (i) that creates or recognizes the right of the spouse, former spouse or child (an "Alternate Payee") of an individual who is granted a Stock Award to an interest in such Stock Award relating to marital property rights or support obligations and (ii) that the administrator of the Plan determines would be a "qualified domestic relations order," as that term is defined in Section 414(p) of the Code and Section 206(d) of the Employee Retirement Income Security Act ("ERISA"), but for the fact that the Plan is not a plan described in Section 3(3) of ERISA.

14. Amendment of the Plan.

(a) The Board at any time, and from time to time, may amend the Plan. However, except as provided in Article II, Section 11 relating to adjustments upon changes in the Common Stock, no amendment shall be effective unless approved by the stockholders of the Company within twelve (12) months before or after the adoption of the amendment, where the amendment will:

(i) increase the number of shares reserved for Stock Awards under the Plan;

(ii) modify the requirements as to eligibility for participation in the Plan (to the extent such modification requires stockholder approval in order for the Plan to satisfy the requirements of Section 422(b) of the Code); or

(iii) modify the Plan in any other way if such modification requires stockholder approval in order for the Plan to satisfy the requirements of Section 422(b) of the Code.

(b) The Board may in its sole discretion submit any other amendment to the Plan for stockholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 162(m) of the Code and the regulations promulgated thereunder regarding the

exclusion of performance-based compensation from the limit on corporate deductibility of compensation to certain executive officers.

(c) It is expressly contemplated that the Board may amend the Plan in any respect the Board deems necessary or advisable to provide optionees with the maximum benefits provided or to be provided under the provisions of the Code and the regulations promulgated thereunder relating to employee Incentive Stock Options and/or to bring the Plan and/or Options granted under it into compliance therewith.

(d) Rights and obligations under any Stock Award granted before amendment of the Plan shall not be impaired by any amendment of the Plan, unless: (i) the Company requests the consent of the person to whom the Stock Award was granted; and (ii) such person consents in writing.

(e) Any amendment of the Plan may be accomplished in a manner calculated to cause such amendment not to constitute an "extension," "renewal" or "modification" (each within the meaning of Code Section 409A) of any Stock Awards that would cause such Stock Awards to be considered "nonqualified deferred compensation" (within the meaning of Code Section 409A). Notwithstanding the foregoing, if at any time the Board or the Committee determines that any Stock Award may be subject to Code Section 409A, the Board or the Committee shall have the right, in its sole discretion, and without a Participant's prior consent to amend the Plan or any Stock Award as it may determine is necessary or desirable either for the Plan and Stock Awards to be exempt from the application of Section 409A or to satisfy the requirements of Section 409A, including by adding conditions with respect to the vesting and/or the payment of the Stock Awards.

15. Termination or Suspension of the Plan.

(a) The Board may suspend or terminate the Plan at any time. Unless sooner terminated, the Plan shall terminate on October 4, 2009. No Stock Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

(b) Rights and obligations under any Stock Awards granted while the Plan is in effect shall not be impaired by suspension or termination of the Plan, except with the consent of the person to whom the Stock Award was granted.

16. Code Section 409A.

Except as may be expressly provided with respect to any Stock Award granted under the Plan, the Plan and the Stock Awards are not intended to constitute a "nonqualified deferred compensation plan" within the meaning of Code Section 409A, but rather are intended to be exempt from the application of Code Section 409A. To the extent that the Plan and/or Stock Awards are nevertheless deemed to be subject to Code Section 409A, the Plan and Stock Awards shall be interpreted in accordance with Code Section 409A and Department of Treasury regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the grant of any Stock Award. Notwithstanding any provision of the Plan or any Stock Award to the contrary, in the event that the Committee determines that any Stock Award may be or become subject to Code Section 409A, the Committee may adopt such amendments to the Plan and the affected Stock Award (as described above) or adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), or take any other actions, that the Committee determines are necessary or appropriate to (a) exempt the Plan and any Stock Award from the application of Code Section 409A and/or preserve the intended tax treatment of the benefits provided with respect to the Stock Award, or (b) comply with the requirements of Code Section 409A.

GRANT OF STOCK OPTION AGREEMENT

THE SPECIFIC TERMS OF YOUR STOCK OPTION ARE FOUND IN THE PAGES RELATING TO THE GRANT OF STOCK OPTIONS FOUND ON MERRILL LYNCH BENEFITS WEBSITE (THE "<u>OPTION NOTICE</u>") WHICH ACCOMPANIES THIS DOCUMENT. THE TERMS OF THE OPTION NOTICE ARE INCORPORATED INTO THIS GRANT OF STOCK OPTIONS.

On the Grant Date, specified in the Option Notice, Amgen Inc., a Delaware corporation (the "<u>Company</u>"), has granted to you, the grantee named in the Option Notice, under the plan specified in the Option Notice (the "<u>Plan</u>"), an option to purchase the number of shares of the \$.0001 par value common stock of the Company (the "<u>Common Stock</u>") specified in the Option Notice, pursuant to the terms set forth in this Stock Option Agreement, any special terms and conditions for your country set forth in the attached Appendix A and the Option Notice (together, the "<u>Agreement</u>"). This option is not intended to qualify and will not be treated as an "incentive stock option" within the meaning of Section 422 of the U.S. Internal Revenue Code of 1986, as amended (together with the regulations and other official guidance promulgated thereunder, the "<u>Code</u>").

The provisions of your option are as follows:

I. Subject to the terms and conditions of the Plan and this Agreement, on each Vesting Date the number of shares of Common Stock indicated on the Vesting Schedule shall vest, provided that you have remained continuously and actively employed with the Company or an Affiliate of the Company (as defined in the Plan) through each applicable Vesting Date, unless your employment has terminated due to your Voluntary Termination (as defined in Section IV(5)) or as otherwise determined by the Company in the exercise of its discretion as provided in Section IV(6). This option may only be exercised for whole shares of the Common Stock, and the Company shall be under no obligation to issue any fractional shares of Common Stock to you. Subject to the limitations contained herein, this option shall be exercisable with respect to each installment on or after the applicable Vesting Date. Notwithstanding anything herein to the contrary, the Vesting Schedule may be accelerated (by notice in writing) by the Company in its sole discretion at any time during the term of this option. In addition, if permitted by local law, vesting may be suspended by the Company in its sole discretion during a leave of absence as provided from time to time according to Company policies and practices.

II. (1) The per share exercise price of this option is the Grant Price as defined in the Option Notice, being not less than the fair market value of the Common Stock on the date of grant of this option.

(2) To the extent permitted by applicable statutes and regulations, payment of the exercise price per share is due in full upon exercise of all or any part of each installment which has become exercisable by you by means of (i) cash or a check, (ii) any cashless exercise procedure through the use of a brokerage arrangement approved by the Company, or (iii) any other form of legal consideration that may be acceptable to the Board or the Committee in their discretion.

(3) If at the time of exercise, the Company's Common Stock is publicly traded and quoted regularly in the <u>Wall Street Journal</u>, payment of the exercise price may be made by delivery of already-owned shares of Common Stock of a value equal to the exercise price of the shares of

Common Stock for which this option is being exercised. The already-owned shares must have been owned by you for the period required to avoid adverse accounting treatment and owned free and clear of any liens, claims, encumbrances or security interests. Payment may also be made by a combination of cash and already-owned Common Stock.

Notwithstanding the foregoing, the Company reserves the right to restrict the methods of payment of the exercise price if necessary or advisable to comply with applicable law or regulation, as determined by the Company in its sole discretion.

III. Notwithstanding anything to the contrary contained herein, this option may not be exercised unless the shares issuable upon exercise of this option are then registered under the U.S. Securities Act of 1933, as amended (the "<u>Act</u>"), or, if such shares are not then so registered, the Company has determined that such exercise and issuance would be exempt from the registration requirements of the Act.

IV. The term of this option commences on the Grant Date and, unless sooner terminated as set forth below or in the Plan, terminates on the seventh (7th) anniversary of the date of this option (the "Expiration Date"). This option shall terminate prior to the Expiration Date as follows: three (3) months after the termination of your employment with the Company or an Affiliate of the Company (as defined in the Plan) for any reason or for no reason, including if your employment is terminated by the Company or an Affiliate without cause, or in the event of any other termination of your employment caused directly or indirectly by the Company or an Affiliate, unless:

(1) such termination of your employment is due to your Permanent and Total Disability (as defined below), in which case the option shall terminate on the earlier of the Expiration Date or five (5) years after termination of your employment and the Vesting Schedule of the unvested portions of the option will be accelerated to vest, subject to your execution of a general release and waiver in a form provided by the Company, as of the day preceding such termination of your employment with respect to the option, except that if the option was granted in the calendar year in which such termination occurs, the option will be accelerated to vest with respect to a number of shares equal to the number of shares subject to the option multiplied by a fraction, the numerator of which is the number of complete months you remained continuously and actively employed during such calendar year, and the denominator of which is twelve (12);

(2) such termination of your employment is due to your death, in which case the option shall terminate on the earlier of the Expiration Date or five (5) years after your death and the Vesting Schedule of the unvested portion of the option will be accelerated to vest as of the day preceding your death with respect to the option, except that if the option was granted in the calendar year in which your death occurs the option will be accelerated to vest with respect to a number of shares equal to the number of shares subject to the option multiplied by a fraction, the numerator of which is the number of complete months you remained continuously and actively employed during such calendar year, and the denominator of which is twelve (12);

(3) during any part of such three (3) month period, this option is not exercisable solely because of the condition set forth in Section III above, in which event this option shall not terminate until the earlier of the Expiration Date or until it shall have been exercisable for an

aggregate period of three (3) months after the termination of your employment;

(4) exercise of this option within three (3) months after termination of your employment with the Company or with an Affiliate would result in liability under Section 16(b) of the U.S. Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>"), in which case this option will terminate on the earlier of: (i) the tenth (10th) day after the last date upon which exercise would result in such liability; (ii) six (6) months and ten (10) days after the termination of your employment with the Company or an Affiliate; or (iii) Expiration Date;

(5) such termination of your employment is due to your voluntary termination (and such voluntary termination is not the result of Permanent and Total Disability (as defined below)) after you are at least sixty five (65) years of age, or after you are at least fifty-five (55) years of age and have been an employee of the Company and/or an Affiliate of the Company for at least ten (10) consecutive years ("<u>Voluntary Termination</u>"), in which case this option shall terminate on the earlier of the Expiration Date or five (5) years after termination of your employment and the unvested portions of this option will become exercisable pursuant to the Vesting Schedule without regard to your Voluntary Termination of your employment prior to the Vesting Date, subject to your execution of a general release and waiver in a form provided by the Company, with respect to the option, except that if the option was granted in the calendar year in which your Voluntary Termination occurs, the option will become exercisable pursuant to the Vesting Schedule only with respect to a number of shares equal to the number of shares subject to the option multiplied by a fraction, the numerator of which is the number of complete months you remained continuously and actively employed during such calendar year, and the denominator of which is twelve (12); or

(6) the Company determines, in its sole discretion at any time during the term of this option, in writing, to otherwise extend the period of time during which this option will vest and may be exercised after termination of your employment.

However, in any and all circumstances and except to the extent the Vesting Schedule has been accelerated by the Company in its sole discretion during the term of this option or as a result of your Permanent and Total Disability or death as provided in Sections IV(1) or IV(2) above, respectively, as a result of your Voluntary Termination as provided in Section IV(5) above or as otherwise determined by the Company in the exercise of its discretion as provided in Section IV(6) above, this option may be exercised following termination of your employment only as to that number of shares as to which it was exercisable on the date of termination of your employment under the provisions of Section I of this option. For purposes of this option, (i) "termination of your employment" shall mean the last date you are either an active employee of the Company or an Affiliate or actively engaged as a consultant or director to the Company or an Affiliate, and (ii) "Permanent and Total Disability" shall have the meaning ascribed to such term under Section 22(e)(3) of the Code and with such permanent and total disability being certified prior to termination of your employment by (i) the Social Security Administration, (ii) the companyle governmental authority applicable to an Affiliate of the Company, (iii) such other body having the relevant decision-making power applicable to an Affiliate of the Company, or (iv) an independent medical advisor appointed by the Company in its sole discretion, as applicable, in any such case.

V. (1) To the extent specified above, this option may be exercised by delivering a Notice of Exercise of Stock Option form in person, by mail, via electronic mail or facsimile or by other authorized method, together with the exercise price to the Secretary of the Company, or to such other person as the Company may designate, during regular business hours, together with such additional documents as the Company may then require pursuant to sub-section 5(f) of the Plan.

(2) Regardless of any action the Company or your actual employer (the "<u>Employer</u>") takes with respect to any or all income tax, social insurance, payroll tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to you ("<u>Tax Obligations</u>"), you acknowledge that the ultimate liability for all Tax Obligations is and remains your responsibility and may exceed the amount actually withheld by the Company and/or your Employer. You further acknowledge that the Company and/or your Employer: (a) make no representations or undertakings regarding the treatment of any Tax Obligations in connection with any aspect of the option grant, including, but not limited to, the grant, vesting or exercise of the option, the subsequent sale of shares of Common Stock acquired pursuant to such exercise and the receipt of any dividends; and (b) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the option to reduce or eliminate your liability for Tax Obligations or achieve any particular tax result. Furthermore, if you become subject to tax in more than one jurisdiction between the Grant Date and the date of any relevant taxable event, you acknowledge that the Company and/or your Employer (or former employer, as applicable) may be required to withhold or account for Tax Obligations in more than one jurisdiction.

Prior to any relevant taxable or tax withholding event, as applicable, you shall pay or make adequate arrangements satisfactory to the Company and/or your Employer to satisfy all Tax Obligations. In this regard, you authorize the Company and/or your Employer, or their respective agents, at their discretion, to satisfy all applicable Tax Obligations by one or a combination of the following:

(a) withholding from your wages or other cash compensation paid to you by the Company and/or your Employer; or

(b) withholding from proceeds of the sale of shares of Common Stock acquired upon exercise of the option either through a voluntary sale or through a mandatory sale arranged by the Company (on your behalf pursuant to this authorization).

To avoid adverse accounting treatment, the Company may withhold or account for Tax Obligations not to exceed the applicable minimum statutory withholding rates or other applicable withholding rates.

Finally, you shall pay to the Company or your Employer any amount of Tax Obligations that the Company or your Employer may be required to withhold or account for as a result of your participation in the Plan that cannot be satisfied by the means previously described. You agree to take any further actions and execute any additional documents as may be necessary to effectuate the provisions of this Section V. Notwithstanding anything to the contrary contained herein, the Company may refuse to issue or deliver the shares or the proceeds of the sale of shares of Common Stock if you fail to comply with your obligations in connection with the Tax Obligations.

VI. This option is not transferable, except by will or the laws of descent and distribution, and is exercisable during your life only by you except if you have named a Trust (as defined in the Plan) as beneficiary of this option, this option may be exercised by the Trust after your death.

VII. Any notices provided for in this option or the Plan shall be given in writing and shall be deemed effectively given upon receipt or, in the case of notices delivered by the Company to you, five (5) days after deposit in the United States mail, postage prepaid, addressed to you at the address specified above or at such other address as you hereafter designate by written notice to the Secretary of the Company.

VIII. This option is subject to all the provisions of the Plan and its provisions are hereby made a part of this option, including without limitation the provisions of Section 5 of the Plan relating to option provisions, and is further subject to all interpretations, amendments, rules and regulations which may from time to time be promulgated and adopted pursuant to the Plan. In the event of any conflict between the provisions of this option and those of the Plan, the provisions of the Plan shall control.

IX. You hereby explicitly and unambiguously consent to the collection, use and transfer, in electronic or other form, of your personal data as described in this option by and among, as applicable, your Employer, the Company, or Affiliates of the Company for the exclusive purpose of implementing, administering and managing your participation in the Plan.

You understand that the Company and your Employer may hold certain personal information about you, including, but not limited to, your name, home address and telephone number, date of birth, social insurance number (to the extent permitted under applicable local law) or other identification number, salary, nationality, job title, residency status, any shares of stock or directorships held in the Company, details of all equity compensation or any other entitlement to shares awarded, canceled, vested, unvested or outstanding in your favor, for the purpose of implementing, administering and managing the Plan ("Data"). You understand that Data may be transferred to Merrill Lynch Bank & Trust Co., FSB, or any third parties assisting in the implementation, administration and management of the Plan, that these recipients may be located in your country or elsewhere including outside the European Economic Area, and that the recipient's country (e.g., the United States) may have different data privacy laws and protections than your country. You understand that you may request a list with the names and addresses of any potential recipients of the Data by contacting your local human resources representative. You authorize your Employer, the Company, Affiliates of the Company, Merrill Lynch Bank & Trust Co., FSB, and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering, and managing your participation in the Plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the sole purposes of implementing, administering and managing your participation in the Plan, including any requisite transfer of such Data as may be required to any other broker, escrow agent or other third party with whom the shares received upon exercise of this option may be deposited. You understand that Data will be held only as long as is necessary to implement, administer and manage your participation in the Plan. You understand that you may, at any time, view Data, request additional information about the storage and processing of Data, require any necessary amendments to Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing your local human resources representative. You understand that refusal or withdrawal of consent may affect your ability to participate in the Plan. For more information on the consequences of your refusal to consent or withdrawal of consent, you understand that you may contact your local human resources representative.

X. The terms of this option shall be governed by the laws of the State of Delaware without giving effect to principles of conflicts of laws. For purposes of litigating any dispute that arises hereunder, the parties hereby submit to and consent to the jurisdiction of the State of Delaware, and agree that such litigation shall be conducted in the courts of the State of Delaware, or the federal courts for the United States for the federal district located in the State of Delaware, and no other courts, where this option is made and/or to be performed.

XI. Notwithstanding any provision of this option to the contrary, if you are employed outside the United States by the Company or an Affiliate, are subject to the laws of any foreign jurisdiction, or relocate to one of the countries included in the attached Appendix A (which constitutes a part of this Agreement), the option granted hereunder shall be subject to any special terms and conditions for your country set forth in Appendix A and the following additional terms and conditions:

- a. the terms and conditions of this option, including Appendix A, are deemed modified to the extent necessary or advisable to comply with applicable foreign laws or facilitate the administration to the Plan;
- b. if applicable, the effectiveness of this option is conditioned upon its compliance with any applicable foreign laws, regulations, rules or local governmental regulatory exemption and subject to receipt of any required foreign regulatory approvals; and
- c. the Company may take any other action before or after the date of this option that it deems advisable to obtain approval or comply with any necessary local governmental regulatory exemptions or approvals.

Notwithstanding the foregoing, the Company may not take any actions hereunder, that would violate the Act, the Exchange Act, the Code, or any other securities or tax or other applicable law or regulation. Notwithstanding anything to the contrary contained herein, the shares issuable upon exercise of this option shall not be issued unless such shares are then registered under the Act, or, if such shares are not then so registered, the Company has determined that such exercise and issuance would be exempt from the registration requirements of the Act.

XII. In accepting this option, you acknowledge that:

(1) the Plan is established voluntarily by the Company, it is discretionary in nature and may be modified, amended, suspended or terminated by the Company at any time, as provided in the Plan;

(2) the grant of this option is voluntary and occasional and does not create any contractual or other right to receive future awards of options, or benefits in lieu of options even if options have been awarded repeatedly in the past;

(3) all decisions with respect to future awards, if any, will be at the sole discretion of the Company;

(4) your participation in the Plan is voluntary;

(5) for labor law purposes, options are an extraordinary item that do not constitute compensation of any kind for services of any kind rendered to the Company or to your Employer, and the grant of this option is outside the scope of your employment contract, if any;

(6) for labor law purposes, the grant of options and the underlying shares of Common Stock are not part of normal or expected compensation or salary for any purposes, including, but not limited to, calculation of any severance, resignation, termination, redundancy, dismissal, end of service payments, bonuses, holiday pay, long-service awards, pension or retirement benefits or similar payment and in no event shall be considered as compensation for, or relating in any way to, past services for the Company or any Affiliate of the Company;

(7) the grant of options and the underlying shares of Common Stock are not intended to replace any pension rights or compensation;

(8) neither the grant of options nor any provision of this option, the Plan or the policies adopted pursuant to the Plan confer upon you any right with respect to employment or continuation of current employment and shall not be interpreted to form an employment contract or relationship with the Company or any Affiliate of the Company;

(9) in the event that you are not an employee of the Company or any Affiliate of the Company, options shall not be interpreted to form an employment contract or relationship with the Company or any Affiliate of the Company;

(10) the future value of the underlying shares is unknown and cannot be predicted with certainty;

(11) if the underlying shares of Common Stock do not increase in value, this option will have no value; if you exercise this option and obtain shares of Common Stock, the value of those shares acquired upon exercise may increase or decrease in value, even below the Grant Price per share;

(12) in consideration of the grant of this option, no claim or entitlement to compensation or damages arises from forfeiture of options resulting from termination of your employment by the Company or an Affiliate of the Company (for any reason whatsoever and whether or not in breach of local labor laws) and you irrevocably release the Company and your Employer from any such claim that may arise; if, notwithstanding the foregoing, any such claim is found by a court of competent jurisdiction to have arisen, you shall be deemed irrevocably to have waived your entitlement to pursue such claim;

(13) in the event of termination of your employment (whether or not in breach of local labor laws), your right to receive options and vest under the Plan, if any, will terminate effective as of the date that you are no longer actively employed and will not be extended by any notice period mandated under local law (e.g., active employment would not include a period of "garden leave" or similar period pursuant to local law). Your right, if any, to exercise the options after termination of employment will be measured by the date of termination of your active employment and will not be extended by any notice period mandated under local law; the Committee shall have the exclusive discretion to determine when you are no longer employed for purposes of your option grant;

(14) the Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding your participation in the Plan, or your acquisition or sale of the underlying shares of Common Stock; and

(15) you are hereby advised to consult with your own personal tax, legal and financial advisors regarding your participation in the Plan before taking any action related to the Plan.

XIII. If one or more of the provisions of this option shall be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby and the invalid, illegal or unenforceable provisions shall be deemed null and void; however, to the extent permissible by law, any provisions which could be deemed null and void shall first be construed, interpreted or revised retroactively to permit this option to be construed so as to foster the intent of this option and the Plan.

XIV. If you have received this option or any other document related to the Plan translated into a language other than English and if the meaning of the translated version is different than the English version, the English version will control.

XV. This option is not intended to constitute "nonqualified deferred compensation" within the meaning of Code Section 409A, but rather is intended to be exempt from the application of Code Section 409A. To the extent that this option is nevertheless deemed to be subject to Code Section 409A for any reason, this option shall be interpreted in accordance with Code Section 409A and Department of Treasury regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the Grant Date. Notwithstanding any provision herein to the contrary, in the event that following the Grant Date, the Committee (as defined in the Plan) determines that this option may be or become subject to Code Section 409A, the Committee may adopt such amendments to the Plan and/or this option or adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), or take any other actions, that the Committee determines are necessary or appropriate to (a) exempt the Plan and/or this option from the application of Code Section 409A and/or preserve the intended tax treatment of the benefits provided with respect to this option, or (b) comply with the requirements of Code Section 409A.

XVI. By electing to accept this option, you acknowledge receipt of this option and hereby confirm your understanding that the terms set forth in this option constitute, subject to the terms of the Plan, which terms shall control in the event of any conflict between the Plan and this option, the entire agreement and understanding of the parties with respect to the matters contained herein and supersede any and all prior agreements, arrangements and understandings, both oral and written, between the parties concerning the subject matter of this option. The Company may, in its sole discretion, decide to deliver any documents related to options awarded under the Plan or future option that may be awarded under the Plan by electronic means or request your consent to participate in the Plan by electronic means. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or another third party designated by the Company.

XVII. The Company reserves the right to impose other requirements on your participation in the Plan, on this option and on any shares of Common Stock acquired under the Plan, to the extent the Company determines it is necessary or advisable in order to comply with local law or facilitate the administration of the Plan, and to require you to sign any additional agreements or undertakings that may be necessary to accomplish the foregoing.

Very truly yours,

AMGEN INC.

Duly authorized on behalf of the Board of Directors

RESTRICTED STOCK UNIT AGREEMENT

THE SPECIFIC TERMS OF YOUR GRANT OF RESTRICTED STOCK UNITS ARE FOUND IN THE PAGES RELATING TO THE GRANT OF RESTRICTED STOCK UNITS FOUND ON MERRILL LYNCH BENEFITS WEBSITE (THE "<u>RSU NOTICE</u>") WHICH ACCOMPANIES THIS DOCUMENT. THE TERMS OF THE RSU NOTICE ARE INCORPORATED INTO THIS RESTRICTED STOCK UNIT AGREEMENT.

On the Grant Date specified in the RSU Notice, Amgen Inc., a Delaware corporation (the "<u>Company</u>"), has granted to you, the grantee named in the RSU Notice, under the plan specified in the RSU Notice (the "<u>Plan</u>"), the Number of Units with respect to the number of shares of the \$.0001 par value common stock of the Company (the "<u>Common Stock</u>") specified in the RSU Notice, on the terms and conditions set forth in this Restricted Stock Unit Agreement, any special terms and conditions for your country set forth in the attached Appendix A and the RSU Notice (together, the "<u>Agreement</u>"). The Units shall constitute stock bonuses under Sections 7 and 10(d) of the Plan, which is incorporated herein by reference. Capitalized terms not defined herein shall have the meanings assigned to such terms in the Plan.

I. Vesting Schedule and Termination of Units.

- a. *General.* Subject to the terms and conditions of this Agreement, on each Vesting Date, the Number of Units indicated on the Vesting Schedule shall vest, provided that you have remained continuously and actively employed with the Company or an Affiliate of the Company (as defined in the Plan) through each applicable Vesting Date, unless your employment has terminated due to your Voluntary Termination (as defined in paragraph (d) of this Section I below) or as otherwise determined by the Company in the exercise of its discretion as provided in paragraph (e) of this Section I. The Units represent an unfunded, unsecured promise by the Company to deliver shares of Common Stock. Only whole shares of Common Stock shall be issued upon vesting of the Units, and the Company shall be under no obligation to issue any fractional shares of Common Stock to you. If your employment with the Company or an Affiliate of the Company is terminated for any reason or for no reason, including if your active employment is terminated by the Company or an Affiliate, except as otherwise provided in paragraphs (b), (c), (d) or (e) of this Section I below, your unvested Units shall automatically expire and terminate on the date of termination of your employment. Notwithstanding anything herein to the contrary, the Vesting Schedule may be accelerated (by notice in writing) by the Company in its sole discretion at any time during the term of the Unit. In addition, if permitted under local law, vesting may be suspended by the Company in its sole discretion during a leave of absence as provided from time to time according to Company policies and practices.
- b. *Permanent and Total Disability*. Notwithstanding the provisions in paragraph (a) above, if your employment with the Company or an Affiliate of the Company terminates due to your Permanent and Total Disability (as defined below), then the vesting schedule of unvested portions of Units granted under this Agreement will be accelerated, subject to your execution of a general release and waiver in a form provided by the Company, to vest as of the day preceding such termination of your employment with respect to all Units granted hereunder, except that if the Units were granted in the calendar year in which such termination occurs, the Units will be accelerated to vest with respect to a number of Units equal to the number of Units subject to this Agreement multiplied by a fraction, the numerator of which is the

number of complete months you remained continuously and actively employed during such calendar year, and the denominator of which is twelve (12).

- c. *Death.* Notwithstanding the provisions in paragraph (a) above, if your employment with the Company or an Affiliate of the Company terminates due to your death, then the vesting schedule of unvested portions of Units granted under this Agreement will be accelerated to vest as of the day preceding your death with respect to all Units granted hereunder, except that if the Units were granted in the calendar year in which your death occurs the Units will be accelerated to vest with respect to a number of Units equal to the number of Units subject to this Agreement multiplied by a fraction, the numerator of which is the number of complete months you remained continuously and actively employed during such calendar year, and the denominator of which is twelve (12).
- d. *Retirement*. Notwithstanding the provisions in paragraph (a) above, if you terminate your employment with the Company or an Affiliate of the Company due to your voluntary termination (and such voluntary termination is not the result of Permanent and Total Disability (as defined below)) after you are at least sixty-five (65) years of age, or after you are at least fifty-five (55) years of age and have been an employee of the Company and/or an Affiliate of the Company for at least ten (10) consecutive years ("<u>Voluntary Termination</u>"), then the Units will vest pursuant to the Vesting Schedule without regard to the termination of employment prior to the Vesting Date, subject to your execution of a general release and waiver in a form provided by the Company, with respect to all Units granted hereunder, except that if the Units were granted in the calendar year in which such termination occurs, the Units will vest pursuant to the Vesting Schedule provided in the RSU Notice only with respect to a number of Units equal to the number of Units subject to this Agreement multiplied by a fraction, the numerator of which is the number of complete months you remained continuously and actively employed during such calendar year, and the denominator of which is twelve (12).
- e. *Continued Vesting*. Notwithstanding the provisions in paragraph (a) above, the Company may in its sole discretion at any time during the term of this Agreement, in writing, otherwise provide that the Units will vest pursuant to the Vesting Schedule without regard to the termination of employment prior to the Vesting Date, subject to any terms and conditions that the Company may determine.

For purposes of this Agreement, (i) "<u>termination of your employment</u>" shall mean the last date that you are either an active employee of the Company or an Affiliate or actively engaged as a consultant or director of the Company or an Affiliate, and (ii) "<u>Permanent and Total Disability</u>" shall have the meaning ascribed to such term under Section 22(e)(3) of the Internal Revenue Code of 1986, as amended (together with the regulations and other official guidance promulgated thereunder, the "<u>Code</u>") and with such permanent and total disability being certified prior to termination of your employment by (i) the Social Security Administration, (ii) the comparable governmental authority applicable to an Affiliate of the Company, (iii) such other body having the relevant decision-making power applicable to an Affiliate of the Company, or (iv) an independent medical advisor appointed by the Company in its sole discretion, as applicable, in any such case. Units that remain unvested as of the date of termination of your employment shall expire and terminate on the date of termination of your employment.

II. <u>Form and Timing of Payment</u>. Subject to satisfaction of tax or similar obligations as provided for in Section III, any vested Units shall be paid by the Company in shares of Common Stock (on a one-to-one basis) on, or as soon as practicable after, the applicable Vesting Date (which, for purposes of this Section II, includes the date of any accelerated vesting under Sections I(b), (c), (d) or (e) above); <u>provided</u>, <u>however</u>, that in no event shall the payment be made after the close of your taxable

year which includes the applicable Vesting Date or, if later, after the 15th day of the third calendar month following the applicable Vesting Date. Shares of Common Stock issued in respect of a Unit shall be deemed to be issued in consideration of past services actually rendered by you to the Company or an Affiliate or for its benefit for which you have not previously been compensated or for future services to be rendered, as the case may be, which the Company deems to have a value at least equal to the aggregate par value thereof.

III. <u>Tax Withholding; Issuance of Certificates</u>. Regardless of any action the Company or your actual employer (the "<u>Employer</u>") takes with respect to any or all income tax (including federal, state and local taxes), social insurance, payroll tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to you ("<u>Tax Obligations</u>"), you acknowledge that the ultimate liability for all Tax Obligations is and remains your responsibility and may exceed the amount actually withheld by the Company and/or your Employer. You further acknowledge that the Company and/or your Employer (i) make no representations or undertakings regarding the treatment of any Tax Obligations in connection with any aspect of the Units, including the grant of the Units, the vesting of Units, the conversion of the Units into shares or the receipt of an equivalent cash payment, the subsequent sale of any shares acquired at vesting and the receipt of any dividends; and (ii) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Units to reduce or eliminate your liability for Tax Obligations or achieve any particular tax result. Furthermore, if you become subject to tax in more than one jurisdiction between the Grant Date and the date of any relevant taxable event, you acknowledge that the Company and/or your Employer (or former employer, as applicable) may be required to withhold or account for Tax Obligations in more than one jurisdiction.

Prior to any relevant taxable or tax withholding event, as applicable, you shall pay, or make adequate arrangements satisfactory to the Company or to your Employer (in their sole discretion) to satisfy all Tax Obligations. In this regard, you authorize the Company and/or your Employer or their respective agents, at their discretion, to satisfy all applicable Tax Obligations by one or a combination of the following:

(a) withholding from your wages or other cash compensation paid to you by the Company and/or your Employer; or

(b) withholding from proceeds of the sale of shares of Common Stock acquired upon vesting or payment of the Units either through a voluntary sale or through a mandatory sale arranged by the Company (on your behalf pursuant to this authorization); or

(c) withholding in shares of Common Stock to be issued upon vesting or payment of the Units, provided that the Company and your Employer shall only withhold an amount of shares of Common Stock with a fair market value equal to the Tax Obligations.

To avoid adverse accounting treatment, the Company may withhold or account for Tax Obligations not to exceed the applicable minimum statutory withholding rates or other applicable withholding rates. If the Tax Obligations are satisfied by withholding in shares of Common Stock, for tax purposes, you are deemed to have been issued the full number of shares of Common Stock subject to the vested Units, notwithstanding that a number of the shares of Common Stock is held back solely for the purpose of paying the Tax Obligations due as a result of any aspect of your participation in the Plan (any shares of Common Stock withheld by the Company hereunder shall not be deemed to have been issued by the Company for any purpose under the Plan and shall remain available for issuance thereunder).

Finally, you shall pay to the Company or your Employer any amount of Tax Obligations that the Company or your Employer may be required to withhold or account for as a result of your participation in the Plan that cannot be satisfied by the means previously described. You agree to take any further

actions and execute any additional documents as may be necessary to effectuate the provisions of this Section III. Notwithstanding Section II above, the Company may refuse to issue or deliver the shares or the proceeds of the sale of shares of Common Stock if you fail to comply with your obligations in connection with the Tax Obligations.

IV. <u>Transferability</u>. No benefit payable under, or interest in, this Agreement or in the shares of Common Stock that are scheduled to be issued to you hereunder shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance or charge and any such attempted action shall be void and no such benefit or interest shall be, in any manner, liable for, or subject to, your or your beneficiary's debts, contracts, liabilities or torts; <u>provided</u>, <u>however</u>, nothing in this Section IV shall prevent transfer (i) by will or (ii) by applicable laws of descent and distribution.

V. <u>Notices</u>. Any notices provided for in this Agreement or the Plan shall be given in writing and shall be deemed effectively given upon receipt or, in the case of notices delivered by the Company to you, five (5) days after deposit in the United States mail, postage prepaid, addressed to you at such address as is currently maintained in the Company's records or at such other address as you hereafter designate by written notice to the Secretary of the Company.

VI. <u>Plan</u>. This Agreement is subject to all the provisions of the Plan, which provisions are hereby made a part of this Agreement, including without limitation the provisions of Section 7 of the Plan relating to stock bonuses, and is further subject to all interpretations, amendments, rules and regulations which may from time to time be promulgated and adopted pursuant to the Plan. In the event of any conflict between the provisions of this Agreement and those of the Plan, the provisions of the Plan shall control.

VII. <u>Governing Law</u>. The terms of this Agreement shall be governed by the laws of the State of Delaware without giving effect to principles of conflicts of laws. For purposes of litigating any dispute that arises hereunder, the parties hereby submit to and consent to the jurisdiction of the State of Delaware, and agree that such litigation shall be conducted in the courts of the State of Delaware, or the federal courts for the United States for the federal district located in the State of Delaware, and no other courts, where this Agreement is made and/or to be performed.

VIII. <u>Code Section 409A.</u> The time and form of payment of the Units is intended to comply with the requirements of Code Section 409A and this Agreement shall be interpreted in accordance with Code Section 409A and Department of Treasury regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the Grant Date. Notwithstanding any provision herein to the contrary, in the event that following the Grant Date, the Committee (as defined in the Plan) determines that it may be necessary or appropriate to do so, the Committee may adopt such amendments to the Plan and/or this Agreement or adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), or take any other actions, that the Committee determines are necessary or appropriate to (a) exempt the Plan and/or the Units from the application of Code Section 409A and/or preserve the intended tax treatment of the benefits provided with respect to this option, or (b) comply with the requirements of Code Section 409A.

IX. <u>Acknowledgement</u>. By electing to accept this Agreement, you acknowledge receipt of this Agreement and hereby confirm your understanding that the terms set forth in this Agreement constitute, subject to the terms of the Plan, which terms shall control in the event of any conflict between the Plan and this Agreement, the entire agreement and understanding of the parties with respect to the matters contained herein and supersede any and all prior agreements, arrangements and understandings, both oral and written, between the parties concerning the subject matter of this Agreement. The Company may, in its sole discretion, decide to deliver any documents related to Units awarded under the

Plan or future Units that may be awarded under the Plan by electronic means or request your consent to participate in the Plan by electronic means. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or another third party designated by the Company.

X. Acknowledgment of Nature of Plan and Units. In accepting this Agreement, you acknowledge that:

(1) the Plan is established voluntarily by the Company, it is discretionary in nature and may be modified, amended, suspended or terminated by the Company at any time, as provided in the Plan;

(2) the grant of the Units is voluntary and occasional and does not create any contractual or other right to receive future awards of Units, or benefits in lieu of Units even if Units have been awarded repeatedly in the past;

(3) all decisions with respect to future awards, if any, will be at the sole discretion of the Company;

(4) your participation in the Plan is voluntary;

(5) for labor law purposes, Units are an extraordinary item that do not constitute wages of any kind for services of any kind rendered to the Company or to your Employer, and the grant of Units is outside the scope of your employment contract, if any;

(6) for labor law purposes, the grant of Units and the shares of Common Stock subject to the Units are not part of normal or expected wages or salary for any purposes, including, but not limited to, calculation of any severance, resignation, termination, redundancy, dismissal, end of service payments, bonuses, holiday pay, long-service awards, pension or retirement benefits or similar payments;

(7) the grant of Units and the shares of Common Stock subject to the Units are not intended to replace any pension rights or compensation;

(8) neither the grant of Units nor any provision of this Agreement, the Plan or the policies adopted pursuant to the Plan confer upon you any right with respect to employment or continuation of current employment and shall not be interpreted to form an employment contract or relationship with the Company or any Affiliate of the Company;

(9) the future value of the underlying shares is unknown and cannot be predicted with certainty;

(10) in consideration of the grant of Units hereunder, no claim or entitlement to compensation or damages arises from termination of Units, and no claim or entitlement to compensation or damages shall arise from forfeiture of the Units resulting from termination of your employment by the Company or an Affiliate of the Company (for any reason whatsoever and whether or not in breach of local labor laws) and you irrevocably release the Company and your Employer from any such claim that may arise; if, notwithstanding the foregoing, any such claim is found by a court of competent jurisdiction to have arisen, you shall be deemed irrevocably to have waived your entitlement to pursue such claim;

(11) in the event of termination of your employment (whether or not in breach of local labor laws), your right to receive Units and vest under the Plan, if any, will terminate effective as of the date that you are no longer actively employed and will not be extended by any notice period mandated under local law (e.g., active employment would not include a period of "garden leave" or similar period pursuant to local law); the Company shall have the exclusive discretion to determine when you are no longer actively employed for purposes of your grant;

(12) the Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding your participation in the Plan, or your acquisition or sale of the underlying shares of Common Stock; and

(13) you are hereby advised to consult with your own personal tax, legal and financial advisors regarding your participation in the Plan before taking any action related to the Plan.

XVIII. <u>Compliance with Laws</u>. Notwithstanding any provision of this Agreement to the contrary, if you are employed outside the United States by the Company or an Affiliate, are subject to the laws of any foreign jurisdiction, or relocate to one of the countries included in the attached Appendix A (which constitutes a part of this Agreement), the Units granted hereunder shall be subject to any special terms and conditions for your country set forth in Appendix A and to the following additional terms and conditions:

- a. the terms and conditions of this Agreement, including Appendix A, are deemed modified to the extent necessary or advisable to comply with applicable foreign laws or facilitate the administration of the Plan;
- b. if applicable, the effectiveness of your award of Units is conditioned upon its compliance with any applicable foreign laws, regulations, rules or local governmental regulatory exemption and subject to receipt of any required foreign regulatory approvals;
- c. to the extent necessary to comply with applicable foreign laws, the payment of any earned Units shall be made in cash or Common Stock, at the Company's election; and
- d. the Company may take any other action, before or after an award of Units is made, that it deems advisable to obtain approval or comply with any necessary local governmental regulatory exemptions or approvals.

Notwithstanding the foregoing, the Company may not take any actions hereunder, that would violate the U.S. Securities Act of 1933, as amended (the "<u>Act</u>"), the U.S. Securities Exchange Act of 1934, as amended, the Code, or any other securities or tax or other applicable law or regulation. Notwithstanding anything to the contrary contained herein, the shares issuable upon vesting of the Unit shall not be issued unless such shares are then registered under the Act, or, if such shares are not then so registered, the Company has determined that such vesting and issuance would be exempt from the registration requirements of the Act.

XII. <u>Data Privacy and Notice of Consent</u>. You hereby explicitly and unambiguously consent to the collection, use and transfer, in electronic or other form, of your personal data as described in this Agreement by and among, as applicable, your Employer, the Company, and Affiliates of the Company for the exclusive purpose of implementing, administering and managing your participation in the Plan.

You understand that the Company and your Employer may hold certain personal information about you, including, but not limited to, your name, home address and telephone number, date of birth, social insurance number (to the extent permitted under applicable local law) or other identification

number, salary, nationality, job title, residency status, any shares of stock or directorships held in the Company, details of all equity compensation or any other entitlement to shares awarded, canceled, vested, unvested or outstanding in your favor, for the purpose of implementing, administering and managing the Plan ("<u>Data</u>"). You understand that Data may be transferred to Merrill Lynch Bank & Trust Co., FSB, or any third parties assisting in the implementation, administration and management of the Plan, that these recipients may be located in your country or elsewhere, including outside the European Economic Area and that the recipient's country (e.g., the United States) may have different data privacy laws and protections than your country. You understand that you may request a list with the names and addresses of any potential recipients of the Data by contacting your local human resources representative. You authorize your Employer, the Company, Affiliates of the Company, Merrill Lynch Bank & Trust Co., FSB, and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering and managing your participation in the Plan to receive, posses, use, retain and transfer the Data, in electronic or other form, for the purposes of implementing, administering and managing your participation in the Plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the purposes of implementing, administering and managing your participation in the Plan. You understand that you may understand that Data will be held only as long as is necessary to implement, administer and manage your participation in the Plan. You understand that you may, at any time, view Data, request additional information about the storage and processing of Data, require any necessary amendments to Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing your local human resources representative. You understand that refusal

XIII. <u>Severability</u>. If one or more of the provisions of this Agreement shall be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby and the invalid, illegal or unenforceable provisions shall be deemed null and void; however, to the extent permissible by law, any provisions which could be deemed null and void shall first be construed, interpreted or revised retroactively to permit this Agreement to be construed so as to foster the intent of this Agreement and the Plan.

XIV. <u>Language</u>. If you have received this Agreement or any other document related to the Plan translated into a language other than English and if the meaning of the translated version is different than the English version, the English version will control.

XV. <u>Imposition of Other Requirements</u>. The Company reserves the right to impose other requirements on your participation in the Plan, on the Units and on any shares of Common Stock acquired under the Plan, to the extent the Company determines it is necessary or advisable in order to comply with local law or facilitate the administration of the Plan, and to require you to sign any additional agreements or undertakings that may be necessary to accomplish the foregoing.

Very truly yours, AMGEN INC.

By: Name: Title:

APPENDIX A

ADDITIONAL TERMS AND CONDITIONS OF THE AMENDED AND RESTATED 1991 EQUITY INCENTIVE PLAN AMGEN INC. AMENDED AND RESTATED 1999 EQUITY INCENTIVE PLAN AMGEN INC. AMENDED AND RESTATED 1999 INCENTIVE STOCK PLAN

GRANT OF STOCK OPTION, RESTRICTED STOCK UNITS AND/OR PERFORMANCE UNITS (NON-U.S.)

TERMS AND CONDITIONS

This Appendix includes additional terms and conditions that govern any option to purchase shares of Common Stock ("Option"), Restricted Stock Units ("RSUs") and Performance Units (any of the above individually, a "Stock Award"; collectively, "Stock Awards") under the Plan <u>if, under applicable law, you</u> <u>are a resident of, or are deemed to be a resident of one of the countries listed below. Furthermore, the additional terms and conditions that govern any</u> <u>Stock Award granted hereunder may apply to you if you relocate to one of the countries listed below</u>. Certain capitalized terms used but not defined in this Appendix A shall have the meanings set forth in the Plan and/or the agreement relating to the grant of Options (the "Option Agreement"), Units (the "RSU Agreement") or Performance Units (the "Performance Unit Agreement", and collectively with the Option Agreement and the RSU Agreement, the "Agreements"), as applicable to which this Appendix is attached.

NOTIFICATIONS

This Appendix also includes notifications relating to exchange control and other issues of which you should be aware with respect to your participation in the Plan. The information is based on the exchange control, securities and other laws in effect in the countries to which this Appendix refers as of October 1, 2008. Such laws are often complex and change frequently. As a result, the Company strongly recommends that you not rely on the notifications herein as the only source of information relating to the consequences of your participation in the Plan because the information may be outdated when you acquire shares of Common Stock under the Plan, or when you subsequently sell shares of Common Stock acquired under the Plan.

In addition, the notifications are general in nature and may not apply to your particular situation, and the Company is not in a position to assure you of any particular result. Accordingly, you are advised to seek appropriate professional advice as to how the relevant laws in your country may apply to your situation. Finally, if you are a citizen or resident of a country other than the one in which you are currently working, the information contained herein may not be applicable to you or you may be subject to the provisions of one or more jurisdictions.

AUSTRALIA

TERMS AND CONDITIONS

RSUs and Performance Units Payable Only in Shares. Notwithstanding any discretion in the Plan or anything to the contrary in the Agreements, the award of RSUs or Performance Units does not provide any right for you to receive a cash payment and shall be paid in shares of Common Stock only.

AUSTRIA

NOTIFICATIONS

Consumer Protection Notification. You may be entitled to revoke acceptance of any Stock Awards granted under the Plan on the basis of the Austrian Consumer Protection Act (the "Act") under the conditions listed below, if the Act is considered to be applicable to the Agreements and the Plan:

- (i) If you accept a Stock Award outside the business premises of the Company, you may be entitled to revoke your acceptance of the Stock Award, provided the revocation is made within one (1) week after such acceptance of a Stock Award.
- (ii) The revocation must be in written form to be valid. It is sufficient if you return the applicable Agreement to the Company or the Company's representative with language which can be understood as a refusal to conclude or honor the applicable Agreement, provided the revocation is sent within the period discussed above.

Exchange Control Notification. When you sell shares of Common Stock acquired under the Plan, there may be exchange control obligations if the cash proceeds are held outside of Austria. If the transaction volume of all accounts abroad exceeds \notin 3,000,000, the movements and balances of all accounts must be reported monthly, as of the last day of the month, on or before the 15th day of the following month, on the prescribed form (*Meldungen SI-Forderungen und/oder SI-Verpflichtungen*).

BELGIUM

TERMS AND CONDITIONS

Tax Considerations. Any Option granted hereunder must be accepted in writing within 60 days of the offer (and will be subject to taxation on the 60th day following the offer date of the Option, the offer date being defined as the date on which these documents have been sent to you). If you do not accept the Option in writing within 60 days of the offer, you will be deemed to have refused the grant. Please refer to the Option acceptance letter that you will receive along with the applicable Agreement for a more detailed description of the tax consequences of choosing to accept the option. You should consult your personal tax advisor regarding completion of the additional forms.

NOTIFICATIONS

Tax Reporting Notification. You are required to report any taxable income attributable to any Option granted hereunder on your annual tax return. You are also required to report any bank accounts opened and maintained outside Belgium on your annual tax return.

CANADA

TERMS AND CONDITIONS

Form of Payment. Due to legal restrictions in Canada, you are prohibited from surrendering shares of the Company's Common Stock that you already own or attesting to the ownership of shares of Common Stock to pay the exercise price or any Tax Obligations in connection with the Option.

Termination of Employment. Section XII(12) of the Option Agreement, Section X(11) of the RSU Agreement and Section VIII(10) of the Performance Units Agreement are amended to read as follows:

In the event of involuntary termination of your employment (whether or not in breach of local labor laws), your right to receive any Stock Awards and vest under the Plan, if any, will terminate effective as of the date that is the earlier of: (1) the date you receive notice of termination of employment from the Company or your Employer, or (2) the date you are no longer actively employed by the Company or your Employer regardless of any notice period or period of pay in lieu of such notice required under local law (including, but not limited to statutory law, regulatory law and/or common law). Your right, if any, to acquire shares of Common Stock pursuant to a Stock Award after termination of employment will be measured by the date of termination of your active employment and will not be extended by any notice period mandated under local law; the Committee shall have the exclusive discretion to determine when you are no longer employed for purposes of your Stock Award grants.

The following provisions will apply to you if you are a resident of Quebec:

Language Consent. The parties acknowledge that it is their express wish that this agreement, as well as all documents, notices, and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English.

Les parties reconnaissent avoir exigé la rédaction en anglais de cette convention ("Agreement"), ainsi que de tous documents exécutés, avis donnés et procédures judiciaries intentées, directement ou indirectement, relativement à ou suite à la présente convention.

Data Privacy Notice and Consent. This provision supplements Section IX of the Option Agreement, Section XII of the RSU Agreement and Section XIII of the Performance Units Agreement:

You hereby authorize the Company and the Company's representative to discuss with and obtain all relevant information from all personnel (professional or not) involved in the administration and operation of the Plan. You further authorize the Company and the Employer to disclose and discuss your participation in the Plan with their advisors. You also authorize the Company and the Employer to record such information and keep it in your employee file.

CZECH REPUBLIC

NOTIFICATIONS

Exchange Control Notification. Proceeds from the sale of shares of Common Stock may be held in a cash account abroad and you are no longer required to report the opening and maintenance of a foreign account to the Czech National Bank (the "CNB"), unless the CNB notifies you specifically that such reporting is required. Upon request of the CNB, you may need to file a notification within 15 days of the end of the calendar quarter in which you acquire shares of Common Stock.

DENMARK

NOTIFICATIONS

Exchange Control Information. If you establish an account holding shares or an account holding cash outside Denmark, you must report the account to the Danish Tax Administration. The form which should be used in this respect can be obtained from a local bank. (These obligations are separate from and in addition to the obligations described below.)

Securities/Tax Reporting Information. If you hold shares of Common Stock acquired under the Plan in a brokerage account with a broker or bank outside Denmark, you are required to inform the Danish Tax Administration about the account. For this purpose, you must file a Form V (*Erklaering V*) with the Danish Tax Administration. The Form V must be signed both by you and by the applicable broker or bank where the account is held. By signing the Form V, the broker or bank undertakes to forward information to the Danish Tax Administration concerning the shares in the account without further request each year. By signing the Form V, you authorize the Danish Tax Administration to examine the account.

In addition, if you open a brokerage account (or a deposit account with a U.S. bank) for the purpose of holding cash outside Denmark, you are also required to inform the Danish Tax Administration about this account. To do so, you must file a Form K (*Erklaering K*) with the Danish Tax Administration. The Form K must be signed both by you and by the applicable broker or bank where the account is held. By signing the Form K, the broker/bank undertakes an obligation, without further request each year, to forward information to the Danish Tax Administration concerning the content of the account. By signing the Form K, you authorize the Danish Tax Administration to examine the account.

If you exercise an Option by means of the cashless method of exercise, you are not required to file a Form V because you will not hold any shares of Common Stock. However, if you open a deposit account with a foreign broker or bank to hold the cash proceeds, you are required to file a Form K as described above.

HONG KONG

NOTIFICATIONS

Securities Law Notification. The offer of a Stock Award subject to the terms and conditions of the Plan and the applicable Agreement, including this Appendix, does not constitute a public offering of securities, and it is available only to employees, directors and consultants of the Company and its Affiliates.

Please be aware that the contents of the applicable Agreement, including this Appendix, and the Plan have not been reviewed by any regulatory authority in Hong Kong. You are advised to exercise caution in relation to any Stock Award granted hereunder and the Plan. If you are in any doubt about any of the contents of the applicable Agreement, including this Appendix, or the Plan, you should obtain independent professional advice.

INDIA

TERMS AND CONDITIONS

Option Exercise Restriction. Due to legal restrictions in India, you will not be permitted to pay the exercise price for shares of Common Stock subject to any Option granted hereunder by a cashless "sell-to-cover" procedure, under which method a number of shares of Common Stock with a value sufficient to cover the exercise price, brokerage fees and any applicable Tax Obligations would be sold upon exercise and you would receive only the remaining shares of Common Stock subject to the exercise price to the exercise price in the future, depending on the development of local law.

Fringe Benefit Tax Obligation. This provision supplements Section V of the Option Agreement, Section III of the RSU Agreement and Section V of the Performance Units Agreement:

Effective for any Stock Awards granted to you on or after October 1, 2008, by accepting the Stock Award, you consent and agree to assume liability for any fringe benefit tax ("FBT") that may be payable by the Company and/or the Employer in connection with the Stock Award. You understand that the grant of any Stock Awards on or after October 1, 2008 is contingent upon your agreement to assume liability for FBT payable on the Stock Awards. Further, by accepting any Stock Awards granted on or after October 1, 2008, you agree that the Company and/or the Employer may collect the FBT from you by any of the means set forth, as applicable, in Section V(2) of the Option Agreement, Section III of the RSU Agreement and/or Section V of the Performance Units Agreement, or by any other reasonable method established by the Company. You also agree to execute promptly any other consents or elections required to accomplish the foregoing, upon request of the Company.

You understand that, for any Option granted hereunder, the FBT will be calculated based on the difference between the exercise price and the fair market value (as determined under Indian law) of the underlying shares of Common Stock at the time of vesting. Therefore, no FBT will be due if the Option is not "in-the-money" at vesting. On the other hand, if the Option is in-the-money at vesting and the fair market value of the shares of Common Stock decreases between vesting and exercise, you will be liable for FBT on a greater amount than the benefit you will receive at exercise.

NOTIFICATIONS

Exchange Control Notification. If you remit funds out of India to purchase shares of Common Stock at exercise of any Option granted hereunder, you are responsible for complying with applicable exchange control regulations.

You must repatriate the proceeds from the sale of shares of Common Stock acquired under the Plan and any dividends received in relation to the shares of Common Stock to India within 90 days after receipt. You must maintain the foreign inward remittance certificate received from the

bank where the foreign currency is deposited in the event that the Reserve Bank of India or your Employer requests proof of repatriation.

IRELAND

TERMS AND CONDITIONS

Restriction on Type of Shares Issued to Directors. Due to legal restrictions in Ireland, any Stock Awards granted hereunder to a director or shadow director¹ of an Irish Affiliate may only be granted in respect of newly issued shares of Common Stock. In no event may treasury shares be issued to a director or shadow director of an Irish Affiliate in connection with the exercise of an option under the Plan.

Nature of Agreement. This provision supplements Section XII of the Option Agreement, Section X of the RSU Agreement and Section VIII of the Performance Units Agreement:

In accepting any Stock Awards granted hereunder, you acknowledge your understanding and agreement that the benefits received under the Plan will not be taken into account for any redundancy or unfair dismissal claim.

NOTIFICATIONS

Director Notification Requirements. If you are a director, shadow director or secretary of an Irish Affiliate, you must notify the Irish Affiliate in writing within five (5) business days of receiving or disposing of an interest in the Company (*e.g.*, an Option or shares of Common Stock) in the Company, or within five (5) business days of becoming aware of the event giving rise to the notification requirement, or within five (5) business days of becoming a director or secretary if such an interest exists at the time. This notification requirement also applies with respect to the interests of a spouse or minor children (whose interests, if any, will be attributed to the director, shadow director or secretary).

ITALY

TERMS AND CONDITIONS

Option Cashless Exercise Restriction. Due to legal restrictions in Italy, you will be required to pay the exercise price for any shares of Common Stock subject to an Option granted hereunder by a cashless sell-all exercise, such that all shares will be sold immediately upon exercise and the cash proceeds of sale, less the exercise price, any Tax Obligations and broker's fees or commissions, will be remitted to you. The Company reserves the right to provide additional methods of exercise depending on local developments.

Data Privacy Consent. The following provision replaces Section IX of the Option Agreement, Section XII of the RSU Agreement and Section XIII of the Performance Units Agreement:

You hereby explicitly and unambiguously consent to the collection, use, processing and transfer, in electronic or other form, of your personal data as described herein by and among, as

¹ A shadow director is an individual who is not on the board of directors of the Irish Affiliate but who has sufficient control such that the board of directors of the Irish Affiliate acts in accordance with the directions or instructions of the individual.

applicable, the Employer, the Company and any Affiliate for the exclusive purpose of implementing, administering, and managing your participation in the Plan.

You understand that your Employer, the Company and any Affiliate may hold certain personal information about you, including, but not limited to, your name, home address and telephone number, date of birth, social insurance (to the extent permitted under Italian law) or other identification number, salary, nationality, job title, any shares or directorships held in the Company or any Affiliate, details of all option granted, or any other entitlement to shares of Common Stock awarded, canceled, exercised, vested, unvested or outstanding in your favor, for the exclusive purpose of implementing, managing and administering the Plan ("Data").

You also understand that providing the Company with Data is necessary for the performance of the Plan and that your refusal to provide such Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to participate in the Plan. The Controller of personal data processing is Amgen Inc., with registered offices at One Amgen Center Drive, Thousand Oaks, California 91320, U.S.A., and, pursuant to Legislative Decree no. 196/2003, its Representative in Italy for privacy purposes is Amgen Dompe S.p.A., with registered offices at Via Tazzoli, 6 – 20154 Milan, Italy.

You understand that Data will not be publicized, but it may be transferred to banks, other financial institutions, or brokers involved in the management and administration of the Plan. You understand that Data may also be transferred to the independent registered public accounting firm engaged by the Company. You further understand that the Company and/or any Affiliate will transfer Data among themselves as necessary for the purpose of implementing, administering and managing your participation in the Plan, and that the Company and/or any Affiliate may each further transfer Data to third parties assisting the Company in the implementation, administration, and management of the Plan, including any requisite transfer of Data to a broker or other third party with whom you may elect to deposit any shares of Common Stock acquired at vesting of the option. Such recipients may receive, possess, use, retain, and transfer Data in electronic or other form, for the purposes of implementing, administering, and managing your participation in the Plan. You understand that these recipients may be located in or outside the European Economic Area, such as in the United States or elsewhere. Should the Company exercise its discretion in suspending all necessary legal obligations connected with the management and administration of the Plan, it will delete Data as soon as it has completed all the necessary legal obligations connected with the management and administration of the Plan.

You understand that Data processing related to the purposes specified above shall take place under automated or non-automated conditions, anonymously when possible, that comply with the purposes for which Data is collected and with confidentiality and security provisions, as set forth by applicable laws and regulations, with specific reference to Legislative Decree no. 196/2003.

The processing activity, including communication, the transfer of Data abroad, including outside of the European Economic Area, as herein specified and pursuant to applicable laws and regulations, does not require your consent thereto, as the processing is necessary to performance of contractual obligations related to implementation, administration, and management of the Plan. You understand that, pursuant to Section 7 of the Legislative Decree no. 196/2003, you have the right to, including but not limited to, access, delete, update, correct, or terminate, for legitimate reason, the Data processing.

Furthermore, you are aware that Data will not be used for direct-marketing purposes. In addition, Data provided can be reviewed and questions or complaints can be addressed by contacting your local human resources representative.

Acknowledgement of Nature of Agreement. By accepting any Stock Awards granted hereunder, you acknowledge that (1) you have received a copy of the Plan, the applicable Agreement and this Appendix; (2) you have reviewed the applicable documents in their entirety and fully understand the contents thereof; and (3) you accept all provisions of the Plan, the applicable Agreement and this Appendix.

For any Option granted, you further acknowledge that you have read and specifically and explicitly approve, without limitation, the following Sections of the Option Agreement: Section V, Section IX (as replaced by the above consent), Section X, Section XII, Section XIV, and Section XVI.

For any RSUs granted, you further acknowledge that you have read and specifically and explicitly approve, without limitation, the following Sections of the RSU Agreement: Section III, Section IX, Section X, Section XII (as replaced by the above consent), and Section XIV.

For any Performance Units granted, you further acknowledge that you have read and specifically and explicitly approve, without limitation, the following Sections of the Performance Agreement: Section III, Section V, Section VIII, Section XIII (as replaced by the above consent), and Section XV.

MEXICO

TERMS AND CONDITIONS

Acknowledgement of the Agreement. In accepting any Stock Awards granted hereunder, you acknowledge that you have received a copy of the Plan, have reviewed the Plan and the applicable Option Agreement, RSU Agreement and/or Performance Units Agreement, including this Appendix, in their entirety and fully understand and accept all provisions of the Plan and the Agreement, including this Appendix. You further acknowledge that you have read and specifically and expressly approve the terms and conditions of Section XII of the Option Agreement, Section X of the RSU Agreement and/or Section VIII of the Performance Units Agreement, as applicable, in which the following is clearly described and established:

- (1) Your participation in the Plan does not constitute an acquired right.
- (2) The Plan and your participation in the Plan are offered by Amgen Inc. on a wholly discretionary basis.
- (3) Your participation in the Plan is voluntary.
- (4) Amgen Inc. and its Affiliates are not responsible for any decrease in the value of the option granted and/or shares of Common Stock issued under the Plan.

Labor Law Acknowledgement and Policy Statement. In accepting any Stock Awards granted hereunder, you expressly recognize that Amgen Inc., with registered offices at One Amgen Center Drive, Thousand Oaks, California 91320, U.S.A., is solely responsible for the administration of the Plan and that your participation in the Plan and acquisition of shares of Common Stock do not constitute an employment relationship between you and Amgen Inc. since you are participating in the Plan on a wholly commercial basis and your sole employer is Amgen Latin America Services, S.A. de C.V. ("Amgen-Mexico"). Based on the foregoing, you expressly recognize that the Plan and the benefits that you may derive from participation in the Plan do not

establish any rights between you and your employer, Amgen-Mexico, and do not form part of the employment conditions and/or benefits provided by Amgen-Mexico and any modification of the Plan or its termination shall not constitute a change or impairment of the terms and conditions of your employment.

You further understand that your participation in the Plan is as a result of a unilateral and discretionary decision of Amgen Inc.; therefore, Amgen Inc. reserves the absolute right to amend and/or discontinue your participation in the Plan at any time without any liability to you.

Finally, you hereby declare that you do not reserve to yourself any action or right to bring any claim against Amgen Inc. for any compensation or damages regarding any provision of the Plan or the benefits derived under the Plan, and you therefore grant a full and broad release to Amgen Inc., its Affiliates, shareholders, officers, agents or legal representatives with respect to any claim that may arise.

Spanish Translation

Reconocimiento del Otorgamiento. Al aceptar cualquier Otorgamiento de Acciones bajo el presente documento, usted reconoce que ha recibido una copia del Plan, que ha revisado el mismo en su totalidad, así como también el Acuerdo de Opción, el Acuerdo de Unidades y/o el Acuerdo de Rendimiento de Unidades, que sean aplicables, incluyendo este Apéndice, además que comprende y está de acuerdo con todas las disposiciones tanto del Plan y del Otorgamiento, incluyendo este Apéndice. Asimismo, usted reconoce que ha leído y manifiesta específicamente y expresamente la conformidad con los términos y condiciones establecidos en la Sección XII del Acuerdo de Opción, la Sección X del Acuerdo de Unidades y/o la Sección VIII del Acuerdo de Rendimiento de Acciones, según sea aplicable, en los que se establece y describe claramente que:

- (1) Su participación en el Plan de ninguna manera constituye un derecho adquirido.
- (2) El Plan y su participación en el mismo son ofrecidos por Amgen Inc. de forma completamente discrecional.
- (3) Su participación en el Plan es voluntaria.
- (4) Amgen Inc. y sus Afiliados no son responsables de ninguna disminución en el valor de la opción otorgada y/o de las Acciones Comunes emitidas mediante el Plan.

Reconocimiento de la Ley Laboral y Declaración de Política. Al aceptar cualquier Otorgamiento de Acciones bajo el presente, usted reconoce expresamente que Amgen Inc., con oficinas registradas localizadas en One Amgen Center Drive, Thousand Oaks, California 91320, U.S.A., es la única responsable de la administración del Plan y que su participación en el mismo y la adquisición de Acciones Comunes no constituyen de ninguna manera una relación laboral entre usted y Amgen Inc., debido a que su participación en el Plan es únicamente una relación comercial y que su único empleador es Amgen Latin America Services, S.A. de C.V. ("Amgen-México"). Derivado de lo anterior, usted reconoce expresamente que el Plan y los beneficios a su favor que pudieran derivar de la participación en el mismo, no establecen ningún derecho entre usted y su empleador, Amgen – México, y no forman parte de las condiciones laborales y/o los beneficios otorgados por Amgen – México, y cualquier modificación del Plan o la terminación del mismo no constituirá un cambio o desmejora de los términos y condiciones de su trabajo.

Asimismo, usted entiende que su participación en el Plan es resultado de la decisión unilateral y discrecional de Amgen Inc., por lo tanto, Amgen Inc. se reserva el derecho absoluto de modificar y/o descontinuar su participación en el Plan en cualquier momento y sin ninguna responsabilidad para usted.

Finalmente, usted manifiesta que no se reserva ninguna acción o derecho que origine una demanda en contra de Amgen Inc., por cualquier compensación o daños y perjuicios, en relación con cualquier disposición del Plan o de los beneficios derivados del mismo, y en consecuencia usted exime amplia y completamente a Amgen Inc. de toda responsabilidad, como así también a sus Afiliadas, accionistas, directores, agentes o representantes legales con respecto a cualquier demanda que pudiera surgir.

NETHERLANDS

TERMS AND CONDITIONS

Nature of Agreement. This provision supplements Section XII of the Option Agreement, Section X of the RSU Agreement and Section VIII of the Performance Units Agreement:

By accepting any Stock Awards granted hereunder, you acknowledge and agree that (i) any Stock Awards granted hereunder are intended as an incentive for you to remain employed with your Employer and are not intended as remuneration for labor performed; (ii) any Stock Awards granted hereunder are not intended to replace any pension rights or compensation; and (iii) the benefits under the Plan will not automatically transfer to another corporation in the case of a merger, take-over or transfer of liability.

NOTIFICATIONS

Securities Law Notification. You should be aware of Dutch insider-trading rules, which may impact the exercise of any Option granted hereunder and the sale of shares of Common Stock acquired under the Plan. In particular, you may be prohibited from effectuating certain transactions if you have insider information regarding the Company.

By accepting any Stock Awards granted hereunder and participating in the Plan, you acknowledge having read and understood this Securities Law Notification and further acknowledge that it is your responsibility to comply with the following Dutch insider trading rules:

Under Article 46 of the Act on the Supervision of the Securities Trade 1995, anyone who has "inside information" related to the Company is prohibited from effectuating a transaction in securities in or from the Netherlands. "Inside information" is knowledge of a detail concerning the issuer to which the securities relate that is not public and which, if published, would reasonably be expected to affect the stock price, regardless of the development of the price.

Given the broad scope of the definition of inside information, certain employees of the Company working at an Affiliate in the Netherlands (including person eligible to participate in the Plan) may have inside information and, thus, would be prohibited from effectuating a transaction in securities in the Netherlands at a time when in possession of such inside information.

POLAND

NOTIFICATIONS

Exchange Control Notification. Polish residents holding foreign securities (including shares of Common Stock) and maintaining accounts abroad must report information to the National Bank of Poland on transactions and balances of the securities and cash deposited in such accounts if the value of such transactions or balances exceeds $\leq 10,000$. If required, the reports are due on a quarterly basis by the 20th day following the end of each quarter. The reports are filed on special forms available on the website of the National Bank of Poland.

PORTUGAL

NOTIFICATIONS

Exchange Control Notification. If you do not hold the shares of Common Stock acquired under the Plan with a Portuguese financial intermediary, you may need to file a report with the Portuguese Central Bank. If the shares are held by a Portuguese financial intermediary, it will file the report for you.

RUSSIA

TERMS AND CONDITIONS

Securities Law Requirements. Any Stock Awards granted hereunder, the applicable Option Agreement, RSU Agreement and/or Performance Units Agreement, including this Appendix, the Plan and all other materials you may receive regarding your participation in the Plan or any Stock Awards granted hereunder do not constitute advertising or an offering of securities in Russia. The issuance of shares of Common Stock under the Plan has not and will not be registered in Russia; therefore, such shares may not be offered or placed in public circulation in Russia.

In no event will shares of Common Stock acquired under the Plan be delivered to you in Russia; all shares of Common Stock will be maintained on your behalf in the United States.

You are not permitted to sell any shares acquired under the Plan directly to a Russian legal entity or resident.

NOTIFICATIONS

Exchange Control Notification. If you remit funds out of Russia to purchase shares of Common Stock, the funds must be remitted from a foreign currency account in your name at an authorized bank in Russia. This requirement does not apply if you use a cashless exercise procedure such that all or part of the shares subject to the option granted hereunder are sold immediately upon exercise and the proceeds of sale remitted to the Company to cover the exercise price for the purchased shares and any Tax Obligations because, in this case, there is no remittance of funds out of Russia.

With respect to any shares acquired under the Plan, you must repatriate the proceeds from the sale of such shares and any dividends received in relation to such shares to Russia within a reasonably short period after receipt. The sale proceeds and any dividends received must be initially credited to you through a foreign currency account opened in your name at an

authorized bank in Russia. After the funds are initially received in Russia, they may be further remitted to a foreign bank subject to the following limitations: (i) the foreign account may be opened only for individuals; (ii) the foreign account may not be used for business activities; (iii) the Russian tax authorities must be given notice about the opening/closing of each foreign account within one month of the account opening/closing; and (iv) the Russian tax authorities must be given notice of the account balances of such foreign accounts as of the beginning of each calendar year.

SLOVAKIA

NOTIFICATIONS

Exchange Control Information. You are required to notify the Slovak National Bank with respect to the establishment of accounts abroad within 15 days of the end of the calendar year. The notification forms may be found at the Slovak National Bank website (<u>www.nbs.sk</u>). You should consult your personal legal advisor to determine which forms you must submit and when such forms will be due.

SPAIN

TERMS AND CONDITIONS

Labor Law Acknowledgement. The following provision supplements Section XII of the Option Agreement, Section X of the RSU Agreement and Section VIII of the Performance Units Agreement:

By accepting any Stock Awards granted hereunder, you consent to participation in the Plan and acknowledge that you have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion decided to grant any Stock Awards under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any Stock Awards granted will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis other than as expressly set forth in the applicable Option Agreement, RSU Agreement or Performance Units Agreement, including this Appendix. Consequently, you understand that any Stock Awards granted hereunder are given on the assumption and condition that they shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. Further, you understand and freely accept that there is no guarantee that any benefit whatsoever shall arise from any gratuitous and discretionary grant of Stock Awards since the future value of any Stock Awards and the underlying shares of Common Stock is unknown and unpredictable. In addition, you understand that any Stock Awards granted hereunder would not be made but for the assumptions and conditions referred to above; thus, you understand, acknowledge and freely accept that, should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, then any grant of Stock Awards or right to Stock Awards shall be null and void.

NOTIFICATIONS

Exchange Control Notification. When receiving foreign currency payments derived from the ownership of shares (*i.e.*, dividends or sale proceeds), you must inform the financial institution receiving the payment of the basis upon which such payment is made. You will need to provide

the institution with the following information: (i) your name, address, and fiscal identification number; (ii) the name and corporate domicile of the Company; (iii) the amount of the payment and the currency used; (iv) the country of origin; (v) the reasons for the payment; and (vi) further information that may be required.

If you acquire shares of Common Stock under the Plan and wish to import the ownership title of such shares (*i.e.*, share certificates) into Spain, you must declare the importation of such securities to the *Direccion General de Política Comercial y de Inversiones Extranjeras* ("DGPCIE").

SWITZERLAND

NOTIFICATIONS

Securities Law Notification. Any Stock Awards offered hereunder are considered a private offering in Switzerland and is, therefore, not subject to registration in Switzerland.

UNITED KINGDOM

TERMS AND CONDITIONS

Tax Withholding. This provision supplements Section V of the Option Agreement, Section III of the RSU Agreement and Section V of the Performance Units Agreement:

If payment or withholding of the Tax Obligations is not made within 90 days of the relevant taxable event (the "Due Date") or such other period specified in Section 222(1)(c) of the U.K. Income Tax (Earnings and Pensions) Act 2003, the amount of any uncollected Tax Obligations will constitute a loan owed by you to the Employer, effective on the Due Date. You agree that the loan will bear interest at the HM Revenue and Customs official rate of interest and will be immediately due and repayable, and that the Company or the Employer may recover it at any time thereafter by any of the means specified in the Section III of the Agreement.

Notwithstanding the foregoing, if you are an officer or a member of the Board (within the meaning of Section 13(k) of the U.S. Securities Exchange Act of 1934, as amended), the terms of this provision will not apply to you. In the event that you are an officer or a member of the Board and Tax Obligations are not collected from or paid by you by the Due Date, the amount of any uncollected Tax Obligations may constitute a benefit to you on which additional income tax and national insurance contributions may be payable; in this case, you agree that, to the extent required under U.K. law, the Company and/or the Employer may collect any income tax and national insurance contributions due on this additional benefit from you by any of the means specified in Section V of the Option Agreement, Section III of the RSU Agreement and/or Section V of the Performance Units Agreement, as applicable.

Joint Election. As a condition of participating in the Plan and vesting in any Stock Awards granted hereunder, you acknowledge and agree that you shall be liable for the Secondary Class 1 National Insurance Contributions which may be payable by the Company or the Employer (or by any successor to the Company or the Employer) with respect to the acquisition of shares pursuant to any Stock Awards, the assignment or release of any Stock Awards for consideration, or the receipt of any other benefit in connection with the Stock Awards and that liability for the Secondary Class 1 National Insurance Contribution payments shall be transferred to you to the fullest extent permitted by law.

Without limitation to the above, you agree to make an election, in the form specified and/or approved for such election by HM Revenue & Customs, that the liability for the Secondary Class 1 National Insurance Contribution payments on any such gains shall be transferred to you (the "Election"). You further agree to execute such other elections as may be required between you and any successor to the Company and/or the Employer. You hereby authorizes the Company and the Employer to withhold such Secondary Class 1 National Insurance Contributions by any of the means set forth in Section V of the Option Agreement, Section III of the RSU Agreement and/or Section V of the Performance Units Agreement, as applicable.

If you do not make an Election, if approval of the Election has been withdrawn by HM Revenue and Customs, or if such Election is jointly revoked by you and the Company or the Employer, as applicable, any Stock Awards granted hereunder shall, at the discretion of the Company, without any liability to the Company or the Employer, cease vesting and become null and void.

AMGEN INC. SUPPLEMENTAL

RETIREMENT PLAN

(As Amended and Restated Effective January 1, 2009)

AMGEN SUPPLEMENTAL RETIREMENT PLAN

(As Amended and Restated Effective January 1, 2009)

ARTICLE I INTRODUCTION AND PLAN PURPOSE

- 1.1 <u>Purpose</u>. The purpose of this Plan is to provide benefits to employees of the Company and certain of its affiliates and subsidiaries whose Matching Contributions and Nonelective Contributions are limited under the Amgen Retirement and Savings Plan (the "Retirement Plan"), whether because of statutory limitations or because of employee deferrals to the Amgen Nonqualified Deferred Compensation Plan (the "NQDC"), or both. The Company intends that the Plan will provide benefits to a select group of management or highly compensated employees.
- 1.2 <u>History and Effective Date</u>. The Amgen Inc. Supplemental Retirement Plan (the "Plan") was established by Amgen Inc. (the "Company") effective as of January 1, 1993, was amended and restated effective January 1, 1998, and again effective November 1, 1999. The Plan was further amended and restated effective January 1, 2005 to document the merger of the Immunex Key Employee Plan with and into this Plan. The Plan, as set forth herein, is further amended and restated, effective January 1, 2009, subject to any earlier date specifically set forth within the Plan, to incorporate amendments adopted after its January 1, 2005 restatement and to adopt provisions intended to comply with Code Section 409A and related Treasury Regulations and guidance. The Plan shall be operated and interpreted in accordance with this intention. However, if your payments commenced prior to January 1, 2009, or if the Committee determines that all of the events necessary to receive payment have occurred prior to January 1, 2009, you shall receive or continue to receive payments in accordance with the Plan terms in effect on December 31, 2008, to the extent that the Company determines that doing so would comply with a reasonable, good-faith interpretation of Code Section 409A and applicable guidance relating to Code Section 409A. Where payments have not commenced on or before December 31, 2008 because you were treated as not having experienced a "separation from service" under a reasonable, good-faith interpretation of Code Section 409A and applicable guidance, but you would be treated as having experienced a "separation from service" under Treasury Regulation Section 1.409A-1(h) on a date that is on or after April 10, 2007 and on or before December 31, 2008, you will be treated as having experienced a separation from service on December 31, 2008.

ARTICLE II DEFINITIONS

For the purposes of this Plan, the following terms, when capitalized, have the following meanings. Any capitalized term in this Plan that is not defined in this Article II has the meaning given such term in the Retirement Plan.

2.1 Account means the Account maintained by the Company in accordance with Article IV with respect to Plan Credits and Earnings.

2.2 <u>Account Balance Plan</u> means any plan, agreement or arrangement of the Company or any of its Affiliates that is an "account balance plan" as defined in Treasury Regulation Section 1.409A-1(c)(2)(A) and (B).

2.3 <u>Affiliate</u> shall mean, with respect to any entity, all other entities with which the subject entity would be aggregated and treated as a single employer under Code Section 414(b) (controlled group of corporations)

and Code Section 414(c) (a group of trades or businesses, whether or not incorporated, under common control), as applicable.

2.4 Beneficiary means the person, persons or entity entitled under Article VI to receive Plan benefits payable in the event of your death.

2.5 Board means the Board of Directors of the Company.

2.6 Code means the Internal Revenue Code of 1986, as amended.

2.7 Committee means the Compensation Committee of the Company's Board.

2.8 <u>Company</u> means Amgen Inc. or any subsidiary or affiliate of Amgen Inc. selected by the Board or the Committee to participate in the Plan and excludes any disregarded entity pursuant to Treasury Regulations section 301.7701-3, unless such disregarded entity is selected by the Board or Committee to participate in the Plan.

2.9 <u>Compensation</u> has the same meaning as such term has under the Retirement Plan, except that, for purposes of this Plan, Compensation is not limited by the Salary Cap, includes amounts that are deferred into the NQDC.

2.10 Earnings means the amount credited to your Account under Section 4.3 of the Plan.

2.11 <u>Employer</u> means, for the purpose of determining whether you have experienced a Separation from Service, the entity for which you perform services and with respect to which the legally binding right to compensation deferred or contributed under this Plan arises and all of its Affiliates.

2.12 Normal Retirement Date means the first day of the month coinciding with or next following your attainment of age 65.

2.13 NQDC means the Amgen Nonqualified Deferred Compensation Plan.

2.14 Participation Agreement means the agreement you file with the Committee acknowledging the terms of the Plan and enrolling in the Plan.

2.15 Plan means this Amgen Inc. Supplemental Retirement Plan.

2.16 Plan Year means a period beginning on January 1 of each calendar year and continuing through December 31 of such calendar year.

2.17 <u>Plan Credits</u> means the amount credited to your Account under Section 4.2 and, where applicable, also includes Core Credits and Matching Credits that were made to your Account for periods prior to January 1, 2005.

2.18 Retirement Plan means the Amgen Inc. Retirement and Savings Plan.

2.19 Salary Cap means the highest level of compensation that can be considered for the purpose of calculating benefits under Section 401(a)(17) of the Code.

2.20 <u>Separation from Service</u> means the termination of services that you provide to your Employer, whether voluntarily or involuntarily, as determined by the Committee in accordance with Treasury

Regulation Section 1.409A-1(h). In determining whether you have experienced a Separation from Service, the following provisions shall apply:

- (a) Except as otherwise provided in Section 2.20(b), a Separation from Service shall occur when you experience a termination of employment with your Employer. You will be considered to have experienced a termination of employment when the facts and circumstances indicate that either (i) you are not reasonably expected to perform further services for the Employer after a certain date, or (ii) that the level of bona fide services you will perform for the Employer after such date (whether as an employee or as an independent contractor) will permanently decrease to no more than 49% of the average level of bona fide services that you performed (whether as an employee or an independent contractor) over the immediately preceding 36-month period (or full period of services to the Employer if you have been providing services to the Employer for less than 36 months).
- (b) If you are on military leave, sick leave, or other bona fide leave of absence, the employment relationship between you and the Employer shall be treated as continuing intact, provided that the period of such leave does not exceed six months, or longer, so long as you retain a right to reemployment with the Employer under an applicable statute or by contract. If the period of leave exceeds six months and you do not retain a right to reemployment under an applicable statute or by contract, you will incur a Separation from Service as of the first day immediately following the end of such six-month period. However, where your leave of absence is due to your "disability" (as defined below), a 29-month period of absence will be substituted for such six-month period. In applying the provisions of this paragraph, a leave of absence shall be considered a bona fide leave of absence only if there is a reasonable expectation that you will return to perform services for the Employer. For purposes of this Section 2.20(b), "disability" shall mean any medically determinable physical or mental impairment resulting in your inability to perform the duties of your position or any substantially similar position, where such impairment can be expected to result in death or can be expected to last for a continuous period of not less than six months. The determination of whether you have a disability shall be made by the Employer's short-term disability insurance carrier or administrator (or, if none, by the Committee).
- (c) Notwithstanding the foregoing, if you provide services to the Employer as both an employee and a member of the Board, then to the extent permitted by Treasury Regulation Section 1.409A-1(h)(5), the services provided by you as a Board member shall not be taken into account in determining whether you experience a Separation from Service as an employee.

2.21 Spouse means your wife or husband who is lawfully married to you at the time of your death.

2.22 <u>Years of Service</u> means, effective April 1, 2004, a continuous period of employment beginning on your date of hire with the Company and ending on the date your employment with the Company terminates for any reason. You will be credited with one Year of Service for each consecutive 12-month-period beginning on your hire date, and each anniversary thereof, that you remain employed with the Company. If your employment with the Company terminates and you are later rehired, your prior Years of Service under the Plan will be disregarded and your Years of Service for purposes of vesting in Plan Credits after the rehire date will be determined from the date of your rehire until your subsequent termination of employment.

ARTICLE III ELIGIBILITY AND PARTICIPATION

3.1 <u>Eligibility</u>. You are eligible to receive credits in your Account as provided in Section 4.2 of the Plan during the time you are eligible to participate in the Retirement Plan and either your Compensation for the relevant calendar year is in excess of the Salary Cap, or you elect to make a deferral into the NQDC, or both.

3.2 <u>Automatic Participation</u>. Once you satisfy the eligibility requirements under Section 3.1, you will automatically be enrolled in the Plan and eligible to receive Plan Credits under Article IV of the Plan.

3.3 <u>Participation</u>. After you first become eligible, you will continue to participate in the Plan (that is, you will receive Earnings on the balance in your Account) as long as you have not received a distribution of your Account, even if you are no longer eligible to receive credits under the Plan.

ARTICLE IV CREDITS TO YOUR ACCOUNT

4.1 <u>Account</u>. For record keeping purposes only, an Account will be established under Section 4.2 below and maintained on your behalf under the Plan. Your Account will be used solely to determine the amounts to be paid to you under the Plan. Your Account will not constitute or be treated as a trust fund for your benefit.

4.2 <u>Credits</u>. For each year you are eligible, the Company will credit your Account with your share of Plan Credits in an amount equal to (i) ten percent (10%), multiplied by (ii) your Compensation for the year that is not recognized under the Retirement Plan either because it is in excess of the Salary Cap, or deferred under the NQDC, or both.

4.3 <u>Earnings</u>. Your Account will be credited with Earnings with respect to the investments of the Plan Credits credited to your Account. Earnings will be credited at the rate declared by the Committee, acting in its sole discretion, after taking into account the investment performance of the investment vehicles selected by the Committee permits, selected by you from among the investment vehicles available under the Retirement Plan (excluding the Amgen Inc. Stock Fund).

4.4 <u>Vesting of Your Account</u>. Your Account will become fully vested upon termination of your employment with the Company on or after (1) your Normal Retirement Date, (2) the date of your Disability, or (3) your death. If your employment with the Company is terminated for any other reason, your Account will be vested in accordance with the following schedule:

Years of Service	Vested Percentage
Less than 3	0%
3 or more	100%

Notwithstanding the foregoing vesting schedule, if a portion of your Compensation for a year consists of amounts that were deferred under the NQDC, then a portion of that year's Plan Credits in an amount equal to

(i) 10%, multiplied by (ii) the amount of Compensation deferred under the NQDC that would have been taken into account under the Retirement Plan if it had not been deferred, shall be immediately vested.

Any portion of your Account that is not vested on your termination of employment will be permanently forfeited. All Accounts will be subject to the creditors of the Company in the event of the insolvency of the Company.

4.5 <u>Payroll Taxes Upon Vesting</u>. When any portion of your Account becomes vested and nonforfeitable, the Company shall withhold from your current Compensation, in a manner determined by the Company, your share of employment taxes under the Federal Insurance Contribution Act (FICA) and other applicable employment taxes. If necessary, and in accordance with Section 5.4(c) below, the Company may reduce the vested and nonforfeitable portion of your Account to comply with this Section 4.5.

4.6 Determination of Accounts. Your Account will consist of all your credited Plan Credits and Earnings.

4.7 <u>Statement of Accounts</u>. Prior to March 1 of each year or at such other time as determined by the Committee, the Committee will distribute statements to you showing the balance of your Account.

ARTICLE V DISTRIBUTIONS

5.1 <u>Distributions</u>. Following your Separation from Service, the Company will pay you the vested balance of your Account under the Plan. The distribution of your Account will be paid to you in a lump-sum payment as soon as administratively practicable during the Plan Year immediately following the Plan Year in which such Separation from Service occurs, unless you have elected on an Election Form, within the time and manner described below, to receive either (i) a lump-sum payment as soon as administratively practicable in the second Plan Year following the Plan Year in which your Separation from Service occurs, or (ii) installment payments described in Section 5.2. Any election pursuant to this Section 5.1 must be made within 30 days after the date that you become eligible to participate in the Plan, provided that you have not been eligible to participate in this Plan or in any other plan that would be aggregated with this Plan under Treasury Regulation Section 1.409A-1(c) at any time during the 24-month period ending on the date you became eligible to participate in the Plan, in accordance with Treasury Regulation Section 1.409A-2(a)(7)(ii).

5.2 Installment Payments. Installment payments will be paid in substantially equal annual payments, commencing as soon as administratively practicable in the Plan Year immediately following the Plan Year in which you experience a Separation from Service, and ending in the Plan Year that you specify in the Election Form, which shall not be later than the Plan Year that includes the ten-year anniversary of your Separation from Service. However, if your aggregate account balance under all Account Balance Plans is \$100,000 or less upon your Separation from Service, your election to receive installment payments will be disregarded and your vested Account will be paid to you as a lump-sum payment as soon as administratively practicable in the Plan Year immediately following the Plan Year in which you incur a Separation from Service. For purposes of this Plan, the right to receive a benefit payment in annual installments shall be treated as the entitlement to a single payment.

5.3 <u>Distribution Election Changes</u>. Subject to Section 5.5, with respect to your distribution election made pursuant to this Article 5, you may have a one-time opportunity to extend the payment date and/or change the form of payment initially designated, provided that: (i) the new distribution election shall have no effect until at least 12 months after the date on which such election is made, (ii) the payment date must be at least

five years after the previously designated payment date and must involve completion of all payments not later than the end of the Plan Year that includes the tenyear anniversary of your Separation from Service, and (iii) the election must be made at least 12 months prior to the previously designated payment date. The "previously designated payment date" in the preceding sentence shall be January 1 of the Plan Year in which the payment was initially scheduled to occur, which, in the case of installment payments, shall include only the first installment payment.

5.4 <u>Six-Month Delayed Payment</u>. If, at the time of your Separation from Service, you are a "specified employee" (within the meaning of Section 409A of the Code and Treasury Regulation Section 1.409A-1(i)), the Company will not pay or provide any "Specified Benefits" (as defined herein) during the six-month period beginning with the date of your Separation from Service (the "409A Suspension Period"). In the event of your death, however, the Specified Benefits shall be paid to your Beneficiary without regard to the 409A Suspension Period. For purposes of this Plan, "Specified Benefits" are any amounts that would be subject to Section 409A additional taxes if the Company were to pay them, pursuant to this Plan, on account of your Separation from Service. During the 409A Suspension Period, your Account will continue to be credited or debited in accordance with Section 4.3 above until your Account is distributed. Within 14 calendar days after the end of the 409A Suspension Period, you shall be paid a lump-sum payment in cash equal to any Specified Benefits delayed during the 409A Suspension Period.

5.5 <u>Special Transition Rule for 2008</u>. This Section is effective October 1, 2008. With respect to Article 5 only, the Committee shall, in accordance with the rules established by the Committee and pursuant to Notice 2007-86, provide a limited period in 2008 in which you may make a distribution election with respect to your Account (unless you were subject to the special grandfathering rules set forth in Section 1.2). On or before the deadline established by the Committee, which in no event shall be later than December 31, 2008, you shall make any such elections on an Election Form that the Committee provides. Any such distribution election that is accepted by the Committee shall not be treated as a change in either the form or timing of your benefit payment for purposes of Code Section 409A or Section 5.3 or 6.5 of the Plan. If any distribution election that you make in accordance with this Section 5.5 either (i) relates to an amount that would otherwise be paid to you in 2008, or (ii) would cause an amount to be paid to you in 2008, such election shall not be effective with respect to such amount.

5.6 <u>Accelerated Distributions</u>. Distributions may not be accelerated, except as provided in this Section 5.6 and in Section 8.2. Distributions may be accelerated under the following circumstances:

- (a) You have elected to receive any payments under the installment method and subsequently elect to change from installments to a lump-sum distribution, provided the change in the distribution election satisfies the requirements set forth in Section 5.3 or 6.5.
- (b) You become liable for FICA taxes with respect to any portion of your Account, provided that if an accelerated distribution is made pursuant to this paragraph, the amount distributed shall not exceed the aggregate of the FICA taxes imposed on your Account plus any income tax withholding required for the FICA withholdings.
- (c) The Plan fails to meet the requirements of Code Section 409A with respect to any portion of your Account, provided that if an accelerated distribution is made pursuant to this paragraph, the amount that shall be distributed shall not exceed the amount required to be included in income as a result of the failure to comply with Code Section 409A.

5.7 Delayed Distributions. Except as provided in Sections 5.3, 5.4, 5.5, 6.5, and this Section 5.7, payments

may not be delayed. Distributions may be delayed under the following circumstances:

- (a) If the Company reasonably anticipates that the Company's deduction with respect to any distribution from this Plan would be limited or eliminated by application of Code Section 162(m), then to the extent permitted by Treasury Regulation Section 1.409A-2(b)(7)(i), payment shall be delayed as deemed necessary to ensure that the entire amount of any distribution from this Plan is deductible. Any amounts for which distribution is delayed pursuant to this Section shall continue to be credited or debited with additional amounts in accordance with Section 4.3. The delayed amounts (as adjusted for any amounts credited or debited thereon) shall be distributed to you (or your Beneficiary in the event of your death) at the earliest date the Company reasonably anticipates that the deduction of the payment of the amount will not be limited or eliminated by application of Code Section 162(m).
- (b) The Committee may delay payment if it reasonably anticipates that making the payment would violate federal securities laws or other applicable law, provided the Company treats all payments to similarly situated Plan participants on a reasonably consistent basis and the payment is made at the earliest date at which the Committee reasonably anticipates that the making of the payment will not cause a violation.

5.8 <u>Withholding Payroll Taxes</u>. The Company will withhold any taxes required to be withheld from payments made from the Plan to satisfy any federal, state, or local requirements regarding tax withholding.

5.9 <u>Payments to Incompetents</u>. Whenever and as often as any person entitled to receive a distribution under the Plan shall be under a legal disability or, in the sole judgment of the Committee, shall otherwise be unable to care for such distributions to the person's own best interest and advantage, the Committee, in the exercise of its discretion, may direct such distributions to be made in any one or more of the following ways:

- (a) directly to such person;
- (b) to such person's spouse;
- (c) to such person's legal guardian or conservator; or
- (d) to any other person to be held and used for such person's benefit.

The decision of the Committee shall, in each case, be final and binding upon all parties, and any distribution made pursuant to the power herein conferred on the Committee shall, to the extent so made, be a complete discharge of the obligations under the Plan of the Company and the Committee with respect to such person.

ARTICLE VI BENEFICIARY DESIGNATION

6.1 <u>Beneficiary Designation</u>. Your Beneficiary under the Plan will be the same Beneficiary you select under the Retirement Plan. If you change your Beneficiary designation under the Plan will automatically change as well.

6.2 <u>No Beneficiary Designation</u>. If you fail to designate a Beneficiary under the Retirement Plan, or if the Beneficiary you designate dise before you or before complete distribution of your benefits, your designated Beneficiary will be the first of the following classes in which there is a survivor:

- (a) your surviving Spouse;
- (b) your children, except if any of the children predecease you but leave surviving issue, then such issue will take by right of representation the share the parent would have taken if living;
- (c) your estate.

6.3 <u>Death Before Commencement of Benefits</u>. Subject to Section 6.4, any amounts payable to your Beneficiary under the Plan shall be paid in a lump sum unless you elect on an Election Form, within the time and manner set forth in Section 5.1, for such amounts to be payable in up to ten substantially equal annual installment payments. However, if your aggregate account balance under all Account Balance Plans is \$100,000 or less upon your death, your election to receive installment payments will be disregarded and your vested Account will be paid to your Beneficiary as a lump-sum payment. Any lump-sum payment made pursuant to this Section 6.3 shall be made, or installment payments shall commence, within 60 days of your death. For purposes of this Plan, the right to receive a benefit payment in annual installments shall be treated as the entitlement to a single payment.

6.4 <u>Death After Commencement of Benefits</u>. If you die after installment payments have commenced but before your Account is paid in full, your remaining installment payments shall continue and shall be paid to your Beneficiary over the remaining number of years and in the same amounts as payments would have been made to you had you survived.

6.5 <u>Distribution Election Changes</u>. Subject to Section 5.5, with respect to your distribution election made pursuant to this Article 6, you may have a one-time opportunity to extend the payment date and/or change the form of payment initially designated, provided that: (i) the new distribution election shall have no effect until at least 12 months after the date on which such election is made, (ii) the payment date must involve completion of all payments not later than the end of the Plan Year that includes the ten-year anniversary of your death, and (iii) the election must be made at least 12 months prior to the previously designated payment date. The "previously designated payment date" in the preceding sentence shall be January 1 of the Plan Year in which the payment was initially scheduled to occur, which, in the case of installment payments, shall include only the first installment payment.

6.6 Effect of Payment. The distribution to the Beneficiary completely discharges Company's obligations under this Plan.

ARTICLE VII ADMINISTRATION

7.1 <u>Committee</u>; <u>Duties</u>. This Plan is administered by the Committee, or its duly appointed delegate or delegates, who may or may not be employees of the Company. The Committee (or its delegates) has the same duties, discretionary and interpretive authority and rights under this Plan as the Global Benefits Committee, and its appointees and delegates, has under the Retirement Plan; provided, however, nothing in this Section 7.1 shall be construed to impose any fiduciary duty on the Committee or its delegates under ERISA. The decisions or actions of the Committee with respect to any question arising out of or in connection with the administration, interpretation or application of the Plan and the rules or regulations promulgated hereunder will be final, conclusive and binding upon all persons having any interest in the Plan.

7.2 <u>Indemnity of Committee</u>. The Company will indemnify and hold harmless the members of the Committee against any and all claims, loss, damage, expense or liability arising from any action or failure to act with respect to this Plan, except in the case of the Committee's gross negligence or willful misconduct.

7.3 <u>Claims Procedure</u>. The Claims Procedure under the Plan is the same as that under the Retirement Plan, including the arbitration requirements set forth therein.

ARTICLE VIII AMENDMENT AND TERMINATION OF PLAN

8.1 Plan Amendment.

- (a) <u>Generally</u>. The Committee may at any time amend the Plan in whole or in part. No amendment may decrease or restrict the amount accrued in any Account maintained under the Plan through the date of amendment.
- (b) <u>Amendment for 409A Compliance</u>. This Plan is intended to comply with Section 409A of the Code, and the Company shall have complete discretion to interpret and construe this Plan and any associated documents in any manner that establishes an exemption from or otherwise conforms them to the requirements of Section 409A. If, for any reason including imprecision in drafting, any Plan provision does not accurately reflect its intended establishment of an exemption from or compliance with Section 409A of the Code, as demonstrated by consistent interpretations or other evidence of intent, the provision shall be considered ambiguous and shall be interpreted by the Company in a fashion consistent herewith, as determined in the sole and absolute discretion of the Company. The Company reserves the right to unilaterally amend this Plan without your consent in order to accurately reflect its correct interpretation and operation, as well as to maintain an exemption from or compliance with Section 409A of the Code.

8.2 <u>Company's Right to Terminate</u>. Although the Company anticipates that it will continue the Plan for an indefinite period of time, there is no guarantee that the Company will continue the Plan or will not terminate the Plan at any time in the future. Accordingly, by action of its Board of Directors or the Committee, the Company reserves the right to discontinue its sponsorship of the Plan and to terminate the Plan at any time in accordance with one of the following circumstances set forth in subsections (a) through (c) below and in Treasury Regulation Section 1.409A-3(j)(4)(ix):

- (a) The Company may terminate the Plan if the termination and liquidation is not proximate to a downturn in the Company's financial health and:
 - (i) The Plan and all other plans maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) are irrevocably terminated;
 - (ii) No payments other than payments that would otherwise be payable under the terms of the Plan are made within 12 months following the date the Company takes all necessary actions to terminate and liquidate the Plan;
 - (iii) Except with respect to the participants who became entitled to benefits under the terms of the Plan and any other plan maintained by the Company that would be

aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) within the first 12 months following the date such plans are irrevocably terminated, all payments to the participants due under the terms of such plans must be made between the first day of the 13th month and the last day of the 24th month following the date such plans terminated; and

- (iv) The Company does not adopt a plan that would be aggregated with this Plan under Treasury Regulation Section 1.409A-1(c) within three years following the date the Plan is terminated.
- (b) The Company terminates and liquidates the Plan pursuant to irrevocable action taken within 30 days preceding or 12 months following a "change in control event" (defined below), provided that the Plan and all other plans maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) are terminated on the same date with respect to each participant in such plans that experienced the "change in control event," and all such participants receive all benefits payable under such plans within 12 months following the termination date. For purposes of this Section 8.2(b), "change in control event" shall have the meaning set forth in Treasury Regulation Section 1.409A-3(i)(5).
- (c) The Company terminates and liquidates the Plan within 12 months of a corporate dissolution taxed under Code Section 331, or with the approval of a bankruptcy court pursuant to 11 U.S.C. § 503(b)(1)(A), provided that all benefits payable under the Plan are distributed to participants during the earlier of (i) the taxable year in which the amount is actually or constructively received, or (ii) the latest of the calendar year in which (a) the Plan is terminated and liquidated; (b) the benefits are no longer subject to a substantial risk of forfeiture; or (c) the payment first becomes administratively practicable.

ARTICLE IX MISCELLANEOUS

9.1 <u>Unfunded Plan</u>. This Plan is intended to be an unfunded plan for tax law purposes and for purposes of Title I of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), maintained primarily to provide benefits for a select group of management or highly compensated employees. This Plan is not intended to create an investment contract, but to provide tax planning opportunities and retirement benefits to eligible individuals who have elected to participate in the Plan. Eligible individuals are members of management who, by virtue of their position with the Company, are uniquely informed as to the Company's operations and have the ability to materially affect the Company's profitability and operations.

9.2 <u>Unsecured General Creditor</u>. Neither you nor your Beneficiaries, heirs, successors and assigns will have any legal or equitable rights, interest or claims in any property or assets of the Company, nor will they be Beneficiaries of, or have any rights, claims or interests in any life insurance policies, annuity contracts or the proceeds therefrom owned or which may be acquired by the Company. Such policies or other assets of the Company will not be held under any trust for your benefit or that of your Beneficiaries, heirs, successors or assigns, or held in any way as collateral security for the fulfilling of the obligations of the Company under this Plan. Any and all of the Company's assets and policies will be, and remain, the general, unpledged, unrestricted assets of the Company. The Company's obligation under the Plan will be that of an unfunded and unsecured promise of the Company to pay money in the future.

9.3 <u>Trust Fund</u>. The Company will pay all Plan benefits. At its discretion, the Company may establish one or more trusts, with such trustees as the Board may approve, for the purpose of providing for the

payment of such benefits. Such trust or trusts may be irrevocable, but the assets thereof will be subject to the claims of the Company's creditors. To the extent any benefits provided under the Plan are actually paid from any such trust, the Company will have no further obligation with respect thereto, but to the extent not so paid, such benefits will remain the obligation of, and paid by, the Company.

9.4 <u>Code Section 409A</u>. Except to the extent specifically provided within this Plan or any separate written agreement between you and the Employer, you shall be solely responsible for the satisfaction of any taxes with respect to the benefits payable to you under this Plan (including, but not limited to, employment taxes imposed on employees and additional taxes on nonqualified deferred compensation). Although the Company intends and expects that the Plan and its payments and benefits will not give rise to taxes imposed under Section 409A of the Code, neither the Company, nor its employees, directors, or agents shall have any obligation to mitigate or to hold you harmless from any or all of such taxes.

9.5 <u>Nonassignability</u>. Neither you nor any other person may commute, sell, assign, transfer, hypothecate or convey in advance of actual receipt the amounts, if any, payable hereunder, or any part thereof, which are expressly declared to be nonassignable and nontransferable. No part of the amounts payable will, prior to actual payment, be subject to seizure or sequestration for the payment of any debts, judgments, alimony or separate maintenance owed by you or any other person (other than amounts owed to the Company's creditors in the event of the Company's insolvency), nor be transferable by operation of law in the event of the bankruptcy or insolvency of you or any other person (other than the Company).

9.6 <u>Not a Contract of Employment</u>. The terms and conditions of this Plan may not be construed to constitute a contract of employment between you and the Company and you (or your Beneficiary) will have no rights against the Company except as otherwise specifically provided herein. Moreover, nothing in this Plan will be deemed to give you the right to be retained in the service of the Company as an employee or otherwise, or to interfere with the right of the Company to discipline or discharge you at any time.

9.7 <u>Cooperation</u>. You are required to cooperate with the Company by furnishing any and all information requested by the Company in order to facilitate the payment of benefits hereunder.

9.8 <u>Terms</u>. Whenever words are used in this Plan in the masculine they will be construed as though they were used in the feminine in all cases where they would so apply; and whenever any words are used in this Plan in the singular or in the plural, they will be construed as though they were used in the plural or the singular, as the case may be, in all cases where they would so apply.

9.9 <u>Captions</u>. The captions of the articles, sections and paragraphs of this Plan are for convenience only and do not control or affect the meaning or construction of any of its provisions.

9.10 <u>Governing Law</u>. The provisions of this Plan are to be construed and interpreted according to the laws of the State of California to the extent that they have not been preempted by federal law.

9.11 <u>Validity</u>. In case any provision of this Plan is found to be held illegal or invalid for any reason, said illegality or invalidity will not affect the remaining parts hereof, but this

[Remainder of this page intentionally left blank]

IN WITNESS WHEREOF, the Company has signed this amended and restated Plan document as of ______, 2008.

"Company"

Amgen Inc., a Delaware corporation

By: /s/ BRIAN MCNAMEE

Title: Senior Vice President, Human Resources

APPENDIX A

Participating Subsidiaries and Affiliates of Amgen Inc.

- 1. Amgen USA Inc. – January 1, 2002
- Immunex Corporation January 1, 2003
- 2. 3. Immunex Manufacturing Corporation – January 1, 2003
- 4. Immunex Rhode Island Corporation – January 1, 2003
- 5. Amgen Worldwide Services, Inc. – January 1, 2004
- 6. Amgen San Francisco, LLC – January 1, 2005
- Tularik Pharmaceutical Company January 1, 2005 Amgen Fremont Inc. July 1, 2006 7.
- 8.
- 9. Amgen Mountain View Inc. – January 1, 2007

AMGEN INC. EXECUTIVE INCENTIVE PLAN

I. PURPOSE

The purpose of the Amgen Inc. Executive Incentive Plan (the "Plan") is to attract and retain highly qualified individuals; to obtain from each the best possible performance; to establish a performance goal based on objective criteria; to further underscore the importance of achieving business objectives for the short and long term; and to include in such individual's compensation package an annual incentive component which is tied directly to the achievement of those objectives. Such component is intended to qualify as performance-based compensation under Section 162(m) of the Internal Revenue Code of 1986, as amended (the "Code"), so as to be fully deductible by Amgen Inc. and its subsidiary companies (collectively, "Amgen").

II. EFFECTIVE DATE; TERM

The Plan is effective as of January 1, 2003, subject to approval by the affirmative vote of a majority of shares of Amgen Inc.'s common stock, \$.0001 par value, voting at Amgen Inc.'s 2002 annual meeting of stockholders, and shall remain in effect until such time as it shall be terminated by the Compensation Committee of the board of directors or any successor thereto (the "Compensation Committee"). The Plan is amended and restated effective as of January 1, 2009.

III. ELIGIBILITY AND PARTICIPATION

Eligibility to participate in the Plan is limited to senior executives of Amgen. Participants in the Plan ("Participants") shall be elected annually by the Compensation Committee from those eligible to participate in the Plan.

IV. BUSINESS CRITERIA

The Plan's performance goal shall be based upon Amgen's Adjusted Net Income. No award shall be paid unless there is positive Adjusted Net Income for such performance period. "Adjusted Net Income" shall mean net income for such performance period computed in accordance with accounting principles generally accepted in the U.S. which may be adjusted by the Compensation Committee, as specified in writing, for each performance period, at the time the goal is established for the performance period, for the following:

- (1) any item of significant gain or loss for the performance period determined to be related to a change in accounting principle as reflected in Amgen's audited consolidated financial statements;
- (2) amortization expenses associated with acquired intangible assets;
- (3) expenses associated with acquired in-process research and development; and
- (4) any other items of significant income or expense which are determined to be appropriate adjustments and are specified in writing by the Compensation Committee at the time the goal is established for the performance period.

V. PERFORMANCE GOAL

By no later than the latest time permitted by Section 162(m) of the Code (generally, for performance periods of one year or more, no later than 90 days after the commencement of the performance period) and while the performance relating to the performance goal remains substantially uncertain within the meaning of Section 162(m) of the Code, the Compensation Committee shall specify the adjustments which shall be included in determining Adjusted Net Income for such performance period pursuant to Section IV, shall establish the Plan's performance goal for such performance period based upon Adjusted Net Income for such performance period and shall adopt targeted awards for Participants for such performance period.

Subject to the foregoing and to the limitations set forth in Section VI, no awards shall be paid to Participants unless and until the Compensation Committee makes a certification in writing with respect to the attainment of the performance goal as required by Section 162(m) of the Code.

VI. DETERMINATION OF AMOUNTS OF AWARDS

The Compensation Committee may grant an award to a Participant which shall be payable if there is positive Adjusted Net Income. The maximum award payable to each of the Chief Executive Officer and President, if each is a Participant for such performance period, shall be 0.25% (twenty-five hundredths of one percent) of Adjusted Net Income for such period, the maximum award payable to an Executive Vice President, if each is a Participant for such performance period, shall be 0.15% (fifteen hundredths of one percent) of Adjusted Net Income for such period, and the maximum award payable to any other individual Participants shall be 0.10% (one tenth of one percent) of Adjusted Net Income for such period. The maximum total awards payable to all Participants shall be 2.0% (two percent) of Adjusted Net Income for such period.

The Compensation Committee shall have authority to exercise discretion in determining the amount of the targeted award granted to each Participant at the beginning of a performance period, provided that no such targeted award shall exceed the foregoing maximum award limits, and to exercise discretion to reduce the amount of a targeted award which shall be payable to each Participant at the end of each performance period, subject to the terms, conditions and limits of the Plan and of any other written commitment authorized by the Compensation Committee. The Compensation Committee may at any time establish (and once established, rescind, waive or amend) additional conditions and terms of payment of awards (including but not limited to the achievement of other financial, strategic or individual goals, which may be objective or subjective) as it deems desirable in carrying out the purposes of the Plan and may take into account such other factors as it deems appropriate in administering any aspect of the Plan. However, the Compensation Committee shall have no authority to increase the amount of a targeted award granted to any Participant or to pay an award under the Plan if the performance goal has not been satisfied. In determining the amount of any award to be granted or to be paid to any Participant, the Compensation Committee shall give consideration to the contribution which may be or has been made by the Participant to achievement of Amgen's established objectives and such other matters as it shall deem relevant.

The payment of an award to a Participant with respect to a performance period shall be conditioned upon the Participant's employment by Amgen on the last day of the performance period; provided, however, that in the discretion of the Compensation Committee, awards may be paid to Participants who have retired or whose employment has terminated after the beginning of the period for which an award is made, or to the designee or estate of a Participant who died during such period.

VII. FORM OF AWARDS

All awards shall be determined by the Compensation Committee and shall be paid in cash. Before the beginning of each performance period, each Participant may elect that all or part of the Participant's award for that period will be deferred and distributed at a later date under the Amgen Inc. Nonqualified Deferred Compensation Plan subject to the terms of the Amgen Inc. Nonqualified Deferred Compensation Plan.

VIII. PAYMENT OF AWARDS

Awards shall be paid promptly following the end of the performance period; provided, however, that no awards shall be paid unless and until the Compensation Committee certifies, in writing, that the amounts payable with respect to each award, and all awards in the aggregate, do not exceed the limitations set forth in Section VI and that the amount payable to each Participant does not exceed the amount of the targeted award granted to the Participant at the beginning of the performance period. If the Compensation Committee deems it appropriate or advisable, it may request a report from a nationally recognized public accounting firm stating the amount of Adjusted Net Income for such performance period. Notwithstanding the foregoing, awards under this Plan shall in any event be paid no later than the fifteenth day of the third month following the later to occur of (i) the close of the Participant's tax year, or (ii) the close of the Company's tax year, in either case, in which the applicable performance period ends (it being understood that such payment date is intended to comply with the "short-term deferral" exemption from the application of Section 409A of the Internal Revenue Code (together with the regulations and other official guidance promulgated thereunder, the "Code")). If, for any reason, any amounts payable under this Plan are nevertheless deemed to constitute "nonqualified deferred compensation" under Code Section 409A for any reason, then, notwithstanding the foregoing, with respect to any such amounts, the specified payment date applicable to such amounts shall be the year immediately following the applicable Plan Year, provided that to the extent that the Committee reasonably determines that the payment of any such amounts during such year would be subject to the deduction limitations imposed by Section 162(m) of the Code in such year, the payment of such amounts may, in the Committee's sole discretion, be delayed until as soon as reasonably practicable following the date that such amounts would no

IX. SPECIAL AWARDS AND OTHER PLANS

Nothing contained in the Plan shall prohibit Amgen from granting awards or authorizing other compensation to any person under any other plan or authority or limit the authority of Amgen to establish other special awards or incentive compensation plans providing for the payment of incentive compensation to employees (including those employees who are eligible to participate in the Plan).

X. STOCKHOLDER APPROVAL

No awards shall be paid under the Plan unless and until Amgen Inc.'s stockholders shall have approved the Plan and the performance goal as required by Section 162(m) of the Code.

XI. ADMINISTRATION, AMENDMENT AND INTERPRETATION OF THE PLAN

The Compensation Committee shall administer the Plan. The Compensation Committee shall consist solely of two or more members of the board of directors who shall qualify as "outside directors" under Section 162(m) of the Code. The Compensation Committee shall have full power to construe and interpret the Plan, establish and amend rules and regulations for its administration, and perform all other acts relating to the Plan, including the delegation of administrative responsibilities, that it believes reasonable and proper and in conformity with the purposes of the Plan.

The Compensation Committee shall have the right to amend the Plan from time to time or to repeal it entirely or to direct the discontinuance of awards either temporarily or permanently; provided, however, that no amendment of the Plan that changes the maximum award payable to any Participant or all Participants in the aggregate, as set forth in Section VI, or materially amends the definition of Adjusted Net Income as used in Section VI, shall be effective before approval by the affirmative vote of a majority of shares voting at a meeting of the stockholders of Amgen Inc.

Any decision made, or action taken, by the Compensation Committee arising out of or in connection with the interpretation and/or administration of the Plan shall be final, conclusive and binding on all persons affected thereby.

XII. RIGHTS OF PLAN PARTICIPANTS

Neither the Plan, nor the adoption or operation of the Plan, nor any documents describing or referring to the Plan (or any part hereof) shall confer upon any Participant any right to continue in the employ of Amgen or shall interfere with or restrict in any way the rights of Amgen, which are hereby expressly reserved, to discharge any Participant at any time for any reason whatsoever, with or without cause.

No individual to whom an award has been made or any other party shall have any interest in the cash or any other asset of Amgen prior to such amount being paid.

No right or interest of any Participant shall be assignable or transferable, or subject to any claims of any creditor or subject to any lien.

XIII. MISCELLANEOUS

Amgen shall deduct all federal, state and local taxes required by law or Amgen policy from any award paid hereunder.

In no event shall Amgen be obligated to pay to any Participant an award for any period by reason of Amgen's payment of an award to such Participant in any other period, or by reason of Amgen's payment of an award to any other Participant or Participants in such period or in any other period. Nothing contained in this Plan shall confer upon any person any claim or right to any payments hereunder. Such payments shall be made at the sole discretion of the Compensation Committee.

The Plan shall be unfunded. Amounts payable under the Plan are not and will not be transferred into a trust or otherwise set aside. Amgen shall not be required to establish any special or separate fund or to make any other segregation of assets to assure the payment of any award under the Plan. Any accounts under the Plan are for bookkeeping purposes only and do not represent a claim against the specific assets of Amgen.

It is the intent of Amgen that the Plan and awards made hereunder shall satisfy and shall be interpreted in a manner that satisfies any applicable requirements as performance-based compensation within the meaning of Section 162(m) of the Code. Any provision, application or interpretation of the Plan that is inconsistent with this intent to satisfy the standards in Section 162(m) of the Code shall be disregarded.

Any provision of the Plan that is prohibited or unenforceable shall be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions of the Plan.

The Plan and the rights and obligations of the parties to the Plan shall be governed by, and construed and interpreted in accordance with, the law of the State of Delaware (without regard to principles of conflicts of law).

Although the Company intends and expects that the Plan and its payments and benefits will not give rise to taxes imposed under Section 409A of the Code, neither the Company, nor its employees, directors, or agents, shall have any obligation to mitigate or to hold any Participant harmless from any or all of such taxes. The Plan is intended to be exempt from Section 409A of the Code, and the Compensation Committee shall have complete discretion to interpret and construe this Plan and any associated documents in any manner that establishes an exemption from or otherwise conforms them to the requirements of Section 409A. If, for any reason including imprecision in drafting, any Plan provision does not accurately reflect its intended establishment of an exemption from or compliance with Section 409A of the Code, as demonstrated by consistent interpretations or other evidence of intent, the provision shall be considered ambiguous and shall be interpreted by the Compensation Committee in a fashion consistent herewith, as determined in the sole and absolute discretion of the Compensation Committee reserves the right to unilaterally amend this Plan without the consent of any Participant in order to accurately reflect its correct interpretation and operation, as well as to maintain an exemption from or compliance with Section 409A of the Code.

To record the amendment and restatement of the Plan as set forth herein, effective as of January 1, 2009, the Company has caused its authorized officer to execute the same this ______ 2008.

AMGEN INC.

By: /s/ BRIAN MCNAMEE Brian McNamee Senior Vice President, Human Resources

AMGEN INC. EXECUTIVE NONQUALIFIED RETIREMENT PLAN

WHEREAS, Amgen Inc., a Delaware corporation (the "Company") established the Amgen Inc. Executive Nonqualified Retirement Plan effective as of January 1, 2001, to provide supplemental retirement income benefits for a select group of management and highly compensated employees through Company contributions; and

WHEREAS, the Company desires to amend and restate the Plan, effective as of January 1, 2009, subject to any earlier date specifically set forth within the Plan, in order to comply with Section 409A of the Internal Revenue Code and related Treasury Regulations, and to incorporate prior amendments;

NOW, THEREFORE, effective as of January 1, 2009, subject to any earlier date specifically set forth within the Plan, the Plan is hereby amended and restated to read as follows:

ARTICLE I.

TITLE AND DEFINITIONS

1.1 <u>Title</u>.

This Plan shall be known as the Amgen Inc. Executive Nonqualified Retirement Plan.

1.2 Definitions.

Whenever the following words and phrases are used in this Plan, with the first letter capitalized, they shall have the meanings specified below.

(a) "Affiliate" shall mean, with respect to any entity, all other entities with which the subject entity would be aggregated and treated as a single employer under Code Section 414(b) (controlled group of corporations) and Code Section 414(c) (a group of trades or businesses, whether or not incorporated, under common control), as applicable.

(b) "Beneficiary" or "Beneficiaries" shall mean the person or persons, including a trustee, personal representative or other fiduciary, last designated in writing by a Participant in accordance with the procedures established by the Committee to receive the benefits specified hereunder in the event of the Participant's death. However, no designation of a Beneficiary other than the Participant's spouse shall be valid unless consented in writing by such spouse. No Beneficiary designation shall become effective until it is filed with the Committee. Any designation shall be revocable at any time through a written instrument filed by the Participant with. the Committee with or without the consent of the previous Beneficiary, (unless such previous Beneficiary was the Participant's spouse). If there is no Beneficiary designation in effect, or the designated Beneficiary does not survive the Participant, then the Participant's spouse shall be the Beneficiary. If there is no surviving spouse, the duly appointed and currently acting personal representative of the Participant's estate duly appointed and acting in that capacity within 90 days after the Participant's death (or such extended period as the Committee determines is reasonably necessary to allow such personal representative to be appointed, but not to exceed 180 days after the Participant's death), then Beneficiary shall mean the person or persons who can verify by affidavit or court order to the satisfaction of the Committee that they are legally entitled to receive the benefits specified hereunder. In the event any amount is payable under the Plan to a minor, payment shall not be made to the minor, but instead be paid (a) to that person's living parent(s) to act as custodian, (b) if that person's parents are then divorced, and

one parent is the sole custodial parent, to such custodial parent, or (c) if no parent of that person is then living, to a custodian selected by the Committee to hold the funds for the minor under the Uniform Transfers or Gifts to Minors Act in effect in the jurisdiction in which the minor resides. If no parent is living and the Committee decides not to select another custodian to hold the funds for the minor, then payment shall be made to the duly appointed and currently acting guardian of the estate for the minor or, if no guardian of the estate for the minor is duly appointed and currently acting within 60 days after the date the amount becomes payable, payment shall be deposited with the court having jurisdiction over the estate of the minor. Payment by the Company pursuant to any unrevoked Beneficiary designation, or to the Participant's estate if no such designation exists, of all benefits owed hereunder shall terminate any and all liability of the Company.

(c) "Board of Directors" or "Board" shall mean the Board of Directors of the Company.

(d) "Cause" shall mean (i) a Participant's conviction of a felony, (ii) the engaging by Participant in conduct that constitutes willful gross neglect or willful gross misconduct in carrying out his or her duties to the Company, resulting, in either case, in material economic harm to the Company, unless the Participant believed in good faith that such conduct was in, or not contrary to, the best interests of the Company, (iii) the Participant's material breach of any of the terms of his or her offer letter agreement or the Proprietary Information and Inventions Agreement or (iv) the Participant's failure to follow any lawful directive of Amgen Inc.'s Chief Executive Officer with respect to the Participant's employment. For purposes hereof, no act, or failure to act, by Participant shall be deemed "willful" unless done, or omitted to be done, by Participant not in good faith.

(e) "Change of Control" shall be as defined under the Amgen Inc. Change of Control Severance Plan.

(f) "Code" shall mean the Internal Revenue Code of 1986, as amended.

(g) "Committee" shall mean the Compensation Committee of the Board.

(h) "Company" shall mean Amgen Inc., and any successor corporations. Company shall also include affiliates and subsidiaries of Amgen Inc., and any successor corporations, if the Committee provides that such corporation shall participate in the Plan.

(i) "Company Discretionary Contributions" shall mean, for each Participant, the discretionary amount that the Company allocates to a Participant under this Plan as determined by the Committee. Such amount may differ from Participant to Participant, including no contributions.

(j) "Crediting Date" shall mean the date, as determined by the Committee, on which a Participant's Nonqualified Retirement Account is credited with the Company Discretionary Amount.

(k) "Disability" shall mean a permanent and total disability that has been certified by the Social Security Administration prior a Participant's Separation from Service.

(1) "Disability Prorated Nonqualified Retirement Account Amount" shall mean portion of the Nonqualified Retirement Account Amount based upon the ratio of (x) the sum of the number of full months of the Participant's active employment with the Company plus 24 months and (y) the number of months between the Participant's first day of participation in the plan and the Crediting Date.

(m) "Effective Date" shall mean January 1, 2009, subject to any earlier date specifically set forth within the Plan.

(n) "Eligible Employee" shall mean individuals selected by the Committee, in its sole discretion, from those staff members of the Company.

(o) "Employer" shall mean, for the purpose of determining whether a Participant has experienced a Separation from Service, the entity for which the Participant performs services and with respect to which the legally binding right to compensation deferred or contributed under this Plan arises and all of its Affiliates.

(p) "Nonqualified Retirement Account" shall mean the bookkeeping account maintained by Company for each Participant that is credited with an amount equal to the Company Discretionary Amount, if any, and any interest credited pursuant to Article 4.

(q) "Participant" shall mean any Eligible Employee who is selected by the Committee, in its sole discretion, to participate in the Plan.

(r) "Plan" shall mean the Amgen Inc. Executive Nonqualified Retirement Plan set forth herein, now in effect, or as amended from time to time.

(s) "Plan Year" shall mean the initial period beginning on January 1, 2001 and ending on December 31, 2001 and thereafter the 12 consecutive month period beginning on each January 1 and ending on each December 31.

(t) "Prorated Nonqualified Retirement Account Amount" shall mean a prorated portion of the Nonqualified Retirement Account Amount based upon the ratio of (i) the number of full months of the Participant's active employment with the Company and (ii) the number of months between the Participant's first day of participation in the Plan and the Crediting Date, provided, however, that if such a termination of employment occurs within 2 years after a Change of Control of the Company, as defined in the Amgen Inc. Change of Control Severance Plan, the Participant shall be paid (i) the Prorated Nonqualified Retirement Account Amount <u>plus</u> (ii) an amount equal to the Discretionary Company Contribution minus the sum of (x) the Prorated Nonqualified Retirement Account Amount and (y) an amount equal to the aggregate spread between the exercise prices of the Participant's unvested Company stock options which are in the money and the vesting of which is accelerated by the Change of Control and the NASDAQ closing price of the Company stock, with such spread being determined as of the date of the Change of Control. (See Appendix C for an example).

(u) "Retirement Date" shall mean the date upon which a Participant completes 10 years of active employment with the Company and attains age sixty (60).

(v) "Separation from Service" shall mean the termination of services provided by a Participant to his or her Employer, whether voluntarily or involuntarily, as determined by the Committee in accordance with Treasury Regulation Section 1.409A-1(h). In determining whether a Participant has experienced a Separation from Service, the following provisions shall apply:

(i) Except as otherwise provided in Section 1.2(v)(ii), a Separation from Service shall occur when a Participant experiences a termination of employment with his or her Employer. A Participant shall be considered to have experienced a termination of employment when the facts and circumstances indicate that either (i) the Participant is not reasonably expected to perform further services for the Employer after a certain date, or (ii) that the level of bona fide services the Participant will perform for the Employer after such date (whether as an employee or as an independent contractor) will permanently decrease to no more than 49% of the average level of bona fide services performed by such Participant (whether as an employee or an independent contractor) over the immediately preceding 36-

month period (or full period of services to the Employer if the Participant has been providing services to the Employer for less than 36 months).

(ii) If a Participant is on military leave, sick leave, or other bona fide leave of absence, the employment relationship between the Participant and the Employer shall be treated as continuing intact, provided that the period of such leave does not exceed six months, or longer, so long as the Participant retains a right to reemployment with the Employer under an applicable statute or by contract. If the period of leave exceeds six months and the Participant does not retain a right to reemployment under an applicable statute or by contract, the Participant will incur a Separation from Service as of the first day immediately following the end of such six-month period. However, where a Participant's leave of absence is due to his or her "disability" (as defined below), a 29-month period of absence only if there is a reasonable expectation that the Participant will return to perform services for the Employer. For purposes of this Section 1.2(v)(ii), "disability" shall mean any medically determinable physical or mental impairment resulting in a Participant's inability to perform the duties of his or her position or any substantially similar position, where such impairment can be expected to result in death or can be expected to last for a continuous period of not less than six months. The determination of whether a Participant is disabled shall be made by the Employer's short-term disability insurance carrier or administrator (or, if none, by the Committee).

(iii) Notwithstanding the foregoing, if a Participant provides services to the Employer as both an Eligible Employee and a member of the Board, then to the extent permitted by Treasury Regulation Section 1.409A-1(h)(5), the services provided by such Participant as a Board member shall not be taken into account in determining whether the Participant experiences a Separation from Service.

ARTICLE II.

PARTICIPATION

2.1 An Eligible Employee shall become a Participant in the Plan if the Committee designates such Eligible Employee, in writing, as a Participant. The Committee shall also designate the date on which an Eligible Employee becomes a Participant.

ARTICLE III.

ACCOUNTS AND TRUST FUNDING

3.1 Nonqualified Retirement Account.

(a) The Committee shall establish and maintain a Nonqualified Retirement Account for each Participant under the Plan, which shall be credited the amount of Company Discretionary Contributions, if any, contributed to the Plan on behalf of such Participant.

3.2 Trust Funding.

The Company shall pay all Plan benefits. At its discretion, the Committee may establish one or more trusts, with such trustees as the Board may approve, for the purpose of providing for the payment of such benefits.

Although the principal of such a trust and any earnings thereon shall be held separate and apart from other funds of Company and shall be used exclusively for the uses and purposes of Plan Participants and Beneficiaries as set forth therein, neither the Participant nor their Beneficiaries shall have

any preferred claim on, or any beneficial ownership in, any assets of the trust prior to the time such assets are paid to the Participants or Beneficiaries as benefits and all rights created under this Plan shall be unsecured contractual rights of Plan Participants and Beneficiaries against the Company. Any assets held in the Trust will be subject to the claims of Company's general creditors under federal and state law in the event of insolvency.

ARTICLE IV.

CREDITING OF ACCOUNTS

4.1 <u>Crediting of Company Discretionary Contributions</u>. If the Participant is actively employed by the company on the Crediting Date, the Company shall credit the Nonqualified Retirement Account with the Company Discretionary Contributions.

4.2 <u>Termination of Employment before Crediting Date</u>. In the event that the Participant's active employment with the Company is terminated before the Crediting Date for any reason, no credits will be made to the Nonqualified Retirement Account and the Participant will not be paid any portion of the Nonqualified Retirement Account, except as set forth below:

(a) If the Participant's employment is terminated by reason of the Participant's Disability before the Crediting Date, the Company shall pay the Participant a Disability Prorated Nonqualified Retirement Account Amount in accordance with the provisions of Article V. No interest shall be credited on any such payment.

(b) If the Participant's employment is terminated by the Company without Cause before the Crediting Date, the Company shall pay the Participant a Prorated Nonqualified Retirement Account Amount in accordance with the provisions of Article V. No interest shall be credited on any such payment.

4.3 <u>Interest</u>. No interest shall be credited to the Nonqualified Retirement Account prior to the Crediting Date, in any event. However, if the Participant is actively employed by the Company on the Crediting Date, from and after the Crediting Date the Company shall credit the Nonqualified Retirement Account with interest as set forth below.

(a) <u>Interest after Retirement Date</u>. If the Participant continues to be actively employed by the Company until his or her Retirement Date, the Company shall credit interest annually on the Nonqualified Retirement Account at a rate equal to 125% of the 10-year moving average yield on 10-year U.S. Treasury notes, adjusted annually and compounded annually, from the Crediting Date until the date upon which the Nonqualified Retirement Account and accrued interest is distributed. In the event that the Participant elects to receive his or her distribution in installments, as provided below in Section 5.1(b), interest will be credited on the declining balance of the Nonqualified Retirement Account until it is finally distributed.

(b) Interest before Retirement Date. If the Participant's employment with the Company is terminated for any reason before his or her Retirement Date, the Company shall credit interest annually on the Nonqualified Retirement Account at a rate equal to 100% of the 10-year moving average yield on 10-year U.S. Treasury notes, adjusted annually and compounded annually, from the Crediting Date until the date upon which the Nonqualified Retirement Account, and accrued interest is distributed to the Participant.

ARTICLE V.

DISTRIBUTIONS

5.1 <u>Distribution of Accounts</u>. The Company shall make distributions from the Nonqualified Retirement Account as set forth below.

(a) <u>Distribution upon Separation from Service before Retirement Date</u>. If the Participant experiences a Separation from Service for any reason before the Participant's Retirement Date, the amount credited to the Participant's Nonqualified Retirement Account, plus interest credited to the date of the Participant's Separation from Service, shall be distributed to the Participant in a lump-sum payment as soon as administratively practicable during the Plan Year immediately following the Plan Year in which such Separation from Service occurs.

(b) <u>Distribution upon Separation from Service after Retirement Date</u>. If the Participant experiences a Separation from Service for any reason after the Participant's Retirement Date, the amount credited to the Participant's Nonqualified Retirement Account, plus interest credited to the date of the Participant's Separation from Service, shall be distributed to the Participant in a lump-sump payment as soon as administratively practicable during the Plan Year immediately following the Plan Year in which such Separation from Service occurs, unless the Participant elects on an Election Form, within the timeframes set forth in Section 5.1(d), to receive substantially equal annual installment payments. Installment payments will commence as soon as administratively practicable in the Plan Year immediately following the Plan Year in which the Participant experiences a Separation from Service, and will end in the Plan Year specified in the Election Form, which shall not be later than the Plan Year that includes the ten-year anniversary of the Participant's Separation from Service. For purposes of this Plan, the right to receive a benefit payment in annual installments shall be treated as the entitlement to a single payment.

(c) <u>Distribution upon Death</u>. In the event of the Participant's death, any unpaid amounts with respect to the Nonqualified Retirement Account shall be paid to the Participant's Beneficiary or Beneficiaries. The Participant shall elect on an Election Form, within the timeframes set forth in Section 5.1(d), whether, in the event the Participant dies before receiving any payments, his or her Nonqualified Retirement Account shall be paid to his or her Beneficiary as a lump sum or in up to ten substantially equal annual installment payments; if the Participant has failed to make such an election, then such amount shall be paid to the Beneficiary in a lump sum. Any lump-sum payment or initial installment payment that is made in accordance with this paragraph shall be paid within 60 days of the Participant's death. If a Participant dies after installment payments have commenced but before his or her Nonqualified Retirement Account is paid in full, then the Participant's remaining installment payments shall continue and shall be paid to the Participant's Beneficiary over the remaining number of years and in the same amounts as payments would have been made to the Participant if the Participant had survived.

(d) <u>Time for Making Elections</u>. Elections permitted pursuant to this Section 5.1 must be made within 30 days after the date that the Participant becomes eligible to participate in the Plan, provided that the Participant has not been eligible to participate in this Plan or in any other plan that would be aggregated with this Plan under Treasury Regulation Section 1.409A-1(c) at any time during the 24-month period ending on the date that he or she becomes eligible to participate in the Plan, in accordance with Treasury Regulation Section 1.409A-2(a)(7)(ii).

5.2 <u>Distribution Election Changes.</u> Subject to Section 5.4, with respect to each distribution election made pursuant to this Article V, a Participant may have a one-time opportunity to extend the payment date and/or change the form of payment initially designated; provided, however, (i) the new distribution election shall have no effect until at least 12 months after the date on which such election is

made, (ii) the payment date must be at least five years after the previously designated payment date (for Separation from Service distribution elections only) and must involve completion of all payments not later than the end of the Plan Year that includes the ten-year anniversary of the Participant's Separation from Service or death, and (iii) the election must be made at least 12 months prior to the previously designated payment date. The "previously designated payment date" in the preceding sentence shall be January 1 of the Plan Year in which the payment was initially scheduled to occur, which shall include only the first payment in the case of installment payments.

5.3 <u>Six-Month Delayed Payment.</u> If, at the time of the Participant's Separation from Service, the Participant is a "specified employee" (within the meaning of Section 409A of the Code and Treasury Regulation Section 1.409A-1(i)), the Company will not pay or provide any "Specified Benefits" (as defined herein) during the six-month period beginning with the date of the Participant's Separation from Service (the "409A Suspension Period"). In the event of a Participant's death, however, the Specified Benefits shall be paid to the Participant's Beneficiary without regard to the 409A Suspension Period. For purposes of this Plan, "Specified Benefits" are any amounts of the Participant's Nonqualified Retirement Account that would be subject to Section 409A additional taxes if the Company were to pay them, pursuant to this Plan, on account of the Participant's Separation from Service. During the 409A Suspension Period, the Nonqualified Retirement Account will continue to be credited or debited in accordance with Article IV above until the Nonqualified Retirement Account is distributed. Within 14 calendar days after the end of the 409A Suspension Period, the Participant shall be paid a lump-sum payment in cash equal to any Specified Benefits delayed during the 409A Suspension Period.

5.4 <u>Special Transition Rule for 2008.</u> This Section is effective October 1, 2008. With respect to Article V only, the Committee shall, in accordance with the rules established by the Committee and pursuant to Notice 2007-86, provide a limited period in 2008 in which Participants may make distribution elections with respect to all or part of their Nonqualified Retirement Accounts. On or before the deadline established by the Committee, which in no event shall be later than December 31, 2008, Participants shall make any such elections on an Election Form that the Committee provides. Any distribution election made by a Participant, and accepted by the Committee, shall not be treated as a change in either the form or timing of a Participant's benefit payment for purposes of Code Section 409A or Section 5.2 of the Plan. If any distribution election submitted by a Participant in accordance with this Section 5.4 either (i) relates to an amount that would otherwise be paid to the Participant in 2008, or (ii) would cause an amount to be paid to the Participant in 2008, such election shall not be effective with respect to such amount.

5.5 <u>Accelerated Distributions</u>. Distributions may not be accelerated, except as provided in this Section 5.5 and Section 7.5. Distributions may be accelerated under the following circumstances:

(a) A Participant who has elected to receive installment payments subsequently elects to change from installments to a lump-sum distribution, provided the change in the distribution election satisfies the requirements set forth in Section 5.2 above.

(b) The Participant becomes liable for FICA taxes with respect to any portion of the Participant's Nonqualified Retirement Account, provided that if an accelerated distribution is made pursuant to this paragraph, the amount distributed shall not exceed the aggregate of the FICA taxes imposed on the Participant's Nonqualified Retirement Account plus any income tax withholding required for the FICA withholdings.

(c) The Plan fails to meet the requirements of Code Section 409A with respect to the Participant, provided that if an accelerated distribution is made pursuant to this paragraph, the amount that shall be distributed shall not exceed the amount required to be included in income as a result of the failure to comply with Code Section 409A.

5.6 <u>Delayed Distributions</u>. Except as provided in Sections 5.2, 5.3, 5.4, and this Section 5.6, payments may not be delayed. Distributions may be delayed under the following circumstances:

(a) If the Company reasonably anticipates that the Company's deduction with respect to any distribution from this Plan would be limited or eliminated by application of Code Section 162(m), then to the extent permitted by Treasury Regulation Section 1.409A-2(b)(7)(i), payment shall be delayed as deemed necessary to ensure that the entire amount of any distribution from this Plan is deductible. Any amounts for which distribution is delayed pursuant to this Section shall continue to be credited or debited with additional amounts in accordance with Article IV. The delayed amounts (as adjusted for any amounts credited or debited thereon) shall be distributed to the Participant (or his Beneficiary in the event of the Participant's death) at the earliest date the Company reasonably anticipates that the deduction of the payment of the amount will not be limited or eliminated by application of Code Section 162(m).

(b) The Committee may delay payment if it reasonably anticipates that making the payment would violate federal securities laws or other applicable law, provided the Company treats all payments to similarly situated Participants on a reasonably consistent basis and the payment is made at the earliest date at which the Committee reasonably anticipates that the making of the payment will not cause a violation.

ARTICLE VI.

ADMINISTRATION

6.1 Powers and Duties of the Committee.

(a) The Committee, on behalf of the Participants and their Beneficiaries, shall enforce the Plan in accordance with its terms, shall be charged with the general administration of the Plan, and shall have all powers necessary to accomplish its purposes as set forth herein, including, but not by way of limitation, the following:

(1) To construe and interpret the terms and provisions of the Plan and to remedy any inconsistencies or ambiguities hereunder;

(2) To select employees eligible to participate in the Plan;

(3) To compute and certify to the amount and kind of benefits payable to Participants and their Beneficiaries;

(4) To maintain all records that may be necessary for the administration of the Plan;

(5) To provide for the disclosure of all information and the filing or provision of all reports and statements to Participants, Beneficiaries or governmental agencies as shall be required by law;

(6) To make and publish such rules for the regulation of the Plan and procedures for the administration of the Plan as are not inconsistent with the terms hereof;

(7) To appoint a plan administrator or any other agent, and to delegate to them such powers and duties in connection with the administration of the Plan as the Committee may from time to time prescribe; and

(8) To take all actions necessary for the administration of the Plan.

6.2 Construction and Interpretation.

The Committee shall have full discretion to construe and interpret the terms and provisions of this Plan, which interpretations or construction shall be final and binding on all parties, including but not limited to the Company and any Participant or Beneficiary. The Committee shall administer such terms and provisions in a uniform and nondiscriminatory manner and in full accordance with any and all laws applicable to the Plan.

6.3 Information.

To enable the Committee to perform its functions, the Company shall supply full and timely information to the Committee on all matters relating to the Compensation of all Participants, their death or other events that cause termination of their participation in this Plan, and such other pertinent facts as the Committee may require.

6.4 Compensation, Expenses and Indemnity.

(a) The members of the Committee shall serve without compensation for their services hereunder.

(b) The Committee is authorized at the expense of the Company to employ such legal counsel and other advisors as it may deem advisable to assist in the performance of its duties hereunder. Expenses and fees in connection with the administration of the Plan shall be paid by the Company.

(c) To the extent permitted by applicable state law, the Company shall indemnify and save harmless the Committee and each member thereof, the Board of Directors and any delegate of the Committee who is an employee of the Company against any and all expenses, liabilities and claims, including legal fees to defend against such liabilities and claims arising out of their discharge in good faith of responsibilities under or incident to the Plan, other than expenses and liabilities arising out of willful misconduct. This indemnity shall not preclude such further indemnities as may be available under insurance purchased by the Company or provided by the Company under any bylaw, agreement or otherwise, as such indemnities are permitted under state law.

6.5 Annual Statements.

Under procedures established by the Committee, a Participant shall receive a statement with respect to such Participant's Accounts on an annual basis as of each December 31.

6.6 Disputes.

(a) <u>Claim</u>.

A person who believes that he or she is being denied a benefit to which he or she is entitled under this Agreement (hereinafter referred to as "Claimant") may file a written request for such benefit with the Company, setting forth his or her claim. The request must be addressed to the Vice President, Human Resources, or his designee, of the Company at its then principal place of business.

(b) Claim Decision.

Upon receipt of a claim, the Company shall advise the Claimant that a reply will be forthcoming within ninety (90) days and shall, in fact, deliver such reply within such period. The

Company may, however, extend the reply period for an additional ninety (90) days for special circumstances.

If the claim is denied in whole or in part, the Company shall inform the Claimant in writing, using language calculated to be understood by the Claimant, setting forth: (i) the specified reason or reasons for such denial; (ii) the specific reference to pertinent provisions of this Agreement on which such denial is based; (iii) a description of any additional material or information necessary for the Claimant to perfect his or her claim and an explanation of why such material or such information is necessary; (iv) appropriate information as to the steps to be taken if the Claimant wishes to submit the claim for review; and (v) the time limits for requesting a review under subsection (c).

(c) Request For Review.

With sixty (60) days after the receipt by the Claimant of the written opinion described above, the Claimant may request in writing that the Committee review the determination of the Company. Such request must be addressed to the Secretary of the Company, as its then principal place of business. The Claimant or his or her duly authorized representative may, but need not, review the pertinent documents and submit issues and comments in writing for consideration by the Committee. If the Claimant does not request a review within such sixty (60) day period, he or she shall be barred and estoppel from challenging the Company's determination.

(d) Review of Decision.

Within sixty (60) days after the Committee's receipt of a request for review, after considering all materials presented by the Claimant, the Committee will inform the Participant in writing, in a manner calculated to be understood by the Claimant, the decision setting forth the specific reasons for the decision contained specific references to the pertinent provisions of this Agreement on which the decision is based. If special circumstances require that the sixty (60) day time period be extended, the Committee will so notify the Claimant and will render the decision as soon as possible, but no later than one hundred twenty (120) days after receipt of the request for review.

ARTICLE VII.

MISCELLANEOUS

7.1 Unsecured General Creditor.

Participants and their Beneficiaries, heirs, successors, and assigns shall have no legal or equitable rights, claims, or interest in any specific property or assets of the Company. No assets of the Company shall be held in any way as collateral security for the fulfilling of the obligations of the Company under this Plan. Any and all of the Company's assets shall be, and remain, the general unpledged, unrestricted assets of the Company. The Company's obligation under the Plan shall be merely that of an unfunded and unsecured promise of the Company to pay money in the future, and the rights of the Participants and Beneficiaries shall be no greater than those of unsecured general creditors. It is the intention of the Company that this Plan be unfunded for purposes of the Code and for purposes of Title I of ERISA.

7.2 Restriction Against Assignment.

The Company shall pay all amounts payable hereunder only to the person or persons designated by the Plan and not to any other person or corporation.

(a) No right, title or interest in the Plan or in any account may be sold, pledged, assigned or transferred in any manner other than by will or the laws of descent and distribution. No right, title or interest in the Plan or in any Account shall be liable for the debts, contracts or engagements of the Participant or his successors in interest or shall be subject to disposition by transfer, alienation, anticipation, pledge, encumbrance, assignment or any other means whether such disposition be voluntary or involuntary or by operation of law by judgment, levy, attachment, garnishment or any other legal or equitable proceedings (including bankruptcy), and any attempted disposition thereof shall be null and void and of no effect, except to the extent that such disposition is permitted by the preceding sentence.

(b) Notwithstanding the provisions of a subsection, a Participant's interest in his Account may be transferred by the Participant pursuant to a domestic relations order that constitutes a "qualified domestic relations order" as defined by the Code or Title I of ERISA.

7.3 Taxes and Withholding.

(a) <u>Participant Responsibility for Taxes</u>. Except to the extent specifically provided within this Plan or any separate written agreement between a Participant and the Employer, a Participant shall be solely responsible for the satisfaction of any taxes with respect to the benefits payable to the Participant under this Plan (including, but not limited to, employment taxes imposed on employees and additional taxes on nonqualified deferred compensation). Although the Company intends and expects that the Plan and its payments and benefits will not give rise to taxes imposed under Section 409A of the Code, neither the Company, nor its employees, directors, or agents shall have any obligation to mitigate or to hold any Participant harmless from any or all of such taxes.

(b) <u>Withholding</u>. The Participant's Employer(s), or the trustee of the trust, shall withhold from any payments made to a Participant under this Plan all federal, state and local income, employment and other taxes required to be withheld by the Employer(s), or the trustee of the trust, in connection with such payments, in amounts and in a manner to be determined in the sole discretion of the Employer(s) and the trustee of the trust, respectively (whichever is making the payment). The Participant's Employer, or the trustee of the trust, shall withhold from any payments made to a Participant under this Plan any garnishment of wages in amounts and in a manner to be determined by the sole discretion of the Employer(s) and the trustee of the trust, respectively (whichever is making the payment).

7.4 Plan Amendment, Modification, or Suspension,

(a) <u>Plan Amendment, Modification or Suspension</u>. The Committee may amend, modify or suspend the Plan in whole or in part, except that no amendment, modification or suspension shall have any retroactive effect to reduce any amounts allocated to a Participant's Accounts.

(b) <u>Amendment for 409A Compliance</u>. This Plan is intended to comply with Section 409A of the Code, and the Company shall have complete discretion to interpret and construe this Plan and any associated documents in any manner that establishes an exemption from or otherwise conforms them to the requirements of Section 409A. If, for any reason including imprecision in drafting, any Plan provision does not accurately reflect its intended establishment of an exemption from or compliance with Section 409A of the Code, as demonstrated by consistent interpretations or other evidence of intent, the provision shall be considered ambiguous and shall be interpreted by the Company in a fashion consistent herewith, as determined in the sole and absolute discretion of the Company. The Company reserves the right to unilaterally amend this Plan without the consent of any Participant in order to accurately reflect its correct interpretation and operation, as well as to maintain an exemption from or compliance with Section 409A of the Code.

7.5 <u>Termination</u>. Although the Company anticipates that it will continue the Plan for an indefinite period of time, there is no guarantee that the Company will continue the Plan or will not terminate the Plan at any time in the future. Accordingly, by action of its Board of Directors or the Committee, the Company reserves the right to discontinue its sponsorship of the Plan and to terminate the Plan at any time in accordance with one of the following circumstances set forth in subsections (a) through (c) below and in Treasury Regulation Section 1.409A-3(j)(4)(ix):

(a) The Company may terminate the Plan if the termination and liquidation is not proximate to a downturn in the Company's financial health and:

(i) The Plan and all other plans maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) are irrevocably terminated;

(ii) No payments other than payments that would otherwise be payable under the terms of the Plan are made within 12 months following the date the Company takes all necessary actions to terminate and liquidate the Plan;

(iii) Except with respect to the Participants who became entitled to benefits under the terms of the Plan and any other plan maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) within the first 12 months following the date such plans are irrevocably terminated, all payments to the Participants due under the terms of such plans must be made between the first day of the 13th month and the last day of the 24th month following the date such plans terminated; and

(iv) The Company does not adopt a plan that would be aggregated with this Plan under Treasury Regulation Section 1.409A-1(c) within three years following the date the Plan is terminated.

(b) The Company terminates and liquidates the Plan pursuant to irrevocable action taken within 30 days preceding or 12 months following a "change in control event" (defined below), provided that the Plan and all other plans maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) are terminated on the same date with respect to each participant in such plans that experienced the "change in control event," and all such participants receive all benefits payable under such plans within 12 months following the termination date. For purposes of this Section 7.5(b), "change in control event" shall have the meaning set forth in Treasury Regulation Section 1.409A-3(i)(5).

(c) The Company terminates and liquidates the Plan within 12 months of a corporate dissolution taxed under Code Section 331, or with the approval of a bankruptcy court pursuant to 11 U.S.C. § 503(b)(1)(A), provided that all benefits payable under the Plan are distributed to Participants during the earlier of (i) the taxable year in which the amount is actually or constructively received, or (ii) the latest of the calendar year in which (a) the Plan is terminated and liquidated; (b) the benefits are no longer subject to a substantial risk of forfeiture; or (c) the payment first becomes administratively practicable.

7.6 Governing Law.

This Plan shall be construed, governed and administered in accordance with the laws of the State of California.

7.7 Payments on Behalf of Persons Under Incapacity.

In the event that any amount becomes payable under the Plan to a person who, in the sole judgment of the Committee, is considered by reason of physical or mental condition to be unable to give a valid receipt therefore, the Committee may direct that such payment be made to any person found by the Committee, in its sole judgment, to have assumed the care of such person. Any payment made pursuant to such termination shall constitute a full release and discharge of the Committee and the Company.

7.8 Limitation of Rights and Employment Relationship.

Neither the establishment of the Plan nor any modification thereof, nor the creating of any fund or account, nor the payment of any benefits shall be construed as giving to any Participant or other person any legal or equitable right against the Company except as provided in the Plan, and in no event shall the terms of employment of any Employee or Participant be modified or in any be effected by the provisions of the Plan.

7.9 Exempt ERISA Plan.

The Plan is intended to be an unfunded plan maintained primarily to provide nonqualified retirement benefits for a select group of management or highly compensated employees within the meaning of Sections 201, 301 and 401 of ERISA and therefore to be exempt from Parts 2, 3 and 4 of Title I of ERISA.

7.10 Notice.

Any notice or filing required or permitted to be given to the Committee under the Plan shall be sufficient if in writing and hand delivered, or sent by registered or certified mail, to the principal office of the Company, directed to the attention of the General Counsel and Secretary of the Company. Such notice shall be deemed given as of the date of delivery or, if delivery is made by mail, as of the date shown on the postmark on the receipt for registration or certification.

7.11 Errors and Misstatements.

In the event of any misstatement or omission of fact by a Participant to the Committee or any clerical error resulting in payment of benefits in an incorrect amount, the Committee shall promptly cause the amount of future payments to be corrected upon discovery of the facts and shall pay or, if applicable, cause the Trustee to pay, the Participant or any other person entitled to payment under the Plan any underpayment in a lump sum or to recoup any overpayment from future payments to the participant or any other person entitled to payment under the Plan in such amounts as the Committee shall direct or to proceed against the Participant or any other person entitled to payment under the Plan in such amounts as the Committee shall direct or to proceed against the Participant or any other person entitled to payment under the Plan for recovery of any such overpayment.

7.12 Pronouns and Plurality.

The masculine pronoun shall include the feminine pronoun, and the singular the plural where the context so indicates.

7.13 Severability.

In the event that any provision of the Plan shall be declared unenforceable or invalid for any reason, such unenforceability or invalidity shall not affect the remaining provisions of the Plan but

shall be fully severable, and the Plan shall be construed and enforced as if such unenforceable or invalid provision had never been included herein.

7.14 <u>Status</u>.

The establishment and maintenance of, or allocations and credits to, the Account of any Participant shall not vest in any Participant any right, title or interest in and to any Plan assets or benefits except at the time or times and upon the terms and conditions and to the extent expressly set forth in the Plan and in accordance with the terms of the trust, if applicable.

7.15 <u>Headings</u>.

Headings and subheadings in this Plan are inserted for convenience of reference only and are not to be considered in the construction of the provisions hereof.

By: /s/ BRIAN MCNAMEE

Its: Senior Vice President, Human Resources

APPENDIX A

Name of Participant	Date of Participation	Crediting Date	Discretionary Company Contribution
Roger Perlmutter	January 16, 2001	September 16, 2007	\$10,000,000
George Morrow	January 19, 2001	January 19, 2006	\$15,000,000

APPENDIX B

Other Participating Companies

Amgen USA Inc., effective January 1, 2002

APPENDIX C

For example:

			Discretionary Company
Name of Participant	Date of Participation	Crediting Date	Contribution
Participant A	January 1, 2001	January 1, 2007	\$1,000,000

If the Company were to terminate the Participant's employment without Cause on July 31, 2003, the Participant would be paid \$430,555 (\$1,000,000 x 31/72). If, however, such a termination were to occur within 2 years after a Change of Control of the Company and on the date of the Change of Control the Participant hold unvested options for 10,000 shares of the Company stock and if each of such options has an exercise price of \$80 and the NASDAQ closing price of the Company stock on the date of the Change of Control were \$120, the Participant would be paid \$600,000: (\$430,555 + (\$1,000,000—(\$430,555 + ((\$120—\$80) x 10,000)))).



As Amended and Restated Effective January 1, 2009

4MGEN

Amgen Nonqualified Deferred Compensation Plan

Plan Document

Purpose		1
ARTICLE 1	Definitions	1
ARTICLE 2 2.1 2.2 2.3 2.4	Selection/Enrollment/Eligibility Selection by Committee Enrollment Requirements Eligibility/Commencement of Participation Termination of Participation and/or Deferrals	5 5 5 5 5
ARTICLE 3 3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9	Deferral Commitments/Company Matching/Crediting/Taxes Maximum Deferrals Election to Defer/Effect of Election Form 401(k) Plan/1165(e) Plan Make Whole Elections Withholding of Annual Deferral Amounts Annual Company Contribution Amount Vesting Crediting/Debiting of Account Balances FICA and Other Taxes Distributions	5 6 6 7 7 8 8
ARTICLE 4 4.1 4.2	Short-Term Payout Short-Term Payout Other Benefits Take Precedence Over Short-Term Payout	9 9 9
ARTICLE 5 5.1 5.2. 5.3.	Distribution of Benefits Following Separation from Service Distributions Installment Payments Distribution Election Changes	9 9 9 9
ARTICLE 6 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9	Survivor Benefits Survivor Benefits Death Before Commencement of Benefits Death After Commencement of Benefits Distribution Election Changes Beneficiary Beneficiary Designation Change/Spousal Consent Acknowledgment No Beneficiary Designation Discharge of Obligations	11 11 11 11 11 11 11 11 12 12 12
ARTICLE 7 7.1	Disability Waiver Disability Waiver	12 12
ARTICLE 8 8.1 8.2 8.3 8.4 8.5 8.6 8.7	Distributions – General Generally Six-Month Delayed Payment Special Transition Rule for 2008 Accelerated Distributions Delayed Distributions Withdrawal/Cancellation of Deferrals for Unforeseeable Financial Emergencies Withholding of Employment Taxes Upon Distribution	12 12 13 13 14 14 14
ARTICLE 9	Leave of Absence	15

i

Amgen Nonqualified Deferred Compensation Plan

Plan Document

9.1 9.2	Paid Leave of Absence Unpaid Leave of Absence	15 15
ARTICLE 10 10.1 10.2 10.3 10.4	Termination/Amendment or Modification Termination Amendment Plan Agreement Effect of Payment	15 15 16 16 16
ARTICLE 11 11.1 11.2 11.3 11.4 11.5 11.6	Administration Committee Duties Administration Upon Change of Control Agents Binding Effect of Decisions Indemnity of Committee Employer Information	17 17 17 17 17 17 17 18
ARTICLE 12 12.1	Other Benefits and Agreements Coordination with Other Benefits	18 18
ARTICLE 13 13.1 13.2 13.3 13.4 13.5	Claims Procedures Presentation of Claim Notification of Decision Review of a Denied Claim Decision on Review Legal Action	18 18 18 18 18 19 19
ARTICLE 14 14.1 14.2 14.3 14.4	Trust Establishment of the Trust Interrelationship of the Plan and the Trust Distributions From the Trust Investment of Trust Assets	19 19 19 19 19 19
ARTICLE 15 15.1 15.2 15.3 15.4 15.5 15.6 15.7 15.8 15.9 15.10 15.11 15.12 15.13	Miscellaneous Status of Plan Unsecured General Creditor Employer's Liability Nonassignability Not a Contract of Employment Furnishing Information Terms Captions Governing Law Notice Successors Spouse's Interest Validity	19 19 20 20 20 20 20 20 20 20 20 20 20 21 21 21
15.14 15.15 15.16	Incompetent Insurance Legal Fees To Enforce Rights After Change of Control	21 21 21

ii



AMGEN NONQUALIFIED DEFERRED COMPENSATION PLAN

As Amended and Restated Effective January 1, 2009

Purpose

The purpose of this Plan is to provide specified benefits to a select group of management or highly compensated Employees who contribute materially to the continued growth, development and future business success of Amgen Inc., a Delaware corporation, and its subsidiaries, if any, that sponsor this Plan. This Plan shall be unfunded for tax purposes and for purposes of Title I of ERISA.

The Plan is hereby amended and restated, effective as of January 1, 2009 except as otherwise provided herein, in order to comply with Code Section 409A and related Treasury Regulations, and to incorporate prior amendments. However, if a Participant's payments commenced prior to January 1, 2009, or if the Committee determines that all of the events necessary to receive payment have occurred prior to January 1, 2009, the Participant shall receive or continue to receive payments in accordance with the Plan terms in effect on December 31, 2008, to the extent that the Company determines that doing so would comply with a reasonable, good-faith interpretation of Code Section 409A and applicable guidance relating to Code Section 409A. Where payments have not commenced on or before December 31, 2008 because a Participant was treated as not having experienced a "separation from service" under a reasonable, good-faith interpretation of Code Section 409A and applicable guidance, but the Participant would be treated as having experienced a "separation from service" under Treasury Regulation Section 1.409A-1(h) on a date that is on or after April 10, 2007 and on or before December 31, 2008, the Participant will be treated as having experienced a separation from service on December 31, 2008.

ARTICLE 1 Definitions

For purposes of this Plan, unless otherwise clearly apparent from the context, the following phrases or terms shall have the following indicated meanings:

- 1.1 "Account Balance" shall mean, with respect to a Participant, a credit on the records of the Employer equal to the sum of (i) the Deferral Account balance and (ii) the vested Company Contribution Account balance. The Account Balance, and each other specified account balance, shall be a bookkeeping entry only and shall be utilized solely as a device for the measurement and determination of the amounts to be paid to a Participant, or his or her designated Beneficiary, pursuant to this Plan.
- 1.2 "Account Balance Plan" shall mean any plan, agreement or arrangement of the Company or any of its Affiliates that is an "account balance plan" as defined in Treasury Regulation Section 1.409A-1(c)(2)(A) and (B).
- 1.3 "Affiliate" shall mean, with respect to any entity, all other entities with which the subject entity would be aggregated and treated as a single employer under Code Section 414(b) (controlled group of corporations) and Code Section 414(c) (a group of trades or businesses, whether or not incorporated, under common control), as applicable.
- 1.4 "Annual Base Salary" shall mean a Participant's compensation consisting only of regular salary paid by any Employer for services rendered during the Plan Year and excluding any other compensation. With respect to any member of the Board, Annual Base Salary shall mean the member's annual retainer, chair fees, Board meeting fees, and Committee meeting fees.
- 1.5 "Annual Bonus" shall mean any compensation earned by a Participant during a Plan Year that constitutes a commission paid to a salesperson or that is paid pursuant to the Amgen Global Management Incentive Plan

Plan Document

(GMIP), the Amgen Inc. Executive Incentive Plan (EIP), or an equivalent bonus program. All other compensation is excluded.

- 1.6 "Annual Company Contribution Amount" shall mean, for any one Plan Year, the amount determined in accordance with Section 3.5.
- 1.7 "Annual Deferral Amount" shall mean that portion of a Participant's Annual Base Salary or Annual Bonus, as applicable, that a Participant elects to have, and is deferred, in accordance with Article 3, for any one Plan Year.
- 1.8 "Annual Installment Method" shall mean the method used to make payments to a Participant who has elected to receive a benefit over a period of years. Under the Annual Installment Method, the amount of each annual payment due to a Participant shall be calculated by multiplying the Participant's Account Balance as of the most recent Valuation Date by a fraction, the numerator of which is one and the denominator of which is the remaining number of annual payments due the Participant. By way of example, if the Participant elects a ten-year Annual Installment Method, the first payment shall be ¹/10 of the Account Balance as of the most recent Valuation Date. The following year, the payment shall be ¹/9 of the Account Balance as of the most recent Valuation Date. For purposes of this Plan, the right to receive a benefit payment in annual installments shall be treated as the entitlement to a single payment.
- 1.9 "Beneficiary" shall mean one or more persons, trusts, estates or other entities, designated in accordance with Article 8, or entitled under Article 8 in the absence of a designation, that are entitled to receive benefits under this Plan upon the death of a Participant.
- 1.10 "Beneficiary Designation Form" shall mean the form established from time to time by the Committee that a Participant completes, signs and returns to the Committee to designate one or more Beneficiaries.
- 1.11 "Board" shall mean the board of directors of the Company.
- 1.12 "Change of Control" shall have the meaning set forth in the Amgen Inc. Change of Control Severance Plan, as it may be amended from time to time.
- 1.13 "Claimant" shall have the meaning set forth in Section 13.1.
- 1.14 "Code" shall mean the Internal Revenue Code of 1986, as it may be amended from time to time.
- 1.15 "Committee" shall mean the committee described in Article 11.
- 1.16 "Company" shall mean Amgen Inc., and any successor to all or substantially all of the Company's assets or business and it shall exclude any disregarded entity pursuant to Treasury Regulations section 301.7701-3, unless such disregarded entity is selected by the Board to participate in the Plan.
- 1.17 "Company Contribution Account" shall mean (i) the sum of the Participant's Annual Company Contribution Amounts, plus (ii) amounts credited (net of amounts debited) in accordance with all the applicable crediting provisions of this Plan that relate to the Participant's Company Contribution Account, less (iii) all distributions made to the Participant or his or her Beneficiary pursuant to this Plan that relate to the Participant's Company Contribution Account.
- 1.18 "Deferral Account" shall mean (i) the sum of all of a Participant's Annual Deferral Amounts, plus (ii) amounts credited (net of amounts debited) in accordance with all the applicable provisions of this Plan that relate to the Participant's Deferral Account, less (iii) all distributions made to the Participant or his or her Beneficiary pursuant to this Plan that relate to his or her Deferral Account.



Amgen Nonqualified Deferred Compensation Plan

Plan Document

- 1.19 "Disability" shall mean any medically determinable physical or mental impairment resulting in a Participant's inability to perform the duties of his or her position or any substantially similar position, where such impairment can be expected to result in death or can be expected to last for a continuous period of not less than six months. The determination of whether a Participant has a Disability shall be made by the Employer's short-term disability insurance carrier or administrator (or, if none, by the Committee).
- 1.20 "Election Form" shall mean the form established from time to time by the Committee that a Participant completes, signs and returns to the Committee to make an election under the Plan.
- 1.21 "Employee" shall mean a person whom an Employer classifies as an employee.
- 1.22 "Employer" shall be defined as follows:
 - (a) Except as otherwise provided in part (b) of this Section, the term "Employer" shall mean the Company and/or any of its subsidiaries or affiliates (now in existence or hereafter formed or acquired) that have been selected by the Board to participate in the Plan, through designation in Appendix A of the Plan, and have adopted the Plan by permitting their Employees to participate in the Plan.
 - (b) For the purpose of determining whether a Participant has experienced a Separation from Service, the term "Employer" shall mean the entity for which the Participant performs services and with respect to which the legally binding right to compensation deferred or contributed under this Plan arises, and all of its Affiliates.
- 1.23 "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as it may be amended from time to time.
- 1.24 "401(k) Plan" shall be that certain Amgen Retirement and Savings Plan adopted by the Company, as it may be amended from time to time.
- 1.25 "Participant" shall mean any Employee (i) who is selected by the Committee from among the highly compensated or management employees of the Employer to participate in the Plan, (ii) who elects to participate in the Plan, (iii) who signs a Plan Agreement, an Election Form and a Beneficiary Designation Form, (iv) whose signed Plan Agreement, Election Form and Beneficiary Designation Form are accepted by the Committee, (v) who commences participate in the Plan, and (vi) whose Plan Agreement has not terminated. In addition, a Participant shall mean any member of the Board (i) who elects to participate in the Plan, (ii) who signs a Plan Agreement, an Election Form and a Beneficiary Form, (iii) whose signed Plan Agreement, Election Form and a Beneficiary Form, (iii) whose signed Plan Agreement, Election Form and Beneficiary Designation in the Plan, and (v) whose Plan Agreement, an Election Form and a Beneficiary Form, (iii) whose signed Plan Agreement, Election Form and Beneficiary Designation Form are accepted by the Committee, (iv) who commences participation in the Plan, and (v) whose Plan Agreement has not terminated. A spouse or former spouse of a Participant shall not be treated as a Participant in the Plan or have an account balance under the Plan, even if he or she has an interest in the Participant's benefits under the Plan as a result of applicable law or property settlements resulting from legal separation or divorce.
- 1.26 "Plan" shall mean the AMGEN NONQUALIFIED DEFERRED COMPENSATION PLAN, as amended and restated effective January 1, 2009, which shall be evidenced by this instrument and by each Plan Agreement, as they may be amended from time to time.
- 1.27 "Plan Agreement" shall mean a written agreement, as may be amended from time to time, which is entered into by and between an Employer and a Participant. Each Plan Agreement executed by a Participant and the Participant's Employer shall provide for the entire benefit to which such Participant is entitled under the Plan; should there be more than one Plan Agreement, the Plan Agreement bearing the latest date of acceptance by the Employer shall supersede all previous Plan Agreements in their entirety and shall govern such entitlement. The terms of any Plan





Plan Document

Agreement may be different for any Participant, and any Plan Agreement may provide additional benefits not set forth in the Plan or limit the benefits otherwise provided under the Plan; provided, however, that any such additional benefits or benefit limitations must be agreed to by both the Employer and the Participant.

- 1.28 "Plan Year" shall mean a period beginning on January 1 of each calendar year and continuing through December 31 of such calendar year.
- 1.29 "Separation from Service" shall mean the termination of services provided by a Participant to his or her Employer, whether voluntarily or involuntarily, as determined by the Committee in accordance with Treasury Regulation Section 1.409A-1(h). In determining whether a Participant has experienced a Separation from Service, the following provisions shall apply:
 - (a) For a Participant who provides services to the Employer as an Employee, except as otherwise provided in Section 1.29(b), a Separation from Service shall occur when such Participant experiences a termination of employment with such Employer. A Participant shall be considered to have experienced a termination of employment when the facts and circumstances indicate that either (i) the Participant is not reasonably expected to perform further services for the Employer after a certain date, or (ii) that the level of bona fide services the Participant will perform for the Employer after such date (whether as an Employee or as an independent contractor) will permanently decrease to no more than 49% of the average level of bona fide services performed by such Participant (whether as an Employee or an independent contractor) over the immediately preceding 36-month period (or full period of services to the Employer if the Participant has been providing services to the Employer for less than 36 months).
 - (b) If a Participant is on military leave, sick leave, or other bona fide leave of absence, the employment relationship between the Participant and the Employer shall be treated as continuing intact, provided that the period of such leave does not exceed six months, or longer, so long as the Participant retains a right to reemployment with the Employer under an applicable statute or by contract. If the period of leave exceeds six months and the Participant does not retain a right to reemployment under an applicable statute or by contract, the Participant will incur a Separation from Service as of the first day immediately following the end of such six-month period. However, where a Participant's leave of absence is due to his or her Disability, a 29-month period of absence will be substituted for such six-month period. In applying the provisions of this paragraph, a leave of absence shall be considered a bona fide leave of absence only if there is a reasonable expectation that the Participant will return to perform services for the Employer.
 - (c) Notwithstanding the foregoing, if a Participant who provides services to the Employer as both an Employee and a member of the Board, then to the extent permitted by Treasury Regulation Section 1.409A-1(h)(5), the services provided by such Participant as a Board member shall not be taken into account in determining whether the Participant experiences a Separation from Service as an Employee, and the services provided by such Participant as an Employee shall not be taken into account in determining whether the Participant in determining whether the Participant as a Board member.
- 1.30 "Short-Term Payout" shall mean the payout set forth in Section 4.1.
- 1.31 "Trust" shall mean one or more trusts established pursuant to that certain Trust Agreement, dated as of January 1, 2002 between the Company and the trustee named therein, as amended from time to time.
- 1.32 "Unforeseeable Financial Emergency" shall mean an unanticipated emergency that is caused by an event beyond the control of the Participant that would result in severe financial hardship to the Participant resulting from (i) a



Amgen Nonqualified Deferred Compensation Plan

Plan Document

sudden and unexpected illness or accident of the Participant, or the Participant's spouse, Beneficiary, or dependent (as defined in Code Section 152, without regard to Code Section 152(b)(1), (b)(2), and (d)(1)(B)), (ii) a loss of the Participant's property due to casualty, or (iii) another similar extraordinary and unforeseeable circumstance arising as a result of events beyond the control of the Participant, all as determined in the sole discretion of the Committee, consistent with Treasury Regulation Section 1.409A-3(i)(3).

1.33 "Valuation Date" shall mean the last day of each Plan Year or any other date as of which the Committee, in its sole discretion, designates as a Valuation Date.

ARTICLE 2 Selection/Enrollment/Eligibility

- 2.1 <u>Selection by Committee</u>. Participation in the Plan shall be limited to a select group of Employees of the Employers, each of whom is a member of management or is highly compensated, and to members of the Board. From the group of employees who are management or highly compensated, the Committee shall select, in its sole discretion, Employees to participate in the Plan, and they shall be designated on Appendix B.
- 2.2 <u>Enrollment Requirements</u>. As a condition to participation, each member of the Board and selected Employee shall complete, execute, and return to the Committee a Plan Agreement, an Election Form and a Beneficiary Designation Form, all within the timeframes set forth in Section 3.2. In addition, the Committee may establish from time to time such other enrollment requirements as it determines in its sole discretion are necessary.
- 2.3 <u>Eligibility/Commencement of Participation</u>. Provided an Employee selected to participate in the Plan or member of the Board has met all enrollment requirements set forth in this Plan and required by the Committee, including returning all required documents to the Committee within the specified time period set forth in Section 2.2, that Employee or Board member shall commence participation in the Plan on the first day of the month following the month in which he or she completes all enrollment requirements or such other date specified by the Committee.
- 2.4 **Termination of Participation and/or Deferrals**. If the Committee determines in good faith that a Participant no longer qualifies as a member of a select group of management or highly compensated employees, as membership in such group is determined in accordance with Sections 201(2), 301(a)(3), and 401(a)(1) of ERISA, the Committee shall have the right, in its sole discretion, to prevent the Participant from making future deferral elections in a subsequent Plan Year.

ARTICLE 3 Deferral Commitments/Company Matching/Crediting/Taxes

3.1 Maximum Deferrals.

(a) <u>Annual Base Salary and Annual Bonus</u>. For each Plan Year, a Participant may elect to defer, as his or her Annual Deferral Amount, Annual Base Salary or Annual Bonus up to the following maximum percentages for each deferral elected as determined by the Committee for each Plan Year:

Deferral	Maximum Percentage
Annual Base Salary	50%
Annual Bonus	100%

(b) Notwithstanding the foregoing, if a Participant first becomes a Participant after the first day of a Plan Year, the maximum Annual Deferral Amount, with respect to Annual Base Salary and Annual Bonus



Plan Document

shall be based on the amount of compensation not yet earned by the Participant as of the date the Participant submits a Plan Agreement and an Election Form to the Committee for acceptance.

(c) Notwithstanding the foregoing, Participants who are members of the Board shall be subject to a 100% maximum deferral percentage with respect to their Annual Base Salary.

3.2 Election to Defer/Effect of Election Form.

- (a) First Plan Year. A Board member or Employee designated in Appendix B who first becomes eligible to participate in the Plan on or after the beginning of a Plan Year, as determined in accordance with Treasury Regulation Section 1.409A-2(a)(7)(ii), may elect to defer the portion of his or her Annual Base Salary and/or Annual Bonus paid for services performed after such election, provided that such Employee or Board member (1) submits an Election Form to the Committee within 30 days after the Employee is selected by the Committee for participation in the Plan or within 30 days of the effective date of the member's appointment to the Board, and (2) has not been eligible to participate in this Plan or in any other plan that would be aggregated with this Plan under Treasury Regulation Section 1.409A-1(c) at any time during the 24-month period ending on the date he or she became eligible to participate in the Plan.
- (b) Subsequent Plan Years. For each succeeding Plan Year, an irrevocable deferral election for that Plan Year, and such other elections as the Committee deems necessary or desirable under the Plan, shall be made by timely delivering a new Election Form to the Committee, in accordance with its rules and procedures and before the end of the Plan Year preceding the Plan Year in which the services are first performed for which the Annual Base Salary and/or Annual Bonus that is subject to the election is paid. If no such Election Form is timely delivered for a Plan Year, the Annual Deferral Amount shall be zero for that Plan Year.
- 3.3 **401(k)** Plan/1165(e) Plan Participation. A Participant who also participates in the 401(k) Plan or in the Retirement and Savings Plan of Amgen Manufacturing, Limited (the "1165(e) Plan") shall have the opportunity to delay the effective date of his or her deferral election until the latest date selected by the Committee for the applicable Plan Year. Elections under this Section 3.3 shall be made on an Election Form in accordance with such rules and procedures the Committee shall establish, within the timeframes set forth in Section 3.2.
- 3.4 <u>Withholding of Annual Deferral Amounts</u>. For each Plan Year and except as provided in Section 3.3, the Annual Base Salary portion of the Annual Deferral Amount for each Participant shall be withheld from each regularly scheduled Annual Base Salary payroll. The Annual Bonus portion of the Annual Deferral Amount shall be withheld at the time the Annual Bonus is or otherwise would be paid to the Participant, whether or not this occurs during the Plan Year itself.
- 3.5 **Annual Company Contribution Amount**. For each Plan Year, an Employer, in its sole discretion, may, but is not required to, credit any amount it desires to any Participant's Company Contribution Account under this Plan, which amount shall be for that Participant the Annual Company Contribution Amount for that Plan Year. The amount so credited to a Participant may be smaller or larger than the amount credited to any other Participant, and the amount credited to any Participant for a Plan Year may be zero, even though one or more other Participants receive an Annual Company Contribution Amount for that Plan Year. The Annual Company Contribution Amount, if any, shall be credited as of the date determined by the Committee in its sole discretion. If a Participant is not employed by an Employer as of the last day of a Plan Year for a reason other than his or her retirement or death while employed, the Annual Company Contribution Amount for that Plan Year in accordance with Articles 5 and 6 of the Plan shall also apply to any Annual Company Contribution Amount made on behalf of the Participant for that Plan Year.



Amgen Nonqualified Deferred Compensation Plan

Plan Document

3.6 <u>Vesting</u>.

- (a) A Participant shall at all times be 100% vested in his or her Deferral Account.
- (b) A Participant shall be vested in his or her Company Contribution Account in accordance with the vesting schedules established by the Committee, in its sole and absolute discretion, for each Annual Company Contribution Amount (and amounts credited or debited thereon) at the time each such Annual Company Contribution Amount is first credited to the Participant's Account Balance under the Plan. The vesting schedules established by the Committee for each Annual Company Contribution Amount may be different for different Participants.
- (c) Notwithstanding anything in this Section to the contrary, except as provided in subsection (d) below, in the event of a Change of Control, a Participant's Company Contribution Account shall immediately become 100% vested (without regard to whether it is already vested in accordance with the above vesting schedules).
- (d) Except as otherwise provided by written agreement between a Participant and his/her Employer, notwithstanding anything in this Section or the Plan to the contrary, the vesting schedule for a Participant's Company Contribution Account shall not be accelerated to the extent that the Committee determines that such acceleration would cause the deduction limitations of Section 280G of the Code to become effective. In the event that any portion of a Participant's Company Contribution Account is not vested pursuant to such a determination, the Participant may request independent verification of the Committee's calculations with respect to the application of Section 280G. In such case, the Committee must provide to the Participant within 15 business days of such a request an opinion from a nationally recognized accounting firm selected by the Participant (the "Accounting Firm"), to the effect that, in the Accounting Firm's opinion that any limitation in the vested percentage hereunder is necessary to avoid the limits of Section 280G, and containing supporting calculations, or, in the absence of such an opinion, shall cause the relevant portion of the Participant's Company Contribution Account to become vested. The cost of such opinion shall be paid for by the Company.
- 3.7 **<u>Crediting/Debiting of Account Balances</u>**. In accordance with, and subject to, the rules and procedures that are established from time to time by the Committee, in its sole discretion, amounts shall be credited or debited to a Participant's Account Balance in accordance with the following rules:
 - (a) **Election of Measurement Funds**. A Participant, in connection with his or her initial deferral election in accordance with Section 3.2(a) above, shall elect, on the Election Form, one or more Measurement Fund(s) to be used to determine the additional amounts to be credited to his or her Account Balance for the first business day in which the Participant commences participation in the Plan and continuing thereafter for each subsequent day in which the Participant participates in the Plan, unless changed in accordance with the next sentence. Commencing with the first business day that follows the Participant's commencement of participation in the Plan and continuing thereafter for each subsequent day in which the Participant may (but is not required to) elect, by submitting an Election Form to the Committee that is accepted by the Committee, to add or delete one or more Measurement Fund(s) to be used to determine the additional amounts to be credited to his or her Account Balance, or to change the portion of his or her Account Balance allocated to each previously or newly elected Measurement Fund. If an election is made in accordance with the previous sentence, it shall apply to the next business day and continue thereafter for each subsequent day in which the Participant participates in the Plan, unless changed in accordance with the previous sentence.
 - (b) **Proportionate Allocation**. In making any election described in Section 3.7(a) above, the Participant shall specify on the Election Form, in increments of five percentage points (5%), the percentage of his or her Account Balance to have gains and losses measured by a Measurement Fund.

Plan Document

- (c) <u>Measurement Funds</u>. From time to time, the Committee in its sole discretion shall select and announce to Participants its selection of mutual funds, insurance company separate accounts, indexed rates or other methods (each, a "Measurement Fund"), for the purpose of providing the basis on which gains and losses shall be attributed to Account Balances under the Plan. The Committee may, in its sole discretion, discontinue, substitute or add a Measurement Fund at any time. Each such action shall take effect after a reasonable period of time following the day on which Participants are given written notice of such change.
- (d) <u>Crediting or Debiting Method</u>. The performance of each elected Measurement Fund (either positive or negative) will be determined by the Committee, in its reasonable discretion, based on available reports of the performance of the Measurement Funds. A Participant's Account Balance shall be credited or debited on a daily basis based on the performance of each Measurement Fund selected by the Participant, as determined by the Committee in its sole discretion, as though (i) a Participant's Account Balance were invested in the Measurement Fund(s) selected by the Participant, in the percentages applicable to such day, as of the close of business on such day, at the closing price on such date; (ii) the portion of the Annual Deferral Amount that was actually deferred during any day were invested in the Measurement Fund(s) selected by the Participant, in the percentages applicable to such day, no later than the close of business on the first business day after the day on which such amounts are actually deferred from the Participant's Annual Base Salary through reductions in his or her payroll and from the Participant's Annual Bonus, at the closing price on such date; and (iii) any distribution made to a Participant that decreases such Participant's Account Balance ceased being invested in the Measurement Fund(s), in the percentages applicable to such day, no later than one business day prior to the distribution, at the closing price on such date.
- (e) No Actual Investment. Notwithstanding any other provision of this Plan that may be interpreted to the contrary, the Measurement Funds are to be used for measurement purposes only, and a Participant's election of any such Measurement Fund, the allocation to his or her Account Balance thereto, the calculation of additional amounts and the crediting or debiting of such amounts to a Participant's Account Balance shall not be considered or construed in any manner as an actual investment of his or her Account Balance in any such Measurement Fund. In the event that the Company or the Trustee (as that term is defined in the Trust), in its own discretion, decides to invest funds in any or all of the Measurement Funds, no Participant shall have any rights in or to such investments themselves. Without limiting the foregoing, a Participant's Account Balance shall at all times be a bookkeeping entry only and shall not represent any investment made on his or her behalf by the Company or the Trust; the Participant shall at all times remain an unsecured creditor of the Company.
- 3.8 **FICA and Other Taxes**. For each Plan Year in which an Annual Deferral Amount is being withheld from a Participant or a portion or all of Annual Company Contribution Amount becomes Vested, the Participant's Employer(s) shall withhold from that portion of the Participant's Annual Base Salary or Annual Bonus that is not being deferred, in a manner determined by the Employer(s), the Participant's share of FICA, other employment taxes and other employee contributions on such Annual Deferral Amount. If necessary, the Committee may reduce the Annual Deferral Amount or Annual Company Contribution Amount in order to comply with this Section.
- 3.9 **Distributions**. The Participant's Employer(s), or the trustee of the Trust, shall withhold from any payments made to a Participant under this Plan all federal, state and local income, employment and other taxes required to be withheld by the Employer(s), or the trustee of the Trust, in connection with such payments, in amounts and in a manner to be determined in the sole discretion of the Employer(s) and the trustee of the Trust, respectively (whichever is making the payment). The Participant's Employer, or the trustee of the Trust, shall withhold from any payments made to a Participant under this Plan any garnishment of wages in amounts and in a manner to be determined by the sole discretion of the Employer(s) and the trustee of the Trust, respectively (whichever is making the payment).



ARTICLE 4 Short-Term Payout

- 4.1 **Short-Term Payout**. In connection with each election to defer an Annual Deferral Amount, a Participant may irrevocably elect, within the timeframe and manner prescribed by Section 3.2, to receive a future "Short-Term Payout" from the Plan with respect to such Annual Deferral Amount. Subject to Article 8 below, the Short-Term Payout shall be a lump-sum payment in an amount that is equal to the Annual Deferral Amount plus amounts credited or debited in the manner provided in Section 3.7 above on that amount, determined at the time that the Short-Term Payout becomes payable (rather than the date of a Separation from Service). Subject to the terms and conditions of the Plan, each Short-Term Payout elected shall be paid out as soon as administratively practicable within the Plan Year designated by the Participant. The Plan Year designated by the Participant must be at least three, but no more than ten, Plan Years after the Plan Year in which the Annual Deferral Amount is actually deferred.
- 4.2 **Other Benefits Take Precedence Over Short-Term Payout**. Should an event occur that triggers a benefit under Article 5 or Article 6, any Annual Deferral Amount, plus amounts credited or debited thereon, that is subject to a Short-Term Payout election under Section 4.1 but not in pay status as of the date of the Participant's Separation from Service or death, shall not be paid in accordance with Section 4.1 but shall be paid in accordance with the other applicable Article.

ARTICLE 5

Distribution of Benefits Following Separation from Service

- 5.1 **Distributions**. Subject to Article 8 below, a Participant shall be entitled to a distribution of the vested interest of his or her Account Balance following Separation from Service. Such amount will be paid in a lump-sum cash payment as soon as administratively practicable during the Plan Year immediately following the Plan Year in which such Separation from Service occurs, unless the Participant has elected on an Election Form, within the time and manner prescribed by Section 3.2, to receive either (i) a lump-sum cash payment as soon as administratively practicable during the second Plan Year following the Plan Year in which the Separation from Service occurs, or (ii) installment payments in accordance with Section 5.2. A Participant is permitted to make a single distribution election with respect to all Annual Deferral Amounts and Annual Company Contribution Amounts, if any, deferred or contributed under the Plan Year's Annual Deferral Amount and Annual Company Contribution Amount, if any.
- 5.2 Installment Payments. In lieu of the lump-sum payment described in Section 5.1, a Participant may elect on an Election Form to have the vested portion of his or her Account Balance paid under the Annual Installment Method following Separation from Service. Payments under the Annual Installment Method will commence as soon as administratively practicable during the Plan Year immediately following the Plan Year in which the Participant experiences a Separation from Service, and will end in the Plan Year specified in the Election Form, which shall not be later than the Plan Year that includes the ten-year anniversary of the Participant's Separation from Service. However, if the Participant's aggregate balance under all Account Balance Plans is \$100,000 or less upon his or her Separation from Service, the Participant's election to receive payments under the Annual Installment Method shall be disregarded and the portion of the Participant's Account Balance that is subject to the election will be paid to the Participant as a lump sum as soon as administratively practicable during the Plan Year immediately following the Plan Year in which the Participant experiences a Separation from Service.
- 5.3 **Distribution Election Changes**. Subject to Section 8.3, with respect to each distribution election made pursuant to this Article 5, a Participant may have a one-time opportunity to irrevocably extend the payment date and/or change the form of payment initially designated, provided that: (i) the new distribution election shall have no effect until at least 12 months after the date on which such election is made, (ii) the payment date must be at least five years after the previously designated payment date and must involve completion of all payments not later than



Plan Document

the end of the Plan Year that includes the ten-year anniversary of the Participant's Separation from Service, and (iii) the election must be made at least 12 months prior to the previously designated payment date. The "previously designated payment date" in the preceding sentence shall be January 1 of the Plan Year in which the payment was initially scheduled to occur, which shall include only the first payment under the Annual Installment Method.

ARTICLE 6 Survivor Benefits

- 6.1 **Survivor Benefits**. If a Participant dies before his or her Account Balance has been distributed in full, the Participant's Beneficiary shall receive a survivor benefit equal to the Participant's Account Balance, payable in accordance with the following provisions of this Article 6. A Participant is permitted to make a single distribution election with respect to all Annual Deferral Amounts and Annual Company Contribution Amounts, if any, deferred or contributed under the Plan prior to 2005. For Plan Years beginning after December 31, 2004, a Participant is permitted to make a separate distribution election with respect to each Plan Year's Annual Deferral Amount and Annual Company Contribution Amount, if any.
- 6.2 **Death Before Commencement of Benefits**. Subject to Section 6.3, a Participant shall elect on an Election Form whether any amounts payable to a Beneficiary under the Plan shall be received by his or her Beneficiary in a lump sum or pursuant to the Annual Installment Method for up to a ten-year period. However, if the Participant's aggregate balance under all Account Balance Plans is \$100,000 or less upon his or her death, the Participant's election to have payments made under the Annual Installment Method shall be disregarded and the portion of the Participant's Account Balance that is subject to the election will be paid to the Beneficiary as a lump sum. If a Participant does not make any election with respect to the payment of his or her Account Balance, then such Account Balance shall be paid to the Beneficiary in a lump sum. Any lump-sum payment made pursuant to this Section 6.2 shall be made, or installment payments shall commence, within 60 days of the Participant's death.
- 6.3 **Death After Commencement of Benefits**. If a Participant dies after installment payments have commenced, but before his or her Account Balance is paid in full, the Participant's remaining installment payments shall continue and shall be paid to the Participant's Beneficiary over the remaining number of years and in the same amounts as payments would have been made to the Participant had the Participant survived.
- 6.4 **Distribution Election Changes.** Subject to Section 8.3, with respect to each distribution election made pursuant to this Article 6, a Participant may have a one-time opportunity to irrevocably extend the payment date and/or change the form of payment initially designated, provided that: (i) the new distribution election shall have no effect until at least 12 months after the date on which such election is made, (ii) the payment date must involve completion of all payments not later than the end of the Plan Year that includes the ten-year anniversary of the Participant's death, and (iii) the election must be made at least 12 months prior to the previously designated payment date. The "previously designated payment date" in the preceding sentence shall be January 1 of the Plan Year in which the payment was initially scheduled to occur, which shall include only the first payment under the Annual Installment Method.
- 6.5 **Beneficiary**. Each Participant shall have the right, at any time, to designate his or her Beneficiary(ies) (both primary and contingent) to receive any benefits payable under the Plan to a beneficiary upon the death of a Participant. The Beneficiary designated under this Plan may be the same as or different from the Beneficiary designation under any other plan of an Employer in which the Participant participates.
- 6.6 **Beneficiary Designation Change/Spousal Consent**. A Participant shall designate his or her Beneficiary by completing and signing the Beneficiary Designation Form, and returning it to the Committee or its designated agent. A Participant shall have the right to change a Beneficiary by completing, signing and otherwise complying with the terms of the Beneficiary Designation Form and the Committee's rules and procedures, as in effect from time to time. A Participant may name someone other than his or her spouse as a Beneficiary only if a spousal consent, in the form designated by the Committee, is signed by that Participant's spouse and returned to the Committee. Upon the acceptance by the Committee of a new Beneficiary Designation Form, all Beneficiary designations previously filed shall be cancelled. The Committee shall be entitled to rely on the last Beneficiary Designation Form filed by the Participant and accepted by the Committee prior to his or her death. Notwithstanding anything in this Section or the Plan to the contrary, a Participant's designation of a spouse as a



Plan Document

Beneficiary shall automatically be cancelled and revoked on the date a Participant's divorce from that spouse becomes final.

- 6.7 **Acknowledgment**. No designation or change in designation of a Beneficiary shall be effective until received and acknowledged in writing by the Committee or its designated agent.
- 6.8 **No Beneficiary Designation**. If a Participant fails to designate a Beneficiary as provided in Sections 6.5, 6.6 and 6.7 above or, if all designated Beneficiaries predecease the Participant or die prior to complete distribution of the Participant's benefits, then the Participant's designated Beneficiary shall be deemed to be his or her surviving spouse. If the Participant has no surviving spouse, the Participant's designated Beneficiary shall be deemed to be the Participant's estate.
- 6.9 **Discharge of Obligations**. The payment of benefits under the Plan to a Beneficiary shall fully and completely discharge all Employers and the Committee from all further obligations under this Plan with respect to the Participant, and that Participant's Plan Agreement shall terminate upon such full payment of benefits.

ARTICLE 7 Disability Waiver

7.1 Disability Waiver.

- (a) <u>Waiver of Deferral</u>. A Participant who is determined by the Committee to be suffering from a Disability shall have no further deferrals of the Annual Deferral Amount that would otherwise have been withheld from a Participant's Annual Base Salary or Annual Bonus for the Plan Year during which the Participant first suffers a Disability. During the period of Disability, the Participant shall not be allowed to make any additional deferral elections, but will continue to be considered a Participant for all other purposes of this Plan. Any cancellation of the Participant's Annual Deferral Amount pursuant to this Section 7.1(a) shall occur by the later of the end of the Plan Year or the 15th day of the third month following the date the Participant incurs a Disability.
- (b) <u>Return to Work</u>. If a Participant returns to employment with an Employer, after a Disability ceases, the Participant may elect to defer an Annual Deferral Amount for the Plan Year following his or her return to employment or service and for every Plan Year thereafter while a Participant in the Plan; provided such deferral elections are otherwise allowed and an Election Form is delivered to and accepted by the Committee for each such election in accordance with Section 3.2 above.

ARTICLE 8 Distributions – General

- 8.1 **Generally**. Except as otherwise provided, any and all distributions pursuant to Articles 4 through 6 shall be subject to the terms and conditions of this Article 8.
- 8.2 Six-Month Delayed Payment. If, at the time of the Participant's Separation from Service, the Participant is a "specified employee" (within the meaning of Section 409A of the Code and Treasury Regulation Section 1.409A-1(i)), the Company will not pay or provide any "Specified Benefits" (as defined herein) during the six-month period beginning with the date of the Participant's Separation from Service (the "409A Suspension Period"). In the event of a Participant's death, however, the Specified Benefits shall be paid to the Participant's Beneficiary without regard to the 409A Suspension Period. For purposes of this Plan, "Specified Benefits" are any amounts of the Participant's Account Balance that would be subject to Section 409A additional taxes if the Company were to pay them, pursuant to this Plan, on account of the Participant's Separation from Service. During the 409A Suspension Period, the Account Balance will continue to be credited or debited in accordance with Section 3.7

Amgen Nonqualified Deferred Compensation Plan

Plan Document

above until the Account Balance is distributed. Within 14 calendar days after the end of the 409A Suspension Period, the Participant shall be paid a lumpsum payment in cash equal to any Specified Benefits delayed during the 409A Suspension Period.

- 8.3 **Special Transition Rule for 2008.** This Section is effective October 1, 2008. The Committee shall, in accordance with the rules established by the Committee and pursuant to Notice 2007-86, provide a limited period in 2008 in which Participants (other than those subject to the special grandfathering rules set forth in the "Purpose" provision before Article 1 above) may make distribution elections with respect to all or part of their Account Balances. On or before the deadline established by the Committee, which in no event shall be later than December 31, 2008, Participants shall make any such elections on an Election Form that the Committee provides. Any distribution election made by a Participant, and accepted by the Committee, shall not be treated as a change in either the form or timing of a Participant's benefit payment for purposes of Code Section 409A or Section 5.3 or 6.4 of the Plan. If any distribution election submitted by a Participant in accordance with this Section 8.3 either (i) relates to an amount that would otherwise be paid to the Participant in 2008, or (ii) would cause an amount to be paid to the Participant in 2008, such election shall not be effective with respect to such amount.
 - (a) <u>Short-Term Payouts</u>. If a Participant makes no election pursuant to this Section 8.3(a), the portion of the Participant's Annual Deferral Amount that the Participant previously elected to receive as a Short-Term Payout shall be payable as a lump sum as soon as administratively practicable following the January 1 designated by the Participant in the Election Form, but no later than the end of the Plan Year in which such date occurs. If the Participant elects to change a previous election with respect to a Short-Term Payout pursuant to this Section 8.3(a), the distribution date designated by the Participant must be at least three but no more than ten Plan Years after the Plan Year in which the Annual Deferral Amount is actually deferred.
 - (b) <u>Distributions Following Separation from Service</u>. If a Participant makes no election pursuant to this Section 8.3(b), the vested portion of the Participant's Account Balance that the Participant previously elected to receive upon termination of employment, or the vested portion of the Participant's Account Balance with respect to which the Participant previously made no election, will be paid as a lump sum as soon as administratively practicable during the Plan Year immediately following the Plan Year in which the Participant incurs a Separation from Service. However, a Participant may make an election pursuant to this Section 8.3(b) to receive such Account Balance as either (i) a lump-sum cash payment as soon as administratively practicable during the second Plan Year following the Plan Year in which the Participant's Separation from Service occurs, or (ii) installment payments under the Annual Installment Method pursuant to Section 5.2.
 - (c) <u>Distributions Following Death.</u> If a Participant makes no election pursuant to this Section 8.3(c), the vested portion of the Participant's Account Balance that the Participant previously elected to be paid to the Participant's Beneficiary upon his or her death will be paid to the Beneficiary as a lump sum within 60 days of the Participant's death. However, pursuant to this Section 8.3(c), a Participant may elect to have such Account Balance paid to his or her Beneficiary in installment payments under the Annual Installment Method pursuant to Section 6.2.
- 8.4 <u>Accelerated Distributions</u>. Distributions may not be accelerated, except as provided in this Section 8.4 and Article 10. Distributions may be accelerated under the following circumstances:
 - (a) A Participant who has elected to receive any Annual Deferrals under the Annual Installment Method subsequently elects to change from installments to a lump-sum distribution, provided the change in the distribution election satisfies the requirements set forth in Section 5.3 or 6.4 above.

Plan Document

- (b) A Participant becomes liable for FICA taxes with respect to any portion of the Participant's Account Balance, provided that if an accelerated distribution is made pursuant to this paragraph, the amount distributed shall not exceed the aggregate of the FICA taxes imposed on the Participant's Account Balance plus any income tax withholding required for the FICA withholdings.
- (c) The Plan fails to meet the requirements of Code Section 409A with respect to a Participant, provided that, if an accelerated distribution is made pursuant to this paragraph, the amount that shall be distributed shall not exceed the amount required to be included in income as a result of the failure to comply with Code Section 409A.
- 8.5 **Delayed Distributions**. Except as provided in Sections 5.3, 6.4, 8.2, 8.3, and this Section 8.5, payments may not be delayed. Distributions may be delayed under the following circumstances:
 - (a) If the Company reasonably anticipates that the Company's deduction with respect to any distribution from this Plan would be limited or eliminated by application of Code Section 162(m), then to the extent permitted by Treasury Regulation Section 1.409A-2(b)(7)(i), payment shall be delayed as deemed necessary to ensure that the entire amount of any distribution from this Plan is deductible. Any amounts for which distribution is delayed pursuant to this Section shall continue to be credited or debited with additional amounts in accordance with Section 3.7. The delayed amounts (as adjusted for any amounts credited or debited thereon) shall be distributed to the Participant (or his Beneficiary in the event of the Participant's death) at the earliest date the Company reasonably anticipates that the deduction of the payment of the amount will not be limited or eliminated by application of Code Section 162(m).
 - (b) The Committee may delay payment if it reasonably anticipates that making the payment would violate federal securities laws or other applicable law, provided the Company treats all payments to similarly situated Participants on a reasonably consistent basis and the payment is made at the earliest date at which the Committee reasonably anticipates that the making of the payment will not cause such a violation.
- 8.6 Withdrawal/Cancellation of Deferrals for Unforeseeable Financial Emergencies. If the Participant experiences an Unforeseeable Financial Emergency, the Participant may petition the Committee to receive a partial or full payout from the Plan. The payout shall not exceed the lesser of the Participant's then vested Account Balance or the amount reasonably needed to satisfy the Unforeseeable Financial Emergency. If, subject to the sole discretion of the Committee, the petition for a payout is approved, any payout shall be made within 60 days of the date of such approval. In addition, if the petition for payout is approved, or if the Participant receives a hardship distribution from the 401(k) Plan, the Participant's deferrals for the remainder of the Plan Year shall be cancelled effective as of the date of such hardship distribution or approval. Any deferral for a subsequent Plan Year must be made in accordance with Section 3.2.
- 8.7 Withholding of Employment Taxes Upon Distribution. The Participant's Employer(s), or the trustee of the Trust, shall withhold from any payments made to a Participant under this Plan all federal, state and local income, employment and other taxes required to be withheld by the Employer(s), or the trustee of the Trust, in connection with such payments, in amounts and in a manner to be determined in the sole discretion of the Employer(s) and the trustee of the Trust, respectively (whichever is making the payment). The Participant's Employer, or the trustee of the Trust, shall withhold from any payments made to a Participant under this Plan any garnishment of wages in amounts and in a manner to be determined by the sole discretion of the Employer(s) and the trustee of the Trust, respectively (whichever is making the payment). Except to the extent specifically provided within this Plan or any separate written agreement between a Participant and the Employer, a Participant shall be solely responsible for the satisfaction of any taxes with respect to the benefits payable to the Participant under this Plan (including, but not limited to, employment taxes imposed on employees and additional taxes on nonqualified deferred compensation). Although the Company intends and expects that the Plan and its payments and benefits will not give rise to taxes imposed under Section 409A of the Code, neither the Company, nor its employees,

Plan Document

directors, or agents shall have any obligation to mitigate or to hold any Participant harmless from any or all of such taxes.

ARTICLE 9 Leave of Absence

- 9.1 Paid Leave of Absence. If a Participant is authorized by the Participant's Employer for any reason to take a paid leave of absence from the employment of the Employer, and such leave of absence does not constitute a Separation from Service, the Participant shall continue to be considered eligible for the benefits provided under the Plan, and the Annual Deferral Amount shall continue to be withheld during such paid leave of absence in accordance with Article 3.
- 9.2 <u>Unpaid Leave of Absence</u>. If a Participant is authorized by the Participant's Employer for any reason to take an unpaid leave of absence from the employment of the Employer, and such leave of absence does not constitute a Separation from Service, the Participant shall continue to be considered employed by the Employer and deferrals shall not be made, in the absence of compensation. Upon such expiration of the unpaid leave and resumption of entitlement to compensation, deferrals shall resume for the remaining portion of the Plan Year in which the return occurs, based on the deferral election, if any, made for that Plan Year. If no election was made for that Plan Year, no deferral shall be withheld.

ARTICLE 10 Termination/Amendment or Modification

- 10.1 <u>**Termination**</u>. Although the Company anticipates that it will continue the Plan for an indefinite period of time, there is no guarantee that the Company will continue the Plan or will not terminate the Plan at any time in the future. Accordingly, by action of its Board of Directors or the Committee, the Company reserves the right to discontinue its sponsorship of the Plan and to terminate the Plan at any time in accordance with one of the following circumstances set forth in subsections (a) through (c) below and in Treasury Regulation Section 1.409A-3(j)(4)(ix):
 - (a) The Company may terminate the Plan if the termination and liquidation is not proximate to a downturn in the Company's financial health and:
 - (i) The Plan and all other plans maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) are irrevocably terminated;
 - (ii) No payments other than payments that would otherwise be payable under the terms of the Plan are made within 12 months following the date the Company takes all necessary actions to terminate and liquidate the Plan;
 - (iii) Except with respect to the Participants who became entitled to benefits under the terms of the Plan and any other plan maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) within the first 12 months following the date such plans are irrevocably terminated, all payments to the Participants due under the terms of such plans must be made between the first day of the 13th month and the last day of the 24th month following the date such plans terminated; and
 - (iv) The Company does not adopt a plan that would be aggregated with this Plan under Treasury Regulation Section 1.409A-1(c) within three years following the date the Plan is terminated.

Plan Document

- (b) The Company terminates and liquidates the Plan pursuant to irrevocable action taken within 30 days preceding or 12 months following a "change in control event" (defined below), provided that the Plan and all other plans maintained by the Company that would be aggregated with the Plan under Treasury Regulation Section 1.409A-1(c) are terminated on the same date with respect to each participant in such plans that experienced the "change in control event," and all such participants receive all benefits payable under such plans within 12 months following the termination date. For purposes of this Section 10.1(b), "change in control event" shall have the meaning set forth in Treasury Regulation Section 1.409A-3(i) (5).
- (c) The Company terminates and liquidates the Plan within 12 months of a corporate dissolution taxed under Code Section 331, or with the approval of a bankruptcy court pursuant to 11 U.S.C. § 503(b)(1)(A), provided that all benefits payable under the Plan are distributed to Participants during the earlier of (i) the taxable year in which the amount is actually or constructively received, or (ii) the latest of the calendar year in which (a) the Plan is terminated and liquidated; (b) the benefits are no longer subject to a substantial risk of forfeiture; or (c) the payment first becomes administratively practicable.
- 10.2 **Amendment**. The Company may, at any time, amend or modify the Plan in whole or in part by the action of the Committee; provided, however, that: (i) no amendment or modification shall be effective to decrease or restrict the value of a Participant's Account Balance in existence at the time the amendment or modification is made, calculated as if the Participant had experienced a Separation from Service as of the effective date of the amendment or modification, (ii) no adverse amendment or modification shall be effective upon or after a Change of Control without the prior written consent of a majority of the Participants, and (iii) no amendment or modification of this Section 10.2 or Section 11.2 of the Plan shall be effective. The amendment or modification of the Plan shall not affect any Participant or Beneficiary who has become entitled to benefits under the terms of the Plan as of the date of the amendment or modification.

This Plan is intended to comply with Section 409A of the Code, and the Company shall have complete discretion to interpret and construe this Plan and any associated documents in any manner that establishes an exemption from or otherwise conforms them to the requirements of Section 409A. If, for any reason including imprecision in drafting, any Plan provision does not accurately reflect its intended establishment of an exemption from or compliance with Section 409A of the Code, as demonstrated by consistent interpretations or other evidence of intent, the provision shall be considered ambiguous and shall be interpreted by the Company in a fashion consistent herewith, as determined in the sole and absolute discretion of the Company. The Company reserves the right to unilaterally amend this Plan without the consent of any Participant in order to accurately reflect its correct interpretation and operation, as well as to maintain an exemption from or compliance with Section 409A of the Code.

- 10.3 **Plan Agreement**. Despite the provisions of Sections 10.1 and 10.2 above, if a Participant's Plan Agreement contains benefits or limitations that are not in this Plan document, the Company may only amend or terminate such provisions with the consent of the Participant.
- 10.4 **Effect of Payment**. The full payment of the applicable benefit under Article 4, 5, or 6 of the Plan shall completely discharge all obligations to a Participant and his or her designated Beneficiaries under this Plan and the Participant's Plan Agreement shall terminate.

ARTICLE 11 Administration

- 11.1 **Committee Duties**. Except as otherwise provided in this Article 11, this Plan shall be administered by the Compensation Committee of the Board, or such committee or delegates as the Compensation Committee of the Board shall appoint. The Committee shall also have the discretion and authority to (i) make, amend, interpret, and enforce all appropriate laws, rules and regulations for the administration of this Plan and (ii) decide or resolve any and all questions including interpretations of this Plan, as may arise in connection with the Plan. Any individual serving on the Committee who is a Participant shall not vote or act on any matter relating solely to himself or herself. When making a determination or calculation, the Committee shall be entitled to rely on information furnished by a Participant, the Company or any Employer.
- 11.2 Administration Upon Change of Control. For purposes of this Plan, the Company, acting through the Committee, shall be the "Administrator" at all times prior to the occurrence of a Change of Control. Upon and after the occurrence of a Change of Control, the "Administrator" shall be an independent third party selected by the Trustee and approved by the individual who, immediately prior to such event, was the Company's Chief Executive Officer or, if not so identified, the Company's highest ranking officer (the "Ex-CEO"). The Administrator shall have the discretionary power to determine all questions arising in connection with the administration of the Plan and the interpretation of the Plan and Trust including, but not limited to benefit entitlement determinations; provided, however, upon and after the occurrence of a Change of Control, the Administrator shall have no power to direct the investment of Plan or Trust assets or select any investment manager or custodial firm for the Plan or Trust. Upon and after the occurrence of a Change of Control, the Company must: (1) pay all reasonable administrative expenses and fees of the Administrator; (2) pursuant to Section 11.5, indemnify the Administrator against any costs, expenses and liabilities including, without limitation, attorney's fees and expenses arising in connection with the performance of the Administrator hereunder, except with respect to matters resulting from the gross negligence or willful misconduct of the Administrator or its employees or agents; and (3) pursuant to Section 11.6, supply full and timely information to the Administrator or all matters relating to the Plan, the Trust, the Participants and their Beneficiaries, the Account Balances of the Participants, the date of circumstances of the retirement, Disability, death or Separation from Service of the Participants, and such other pertinent information as the Administrator may reasonably require. Upon and after a Change of Control, the Administrator may be terminated (and a replacement appointed) by the Trustee only with the approval of the Ex-CEO. Upon and after a Change of Control, the Administrator may not be terminated by the Company.
- 11.3 <u>Agents</u>. In the administration of this Plan, the Committee and the Administrator may, from time to time, employ agents and delegate to them such of their respective administrative duties as they see fit (including acting through a duly appointed representative) and may from time to time consult with counsel who may be counsel to any Employer.
- 11.4 **Binding Effect of Decisions**. The decisions or actions of the Committee, the Administrator and/or their respective delegates, with respect to any question arising out of or in connection with the administration, interpretation and application of the Plan and the rules and regulations promulgated hereunder shall be final and conclusive and binding upon all persons having any interest in the Plan.
- 11.5 **Indemnity of Committee**. All Employers shall indemnify and hold harmless the members of the Committee, and any Employee to whom the duties of the Committee may be delegated, and the Administrator against any and all claims, losses, damages, expenses or liabilities arising from any action or failure to act with respect to this Plan, except in the case of willful misconduct by the Committee, any of its members, any such Employee or the Administrator.

Amgen Nonqualified Deferred Compensation Plan

Plan Document

11.6 **Employer Information**. To enable the Committee and Administrator to perform their respective functions, the Company and each Employer shall supply full and timely information to the Committee or Administrator, as the case may be, on all matters relating to the compensation of its Participants, the date and circumstances of the retirement, Disability, death or Separation from Service of its Participants, and such other pertinent information as the Committee or Administrator may reasonably require.

ARTICLE 12 Other Benefits and Agreements

12.1 <u>Coordination with Other Benefits</u>. The benefits provided for a Participant and Participant's Beneficiary under the Plan are in addition to any other benefits available to such Participant under any other plan or program for employees of the Participant's Employer. The Plan shall supplement and shall not supersede, modify or amend any other such plan or program except as may otherwise be expressly provided.

ARTICLE 13 <u>Claims Procedures</u>

- 13.1 **Presentation of Claim**. Any Participant or Beneficiary of a deceased Participant (such Participant or Beneficiary being referred to below as a "Claimant") may deliver to the Committee a written claim for a determination with respect to the amounts distributable to such Claimant from the Plan. If such a claim relates to the contents of a notice received by the Claimant, the claim must be made within 60 days after such notice was received by the Claimant. All other claims must be made within 180 days of the date on which the event that caused the claim to arise occurred. The claim must state with particularity the determination desired by the Claimant.
- 13.2 **Notification of Decision**. The Committee shall consider a Claimant's claim within a reasonable time, and shall notify the Claimant in writing:
 - (a) that the Claimant's requested determination has been made, and that the claim has been allowed in full; or
 - (b) that the Committee has reached a conclusion contrary, in whole or in part, to the Claimant's requested determination, and such notice must set forth in a manner calculated to be understood by the Claimant:
 - (i) the specific reason(s) for the denial of the claim, or any part of it;
 - (ii) specific reference(s) to pertinent provisions of the Plan upon which such denial was based;
 - (iii) a description of any additional material or information necessary for the Claimant to perfect the claim, and an explanation of why such material or information is necessary; and
 - (iv) an explanation of the claim review procedure set forth in Section 13.3 below.
- 13.3 **Review of a Denied Claim**. Within 60 days after receiving a notice from the Committee that a claim has been denied, in whole or in part, a Claimant (or the Claimant's duly authorized representative) may file with the Committee a written request for a review of the denial of the claim. Thereafter, but not later than 30 days after the review procedure began, the Claimant (or the Claimant's duly authorized representative):
 - (a) may review pertinent documents;
 - (b) may submit written comments or other documents; and/or
 - (c) may request a hearing, which the Committee, in its sole discretion, may grant.

Amgen Nonqualified Deferred Compensation Plan

Plan Document

- 13.4 **Decision on Review**. The Committee shall render its decision on review promptly, using an abuse of discretion standard of review, and shall render its decision not later than 60 days after the filing of a written request for review of the denial, unless a hearing is held or other special circumstances require additional time, in which case the Committee's decision must be rendered within 120 days after such date. Such decision must be written in a manner calculated to be understood by the Claimant, and it must contain:
 - (a) specific reasons for the decision;
 - (b) specific reference(s) to the pertinent Plan provisions upon which the decision was based; and
 - (c) such other matters as the Committee deems relevant.
- 13.5 **Legal Action**. A Claimant's compliance with the foregoing provisions of this Article 13 is a mandatory prerequisite to a Claimant's right to commence any legal action with respect to any claim for benefits under this Plan. Effective October 15, 2004, if Claimant has entered into an arbitration agreement with the Company or an Employer, the provisions of that arbitration agreement will govern following a Claimant's compliance with the foregoing provisions of this Article 13, and shall be the sole and exclusive remedy following compliance with the foregoing provisions.

ARTICLE 14 <u>Trust</u>

- 14.1 **Establishment of the Trust**. The Company may establish the Trust, and each Employer may transfer over to the Trust such assets as the Employer determines, in its sole discretion, to provide for its respective future liabilities created with respect to the Annual Deferral Amounts and Annual Company Contribution Amounts, for such Employer's Participants for all periods prior to the transfer, as well as any debits and credits to the Participants' Account Balances for all periods prior to the transfer, taking into consideration the value of the assets in the trust at the time of the transfer.
- 14.2 **Interrelationship of the Plan and the Trust**. The provisions of the Plan and the Plan Agreement shall govern the rights of a Participant to receive distributions pursuant to the Plan. The provisions of the Trust shall govern the rights of the Employers, Participants and the other creditors of the Employers to the assets transferred to the Trust. Each Employer shall at all times remain liable to carry out its obligations under the Plan.
- 14.3 **Distributions From the Trust**. Each Employer's obligations under the Plan may be satisfied with Trust assets distributed pursuant to the terms of the Trust, and any such distribution shall reduce the Employer's obligations under this Plan.
- 14.4 **Investment of Trust Assets**. The Trustee of the Trust shall be authorized, upon written instructions received from the Committee or investment manager appointed by the Committee, to invest and reinvest the assets of the Trust in accordance with the applicable Trust Agreement.

ARTICLE 15 Miscellaneous

15.1 <u>Status of Plan</u>. The Plan is intended to be a plan that is not qualified within the meaning of Code Section 401(a) and that "is unfunded and is maintained by an employer primarily for the purpose of providing deferred compensation for a select group of management or highly compensated employees" within the meaning of ERISA Sections 201(2), 301(a)(3) and 401(a)(1). The Plan shall be administered and interpreted to the extent possible in a manner consistent with that intent. The Plan is an unfunded, nontax-qualified, individual account, profit sharing plan. Plan benefits shall only accrue immediately before they are paid and may be paid directly by the Company. By electing to contribute to this Plan, each Participant acknowledges that this Plan is subject to ERISA but



Amgen Nonqualified Deferred Compensation Plan

Plan Document

exempted from all of ERISA's substantive requirements because it is a "top-hat plan," acknowledges that the Company would not have implemented or continued this Plan but for its good-faith belief that it is a top-hat plan, agrees that all Plan benefits shall be contingent on the Plan being a top-hat plan and promises never to assert otherwise.

- 15.2 **Unsecured General Creditor**. Participants and their Beneficiaries, heirs, successors and assigns shall have no legal or equitable rights, interests or claims in any property or assets of an Employer. For purposes of the payment of benefits under this Plan, the Employer's assets shall be, and remain, neither pledged nor restricted under or as a result of this Plan. An Employer's obligation under the Plan shall be merely that of an unfunded and unsecured promise to pay money in the future.
- 15.3 **Employer's Liability**. An Employer's liability for the payment of benefits shall be defined only by the Plan and the Plan Agreement, as entered into between the Employer and a Participant. An Employer shall have no obligation to a Participant under the Plan except as expressly provided in the Plan and his or her Plan Agreement.
- 15.4 **Nonassignability**. Neither a Participant nor any other person shall have any right to commute, sell, assign, transfer, pledge, anticipate, mortgage or otherwise encumber, transfer, hypothecate, alienate or convey in advance of actual receipt, the amounts, if any, payable hereunder, or any part thereof, which are, and all rights to which are expressly declared to be, unassignable and non-transferable. No part of the amounts payable shall, prior to actual payment, be subject to seizure, attachment, garnishment or sequestration for the payment of any debts, judgments, alimony or separate maintenance owed by a Participant or any other person, be transferable by operation of law in the event of a Participant's or any other person's bankruptcy or insolvency or be transferable to a spouse as a result of a property settlement or otherwise.
- 15.5 **Not a Contract of Employment**. The terms and conditions of this Plan shall not be deemed to constitute a contract of employment between any Employer and the Participant. Such employment is hereby acknowledged to be an "at will" employment relationship that can be terminated at any time for any reason, or no reason, with or without cause, and with or without notice, except to the extent expressly provided in a written employment agreement, if any. Nothing in this Plan shall be deemed to give a Participant the right to be retained in the service of any Employer or to interfere with the right of any Employer to discipline or discharge the Participant at any time.
- 15.6 **Furnishing Information**. A Participant or his or her Beneficiary, as a condition to entitlement to benefits hereunder, shall cooperate with the Committee by furnishing any and all information requested by the Committee and take such other actions as may be requested in order to facilitate the administration of the Plan and the payments of benefits hereunder, including but not limited to taking such physical examinations as the Committee may deem necessary.
- 15.7 **Terms**. Whenever any words are used herein in the masculine, they shall be construed as though they were in the feminine in all cases where they would so apply; and whenever any words are used herein in the singular or in the plural, they shall be construed as though they were used in the plural or the singular, as the case may be, in all cases where they would so apply.
- 15.8 **<u>Captions</u>**. The captions of the articles, sections and paragraphs of this Plan are for convenience only and shall not control or affect the meaning or construction of any of its provisions.
- 15.9 <u>**Governing Law.**</u> Subject to ERISA, the provisions of this Plan shall be construed and interpreted according to the internal laws of the State of California without regard to its conflicts of laws principles.

Amgen Nonqualified Deferred Compensation Plan

Plan Document

15.10 **Notice**. Any notice or filing required or permitted to be given to the Committee under this Plan shall be sufficient if in writing and hand-delivered, or sent by registered or certified mail, to the address below:

Amgen Inc. Nonqualified Deferred Compensation Plan Committee Amgen Inc. One Amgen Center Drive Thousand Oaks, CA 91320-1799

Such notice shall be deemed given as of the date of delivery or, if delivery is made by mail, as of the date shown on the postmark on the receipt for registration or certification.

Any notice or filing required or permitted to be given to a Participant under this Plan shall be sufficient if in writing and hand-delivered, or sent by mail, to the last address of the Participant shown on the records of the Company.

- 15.11 <u>Successors</u>. The provisions of this Plan shall bind and inure to the benefit of the Participant's Employer and its successors and assigns and the Participant and the Participant's designated Beneficiaries.
- 15.12 **Spouse's Interest**. The interest in the benefits hereunder of a spouse of a Participant who has predeceased the Participant shall automatically pass to the Participant and shall not be transferable by such spouse in any manner, including but not limited to such spouse's will, nor shall such interest pass under the laws of intestate succession.
- 15.13 **Validity**. In case any provision of this Plan shall be illegal or invalid for any reason, said illegality or invalidity shall not affect the remaining parts hereof, but this Plan shall be construed and enforced as if such illegal or invalid provision had never been inserted herein.
- 15.14 **Incompetent**. If the Committee determines in its discretion that a benefit under this Plan is to be paid to a minor, a person declared incompetent or to a person incapable of handling the disposition of that person's property, the Committee may direct payment of such benefit to the guardian, legal representative or person having the care and custody of such minor, incompetent or incapable person. The Committee may require proof of minority, incompetence, incapacity or guardianship, as it may deem appropriate prior to distribution of the benefit. Any payment of a benefit shall be a payment for the account of the Participant and the Participant's Beneficiary, as the case may be, and shall be a complete discharge of any liability under the Plan for such payment amount.
- 15.15 **Insurance**. The Employers, on their own behalf or on behalf of the trustee of the Trust, and, in their sole discretion, may apply for and procure insurance on the life of the Participants, in such amounts and in such forms as the Trust may choose. The Employers or the trustee of the Trust, as the case may be, shall be the sole owner and beneficiary of any such insurance. The Participants shall have no interest whatsoever in any such policy or policies, and at the request of the Employers shall submit to medical examinations and supply such information and execute such documents as may be required by the insurance company or companies to whom the Employers have applied for insurance.
- 15.16 **Legal Fees To Enforce Rights After Change of Control**. The Company and each Employer is aware that upon the occurrence of a Change of Control, the Board or the board of directors of a Participant's Employer (which might then be composed of new members) or a shareholder of the Company or the Participant's Employer, or of any successor corporation might then cause or attempt to cause the Company, the Participant's Employer or such successor to refuse to comply with its obligations under the Plan and might cause or attempt to cause the Company or the Participant's Employer to institute, or may institute, litigation seeking to deny Participants the benefits



Plan Document

intended under the Plan. In these circumstances, the purpose of the Plan could be frustrated. Accordingly, if, following a Change of Control, it should appear to any Participant that the Company, the Participant's Employer or any successor corporation has failed to comply with any of its obligations under the Plan or any agreement thereunder or, if the Company, such Employer or any other person takes any action to declare the Plan void or unenforceable or institutes any litigation or other legal action designed to deny, diminish or to recover from any Participant the benefits intended to be provided, then the Company and the Participant's Employer irrevocably authorize such Participant to retain counsel of his or her choice at the expense of the Company and the Participant's Employer (who shall be jointly and severally liable) to represent such Participant in connection with the initiation or defense of any litigation or other legal action, whether by or against the Company, the Participant's Employer or any director, officer, shareholder or other person affiliated with the Company, the Participant's Employer or any successor thereto in any jurisdiction. The Company or the Participant's Employer will pay all expenses described in this Section 15.16 no later than the last day of the Participant's taxable year immediately following the taxable year. Notwithstanding anything in this Section or the Plan to the contrary, the Company and/or the Participant's Employer shall have no obligation for any unpaid expenses under this Section, and the Participant shall reimburse the Company and/or the Participant's Employer for expenses already paid, to the extent there is a judicial determination or final arbitration decision that the litigation or other legal action brought by the Participant is frivolous.

Amgen Nonqualified Deferred Compensation Plan Plan Document

IN WITNESS WHEREOF, the Company has signed this amended and restated Plan document as of ______, 2008.

"Company"

Amgen Inc., a Delaware corporation

 By:
 /s/ BRIAN MCNAMEE

 Title:
 Senior Vice President, Human Resources

23

Amgen Nonqualified Deferred Compensation Plan Plan Document

Appendix A

The following subsidiaries and affiliates of Amgen Inc. are designated as Employers:

Amgen Fremont Inc. Amgen Manufacturing, Limited Amgen Mountain View Inc. Amgen SF, LLC Amgen USA Inc. Amgen Worldwide Services, Inc. Immunex Corporation Immunex Manufacturing Corporation Immunex Rhode Island Corporation Tularik Pharmaceutical Company

24

Amgen Nonqualified Deferred Compensation Plan Plan Document

Appendix B

Subject to the other terms and conditions of the Plan, the following management-level Employees shall be eligible to participate in the Plan:

1. Those management-level Employees at Job Level 7 or higher.

2. Those management-level Employees at Job Level 6 who, prior to the implementation of the Global Career Framework, were participating in the Plan.

AMENDED AND RESTATED AMGEN INC. PERFORMANCE AWARD PROGRAM

(Amended and Restated Effective October 1, 2008)

ARTICLE I

PURPOSE

The purpose of this document is to set forth the general terms and conditions applicable to the Performance Award Program (the "<u>Program</u>") established by the Compensation and Management Development Committee of the Board of Directors of Amgen Inc. (the "<u>Company</u>") pursuant to, and in implementation of, Section 10(d) of the Company's Amended and Restated 1991 Equity Incentive Plan, as amended (the "<u>1991 Plan</u>"). The Program is intended to carry out the purposes of the 1991 Plan and provide a means to reinforce objectives for sustained long-term performance and value creation by awarding selected key employees of the Company with payments in Company stock based on the level of achievement of pre-established performance goals during performance periods, subject to the restrictions and other provisions of the Program and the 1991 Plan.

ARTICLE II

DEFINITIONS

Unless otherwise defined herein, capitalized terms used herein shall have the same definitions as such terms are defined in the 1991 Plan.

"Award" shall mean the earned Performance Units payable in Common Stock under the Program for a Performance Period.

"Board" shall mean the Board of Directors of the Company.

"Code" shall mean the Internal Revenue Code of 1986, as amended, together with the regulations and official guidance promulgated thereunder.

"<u>Committee</u>" shall mean the Compensation and Management Development Committee of the Board, appointed by the Board from among its members to administer the 1991 Plan in accordance with Section 2 thereof.

"Common Stock" shall mean the common stock, par value \$0.0001 per share, of the Company.

"Determination Date" shall have the meaning ascribed to it in Section 4.1.

"Participant" shall mean a key employee of the Company or an Affiliate who participates in this Program pursuant to the provisions of Article III hereof.

"<u>Performance Period</u>" shall mean a period of time with respect to which performance is measured as determined by the Committee. Performance Periods may overlap.

"Performance Goals" shall have the meaning ascribed to it in Section 5.2.

"<u>Performance Unit</u>" shall mean a right granted to a Participant pursuant to the Program to receive Common Stock, the payment of which is contingent upon achieving the Performance Goals.

"<u>Permanent and Total Disability</u>" shall have the meaning ascribed to such term under Section 22(e)(3) of the Code and with such permanent and total disability being certified prior to termination of a Participant's employment by (i) the Social Security Administration, (ii) the comparable governmental authority applicable to an Affiliate of the Company, (iii) such other body having the relevant decision-making power applicable to an Affiliate of the Company, or (iv) an independent medical advisor appointed by the Company in its sole discretion, as applicable, in any such case.

"<u>Retirement-Eligible</u>" shall mean when a Participant is at least sixty-five (65) years of age, or when a Participant is at least fifty-five (55) years of age and has been an employee of the Company and/or an Affiliate of the Company for at least ten (10) consecutive years.

"<u>Section 162(m) Participant</u>" shall mean any Participant designated by the Committee as a "covered employee" within the meaning of Section 162(m) of the Code whose compensation for the fiscal year in which the Participant is so designated or a future fiscal year may be subject to the limit on deductible compensation imposed by Section 162(m) of the Code.

"Voluntary Retirement" shall mean voluntary termination of employment that is not the result of Permanent and Total Disability.

ARTICLE III

PARTICIPATION

3.1 <u>Participants</u>. Participants for any Performance Period shall be those active key employees of the Company or an Affiliate who are designated in writing as eligible for participation by the Committee within the first ninety (90) days of such Performance Period.

3.2 <u>No Right to Participate</u>. No Participate or other employee of the Company or an Affiliate shall, at any time, have a right to participate in this Program for any Performance Period, notwithstanding having previously participated in this Program.

ARTICLE IV

ADMINISTRATION

4.1 <u>Generally</u>. The Committee shall establish the basis for payments under this Program in relation to specified Performance Goals, as more fully described in Article V hereof. With respect to the 162(m) Participants, the Committee shall establish the basis for payments under this Program in relation to specified Performance Goals within the first ninety (90) days of each Performance Period, but in no event after 25 percent of the Performance Period has lapsed. Following the end of each Performance Period, once all of the information necessary for the Committee to determine the Company's performance is made available to the Committee, the Committee shall determine the amount of the Award payable to each Participant; *provided, however*, that any such determination shall be made no later than six months following the end of such Performance Period (the date of such determination shall hereinafter be called the "Determination Date"). The Committee shall have the power and authority granted it under Section 2 of the 1991 Plan, including, without limitation, the authority to construe and interpret this Program, to prescribe, amend and rescind rules, regulations and procedures relating to its administration and to make all other determinations necessary or advisable for administration of this Program. Decisions of the Committee in accordance with the authority granted hereby shall be conclusive and binding. Subject

only to compliance with the express provisions hereof, the Committee may act in its sole and absolute discretion with respect to matters within its authority under this Program.

4.2 <u>Provisions Applicable to Section 162(m) Participants</u>. Subject to the sole discretion of the Committee, any Awards paid hereunder to a Section 162(m) Participant shall satisfy and shall be interpreted in a manner that satisfies any applicable requirements as "qualified performance-based compensation" within the meaning of Section 162(m) of the Code and any provisions, application or interpretation of the Program or the 1991 Plan that is inconsistent with this intent shall be disregarded. To the extent that any Award (i) is deemed to constitute "nonqualified deferred compensation" (within the meaning of Code Section 409A) and (ii) would nevertheless be subject to the deduction limitations imposed by Section 162(m) of the Code in the year in which such Award would otherwise be paid under this Program, the payment of such Award may, in the Committee's discretion, be delayed until the earlier of (A) the first year in which such Award would not be subject to the deduction limitations imposed by Section 162(m) or (B) such time as the Participant ceases to be a "service provider" to the Company (within the meaning of Section 409A of the Code).

4.3 <u>Provisions Applicable to Participants in Foreign Jurisdictions</u>. Notwithstanding any provision of the Program to the contrary, in order to comply with the laws in other countries in which the Company and its Affiliates operate or have employees, the Committee, in its sole discretion, shall have the power and authority to:

(i) modify the terms and conditions of any award of Performance Units granted to employees outside the United States to comply with applicable foreign laws;

(ii) condition the effectiveness of any award of Performance Units upon approval or compliance with any applicable foreign laws, regulations, rules or local governmental regulatory exemption or approvals;

(iii) provide for payment of any Award in cash or Common Stock, at the Company's election, to the extent necessary to comply with applicable foreign laws; and

(iv) take any other action, before or after an award of Performance Units is made, that it deems advisable to obtain approval or comply with any necessary local governmental regulatory exemptions or approvals.

Notwithstanding the foregoing, the Committee may not take any actions hereunder, and no award of Performance Units shall be granted, that would violate the Securities Act of 1933, as amended, Securities Exchange Act of 1934, as amended, the Code, or any other securities or tax or other applicable law or regulation.

ARTICLE V

AWARD DETERMINATIONS

5.1 <u>Award of Performance Units</u>. The Committee shall determine the number of Performance Units (rounded down to the nearest whole number) to be awarded under this Program to each Participant with respect to such Performance Period. With respect to the Section 162(m) Participants, the Committee shall determine the number of Performance Units (rounded down to the nearest whole number) to be awarded under this Program to each Section 162(m) Participant with respect to such Performance Period down to the nearest whole number) to be awarded under this Program to each Section 162(m) Participant with respect to such Performance Period within the first ninety (90) days of such Performance Period, but in no event after 25 percent of the Performance Period has elapsed. Performance Units granted under the Program shall constitute stock bonuses under Sections 7 and 10(d) of the 1991 Plan.

5.2 <u>Performance Requirements</u>. The Committee shall approve the performance goals (collectively, the "<u>Performance Goals</u>") with respect to any of the business criteria permitted under Section 10(d) of the 1991 Plan), each subject to such adjustments as the Committee may specify in writing at such time, and shall establish a formula, standard or schedule which aligns the level of achievement of the Performance Goals with the earned Performance Units.

With respect to the Section 162(m) Participants, the Committee shall approve the Performance Goals within the first ninety (90) days of such the Performance Period, but in no event after 25 percent of the Performance Period has elapsed, and the Performance Goals may not be changed during the Performance Period, but the thresholds, targets and multiplier measures of the Performance Goals shall be subject to such adjustments as the Committee may specify in writing within the first ninety (90) days of the Performance Period, but in no event after 25 percent of the Performance Period, but in no event after 25 percent of the Performance Period.

ARTICLE VI

PAYMENT OF AWARDS

6.1 Form and Timing of Payment. Except as set forth in Section 8.1 below, no Award payable pursuant to this Program shall be paid unless and until the Committee certifies, in writing, the extent to which the Performance Goals have been achieved and the corresponding number of Performance Units earned. The specified payment date applicable to such Awards shall be the year immediately following the tax year including the end of the Performance Period. Shares of Common Stock issued in respect of an Award shall be deemed to be issued in consideration for future services to be rendered or past services actually rendered to the Company or for its benefit, by the Participant, which the Committee deems to have a value at least equal to the aggregate par value thereof.

6.2 <u>Tax Withholding</u>. Regardless of any action the Company or its Affiliate takes with respect to any or all income tax (including federal, state and local taxes), social insurance, payroll tax, payment on account or other tax-related items related to participation in the Program and legally applicable to the Participant ("<u>Tax Obligations</u>"), the Participant acknowledges that the ultimate liability for all Tax Obligations is and remains the Participant's responsibility and may exceed the amount actually withheld by the Company and/or its Affiliate. The Participant further acknowledges that the Company and/or its Affiliate (i) make no representations or undertakings regarding the treatment of any Tax Obligations in connection with any aspect of the Performance Units, including the grant of the Performance Units, the vesting of Performance Units, the conversion of the Performance Units into shares or the receipt of an equivalent cash payment, the subsequent sale of any shares acquired at vesting and the receipt of any dividends; and (ii) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Performance Units to reduce or eliminate the Participant's liability for Tax Obligations or achieve any particular tax result. Furthermore, if the Participant becomes subject to tax in more than one jurisdiction between the Grant Date and the date of any relevant taxable event, the Participant acknowledges that the Company and/or its Affiliate may be required to withhold or account for Tax Obligations in more than one jurisdiction.

Prior to any relevant taxable or tax withholding event, as applicable, the Participant shall pay, or make adequate arrangements satisfactory to the Company or to its Affiliate (in their sole discretion) to satisfy all Tax Obligations. In this regard, the Participant authorizes the Company and/or its Affiliate or their respective agents, at their discretion, to satisfy all applicable Tax Obligations by one or a combination of the following:

(a) withholding from the Participant's wages or other cash compensation paid to the Participant by the Company and/or its Affiliate; or

(b) withholding from proceeds of the sale of shares of Common Stock acquired upon vesting or payment of the Performance Units either through a voluntary sale or through a mandatory sale arranged by the Company (on the Participant's behalf pursuant to this authorization); or

(c) withholding in shares of Common Stock to be issued upon vesting or payment of the Performance Units, provided that the Company and its Affiliate shall only withhold an amount of shares of Common Stock with a fair market value equal to the Tax Obligations.

To avoid adverse accounting treatment, the Company may withhold or account for Tax Obligations not to exceed the applicable minimum statutory withholding rates or other applicable withholding rates. If the Tax Obligations are satisfied by withholding in shares of Common Stock, for tax purposes, the Participant is deemed to have been issued the full number of shares of Common Stock subject to the vested Performance Units, notwithstanding that a number of the shares of Common Stock is held back solely for the purpose of paying the Tax Obligations due as a result of any aspect of the Participant's participation in the Program (any shares of Common Stock withheld by the Company hereunder shall not be deemed to have been issued by the Company for any purpose under the Program and shall remain available for issuance thereunder).

Finally, the Participant shall pay to the Company or its Affiliate any amount of Tax Obligations that the Company or its Affiliate may be required to withhold or account for as a result of the Participant's participation in the Program that cannot be satisfied by the means previously described. The Participant agrees to take any further actions and execute any additional documents as may be necessary to effectuate the provisions of this Section III. Notwithstanding Section II above, the Company may refuse to issue or deliver the shares or the proceeds of the sale of shares of Common Stock if the Participant fails to comply with its obligations in connection with the Tax Obligations.

ARTICLE VII

TERMINATION OF EMPLOYMENT

7.1 Termination of Employment During Performance Period.

(a) In the event that a Participant's employment with the Company or an Affiliate is terminated prior to the last business day of a Performance Period by reason of such Participant's Voluntary Retirement and such Participant is Retirement-Eligible on the date of such termination, the full or prorated amount of such Participant's Award, if any, applicable to such Performance Period shall be paid in accordance with the provisions of Article VI above, *provided*, that if (i) amounts payable under this Program are deemed to constitute "nonqualified deferred compensation," and (ii) a Participant is deemed to be a "specified employee" (within the meaning of Code Section 409A), then amounts payable under this Program shall not be paid until the later of (A) the payment date described in Article VI above, or (B) the date that is six months after the date of termination (or the date on which such Participant dies, if earlier), to the extent required by Code Section 409A. For purposes of the foregoing, the amount of the Participant's Award (rounded down to the nearest whole number) shall be determined based on the Company's performance as compared to the Performance Goals for such Performance Period and (i) if the Award was granted with respect to a Performance Period commencing in a calendar year prior to the calendar year in which such Voluntary Retirement occurs, the full amount of the Award otherwise payable is multiplied by a fraction (rounded to two decimal places), the numerator of which is the number of complete months of employment during the Performance Period, and the denominator of which is twelve (12); provided

however, that prior to termination of a Participant's employment with the Company or an Affiliate, such Participant signs a general release in a form provided by the Company.

(b) In the event that a Participant's employment with the Company or an Affiliate is terminated prior to the last business day of a Performance Period by reason of such Participant's death or Permanent and Total Disability, the full or prorated amount of such Participant's Award, if any, applicable to such Performance Period shall be paid in accordance with the provisions of Article VI above, *provided*, that if (i) a Participant's employment terminates due to Permanent and Total Disability, (ii) amounts payable under this Program are deemed to constitute "nonqualified deferred compensation," and (iii) the Participant is deemed to be a "specified employee" (within the meaning of Code Section 409A), then amounts payable under this Program shall not be paid until the later of (A) the payment date described in Article VI above, or (B) the date that is six months after the date of termination (or the date on which such Participant dies, if earlier), to the extent required by Code Section 409A. For purposes of the foregoing, the amount of the Participant's Award (rounded down to the nearest whole number) shall be determined based on the Company's performance as compared to the Performance Goals for such Performance Period and (i) if the Award was granted with respect to a Performance Period commencing in a calendar year prior to the calendar year in which such termination occurs, the Award is payable, and (ii) if the Award was granted with respect to the Performance Period commencing in the calendar year in which such termination occurs, the Award otherwise payable is multiplied by a fraction (rounded to two decimal places), the numerator of which is the number of complete months of employment during the Performance Period, and the denominator of which is twelve (12). Notwithstanding the foregoing, a Participant's Permanent and Total Disability, such Participant signs a general release in a form provided by the Company.

(c) In the event that a Participant's employment with the Company or an Affiliate is terminated prior to the last business day of a Performance Period for any reason other than as specified in Sections 7.1(a) and (b) above, all of such Participant's rights to an Award for such Performance Period shall be forfeited, unless the Committee approves, based upon the recommendation of the Company's Chief Executive Officer which are based on valid business reasons, the payment of a prorated amount of the Participant's Award, if any, applicable to such Performance Period shall be paid in accordance with the provisions of Article VI above. For purposes of the foregoing, the amount of the Participant's Award (rounded down to the nearest whole number) shall be determined based on the Company's performance as compared to the Performance Goals for such Performance Period and the Award otherwise payable is multiplied by a fraction (rounded to two decimal places), the numerator of which is the number of complete months of employment during the Performance Period, and the denominator of which is the number of months in the Performance Period; provided however, that prior to termination of a Participant's employment with the Company or an Affiliate, such Participant signs a general release in a form provided by the Company.

7.2 <u>Termination of Employment After End of Performance Period</u>. In the event that a Participant's employment with the Company or an Affiliate is terminated on or after the last business day of the applicable Performance Period but prior to the Determination Date for any reason, the amount of any Award applicable to such Performance Period shall be paid to the Participant in accordance with the provisions of Article VI above.

ARTICLE VIII

CHANGE OF CONTROL

8.1 Change of Control During Performance Period.

(a) Notwithstanding anything to the contrary in the Program, in the event of a Change of Control that occurs during the first fiscal year of a Performance Period that began prior to January 1, 2008, such Performance Period shall be shortened and shall terminate as of the last business day of the last completed fiscal quarter preceding the date of such Change of Control and each Participant employed by the Company immediately prior to such Change of Control shall be entitled to a payment equal to the amount of the Participant's Award (rounded down to the nearest whole number) he or she would have received for such shortened Performance Period using the assumption that the target levels with respect to the Company's Revenue CAGR and EPS CAGR of the Performance Goals have been satisfied. Any such payment shall be made as soon as practicable following such Change of Control (provided, that the Company may elect, in its sole discretion, to make any such payments in a manner that will not subject the payments to penalties under Code Section 409A) and, in the Committee's sole discretion, may be paid in cash.

(b) Notwithstanding anything to the contrary in the Program, in the event of a Change of Control that occurs during the second or third fiscal year of a Performance Period that began prior to January 1, 2008, such Performance Period shall be shortened and shall terminate as of the last business day of the last completed fiscal quarter preceding the date of such Change of Control and each Participant employed by the Company immediately prior to such Change of Control shall be entitled to a payment equal to the greater of (i) the amount of the Participant's Award (rounded down to the nearest whole number) he or she would have received for such shortened Performance Period using the assumption that the targets levels with respect to the Company's Revenue CAGR and EPS CAGR of the Performance Goals have been satisfied, or (ii) the amount of the Participant's Award (rounded down to the nearest whole number) he or she would have been entitled to receive for such shortened Performance Period, determined based on the Company's performance as determined by the Amgen Revenue CAGR and Amgen EPS CAGR and comparative performance as determined by the Peer Group Revenue CAGR and Peer Group EPS CAGR (for the 2006-2008 Performance Period) or the Company's performance as determined by the Amgen Revenue CAGR and Amgen EPS CAGR and Total Stockholder Return (for the 2007-2009 Performance Period) for such shortened Performance Period. Any such payment shall be made as soon as practicable following such Change of Control (provided, that the Company may elect, in its sole discretion, to make any such payments in a manner that will not subject the payments to penalties under Code Section 409A) and, in the Committee's sole discretion, may be paid in cash.

(c) Notwithstanding anything to the contrary in the Program, for Performance Periods beginning on or after January 1, 2008, the Committee shall set forth the terms of any Award payable in the event of Change of Control that occurs during a Performance Period in the Performance Goals.

8.2 <u>Change of Control After End of Performance Period</u>. Notwithstanding anything to the contrary in the Program, in the event of a Change of Control that occurs after the end of the applicable Performance Period but prior to the Determination Date, the amount of any Award applicable to such Performance Period shall be paid to the Participant in accordance with the provisions of Article VI above.

ARTICLE IX

MISCELLANEOUS

9.1 <u>Plan</u>. The Program is subject to all the provisions of the 1991 Plan and its provisions are hereby made a part of the Program, including without limitation the provisions of Sections 7 and 10(d) thereof (relating to stock bonuses) and Section 11 thereof (relating to adjustments upon changes in the Common Stock), and is further subject to all interpretations, amendments, rules and regulations which may from time to time be promulgated and adopted pursuant to the 1991 Plan. In the event of any conflict between the provisions of the Program and those of the 1991 Plan, the provisions of the 1991 Plan

shall control. Notwithstanding any provision of the Program to the contrary, any earned Performance Units paid in cash rather than shares of Common Stock shall not be deemed to have been issued by the Company for any purpose under the 1991 Plan.

9.2 <u>Amendment and Termination</u>. Notwithstanding anything herein to the contrary, the Committee may, at any time, terminate, modify or suspend this Program; *provided, however*, that, without the prior consent of the Participants affected, no such action may adversely affect any rights or obligations with respect to any Awards theretofore earned but unpaid for a completed Performance Period, whether or not the amounts of such Awards have been computed and whether or not such Awards are then payable. Notwithstanding the forgoing, at any time the Committee determines that the Performance Units may be subject to Section 409A of the Code, the Committee shall have the right, in its sole discretion, and without a Participant's prior consent to amend the Program as it may determine is necessary or desirable either for the Performance Units to be exempt from the application of Section 409A or to satisfy the requirements of Section 409A, including by adding conditions with respect to the vesting and/or the payment of the Performance Units, provided that no such amendment may change the Program's "performance goals," within the meaning of Section 162(m) of the Code, with respect to any person who is a "covered employee," within the meaning of Section 162(m) of the Code.

9.3 <u>No Contract for Employment</u>. Nothing contained in this Program or in any document related to this Program or to any Award shall confer upon any Participant any right to continue as an employee or in the employ of the Company or an Affiliate or constitute any contract or agreement of employment for a specific term or interfere in any way with the right of the Company or an Affiliate to reduce such person's compensation, to change the position held by such person or to terminate the employment of such person, with or without cause.

9.4 <u>Nontransferability</u>. No benefit payable under, or interest in, this Program shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance or charge and any such attempted action shall be void and no such benefit or interest shall be, in any manner, liable for, or subject to, debts, contracts, liabilities or torts of any Participant or beneficiary; *provided*, *however*, that, nothing in this Section 9.4 shall prevent transfer (i) by will, or (ii) by applicable laws of descent and distribution.

9.5 <u>Nature of Program</u>. No Participant, beneficiary or other person shall have any right, title or interest in any fund or in any specific asset of the Company or any Affiliate by reason of any award hereunder. There shall be no funding of any benefits which may become payable hereunder. Nothing contained in this Program (or in any document related thereto), nor the creation or adoption of this Program, nor any action taken pursuant to the provisions of this Program shall create, or be construed to create, a trust of any kind or a fiduciary relationship between the Company or an Affiliate and any Participant, beneficiary or other person. To the extent that a Participant, beneficiary or other person acquires a right to receive payment with respect to an Award hereunder, such right shall be no greater than the right of any unsecured general creditor of the Company or other employing entity, as applicable. All amounts payable under this Program shall be paid from the general assets of the Company or employing entity, as applicable, and no special or separate fund or deposit shall be established and no segregation of assets shall be made to assure payment of such amounts. Nothing in this Program shall be deemed to give any employee any right to participate in this Program except in accordance herewith.

9.6 <u>Governing Law</u>. This Program shall be construed in accordance with the laws of the State of Delaware, without giving effect to the principles of conflicts of law thereof.

Note: Redacted portions have been marked with [*]. The redacted portions are subject to a request for confidential treatment that has been filed with the Securities and Exchange Commission.

AMENDMENT NO. 2 TO AMENDMENT NO. 6 TO THE ENBREL[®] SUPPLY AGREEMENT

This Amendment No. 2 to Amendment No. 6 ("Amendment No. 6.2") is made this 26th day of August, 2008 (the "Amendment No. 6.2 Effective Date") by and among **IMMUNEX CORPORATION**, a corporation of the State of Washington, having its principal place of business at One Amgen Center Drive, Thousand Oaks, California, 91320, U.S.A., together with its Affiliates ("Immunex"), **WYETH** (formerly known as American Home Products Corporation), a corporation of the State of Delaware having its corporate headquarters at Five Giralda Farms, Madison, New Jersey 07940, U.S.A. and acting through its Wyeth Pharmaceuticals Division ("Wyeth"), and **BOEHRINGER INGELHEIM PHARMA GMBH & CO. KG**, a German corporation having a place of business at Birkendorfer Straße 65, 88397 Biberach an der Riss, Federal Republic of Germany ("BIP"), and amends the *Enbrel* Supply Agreement effective as of November 5, 1998, by and among Immunex, Wyeth, and BIP, and as amended (the "Agreement").

WHEREAS, Immunex, Wyeth and BIP have entered into a certain Agreement for BIP's supply of *Enbrel®* (etanercept) to Immunex and Wyeth; and

WHEREAS, the Parties originally intended to transfer the manufacturing process to a Second Generation Process, called the "T2 Process"; and

WHEREAS, the Parties have agreed that an alternative Second Generation Process, herein referred to as "SFP", shall be used instead of the T2 Process to manufacture *Enbrel* and for such purpose the Parties have agreed to Amendment No. 6, effective as of 27 November 2007; and

WHEREAS, [*]; and

WHEREAS, [*]; and

WHEREAS, [*]; and

WHEREAS, pursuant to Section 23.9 of the Agreement, the Agreement may only be amended and supplemented by a written instrument signed by the Parties.

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto, each intending to be legally bound, hereby agree as follows:

- 1. <u>Capitalized Terms</u>. All initially capitalized terms used herein and not defined shall have the meanings set forth in the Agreement as amended.
- 2. <u>Section 5.3(c)[*]</u>
- 3. <u>Effect of Amendment No. 6.2 on Agreement</u>. Except as otherwise set forth in this Amendment No. 6.2, all other terms and provisions of the Agreement shall remain in full force and effect. In the event of any conflict between the terms and conditions of the Agreement and the terms and conditions of this Amendment No. 6.2, the terms and conditions of this Amendment No. 6.2 shall control.

page: 1 / 2

4. <u>Counterparts</u>. This Amendment No. 6.2 may be executed in counterparts, each of which shall be deemed an original and all of which shall constitute together one and the same instrument.

IN WITNESS WHEREOF, the Parties have, by their duly authorized persons, executed this Amendment No. 6.2 as of the Amendment No. 6.2 Effective Date.

Boehringer Ingelheim Pharma GmbH & Co. KG

By:	/s/ Dr. Uwe Bucheler
Name:	Dr. Uwe Bucheler
Title:	Senior Vice President Biopharmaceuticals & Site
	Management
Date:	September 15, 2008
-	

By:/s/ DR. HANS MICHELBERGERName:Dr. Hans MichelbergerTitle:Vice President LegalDate:September 15, 2008

Immunex Corporation

By:	/s/ MADHAVAN BALACHANDRAN
Name:	Madhavan Balachandran
Title:	Senior Vice President Manufacturing
Date:	September 3, 2008

Wyeth, acting through its Wyeth Pharmaceuticals division

By:	/s/ Robert A. Dougan
Name:	Robert A. Dougan
Title:	Senior Vice President
Date:	August 26, 2008

page: 2 / 2

CERTIFICATIONS

I, Kevin W. Sharer, Chairman of the Board, Chief Executive Officer and President of Amgen Inc., certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Amgen Inc.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this quarterly report based on such evaluation; and
 - (d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 7, 2008

/S/ KEVIN W. SHARER

Kevin W. Sharer Chairman of the Board, Chief Executive Officer and President

CERTIFICATIONS

- I, Robert A. Bradway, Executive Vice President and Chief Financial Officer of Amgen Inc., certify that:
 - 1. I have reviewed this Quarterly Report on Form 10-Q of Amgen Inc.;
 - 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
 - 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this quarterly report based on such evaluation; and
 - (d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 7, 2008

/S/ ROBERT A. BRADWAY

Robert A. Bradway Executive Vice President and Chief Financial Officer

Certification of Chief Executive Officer

Pursuant to 18 U.S.C. § 1350, as created by Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of Amgen Inc. (the "Company") hereby certifies that:

- (i) the accompanying Quarterly Report on Form 10-Q of the Company for the period ended September 30, 2008 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and
- (ii) information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

 Date:
 November 7, 2008
 /S/ KEVIN W. SHARER

Kevin W. Sharer Chairman of the Board, Chief Executive Officer and President

A signed original of this written statement required by Section 906 of the Sarbanes-Oxley Act of 2002 ("Section 906"), or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Amgen Inc. and will be retained by Amgen Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

Certification of Chief Financial Officer

Pursuant to 18 U.S.C. § 1350, as created by Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of Amgen Inc. (the "Company") hereby certifies that:

- (i) the accompanying Quarterly Report on Form 10-Q of the Company for the period ended September 30, 2008 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and
- (ii) information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 7, 2008

/S/ ROBERT A. BRADWAY

Robert A. Bradway Executive Vice President and Chief Financial Officer

A signed original of this written statement required by Section 906 of the Sarbanes-Oxley Act of 2002 ("Section 906"), or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Amgen Inc. and will be retained by Amgen Inc. and furnished to the Securities and Exchange Commission or its staff upon request.